Corporate Climate Responsibility Monitor 2023

ASSESSING THE TRANSPARENCY AND INTEGRITY OF COMPANIES' EMISSION REDUCTION AND NET-ZERO TARGETS

February 2023







© NewClimate Institute 2023



Authors

Thomas Day, Silke Mooldijk, Frederic Hans, Sybrig Smit, Eduardo Posada, Reena Skribbe, Santiago Woollands, Harry Fearnehough, Takeshi Kuramochi, Carsten Warnecke, Aki Kachi, Niklas Höhne.

This document was prepared in collaboration with Carbon Market Watch.

Disclaimer

The Corporate Climate Responsibility Monitor represents the authors' views and interpretations of publicly available information that is self-reported by the companies assessed. Due to the fragmentation, inconsistency and ambiguity of some of the information provided by the assessed companies, as well as the fact that the authors did not seek to validate the public self-reported information provided by those companies, the authors cannot guarantee the factual accuracy of all information presented in this report. Therefore, neither the authors nor NewClimate Institute makes representations or warranties as to the accuracy or reliability of any information in this report. The authors and NewClimate Institute expressly assume no liability for information used or published by third parties with reference to this report.

Design

Quentin Geluyckens, Curious Cowboy SRL – www.curiouscowboy.com Cover image: Matthew Henry

J Download the report: http://newclimate.org/publications/

Table of Contents

Summary

About the corporate climate responsibility monitor

04 13

SECTION A – Good practice overview	16
1 Tracking and disclosure of emissions	18
1.1 Snapshot summary	19
1.2 Principles for good practice	21
1.2.1 Tracking and disclosure of emissions	21
1.2.2 Good practice assessment criteria	22
1.3 Trends, role models, and bad practice	23
2 Setting specific and substantiated targets	25
2.1 Snapshot summary	26
2.2 Principles for good practice	28
2.2.1 Corporate target setting for the short-, medium-, and long-term	28
2.2.2 Coverage of emission sources in targets	28
2.2.3 Emission reduction commitments	29
2.2.4 Good practice assessment criteria	29
2.3 Trends, role models, and bad practice	30
3 Reducing emissions	36
3.1 Snapshot summary	37
3.2 Principles for good practice	39
3.2.1 Emission reduction measures	39
3.2.2 Renewable electricity generation and procurement	40
3.2.3 Good practice assessment criteria	43
3.3 Trends, role models, and bad practice	44
4 Climate contributions and offsetting	47
4.1 Snapshot summary	48
4.2 Principles for good practice	50
4.2.1 Climate contributions without a neutralisation claim	50
4.2.2 Offsetting claims	51
4.2.3 Good practice assessment criteria	55
4.3 Trends, role models, and bad practice	56
4.4 Spotlight on key questions	61
4.4.1 How to assure offset additionality in the context of the Paris Agreement?	61
4.4.2 Is it possible to mitigate the non-permanence of climate impacts from biological carbon dioxide removals?	62
4.4.3 Is the tech industry really carbon neutral?	64
4.4.4 What does insetting and climate positive really mean?	66
4.4.5 Are companies ready for a shift to the climate contribution approach?	67

SECTION B — Company assessments	69
Ahold Delhaize	72
Amazon	74
American Airlines	76
Apple	78
ArcelorMittal	80
Carrefour	82
Deutsche Post DHL	84
Fast Retailing	86
Foxconn	89
Google	90
H&M Group	92
Holcim	94
Inditex	96
JBS	98
Maersk	100
Mercedes-Benz Group	102
Microsoft	104
Nestlé	106
PepsiCo	108
Samsung Electronics	110
Stellantis	112
Thyssenkrupp	114
Volkswagen Group	116
Walmart	118

Glossary and abbreviations	120
Annex I Companies assessed in this report	125
Annex II Target integrity assessments	127
Annex III Comparison of SBTi verifications with CCRM assessment	162
Annex IV Assumptions for aggregated analysis	164
References	165

Summary

About the Corporate Climate Responsibility Monitor

The Corporate Climate Responsibility Monitor evaluates the transparency and integrity of companies' climate pledges.

Companies around the world are increasingly alert to the climate emergency. They face calls from a growing range of stakeholders to take responsibility for the impact of their activities. Most large companies now have public climate strategies and targets, many of which include pledges that appear to significantly reduce, or even eliminate, their contributions to global warming. The rapid acceleration of corporate climate pledges, combined with the fragmentation of approaches, means that it is more difficult than ever to distinguish between real climate leadership and unsubstantiated greenwashing. This is compounded by a general lack of regulatory oversight at international, national and sectoral levels. Identifying and promoting real climate leadership, and sorting it from greenwashing, is a key challenge that, where addressed, has the potential to unlock greater global climate change mitigation ambition.

The 2023 Corporate Climate Responsibility Monitor assesses the climate strategies of 24 major global companies, critically analysing the extent to which they demonstrate corporate climate leadership (Section A, summarised in Table S.1). We evaluate the integrity of climate pledges against good practice criteria to identify examples for replication, and highlight areas where improvement is needed (Section B, summarised in Table S.2). This is the second iteration of the *Corporate Climate Responsibility Monitor*, whereby the 2022 analysis revealed a number of issues with corporate climate strategies.

We assess and draw insights on transparency and integrity in four main areas of corporate climate action:

Tracking and disclosure of emissions (section A1)

Setting emission reduction targets (section A2)

Reducing own emissions (section A3)

Climate contributions and offsetting claims (section A4).

The 24 companies assessed in this report are major multinational companies. They comprise of the largest three global companies with bold climate pledges from eight consumer-facing and industrial sectors (see section B for further details on selection criteria). They reported combined revenues of USD 3.16 trillion in 2021, approximately 10% of the total revenue from the world's largest 500 companies. Their total self-reported GHG emission footprint in 2019, including upstream and downstream emissions (scope 3) that may include a marginal degree of overlap, amount to approximately 2.2 GtCO₂e. This is equivalent to roughly 4% of global GHG emissions in 2019. Ten of the 24 companies (selected through the process described in Section B) were also assessed in the *2022 Corporate Climate Responsibility Monitor*. The repeat analysis of this small sub-set of companies offers insights into what progress has been made over the past year.



Key insights

Most companies' climate strategies are mired by ambiguous commitments, offsetting plans that lack credibility and emission scope exclusions, but replicable good practice can be identified from a minority.

The companies analysed in the 2023 Corporate Climate Responsibility Monitor have put themselves forward as climate leaders. Of eight major-emitting sectors, we assessed the largest three global companies that are members of an initiative affiliated with the Race to Zero campaign, creating a sample of 24 companies. Through this affiliation, these companies have committed themselves to preparing and implementing decarbonisation plans that align with the objective to limit warming to 1.5°C. These companies serve as role models for other large, medium, and small companies around the world. The analysis of these companies should provide the best prospects for the identification of replicable good practice. Scrutiny of their plans is also necessary to identify whether these are truly influential leaders that set the right examples.

Overall, we find the climate strategies of 15 of the 24 companies to be of low or very low integrity. We found that most of the companies' strategies do not represent examples of good practice climate leadership. Companies' climate change commitments do not add up to what their pledges might suggest (Figure S1). Their combined emission reduction commitments are wholly insufficient to align with 1.5°C-compatible decarbonisation trajectories; targets and potential offsetting plans remain ambiguous; and the exclusion of emission scopes severely undermines the targets of several companies. The integrity of emissions disclosure practices is more encouraging. We found most companies' disclosure to have at least a moderate level of transparency. We also found examples of companies with credible decarbonisation commitments, and companies taking proactive and innovative action to reduce GHG emissions, but these good practice examples represent a minority.

We identify limited progress in the transparency or integrity of companies' climate strategies over the past year. Since the publication of the 2022 Corporate Climate Responsibility Monitor, there have been a number of important developments in the guidelines and governance of corporate climate strategies: the Science Based Targets initiative's (SBTi) Net Zero Standard entered a new phase of implementation, a UN-convened high-level expert group (HLEG) published recommendations for credible corporate climate targets, while the International Standards Organisation (ISO) published guidelines for net-zero targets. Yet, for the ten companies that we also analysed in the previous iteration of the Corporate Climate Responsibility Monitor, we have identified only limited signs of improvement in the transparency or integrity of some companies' strategies, while many of the key issues previously highlighted persist.

The following pages provide a summary of our key insights.



Figure S1: Net-zero pledges break down to only moderate emission reductions alongside offseting and scope exclusions

The 24 companies assessed in this report are not necessarily a representative sample of all corporate actors with net-zero targets. They represent 24 of the largest companies in the world, accounting for approximately 4% of global GHG emissions and revenues of USD 3.16 trillion in 2021. We anticipate that any overlap in the scope 1, scope 2 and scope 3 emissions of these companies is marginal and of limited significance to the key insights derived from this report.

There is a critical need to shift attention to the 2030 blind spot. Companies' climate pledges for 2030 fall well short of the required ambition and are inappropriately verified.

Companies' 2030 targets cannot be taken at face value. Nearly all the 24 companies that we assessed have pledged 2030 targets, but we find that these targets can rarely be taken at face value. For many companies, 2030 targets address only a limited scope of emission sources, such as only direct emissions (scope 1) or emissions from procured energy (scope 2) and only selected other indirect emission categories (scope 3). Scope 3 emissions account for over 90% of the GHG emission footprints for most of the companies we have assessed. For others, 2030 targets are misleading due to reliance on offsetting.

Climate pledges for 2030 fall well short of the economy-wide emission reductions required to stay below the 1.5°C

temperature limit. For the 22 companies with targets for 2030, we find that these targets translate to a median absolute emission reduction commitment of just 15% of the full value chain emissions between 2019 and 2030. This may increase to 21% under the most optimistic scenario that emission intensity targets translate to equivalent absolute emission reductions (Figure S2). This compares to the need to cut global GHG and CO_2 emissions by 43% and 48% between 2019 and 2030 respectively, to be in line with the goal to limit the global temperature increase to 1.5°C (IPCC, 2022).

Third-party certifications lend credibility to companies whose targets are highly insufficient. The SBTi has certified the 2030 targets of 16 of the 24 companies included in this analysis as aligned with a pathway to limit global temperature increase to 1.5°C, 'well-below 2°C', or 2°C. Another five companies are listed on SBTi's website as 'committed' to science-based targets for 2030. Most of these companies highlight their SBTi certifications prominently in their climate-related communications as well as in litigation processes, to defend targets that are highly insufficient and sometimes misleading. We find the majority of these companies' 2030 targets to be of poor integrity, due to them not meeting the 1.5°C compatible benchmarks that scientific and grey literature provide, such as from SBTi's own methodologies (see Table 9 in section 2.3 and Table 19 in the Annex for company-specific comparisons).

Figure S2: The median commitment to emission reductions between 2019 and 2030 is just 15-21%

This chart shows the **proportion of full value chain GHG emissions that companies commit to reduce between 2019 and 2030**. Data includes 22 companies. 2 companies without clear commitments for 2030 are not included.



LEGEND OPoor integrity Moderate integrity High integrity

The colour of the data points represents our assessment of the integrity of company's 2030 targets, based on their sufficiency compared to sector-specific 1.5 °C aligned benchmarks, and the appropriateness of the terminology used in the pledge communication. Data presented on a single x-axis. Placement on y-axis has no significance.

The data points in this graphic represent the authors' most optimistic interpretations of companies' absolute emission reduction commitments, including emission intensity targets, based on publicly available information. The median is provided for scenarios with and without the optimistic assumption that intensity targets result in equivalent absolute emission reductions. Targets that are reliant on offsets to an undefined extent are marked as ambiguous. Further details and explanation on the authors' interpretation of companies' targets can be found in the individual company cases in Section B and in Annex II.

Only a minority of net-zero pledges represent credible commitments to deep decarbonisation, while many remain highly ambiguous.

Only five of 24 companies' net-zero pledges represent a commitment to deep decarbonisation. All the 24 companies that we assessed have pledged to reach net-zero emissions, carbon neutrality or other pledges of equivalent terminology, but the general quality of those pledges remains poor. Just five out of the 24 companies – H&M Group, Holcim, Stellantis, Maersk, and Thyssenkrupp – commit to decarbonise their emissions by at least around 90% by their respective net-zero target years. We find the long-term targets of 17 companies to be of poor integrity, due to the inadequacy or complete lack of explicit emission reduction commitments alongside ambiguous net-zero pledges (Figure S3). Overall, the net-zero pledges of the 24 companies translate to a commitment to reduce just 36% of the companies' combined GHG emission footprint, by the respective net-zero target years (Figure S1).

SBTi's Net Zero Standard may provide a framework for credible long-term targets, but compliance must be checked

thoroughly and monitored in the future. Two of the companies we assessed – H&M Group and Holcim – have had their netzero pledges certified by SBTi under its Net Zero Standard. We found the net-zero pledges of these companies to be of high and moderate integrity, respectively. The SBTi's Net Zero Standard requires that net-zero pledges should equate to at least 90% emission reductions across the full value chain. This directly addresses the key issue that we identify for most companies' net-zero pledges, which are either ambiguous or do not translate to commitments for deep decarbonisation. However, the SBTi's Net Zero Standard is not watertight and depends on companies' willingness to comply with the framework fully and transparently: the *2022 Corporate Climate Responsibility Monitor* (Day *et al.*, 2022) found that the SBTi-certified net-zero pledge of CVS Health was undermined by highly contentious baseline setting. The credibility and real ambition of H&M Group's and Holcim's targets could be undermined by the companies' plans to use bioenergy and RECs, and CCS technologies, respectively. Finally, rigorous verified targets are only one part of the overall integrity assessments, as these need to be backed by credible and detailed plans to achieve them and accompanied by a satisfactory level of transparency. This is why some companies can score well at the 'target setting' level, while still performing moderately or poorly in the overall assessment.

Figure S3: Just 5 of 24 companies commit to deep decarbonisation with their net-zero pledges

This chart shows the **proportion of full value chain GHG emissions that companies commit to reduce with their net-zero pledges.** Data includes 12 companies. For 12 other companies the meaning of the net-zero target is ambiguous.



LEGEND

O Poor integrity Moderate integrity High integrity

The colour of the data points represents our assessment of the integrity of company's long-term targets, based on their sufficiency compared to sector-specific 1.5 °C aligned benchmarks, and the appropriateness of the terminology used in the pledge communication. Data presented on a single x-axis. Placement on y-axis has no significance.

The data points in this graphic represent the authors' most optimistic interpretations of companies' emission reduction commitments, including emission intensity targets, based on publicly available information. The chart includes emission reduction commitments under net-zero targets, carbon neutrality pledges and other pledges with equivalent terminology, for the respective target year, which ranges between 2030 and 2050. Targets that are reliant on offsets to an undefined extent are marked as ambiguous. Further details and explanation on the authors' interpretation of companies' targets can be found in the individual company cases in Section B and in Annex II.

Offsetting plans – under various guises – still remain a major stumbling block for the credibility of corporate climate strategies, but the stage is set for a transition to the climate contribution approach.

Developments in 2022 charted a clearer path for a transition from offsetting to a climate contribution claim approach, but companies' climate strategies appear to be behind the curve of these developments. Misleading offsetting claims are increasingly recognised as a legal liability. The climate contribution approach has traction towards wider implementation through the COP 27 decision to create a 'mitigation contribution' unit under Article 6.4 (UNFCCC, 2022), as well as the first-mover announcement from myclimate that it will discontinue its climate neutrality label and transition to a climate contribution model (myclimate, 2022). Companies' climate strategies are not yet in step with these developments; just four of the 24 companies we assessed reported activities or donations that could be interpreted as climate contributions without a neutralisation claim. Meaningful developments in guidelines and frameworks for the climate contribution model may facilitate progress in 2023.

Transparency and integrity of carbon neutrality claims remain critically low. Half of the 24 companies assessed – including most consumer-facing brands – made some form of carbon neutrality claim in 2021 and 2022. These are supposed to be reflections of a company's current impact, as opposed to a distant net-zero target. We found the integrity of every one of those claims to be poor. The integrity of carbon neutrality claims is compromised by various issues, including the scope coverage, and the quality of the carbon credits procured. From the companies assessed, the average company's carbon neutrality claims apply to a company's entire business. We find that the offsetting projects that companies select are highly contentious as they are neither additional nor likely to result in permanent emission removals.

The 24 sampled companies plan to offset 23–45% of their combined 2019 emission footprint to claim achievement of their long-term net-zero pledges. Twenty-three of the 24 companies will rely on some form of offsetting towards their targets in the future. The potential role for offsets is left uncertain by many companies, but we estimate from the available information that companies jointly plan to offset at least 23% and up to 45% of their combined emissions footprint. This is far in excess of the maximum role for offsetting (albeit under different terminologies) indicated by SBTi's Net Zero Standard (10%, SBTi, 2021c) and the ISO Net Zero Guidelines (case specific, but less than 5% in most cases, ISO, 2022a, 2022b).

Planned reliance on forestry- and land-related offsets outstrips the technical potential of the world's natural resources and is fundamentally flawed due to the non-permanence of biological carbon sinks. We find that at least three quarters of the 24 sampled companies rely on forestry and land-use related offsets. The demand for such carbon dioxide removals would exceed the potential of the world's natural resource base by around 2-4 times (see section A 4.3), if these practices would be replicated by other companies. Moreover, these plans demonstrate the widespread lack of awareness that the biological storage of carbon is fundamentally unsuitable for offsetting claims due to the non-permanence of the climate impact. Companies are increasingly proposing plans to overcome the issue of non-permanence, but we find that these plans are implausible (see **Spotlight 4.4.2**: *'Can the non-permanence of climate impacts from nature-based solutions be mitigated?'*).

Offsetting under the guise of 'insetting' is gaining traction and legitimacy, although this practice leads to low credibility offsetting claims and the double counting of emission reductions. 'Insetting' is a business-driven concept with no universally accepted definition. Several companies are advocating for 'insetting' as an alternative to offsetting, but the insetting measures that we have identified amount de facto to the unregulated offsetting of emissions, usually through biological carbon dioxide removals within the value chain (see *Spotlight 4.4.4*: '*What does insetting really mean?*'). This illegitimate concept has gained considerable traction over the past year, and its potential to significantly undermine corporate strategies is already being realised. Nestlé, PepsiCo, JBS and Deutsche Post DHL already employ 'insetting' today or plan for it to be a significant component of future pledges to illegitimately claim that their emissions have been or will be offset. Some of these companies actively distance themselves from 'offsetting' and lobbying for the recognition of that approach. The new SBTi FLAG guidance allows companies to claim the achievement of their emission reduction targets through 'insetting', breaking from the long-held SBTi position that emission reduction targets should only be achieved through emission reductions. SBTi explicitly acknowledges that the definition of insetting and its suitability towards emission reduction targets remains uncertain, while still allowing its use (SBTi, 2021c, p. 30 Box 3). This may have grave consequences for the credibility of climate targets from companies with significant land use emissions, including not only agri-businesses but also retailers and the fashion industry.

Companies' proposed measures are insufficient to catalyse the transformational change that is necessary to limit global warming to 1.5°C.

The majority of companies in this report present measures targeting emissions across the value chain, but their plans usually lack substance. Of the 24 companies assessed, 18 present reduction plans for upstream and downstream scope 3 emissions. Another four companies present measures to reduce upstream emissions only, while two companies (JBS and ArcelorMittal) present no measures at all for scope 3. These findings indicate that companies recognise their responsibility for emissions upstream and downstream the value chain. However, in almost all cases, it remains unclear at what scale measures are implemented and what share of current GHG emissions they address.

While most companies state to work on developing nascent zero-carbon technologies, these generally do not address company's key emission sources. Although 20 of 24 companies state that they invest in new solutions for zero-carbon technologies, we identified only a handful of companies that are working on technologies that address key emission sources. These include, for instance, Maersk, which invests in alternative fuels and vessels, Google, which is pioneering 24/7 monitoring and matching renewable energy generation with consumption, and Deutsche Post, which invests in electrifying its fleet and scaling up the production of low-carbon fuels for road transport.

Companies currently use bioenergy to reduce their emissions footprint, but large-scale reliance on bioenergy is very likely to have negative sustainability implications. Bioenergy use is associated with a range of sustainability issues, including deforestation, biodiversity loss, GHG emissions, and food insecurity. Despite those drawbacks, at least 11 companies plan on using bioenergy to decrease their emissions, including all companies in the transport and heavy industry sectors. Most of them claim that they use 'sustainable' bioenergy, but even the best available options for bioenergy can still exacerbate sustainability problems. The potential for sustainable bioenergy is limited and outweighed by demand, so companies buying sustainable bioenergy, push other companies to use non-sustainable bioeness.

A few companies demonstrate leadership with innovative and higher-quality approaches for sourcing renewable electricity, but overall, the integrity of companies' procurement constructs remains low.

Some of the tech companies in this report are pioneering new approaches to securing 100% renewable electricity consumption on an hourly basis. On-site installation and PPAs on the local grid are generally the most impactful procurement constructs that companies can pursue. However, there is a mismatch between renewable electricity generation and consumption. For instance, a company that has signed a PPA for a wind park cannot claim to use renewable energy on windless days. Matching renewable energy generation with production around the clock addresses this problem. Google and Microsoft have both pledged that 100% of their energy consumption will be matched 24/7 with renewable energy generation by local installations by 2030.

Despite these positive examples, we find that most companies' approaches to securing renewable energy are shallow and unlikely to result in significant additional renewable energy capacity. At least 14 of the 24 companies in the report rely on Renewable Electricity Certificates (RECs) to make bold claims about how they reduce electricity-related emissions. RECs are, however, unlikely to send a signal to the market that there is demand for renewable electricity; they will most likely not result in the installation of new renewable electricity capacity.

We see some traction for taking responsibility for energy-related emissions in the value chain. Apple, Walmart, Foxconn and H&M Group support their suppliers in purchasing renewable electricity – financially, through guidance, or by facilitating arrangements. This is an important component of decarbonising the supply chain, although the real impact of this support depends on the procurement constructs that suppliers end up using. Support for suppliers is critical not only because many suppliers may be too small to set up PPAs or on-site installation themselves, but also because renewable energy policies are underdeveloped in some countries, which hinders suppliers in procuring renewable electricity. This problem is particularly relevant in many Asian countries with large manufacturing industries.

On a more positive note, companies' emission disclosure in 2022 indicates a solidifying consensus and level of understanding on good practice.

Companies exhibit a moderate degree of transparency in their emission disclosure. Of the 24 companies we assessed, we found that 16 companies disclosed emissions data with at least a moderate level of transparency and integrity, for all emission scopes. These trends indicate that it is now common practice for companies we have assessed to acknowledge full responsibility for upstream and downstream scope 3 emissions, and emissions from subsidiaries. This reflects the increasing level of alignment and clarity on these issues among existing and newly published guidelines. Moreover, we see signs that this moderate level of transparency continues to improve.

Companies exhibiting bad practice for emission disclosure are in the minority, but these bad practices remain significant. A minority of companies still publish highly misleading emissions data, usually by excluding major emission sources from their public documentation. Even if fully disclosing all relevant emission scopes, some companies' estimates could remain contentious since they may significantly *underreport* certain emission scopes. For example, recent analysis suggests that automobile manufactures Volkswagen, Mercedes-Benz and Stellantis all underreport their disclosed life-cycle emissions of sold vehicles by more than 50% due to unrealistic assumptions on vehicle lifetimes (Bonaccorsi *et al.*, 2022). We did not identify any significant improvements over the past year from the minority of companies with poor emission disclosure practices.

2023 must be the year for regulators, companies and voluntary initiatives to align with the solidifying consensus on what constitutes good practice for corporate climate responsibility.

Companies must play a central role in finding and scaling up solutions for deep decarbonisation, but their efforts need urgent acceleration and appropriate regulatory frameworks. The findings of this report indicate that regulators cannot rely on consumer and shareholder pressure to drive corporate action, nor can they rely on existing voluntary initiatives to ensure compliance with the necessary standards for credible and transparent corporate climate action. In particular, companies' plans for the period up to 2030 fall far short of the efforts needed in this crucial decade for climate action to stand a reasonable chance of limiting global warming to 1.5°C. Forthcoming regulation, for example the EU's Corporate Sustainability Reporting Directive entering into force in 2023 (EU, 2022), will introduce tighter requirements for corporate climate strategies, but their final implementation will need to be closely monitored to ensure a high standard of compliance. The publication of the UN HLEG recommendations and the ISO Net Zero Guidelines at COP 27 demonstrated the solidifying consensus on what constitutes good practice for corporate climate responsibility. Although these publications largely took stock of the existing consensus on good practice, rather than proposing new and more ambitious benchmarks, most companies' strategies fall far short of the established recommendations.

Urgent action is needed to unlock the potential of corporate climate leadership in this crucial decade of climate action towards 2030. 2023 is an important year for regulators, companies, and the standard-setting initiatives to step up and align with the requirements set out in the scientific literature for immediate action towards deep decarbonisation. Having focused largely on the integrity of net-zero pledges in recent years, regulators and voluntary initiatives must place a renewed and urgent focus on the integrity of companies' emission reduction plans up to 2030, ensuring that the discourse on longer-term net zero does not distract from this most immediate and unfulfilled objective.

•

10

Table S1: Overview of corporate climate responsibility good practice assessment for 24 companies in 2023

1 TRACKING AND DISCLOSING EMISSIONS		GOOD PRACTICE	TRANSPARENC	Y & INTEGRITY
Comprehensiveness of disclosure	~	Disclose full details on their GHG emissions on an annual basis, with a breakdown of the data to specific emission sources (including scope 1, 2, 3 and non-GHG climate forcers) and the presentation of historical data for each emission source.	Moder	rate 🔺
2 SETTING SPECIFIC AND		GOOD PRACTICE	AVERAGE PERFORMANCE	* & CHANGE SINCE 2022
SUBSTANTIATED TARGETS			TRANSPARENCY⁺	INTEGRITY⁺
Short- & medium-term targets towards 2030	~	Set short- and medium-term emission reduction targets <i>towards</i> 2030 within five-year intervals that reflect a commitment to immediate action and accountability. Targets should be independent from offsetting and aligned with 1.5°C-compatible trajectories in the sector, across all emission scopes.	Moderate 🔺	Very low —
Long-term targets beyond 2030	~	Set specific long-term emission reduction targets <i>beyond 2030</i> that are independent from offsetting and aligned with 1.5°C-compatible trajectories in the sector, across all emission scopes, as a <i>vision</i> for deep decarbonisation.	Low —	Very low —
3 REDUCING EMISSIONS		GOOD PRACTICE	AVERAGE PERFORMANCE	* & CHANGE SINCE 2022
3 REDUCING EMISSIONS		GOOD FRACTICE	TRANSPARENCY ⁺	INTEGRITY ⁺
Emission reduction measures	~	Implement encompassing and deep decarbonisation measures and disclose details of those measures to support replication.	Moderate 🔺	Low —
	~	Refrain from using bioenergy where alternatives to combustion exist, and ensure that any bioenergy they use does not have negative sustainability implications.	-	
Renewable energy procurement	~	Procure the highest quality renewable electricity available and disclose the details of that procurement.	Low —	Very low —
4 CLIMATE CONTRIBUTIONS		GOOD PRACTICE	AVERAGE PERFORMANCE	
AND OFFSETTING			TRANSPARENCY⁺	INTEGRITY ⁺
Responsibility for unabated emissions	~	Pursue high transparency and integrity on climate contributions and any neutralisation claims made today (see criteria below).	Low —	Very low —
Climate contributions	~	Provide an ambitious volume of financial support to climate change mitigation activities beyond the value chain, without claiming to neutralise the company's own emissions.	Low 🔺	Very low —
Offsetting claims today		Clearly disclose offsetting claims and plans; avoid misleading pledges and claims; avoid risk of distraction by also committing to measures for deep emission	Low —	Very low —
Offsetting plans for the future	~	reductions; commit to procure only high-quality credits from ambitious projects with a permanent climate impact; and commit to preventing any form of double-counting of climate impacts.	Low —	Very low —

* Transparency and integrity columns: the bar indicates the distribution of our rating of the 24 companies (**Poor Moderate High n.a**.); the text above the shaded bars represents the average rating across all the companies we assessed, calculated excluding non-applicable cases, on a 5-point scale (**Very low, Low, Moderate, Reasonable, High**); and an indication of progress since the last analysis in 2022 (▲ — ▼), based on the authors' interpretation of progress from the companies that were analysed also in 2022, against the current methodology version. Good practices were derived from the principles elaborated in the following subsections, and from a compilation of the practices identified from existing company strategies. Full details on the assessment methodology can be found in the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0* (NewClimate Institute, 2023b).

Table S2: Overview of companies assessed in the Corporate Climate Responsibility Monitor 2023

HIGH INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
No companies achieved a high integrity rating				
REASONABLE INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
Maersk	Net zero by 2040			p. 100
MODERATE INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
Apple	Carbon neutral by 2030			p. 78
Arcelor Mittal	Net zero by 2050	\bigcirc		p. 80
Google	Net zero by 2030			p. 90
H&M Group	Net zero by 2040			p. 92
Holcim	Net zero by 2050			p. 94
Microsoft	Carbon negative by 2030		\bigcirc	p. 104
Stellantis	Net-zero carbon by 2038			p. 112
Thyssenkrupp	Climate neutral by 2050			p. 114
C LOW INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
Ahold Delhaize	Net zero by 2050		C	p. 72
Amazon	Net-zero carbon by 2040	C	C	p. 74
Deutsche Post DHL	Net zero by 2050	\bigcirc	C	p. 84
Fast Retailing	2030 emission reduction targets	\bigcirc	G	p. 86
Foxconn	Net zero by 2050	G	G	p.88
Inditex	Net zero by 2040		G	p. 96
Mercedes-Benz	Carbon neutral vehicles by 2039	O	G	p. 102
Nestlé	Net zero by 2050	O	G	p. 106
PepsiCo	Net zero by 2040	O	G	p. 108
Volkswagen	Carbon neutral by 2050		G	p. 116
Walmart	Zero emissions by 2040		C	p. 118
VERY LOW INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
American Airlines	Net zero by 2050		0	p. 76
Carrefour	Carbon neutral by 2040	0	0	p. 82
JBS	Net zero by 2040	\circ	0	p. 98
Samsung Electronics	Net-zero carbon by 2050	G	0	p. 110

RATINGS 5-point scale High Reasonable Moderate C Low Very low. See individual company analyses.

Assessments were made based on public information identified by the authors. A poor rating may not necessarily be an indication that a company's climate strategy is weak, but could also indicate that the information was insufficient to confirm good practice. Ambitious companies can improve their ratings by ensuring that all aspects of their climate responsibility strategies are transparently and accurately disclosed, and in the public domain.

About the corporate climate responsibility monitor

The need for scrutiny on corporate climate action

Many companies are putting themselves at the forefront of climate action. The rate of corporate climate pledge setting is accelerating exponentially: by January 2023, nearly 9,000 companies had joined the UNFCCC's Race to Zero campaign (UNFCCC, 2023), up from just over 3,000 one year earlier (Day *et al.*, 2022).

Civil society's increasing concern with the urgency of the climate crisis is resulting in more pressure from consumers, shareholders and regulators for companies to decarbonise. In parallel, companies realise that the direction of travel is set for the decarbonisation of the global economy, and it is increasingly attractive for them to assume a leading role in that new paradigm. Many companies are scrambling for new approaches and narratives to demonstrate their climate leadership, recognising that historical approaches face limitations in today's context.

The rapid acceleration of corporate climate pledge setting, combined with the fragmentation of approaches and the general lack of regulation or oversight, means that it is more difficult than ever to distinguish between real climate leadership and unsubstantiated greenwashing.

The goalpost of what constitutes good practice climate action for companies has shifted in the era of the Paris Agreement and the increasingly clear scientific evidence that underpins its urgency. With the objectives of the Paris Agreement, greenhouse gas emissions need to be reduced at speed, in all countries and in all sectors. The 1.5° C limit requires a reduction in global greenhouse gases and CO₂ emissions by 43% and 48% respectively from 2019 levels by 2030, to reach a state of net-zero global CO₂ emissions by around 2050, net-zero emissions of all greenhouse gases by around 2070, and netnegative emissions thereafter (IPCC, 2022). Company actions that were considered viable in the era of the Kyoto Protocol only ten years ago are no longer sufficient. For example, it is no longer sufficient for companies to only address their own direct emissions; rather, companies now need to address upstream and downstream emissions as well. It is no longer good practice for a company to compensate for emissions by reducing or removing emissions elsewhere; rather, emission reductions and removals 'elsewhere' need to be enhanced in parallel to the company's emission reductions, to reach global net zero.

A new mindset and evaluation standard for companies is emerging. While in the Kyoto era only some countries were required to act, companies now need to ask themselves: 'Would we reach global net-zero emissions if all would do what we are doing?'

The difficulty of distinguishing real climate leadership from greenwashing is a key challenge that, where addressed, has the potential to unlock more substantial global climate change mitigation. Corporate climate action is key to closing the emissions gap to a 1.5°C-aligned emissions pathway. In a short space of time, and in the absence of sufficient top-down regulation, consumers' and shareholders' expectations have become a major driver for enhanced corporate climate action. Companies appear to be responding. To facilitate this important bottom-up pressure mechanism, it is essential that the credibility of companies' strategies is transparent and can be understood by their target audiences.

The Corporate climate responsibility monitor

The Corporate Climate Responsibility Monitor evaluates the transparency and integrity of companies' climate pledges. The objectives of the Corporate Climate Responsibility Monitor are:

- Identify and highlight good practice approaches that can be replicated by other companies, recognising that companies are experimenting to work out what is constructive and credible practice.
- Reveal the transparency and integrity of major companies' climate leadership claims and provide a structured methodology for others to replicate such an evaluation. **Transparency** refers to the extent to which a company publicly discloses the information necessary to fully understand the integrity of that company's approaches towards the various elements of corporate climate responsibility. **Integrity**, in this context, is a measure of the quality, credibility and comprehensiveness of those approaches.
- Scrutinise the credibility of companies' plans for offsetting their emissions through carbon dioxide removals or emission reduction credits, recognising that voluntary carbon markets are highly fragmented and there remains a lot of uncertainty on credible good practice.

The Corporate Climate Responsibility Monitor focuses on four main areas of corporate climate action: tracking and disclosure of emissions (section A1), setting emission reduction targets (section A2), reducing own emissions (section A3) and taking responsibility for unabated emissions through climate contributions or offsetting (section A4). Evaluations for 24 major global companies are set out in Section B. The Corporate Climate Responsibility Monitor is prepared by **NewClimate Institute** and **Carbon Market Watch**. The consortium partners combine years of experience with the independent critical analysis of corporate climate action and carbon market mechanisms. NewClimate Institute and Carbon Market Watch are both not-for-profit organisations. Neither of the institutions hold private commercial interests in voluntary carbon credit markets.

Development of the Corporate Climate Responsibility Monitor

The Corporate Climate Responsibility Monitor is based on the guiding principles for good practice corporate climate responsibility set out in the accompanying methodology document: Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0 (NewClimate Institute, 2023b). We have drawn these guiding principles from a combination of scientific literature review, previous work by the authors, and the identification of existing good practices from company case studies.

The guiding principles identified in this document and the accompanying methodology document relate to issues where the state of scientific knowledge and debate is rapidly evolving. The contents of this document represent the views of the authors, based on our interpretation of existing research and current developments. Our assessments of specific companies are based upon these perspectives and interpretations, which may not be universally held views.

The *Corporate Climate Responsibility Monitor* promotes transparency with the philosophy that consumers, shareholders, regulators and wider observers should be able to follow and assess the integrity of companies' claims. Accordingly, the company assessments in section B are based only on publicly available information that the authors were able to identify (see Annex-Data Sources in the Methodology document). Each rating represents the authors' understanding of the publicly available information. In some cases, company information was scattered across different sources (e.g., annual reports, press releases and statements, webpages, or other marketing materials); it is possible in this process that information may have been misinterpreted, or that relevant information was overlooked. Companies should consider how to present information as transparently as possible, to ensure that observers are able to identify all the relevant information necessary to understand their climate strategies.

We assess the transparency and integrity of companies' strategies based on the information that is self-reported by the companies. We do not assess or certify the accuracy or truth of the information provided by companies, including their GHG emission reporting. In specific cases, we supplement self-reported information from the companies with information that we have identified from other sources, but we cannot guarantee the accuracy of that information.

 \rightarrow See also the assessment methodology for the Corporate Climate Responsibility Monitor. Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0 (NewClimate Institute, 2023b).

SECTION A Key elements of corporate climate responsibility

This section provides an overview of the good practice criteria and assessment methodology for the *Corporate Climate Responsibility Monitor*, based on the specific principles for credibility in four focus areas: tracking and disclosure of emissions; setting specific and credible targets; reducing own emissions; and responsibility for unabated emissions. For each of these four focus areas, we discuss trends, good practices and challenges identified from the companies we assessed.

Good practice overview

Corporates looking to take a position of climate leadership can learn from each other to replicate good practice approaches that are transparent, constructive and robust. The *Corporate Climate Responsibility Monitor 2023* assesses 24 major global companies to draw out good practice in four key areas:

1. Tracking and disclosure of emissions (section 1)

To develop a comprehensive and robust climate strategy, it is key that companies understand and are transparent about their GHG emission footprints and their trajectories. Section 1 presents good practice principles and trends for tracking and disclosure of emissions.

2. Setting specific and substantiated targets (section 2)

Companies' headline climate change pledges encompass a broad range of target setting approaches. Regardless of the type of target and the terminology used, the commitments should send a clear signal for immediate action to decarbonise the value chain, and should avoid misleading consumers, shareholders, observers and regulators. Section 2 presents good practice principles and trends for setting specific and substantiated targets, considering the coverage of emission sources, the explicit specification of an emission reduction target as part of the headline pledge, and the substantiation of long-term visions through interim targets.

3. Reducing emissions (section 3)

16

Encompassing measures for deep emission reductions are the backbone of ambitious corporate climate targets. Section 3 presents good practice principles and trends for reducing emissions, including a special focus on good practice for sourcing renewable electricity.

4. Climate contributions and offsetting (section 4)

•••••

Corporate climate leadership includes not only ambitious target setting, but also taking responsibility for unabated emissions. Section 4 explores good practice and trends related to two distinct approaches for assuming responsibility for unabated emissions: climate contributions and offsetting claims.

The specific assessments include a rating of the transparency and integrity of companies' approaches. **Transparency** refers to the extent to which a company publicly discloses the information necessary to fully understand the integrity of that company's approaches towards the various elements of corporate climate responsibility. **Integrity**, in this context, is a measure of the quality, credibility and comprehensiveness of those approaches.

Table 1 provides an overview of good practice corporate climate responsibility and the rating methodology for the *Corporate Climate Responsibility Monitor 2023* for transparency and integrity in each of these four areas. Full details on the methodology can be found in the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0* (NewClimate Institute, 2023b).

Table 1: Overview of corporate climate responsibility good practice assessment for 24 companies in 2023

1 TRACKING AND DISCLOSING EMISSIONS		GOOD PRACTICE	TRANSPARENC	Y & INTEGRITY
Comprehensiveness of disclosure	~	Disclose full details on their GHG emissions on an annual basis, with a breakdown of the data to specific emission sources (including scope 1, 2, 3 and non-GHG climate forcers) and the presentation of historical data for each emission source.	Moder	rate 🔺
2 SETTING SPECIFIC AND SUBSTANTIATED TARGETS		GOOD PRACTICE	AVERAGE PERFORMANCE	* & CHANGE SINCE 2022 INTEGRITY*
Short- & medium-term targets towards 2030	~	Set short- and medium-term emission reduction targets <i>towards</i> 2030 within five-year intervals that reflect a commitment to immediate action and accountability. Targets should be independent from offsetting and aligned with 1.5°C-compatible trajectories in the sector, across all emission scopes.	Moderate 🔺	Very low —
Long-term targets beyond 2030	~	Set specific long-term emission reduction targets <i>beyond</i> 2030 that are independent from offsetting and aligned with 1.5°C-compatible trajectories in the sector, across all emission scopes, as a <i>vision</i> for deep decarbonisation.	Low —	Very low —
3 REDUCING EMISSIONS		GOOD PRACTICE	AVERAGE PERFORMANCE	* & CHANGE SINCE 2022
Emission reduction measures	~ ~	Implement encompassing and deep decarbonisation measures and disclose details of those measures to support replication. Refrain from using bioenergy where alternatives to combustion exist, and ensure that any bioenergy they	Moderate 🔺	Low —
Renewable energy procurement	~	use does not have negative sustainability implications. Procure the highest quality renewable electricity available and disclose the details of that procurement.	Low —	Very low —
4 CLIMATE CONTRIBUTIONS AND OFFSETTING		GOOD PRACTICE	AVERAGE PERFORMANCE	* & CHANGE SINCE 2022 INTEGRITY*
Responsibility for unabated emissions	~	Pursue high transparency and integrity on climate contributions and any neutralisation claims made today (see criteria below).	Low —	Very low —
Climate contributions	~	Provide an ambitious volume of financial support to climate change mitigation activities beyond the value chain, without claiming to neutralise the company's own emissions.	Low 🔺	Very low —
Offsetting claims today		Clearly disclose offsetting claims and plans; avoid misleading pledges and claims; avoid risk of distraction by also committing to measures for deep emission	Low —	Very low —
Offsetting plans for the future	~	reductions; commit to procure only high-quality credits from ambitious projects with a permanent climate impact; and commit to preventing any form of double-counting of climate impacts.	Low —	Very low —

⁺ Transparency and integrity columns: the bar indicates the distribution of our rating of the 24 companies (**Poor Moderate High n.a.**); the text above the shaded bars represents the average rating across all the companies we assessed, calculated excluding non-applicable cases, on a 5-point scale (**Very low, Low, Moderate, Reasonable, High**); and an indication of progress since the last analysis in 2022 ($\Delta - \nabla$), based on the authors' interpretation of progress from the companies that were analysed also in 2022, against the current methodology version. Good practices were derived from the principles elaborated in the following subsections, and from a compilation of the practices identified from existing company strategies. Full details on the assessment methodology can be found in the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0* (NewClimate Institute, 2023b).

Tracking and disclosure of emissions

To develop a comprehensive and robust climate strategy, it is key that companies understand and are transparent about their GHG emission footprints and their trends over time. A complete and transparent overview of a company's emissions footprint is crucial to understand a company's scope of influence, to grasp the relevance of its climaterelated targets, and to determine whether emission reduction measures are appropriate and comprehensive.

This section assesses the comprehensiveness of companies' GHG emission tracking and disclosure for specific emission scopes and for subsidiary companies. This report does not assess the rigorousness and accuracy of companies' calculations when quantifying emissions from each emissions scope; quantified GHG emissions throughout this document are self-reported by the companies and not verified by the authors. Rather, we assess how comprehensive the companies' own disclosure is in terms of the coverage of emission sources.

Table 2 presents a summary overview of principles for good practice (section 1.2) as well as a summary trends, promising examples and bad practice identified from the company assessments (section 1.3).



1.1 Snapshot summary

Table 2: Summary of good practice and trends for GHG emission tracking and disclosure

SUMMARY OF GOOD PRACTICE AND TRENDS FOR GHG EMISSION TRACKING AND DISCLOSURE					
GOOD PRACTICE FOR EMISSION DISCLOSURE		ge performance* ENCY & INTEGRITY*			
 Disclose full details on their GHG emissions on an annual basis in public documents. 	Scope 1	Reasonable —			
 Include a breakdown of the data to specific emission sources (including scope 1, 2, 3 and non-GHG climate forcers) 	Scope 2	Moderate			
 Present of historical data for each emission source. Ensure consistency of emission disclosure across documents. 	Scope 3 upstream	Moderate —			
	Scope 3 downstream	Moderate —			
(See section 1.2.2 for further detail)	Inclusion of subsidiaries	High —			

* Transparency and integrity columns: the bar indicates the distribution of our rating of the 24 companies (**Poor Moderate High n.a.**); the text above the shaded bars represents the average rating across all the companies we assessed, calculated excluding non-applicable cases, on a 5-point scale (**Very low, Low, Moderate, Reasonable, High**); and an indication of progress since the last analysis in 2022 (**M V**), based on the authors' interpretation of progress from the companies that were analysed also in 2022, against the current methodology version.

SUMMARY OF TRENDS, ROLE MODELS AND BAD PRACTICE (SECTION 1.3)

Moderate transparency in emission disclosure: Companies' emission disclosure in 2022 indicates a solidifying consensus and level of understanding on good practice. Most companies disclose emissions with at least a moderate degree of transparency.

Disclosure of indirect (scope 3) emissions and emissions from subsidiaries: Reflecting a consensus across existing and new guidance, it is now established common practice for companies to take responsibility for scope 3 emissions, and emissions from subsidiaries.

Companies exhibiting bad practice for emission disclosure are in the minority, but misleading scope 3 emission disclosure remains a critical issue.

PROMISING EXAMPLES

Emissions breakdown: Apple, Microsoft, Holcim, Maersk and Inditex provide a breakdown of emissions for each emissions scope, which allows for an understanding of emission sources. Microsoft provides a transparent breakdown of scope 3 emissions into the 15 categories of the GHG Protocol and shows whenever a category is not relevant.

Scope 2 emissions reporting: Holcim, Foxconn, Fast Retailing and Mercedes-Benz use the highest estimate for scope 2 emissions as part of their emissions aggregates, providing incentives for improvements in both renewable energy procurement and energy efficiency.

BAD PRACTICE

Inconsistent reporting: Some companies misuse CDP certifications to claim the legitimacy of inconsistent and incomplete public disclosure.

Hidden scope exclusions: Carrefour use footnoted scope clarifications to exclude the majority of their emissions from their public emissions disclosure.

SUPPORTING COMPANIES TO IMPROVE THEIR EMISSION DISCLOSURE

Sector-specific reporting boundaries: There remains a lack of guidance on the appropriate boundaries of downstream scope 3 emission reporting for some sectors, including the steelmaking sector.

Indirect product use phase emissions: There is no clear consensus on the extent to which it is constructive and appropriate for companies to disclose indirect product use phase emissions (e.g. emissions from operating washing machines as downstream emissions for clothing), which are labelled as optional by the GHG Protocol. In some cases 'over-reporting' emissions can undermine targets (see section 1.2.1).

Table 3: Summary of developments in 2022 related to disclosure of GHG emissions

DEVELOPMENTS IN 2022 FOR CORPORATE EMISSION DISCLOSURE PROGRESS IN COMPANIES' CLIMATE STRATEGIES (INCLUDING THE 10 COMPANIES ASSESSED IN BOTH THE 2022 AND 2023 CCRM) ALL (AVERAGE) Amazon \square Apple Carrefour Deutsche Post DHL Google \bigcirc JBS Maersk The overall transparency of companies' disclosure remains moderate; some companies improved aspects of their Nestlé disclosure but the minority of companies with the lowest Volkswagen performance on emissions disclosure did not make significant improvements to address identified issues. Walmart \cap

+ Authors' interpretation of companies' progress against the current methodology version since the last analysis in 2022 (

NEW GUIDELINES AND FRAMEWORKS RELATED TO CORPORATE GHG EMISSION DISCLOSURE IN 2022			
ISO Net Zero guidelines (ISO, 2022b)	Scope 1, 2, and 3 emissions should be reported in full detail, including a breakdown of the emission scopes by GHG or activity, including and differentiating between all specific scope 3 categories (ISO, 2022b, p. 30).		
	Focus on the higher scope 2 emission accounting method: Companies should use both the location- and market-based methods to derive a scope 2 emissions estimate, but use the highest of the two for aggregate emission reporting and target setting (ISO, 2022b, p. 18), in order to adequately prioritise energy efficiency as well as renewable energy procurement.		
UN High-Level Expert Group (UN HLEG, 2022) Race to Zero Criteria 3.0 (Race to Zero, 2022)	Full inclusion of all emission sources and business activities in climate strategies: The HLEG recommendations and the Race to Zero criteria are both specific to target setting rather than emission disclosure (see section 2). Nevertheless, the HLEG recommendation and Race to Zero criteria both state that net-zero targets must include all emission scopes, including all company divisions and subsidiaries, is a clear signal that companies should be reporting emissions in this way as well.		
Other developments	Regulation on emission disclosure: The European Parliament and Council approved a <i>Corporate</i> <i>Sustainability Reporting Directive</i> in November 2022 (Directive 2022/2464 (EU, 2022)), which requires companies to report emissions related both to their own operations and their supply chain, and to prioritise disclosing emission estimates for significant scope 3 emission categories. In the United States, the <i>Climate Risk Disclosure Act</i> , which would require all issuers of securities to disclose direct and indirect GHG emissions, has stalled in Congress (Congress.gov, 2021). Nevertheless, since November 2022 the US Federal Government requires its major suppliers and contractors to publicly disclose scope 1, 2 and 3 emissions (The White House, 2022).		

.

.

....

.

.

Note: The summaries in this table represent the authors' interpretations of the guidelines and developments presented.

20

1.2 Principles for good practice

This section includes a summary of guiding principles and assessment criteria for tracking and disclosure of emissions from section 1 of the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0.*

1.2.1 Tracking and disclosure of emissions

Companies should annually disclose information on the GHG emissions from their full value chain and provide details that allow for a complete understanding of key emission sources and trends. Meaningful planning for complete decarbonisation depends on a thorough and granular understanding of a company's emission sources. Complete and transparent disclosure covers all direct emissions (scope 1), indirect energy-use emissions (scope 2), and other upstream and downstream indirect emissions (scope 3). Companies should report these emission scopes separately and break them down into GHG, activity or emissions source, while providing historical data. Where relevant, companies should also include non-GHG climate forcers in their disclosure. Companies' disclosure should be accompanied by the methodologies used to derive the emission estimates, especially for emission sources where there remains uncertainty in estimation approaches.

Companies should report scope 2 emissions using both the location-based and market-based method, taking the highest of the two values for their calculation of their total emissions footprint. According to the GHG Protocol and the ISO's Net Zero Guidelines, companies should report on scope 2 emissions using both the location-based and market-based accounting methods (GHG Protocol, 2015; ISO, 2022b, p. 18). The location-based method reflects the average emissions intensity of grids on which energy consumption occurs. The market-based method reflects emissions from electricity procurement constructs that companies have purposefully chosen, deriving emission factors from contractual renewable electricity procurement instruments. Both accounting approaches have the potential to misrepresent the emission footprint of electricity consumption in different circumstances. To create a clear incentive to both maximise energy efficiency improvements and to procure renewable electricity, companies should report on both market-based and location-based scope 2 emissions, and use the larger of the two values towards their aggregated total emission estimates (ISO, 2022b, p. 18).

Companies should report on all upstream and downstream indirect emissions, including emission sources deemed minor or irrelevant. The ISO's Net Zero Guidelines require companies to provide separate data for the different scope 3 categories (ISO, 2022b, p. 30), such as emissions from procured products and services, investments, waste, upstream and downstream transport and distribution, and emissions from product use. The GHG Protocol's Scope 3 Standard identifies 15 distinct reporting categories for scope 3 emission sources and requires companies to quantify and report scope 3 emissions from each (WRI and WBCSD, 2013). For transparency, companies should disclose data or at least explanatory information for all 15 of these mandatory scope 3 emission categories, even those deemed minor or irrelevant. Different interpretations of what constitutes a 'minor' or 'irrelevant' emission source could lead to inconsistencies between companies' reporting. Reporting on scope 3 emissions outside of the 15 mandatory categories, such as indirect use-phase emissions and emissions from products that are not sold to end users, is in some cases crucial to represent a company's complete climate impact, while in other cases it may distract from a company's mandatory emissions scope and make its targets disingenuous (see section 1.1 of the accompanying methodology document, Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0).

Companies' disclosure should include the emissions associated with subsidiary companies. Companies may depend on emission-intensive assets and infrastructure that are held in subsidiary companies. Transparent and complete reporting includes these emissions, which should be integrated into the company's scope 1, 2 and 3 emissions. If companies report transparently on the emissions of all subsidiaries, they have a stronger incentive to make a real shift away from emission-intensive activities and assets, rather than continuing those activities through subsidiaries.

1.2.2 Good practice assessment criteria

The criteria for good practice in Table 4 forms the basis for the company assessments in section B. Full details on the methodology for rating companies' tracking and disclosure can be found in the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0, section 1.*

Table 4: Good practice for tracking and disclosing emissions.

DISCLOSING EMISSIONS CORPORATE CLIMATE LEADERS EXHIBITING GOOD PRACTICE..

22

Provide in their public-facing documents:

- An annual disclosure of their full value chain emissions (scopes 1, 2, and 3), including emissions from subsidiaries.
- A breakdown of data for each scope to specific emission sources, including for scope 3 the 15 mandatory categories from the GHG protocol.
- For scope 3, emission estimates for other 'optional' sources, such as indirect use-phase emissions, only if they are crucial to represent a company's complete climate impact.
- Historical data for the same emission sources.
- Explanations on why omitted emission sources are not tracked.
- A disclosure of non-GHG climate forcers, if relevant.
- For scope 2, both market- and location-based emission estimates, while using the highest estimate for emission aggregates.

1.3 Trends, role models, and bad practice

Companies' emission disclosure in 2022 indicates a solidifying consensus and level of understanding on good practice. Of the 24 companies we assessed in this report, we found that 16 disclosed emissions data with at least a moderate level of transparency and integrity, for all emission scopes. These trends indicate that it is now common practice for the companies we assessed to acknowledge full responsibility for upstream and downstream scope 3 emissions, and emissions from subsidiaries. This finding reflects an increasing level of alignment and clarity on these issues among existing and newly published guidelines.

Companies exhibiting bad practice for emission disclosure are in the minority, but these bad practices remain significant. A minority of companies still publish highly misleading emissions data, usually by excluding major emission sources from their public documentation. Of the companies that we assessed in the previous iteration of the *Corporate Climate Responsibility Monitor*, we did not identify any significant improvements from those with poor emission disclosure practices.

Companies' emission disclosure in 2022 indicates a solidifying consensus and level of understanding on good practice.

Most of the 24 companies disclose emissions data with at least a moderate level of transparency and integrity, for all emission scopes. Although we identified room for improvement in the emission disclosure practices of all companies, we find that the disclosure of most companies is of at least moderate transparency, and we see signs of further improvement. Three companies, **Apple**, **Maersk** and **Microsoft**, were found to report and disclose with high transparency and integrity for at least three of the four emission scopes we assessed (scope 1, 2, upstream scope 3 and downstream scope 3).

It is now established common practice for companies to take full responsibility for scope 3 emissions, and emissions from subsidiaries. The majority of companies report on scope 3 emissions with at least a moderate level of transparency and integrity, demonstrating that it is now common practice for the companies that we assessed to do so. Likewise, almost all companies report on their subsidiaries' emissions and make the inclusion explicit. These positive findings reflect the improving degree of consensus across guidelines and standards on the issue of scope 3 emissions. In addition to the existing guidelines of initiatives like the Race to Zero campaign, the Science Based Targets Initiative and the GHG Protocol, the requirement for companies to include full scope 3 emissions and subsidiaries in their climate strategies was underlined by the new ISO Net Zero Guidelines (ISO, 2022b), the recommendations of the high-level expert group (UN HLEG, 2022), the European Corporate Sustainability Reporting Directive (Directive 2022/2464 (EU, 2022)), and the procurement conditions of the U.S. Federal Government (The White House, 2022). Despite these developments and positive trends, poor scope 3 disclosure remains a critical issue for a minority of companies (see below).

There are signs of improvement in the disclosure of emissions associated with procured energy (scope 2). Ten out of the 24 companies studied provided a breakdown of scope 2 emissions. Most companies assessed (14 out of 24) report both the location- and the market-based scope 2 emission estimates (see section 1.2.1), although some do not clarify the method used. Of the companies that report both estimates, four - American Airlines, Fast Retailing, Foxconn and Holcim - used the highest estimate towards their emission aggregates. Although it remains only a minority of companies that pursue this most transparent approach, this handful of positive examples, alongside the release of the ISO guidelines, may provide an encouraging sign and inspiration for other companies to replicate this transparent practice, for an emission scope that has been rather untransparent in the past.



Companies exhibiting bad practice for emission disclosure are in the minority, but misleading scope 3 emission disclosure remains a critical issue.

A minority of companies are still not transparent about their scope 3 emissions. We found that 8 of the 24 companies disclose emissions with a low, or very low, level of transparency and integrity. The most critical issues identified from these companies often relate to the untransparent disclosure of indirect upstream and downstream scope 3 emissions, which account for over 90% of most companies' emission footprints. **Carrefour**, **Foxconn**, **Samsung**, **Stellantis** and **Walmart** do not clearly disclose most of their scope 3 emissions in their public-facing sustainability reports, giving the impression that their emission footprint is just a small fraction of the reality, and depriving consumers and investors of key information on the emissions profile associated with specific products and business models.

Some companies misuse CDP certifications to claim the legitimacy of inconsistent and incomplete disclosure. We found that some companies report more comprehensive data in non-public CDP responses, while presenting a much smaller scope of emissions data in public-facing sustainability reports, alongside the statement that the transparency of disclosure has been verified by CDP. This highly misleading practice calls into question the value of climate transparency certifications for non-public information. It may be more effective for assessments and certifications of transparency to be restricted to public information contained in or attached to companies' annual reports.

Guidance for scope 3 disclosure remains lacking in some sectors, including steelmaking. The disclosure of downstream scope 3 emissions from **ArcelorMittal** and **Thyssenkrupp** appears inconsistent and untransparent. ArcelorMittal does not provide a clear breakdown of scope 3 emissions upstream and downstream; Thyssenkrupp reported no scope 3 emissions from the use phase of sold products in 2021, compared to around 780 MtCO₂e in 2017. These issues appear to be owing partly to the poor availability of guidance on the appropriate boundaries of downstream scope 3 emissions for steel companies. There remains no clear consensus on the extent to which steelmaking companies should be required to take responsibility for indirect product use phase emissions.



2 Setting specific and substantiated targets

Corporate climate targets must provide a clear signal for immediate action to reduce emissions across the value chain paired with a longer-term vision for deep decarbonisation. Corporates ought to set both shortand medium-term climate targets towards 2030 *and* long-term climate targets beyond 2030 in line with sector-specific 1.5°C Paris Agreement-aligned emission pathways.

This section assesses the transparency and integrity of corporate climate targets towards 2030 and beyond. Table 5 presents a summary overview of principles for good practice (section 2.2) as well as a summary trends, promising examples and bad practice identified from the company assessments (section 2.3).

2.1 Snapshot summary

Table 5: Summary of good practice and trends for corporate target setting

SUMMARY OF GOOD PRACTICE PRINCIPLES (SECTION 2.2) AND PERFORMANCE FROM THE 24 COMPANIES ASSESSED

GOOD PRACTICE FOR TARGET SETTING		AVERAGE PERFORMANCE*	
		TRANSPARENCY⁺	INTEGRITY ⁺
Short- &	Companies should set both		
medium-term targets towards 2030	 short- and medium-term emission reduction targets towards 2030 within 5-year intervals, 	Moderate 🔺	Very low —
	 and specific long-term emission reduction targets beyond 2030 as a long-term vision for deep decarbonisation. 	-	
Long-term targets	All short-, medium-, and long-term targets should		
beyond 2030	\checkmark be independent from offsetting claims,	pendent from offsetting claims,	
	 align with 1.5°C-compatible trajectories and benchmarks for the sector, and 		Very low —
	✓ cover all scope 1, 2 & 3 emissions & non-GHG climate forcers, where relevant.		

* Transparency and integrity columns: the bar indicates the distribution of our rating of the 24 companies (**Poor Moderate High n.a.**); the text above the shaded bars represents the average rating across all the companies we assessed, calculated excluding non-applicable cases, on a 5-point scale (**Very low, Low, Moderate, Reasonable, High**); and an indication of progress since the last analysis in 2022 (**— V**), based on the authors' interpretation of progress from the companies that were analysed also in 2022, against the current methodology version.

SUMMARY OF TRENDS, ROLE MODELS AND BAD PRACTICE (SECTION 2.3)

2030 climate targets fall well short of the required ambition: The emission reduction commitment across all value chain emissions is estimated to be 15-21% between 2019 and 2030. This compares to the need to cut global GHG and CO₂ emissions by 43% and 48% between 2019 and 2030 respectively, to stand a reasonable chance of limiting global warming to 1.5° C.

Only a minority of net-zero pledges represent credible commitments to deep decarbonisation: Only half of the companies' net-zero or carbon neutrality pledges commit to an explicit emission reduction target alongside these pledges. Collectively, these pledges translate to a commitment to reduce only 36% of the companies' combined GHG emission footprint by the respective net-zero target years.

Third-party certifications lend credibility to highly insufficient targets: SBTi has certified most of the assessed companies' 2030 climate targets ('1.5°C', 'well-below 2°C', '2°C') despite most companies falling way short of the required ambition across the entire value chain according to latest science.

PROMISING EXAMPLES

Deep decarbonisation commitments: Holcim, H&M Group, Maersk, Stellantis, and ThyssenKrupp explicitly commit to reduce emissions by at least around 90% across the entire value chain next to their net-zero pledges.

Meeting 1.5°C-aligned decarbonisation milestones for 2030: Apple, H&M Group, and Maersk meet 1.5°C-aligned decarbonisation milestones for their respective sectors, while ArcelorMittal, Foxconn, Holcim, Stellantis, and ThyssenKrupp come close to meeting them.

Voluntary compliance to net-zero guidelines: Holcim's and H&M Group's net-zero targets have been verified under the SBTi's Net Zero Standard; we find them to be of moderate and high integrity, respectively. Ahold Delhaize and Maersk have committed to the Net Zero Standard (without yet being verified by SBTi); we find these pledges to be of moderate and high integrity, respectively.

BAD PRACTICE

Unsubstantiated net-zero pledges: Half of the 24 companies make no specific emission reduction commitment for the net-zero target year. Five other companies commit to less than 40% emission reductions across the entire value chain. **Carrefour** and **Walmart** commit to only 1% and 9% emission reductions respectively by 2050 below 2019 levels, due to scope exclusions.

Insufficient 2030 targets: Fifteen out of 24 companies clearly fall short of meeting 1.5°C-aligned decarbonisation milestones for their respective sectors despite being verified by the SBTi in most cases. **Mercedes-Benz** and **Volkswagen**, for example, both do not set 1.5°C-aligned phase-out dates for internal combustion engines in line with latest scientific findings.

SUPPORTING COMPANIES TO IMPROVE THEIR CORPORATE TARGET SETTING

A renewed focus on ratcheting-up 2030 targets through improved guidance and mandatory regulation that cover the full value chain, align with the 1.5°C temperature limit, and are independent of offsets is of most critical importance.

A more transparent presentation and summary of the increasing body of literature defining sector-specific 1.5°C-aligned decarbonisation milestones can support companies, regulators and the wider public to set and evaluate corporate climate targets.

Table 6: Summary of developments in 2022 related to corporate target setting



+ Authors' interpretation of companies' progress against the current methodology version since the last analysis in 2022 (🛕 — 🛡).

NEW GUID	ELINES AND FRAMEWORKS RELATED TO CORPORATE TARGET SETTING IN 2022
UN High-Level Expert Group (UN HLEG, 2022)	The emission coverage of emission reduction targets should include all emissions along a company's entire value chain (scope 1, 2 and 3) and transparently explain any missing emission sources.
	Companies should set 1.5-compatible absolute emission reduction targets for the short-, medium- and long-term informed by modelled 1.5°C emission pathways with no or limited overshoot, and where global emissions decline at least 50% below 2020 levels by 2030, reaching net-zero by 2050 or sooner.
	A target should be set for 2025 with subsequent targets following in five-year intervals for 2030 and 2035.
ISO Net Zero guidelines (ISO, 2022b)	The emission coverage of emission reduction targets should include all emissions along the value chain (scope 1, 2 and 3), and set emission reduction targets for each emission scope separately using the same base year.
	Companies should set 1.5-compatible absolute emission reduction targets informed by sector-specific modelled emission pathways with a high likelihood of limiting warming to 1.5°C (e.g., 2021 IEA <i>Net Zero by 2050</i> report).
	Companies should restrict the use of carbon dioxide removal offsets—both inside and outside the value chain—to only counterbalance residual emissions towards net-zero claims. The guidance shows that its definition of <i>residual</i> emissions likely accounts for 0-5% of emission levels today for most sectors.
	The use of a net-zero terminology for companies operating in sectors with significant technological challenges might lead to false claims whereas companies operating in such sectors should rather set <i>achievable</i> targets.
	Corporates should set interim emission reduction targets within two-year to five-year intervals.
Other developments	An increasing body of literature defines sector-specific 1.5°C-aligned decarbonisation milestones. Over the course of 2022, the Science-Based Targets Initiative has published new guidance for the cement, maritime, and FLAG sectors (SBTi, 2022a, 2022b, 2022e), the Transition Pathways Initiative has published new and updated guidance for several sectors including food producers and cement (Dietz, Gardiner, <i>et al.</i> , 2022; Dietz, Harvey, <i>et al.</i> , 2022), and both Teske (2022) and Boehm <i>et al.</i> (2022) provide 1.5°C-aligned decarbonisation milestones for multiple sectors. This literature facilitates an improved understanding of 1.5°C-aligned sectoral climate action by corporates.
	Companies increasingly face risks of litigation over the sufficiency of their climate action. The appeal procedure against the Hague District Court's landmark ruling for Shell's emission reduction targets is ongoing, with a final verdict expected in 2024 (Setzer and Higham, 2022). Oil and gas companies remain the most frequent defendants in climate litigation cases against corporates (Higham and Kerry, 2022; Setzer and Higham, 2022). German NGOs brought forward court cases against German automobile manufacturers Mercedes-Benz, BMW, and Volkswagen to mandate them to phase out internal combustion engines by 2030 latest in line with latest available science (Setzer and Higham, 2022).

Note: The summaries in this table represent the authors' interpretations of the guidelines and developments presented.

2.2 Principles for good practice

2.2.1 Corporate target setting for the short-, medium-, and long-term

Corporate climate targets must provide a clear signal for immediate action to reduce emissions along the entire value chain paired with a longer-term vision for deep decarbonisation. For this reason, corporates should set both short- and medium-term climate targets towards 2030 and long-term climate targets beyond 2030.

Credible short- and medium-term targets towards 2030 ensure that corporate emissions decrease in line with what limiting global temperature increase to 1.5° C requires by 2030. The pathway to net-zero is crucial: a 1.5° C limit requires immediate action to achieve a reduction in global GHG and CO₂ emissions of about 43% and 48% respectively from 2019 levels by 2030 (IPCC, 2022). Further delay in emission reductions put the Paris Agreement objectives beyond reach. Well-defined short- and medium-term targets set within five-year intervals ensure such immediate action and provide accountability. The HLEG recommendations and ISO Net Zero Guidelines

2.2.2 Coverage of emission sources in targets

Short-, medium- and long-term targets should be explicit in their coverage of the complete spectrum of emission sources and greenhouse gases, to maximise impact and avoid misleading communication. The most comprehensive targets cover the full GHG emission footprint of a company across its entire value chain, including upstream and downstream scope 3 emissions, and non-GHG climate forcers where relevant. When setting multiple short- and medium-term targets, for example targeting specific emission scopes, the company ought to transparently explain how much of its emissions across the entire value chain these targets cover.

Companies should explicitly set out the coverage of their headline climate pledges to avoid misinterpretation and to ensure accountability. For net-zero targets, as well as for short- and medium-term targets, the United Nations' High-Level Expert Group (HLEG) recommendations and ISO Net Zero Guidelines both mandate the coverage of all emission scopes (ISO, 2022b, pp. 16–19; UN HLEG, 2022, pp. 17–18). Targets with partial scope coverage have the potential to mislead: disclaimers are easily overlooked or may not be well understood by the audiences of climate pledge communications. mandate companies to set short-term targets within the next two to five years and align their medium-term 2030 targets with the latest science using 1.5°C-aligned pathways with no or limited overshoot (ISO, 2022b, p. 11; UN HLEG, 2022, p. 17).

Long-term targets beyond 2030 set out a vision towards deep decarbonisation. Such targets must provide a clear indication of what the company aims to achieve in the long-term, to inform today's management and investment decisions. Limiting global temperature increase to 1.5° C requires the rapid decarbonisation of all sectors to reach a state of global net-zero CO₂ emissions by around 2050, net-zero GHG emissions by around 2070, and net-negative emissions thereafter (IPCC, 2022). Long-term targets beyond 2030 ought to be presented in five-year intervals over time and align with sector-specific emission pathways towards full decarbonisation.

Coverage of all mandatory scope 3 emission categories is highly relevant, despite uncertainties and indirect influence. Scope 3 emissions can entail a degree of uncertainty, particularly for complex emission sources related to land-use such as upstream food processing, and downstream emissions associated with consumer behaviour and product use. The decarbonisation of these emissions may also depend partially on actions taken by others. Despite these uncertainties, the inclusion of all mandatory scope 3 emission sources from the GHG Protocol's Scope 3 Standard in companies' targets is crucial (WRI and WBCSD, 2013). This provides a clear incentive for all actors with a potential influence on the decarbonisation of emission sources to take measures to do so. For manufacturers of cars, electric appliances, or electronic devices, scope 3 emissions often account for the major share of those companies' emissions, and the companies are the actors with the greatest influence to decarbonise those emission sources, by manufacturing products with alternative or more efficient technologies. Even in the cases where companies have a lower degree of influence in the reduction of scope 3 emissions, this does not justify their exclusion from targets; the full inclusion of scope 3 emissions in targets can incentivise companies to cooperate with suppliers and consumers to mutually support each other to reduce emissions, including to seek out new solutions where needed. Targets that omit scope 3 emissions carry a significant potential to mislead, since scope 3 emissions account for a large portion of most companies' climate impact.

2.2.3 Emission reduction commitments

Headline pledges only send a meaningful signal for decarbonisation if they explicitly include deep emission reduction commitments that are independent of offsetting and carbon dioxide removals. Some companies' headline pledges are directly specified in the form of emission reduction targets, some are *accompanied* by such targets, while others do not specify any emission reduction targets at all. Headline pledges only contribute to the Paris Agreement objectives in a meaningful way if they put emission reductions across the entire value chain in the spotlight. Such pledges are also more constructive if they avoid ambiguous terminology that can distract from this focus, for example by remaining unspecific on emissions reductions to be achieved without relying on offsets or carbon dioxide removal. Both the SBTi's Net Zero Standard and the ISO Net Zero Guidelines require companies with net-zero targets from any sector—except the forestry, land-use, and agriculture sectors—to explicitly commit to emission reductions of at least 90% below 2019 levels across all emission scopes (SBTi, 2021c; ISO, 2022b, pp. 16–17). The commitment to such deep emission reductions ensures that the net-zero terminology is not misleading, regardless of the target year, but it is not alone a measurement of sufficiency in terms of 1.5°C compatibility.

Short, medium, and long-term targets must be ambitious enough to align with 1.5°C compatible emission pathways with no or limited overshoot. To stand a reasonable chance of limiting global warming to 1.5°C, global GHG and CO₂ emissions must decrease by around 43% and 48% respectively between 2019 and 2030 under modelled pathways with no or limited overshoot (IPCC, 2022). Global CO₂ and GHG emissions must further reach net-zero by around 2050 and around 2070 respectively. Both the HLEG recommendations and ISO Net Zero Guidelines emphasise the need for companies to align their targets with those findings (ISO, 2022b, pp. 19–20; UN HLEG, 2022, p. 17).

Sector-specific decarbonisation milestones compatible with the Paris Agreement's 1.5°C temperature limit provide relevant benchmarks to inform and evaluate corporate target setting. These benchmarks informed by latest science help to determine ambition levels of corporate climate action across different sectors in line with global efforts. For automobile manufacturers, for example, several studies identify 1.5°C-aligned decarbonisation milestones for the phase out of internal combustion engines (ICEs) replaced by electric and low-emission vehicles at the global and regional level (CAT, 2020a; IEA, 2021; UNFCCC, 2021; Boehm *et al.*, 2022; Teske *et al.*, 2022; WBA, 2022). Table 3 in the report's accompanying methodology provides an overview of sector-specific decarbonisation benchmarks available in existing literature as of February 2022 for all eight sectors covered in the report.

2.2.4 Good practice assessment criteria

The criteria for good practice in Table 7 form the basis for the company assessments in section B. Full details on the methodology for rating companies' corporate target setting can be found in the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0, section 2.*

Table 7: Good practice for setting specific and substantiated targets

SETTING SPECIFIC AND SUBSTANTIATED TARGETS CORPORATE CLIMATE LEADERS EXHIBITING GOOD PRACTICE			
A. Short- & medium-term targets towards 2030	 Set short- and medium-term emission reduction targets towards 2030 within five-year intervals that reflect a commitment to immediate action and accountability. Set all short- and medium-term targets towards 2030 independent from direct offsetting or offsetting under the guise of 'insetting' (sometimes also referred to as within value chain removals). Align their short- and medium-term targets with sector-specific 1.5°C-compatible decarbonisation milestones in the literature. Cover all emission scopes along the entire value chain and —if setting multiple short- and medium-term targets cover. 		
B. Long-term targets beyond 2030	 Set specific long-term emission reduction targets beyond 2030 to provide a vision for deep decarbonisation towards mid-century with five-year intervals. Set all long-term targets independent from offsetting or offsetting under the guise of 'insetting' (sometimes also referred to as within value chain removals). Align their long-term targets with sector-specific 1.5°C-compatible decarbonisation milestones in the literature. Cover all emission scopes along the entire value chain. 		

2.3 Trends, role models, and bad practice

Summary of rends in 2022 Several new net-zero guidelines and target setting methodologies have been released for corporates over the course of 2022. These recent developments point to the emerging consensus on the need for robust, credible, and transparent emissions reduction targets in the short-, medium-, and long-term. Despite these recent developments, we find that the 24 companies' 2030 targets jointly fall well short of the required ambition to cut global GHG and CO_2 emissions by 43% and 48% between 2019 and 2030 respectively, to stand a reasonable chance of limiting global warming to 1.5°C. Companies' 2030 targets translate to a median commitment of only 15–21% emission reductions across their full value chain emissions between 2019 and 2030. In this context, SBTi certifications for short- and medium-term targets towards 2030 lend credibility to companies whose targets are highly insufficient. Net-zero targets of most of the 24 companies remain highly ambiguous. Only a small minority of companies follow recent recommendations and commit to at least around 90% emission reduction by their respective net-zero target years and meet 1.5°C-aligned decarbonisation milestones for their respective sectors. None of the ten companies analysed in the previous iteration of the *Corporate Climate Responsibility Monitor* improved on the integrity of their target setting over the course of 2022.

Only a minority of net-zero pledges represent credible commitments to deep decarbonisation, while most others remain highly ambiguous due to the lack of credible emission reduction targets.

Just five of 24 companies' net-zero pledges represent a commitment to deep decarbonisation. Only a small minority of companies-covering a diverse range of sectors-comply with latest recommendations by standard setters to substantiate corporate net-zero targets with deep decarbonisation targets aligned with sector-specific 1.5°C milestones (SBTi, 2021c; ISO, 2022b; UN HLEG, 2022). Holcim, H&M Group, Maersk, Stellantis and Thyssenkrupp commit to reducing their emissions by at least around 90% by their respective net-zero target years (Figure 1). Among these five companies, we evaluate H&M Group's and Maersk's net-zero targets of high integrity, as these targets fully align with available 1.5°C-aligned decarbonisation milestones in the literature. Stellantis only meets some of the 1.5°C-aligned milestones for automobile manufacturers; Thyssenkrupp likely aligns with 1.5°C-compatible milestones for steelmaking but uncertainty remains regarding the coverage and relevance of downstream scope 3 emissions in its climate strategy; while Holcim's net-zero target hinges on intensity targets and the extensive use of CCUS, which makes the appropriateness of the net-zero terminology contentious. For this reason, we evaluate the latter companies' net-zero targets to have moderate integrity.

Overall, we evaluate the large majority of net-zero targets to be of poor integrity due to the inadequacy or complete lack of explicit emission reduction commitments. Despite all 24 companies committing to net zero or carbon neutrality, only half of them set an explicit emission reduction target next to these pledges. For the other half, net-zero pledges represent ambiguous commitments, as they do not explicitly specify the extent to which companies intend to reduce emissions by the respective target years. Five out of 24 companies only set emission reduction targets of less than 40% across the value chain; none of these comply with sector-specific 1.5°C milestones. Three companies (Apple, Google, Microsoft) commit to net-zero targets by 2030 already, but these entail only about 63%, 37% and 38% emission reductions across the entire value chain below 2019 emission levels, respectively. Those three companies do not commit to any further emission reductions beyond 2030.

Figure 1: Just 5 of 24 companies commit to deep decarbonisation with their net-zero pledges

This chart shows the **proportion of full value chain GHG emissions that companies commit to reduce with their net-zero pledges.** Data includes 12 companies. For 12 other companies the meaning of the net-zero target is ambiguous.



compared to sector-specific 1.5 °C aligned benchmarks, and the appropriateness of the terminology used in the pledge communication. Data presented on a single x-axis. Placement on y-axis has no significance.

The data points in this graphic represent the authors' most optimistic interpretations of companies' emission reduction commitments, including emission intensity targets, based on publicly available information. The chart includes emission reduction commitments under net-zero targets, carbon neutrality pledges and other pledges with equivalent terminology, for the respective target year, which ranges between 2030 and 2050. Targets that are reliant on offsets to an undefined extent are marked as ambiguous. Further details and explanation on the authors' interpretation of companies' targets can be found in the individual company cases in Section B and in Annex II.

Collectively the net-zero pledges of the 24 companies translate to a commitment to reduce just 36% of the companies' combined GHG emission footprint at a maximum, by the respective net-zero target years. Net-zero targets thus on average fall way short of a commitment to deep decarbonisation towards near-zero emissions, as suggested by the term 'net-zero' (Figure 2).

Half of the net-zero targets cover emissions across all scopes, but 2030 targets often do not align with this emission source coverage. Eleven companies explicitly cover 100% of their emissions footprint in their net-zero targets. While three companies (Carrefour, Samsung, Walmart) exclude more than 80% of their emissions across the value chain from their net-zero targets, eight companies do not disclose their emission coverage at all. Limited emissions coverage is often not presented transparently and has significant potential to mislead. Companies active in the aviation sector (American Airlines and Deutsche Post DHL) focus on reporting CO₂ emissions but miss to transparently cover all non-GHG climate forcers, the impacts of which are larger than the CO₂ impact for aviation (Lee *et al.*, 2021). Short- and medium-term targets towards 2030 often explicitly leave out certain emission scopes covered by net-zero targets. Only five companies (Ahold Delhaize, Foxconn, H&M Group, PepsiCo, and Stellantis) cover 100% of their emissions both under their 2030 emission reduction targets and their net-zero targets.



Figure 2: Net-zero pledges break down to only moderate emission reductions



The 24 companies assessed in this report are not necessarily a representative sample of all corporate actors with net-zero targets. They represent 24 of the largest companies in the world, accounting for approximately 4% of global GHG emissions and revenues of USD 3.16 trillion in 2021. We anticipate that any overlap in the scope 1, scope 2 and scope 3 emissions of these companies is marginal and of limited significance to the key insights derived from this report.

Inadequate and ambiguous net-zero target setting distracts from the most important issue at hand: immediate and economy-wide emission reductions towards 2030. Companies' climate pledges for 2030 fall well short of the required ambition according to latest science.

Nearly all the 24 companies that we assessed have pledged 2030 targets, but we find that these targets can rarely be taken at face value. For many companies, 2030 targets address only a limited scope of emission sources, such as scope 1 and 2 emissions, or selected scope 3 emission categories. For others, 2030 targets are misleading due to reliance on direct offsetting or offsetting under the guise of 'insetting'.

Climate pledges for 2030 fall well short of the economywide emission reductions required to stay below the **1.5°C temperature limit.** For the 22 companies with targets for 2030, we find that these targets translate to a median absolute emission reduction commitment of just 15% of the of full value chain emissions between 2019 and 2030, or 21% under the most optimistic scenario that emission intensity targets translate to equivalent absolute emission reductions (Figure 3). This compares to the IPCC Sixth Assessment Report's findings that global CO₂ and GHG emissions must decrease by 48% and 43% between 2019 and 2030 respectively to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022). Emission reduction targets for the crucial decade towards 2030 thus fall way short of the ambition required to align with the internationally agreed goals of the Paris Agreement, as emphasised by recent recommendations by the UN High-Level Expert Group and the ISO Net Zero Guidelines (ISO, 2022b; UN HLEG, 2022).

We find only three companies (Apple, H&M Group, and Maersk) to meet 1.5°C-aligned decarbonisation milestones for 2030 in the literature. We evaluate these companies' targets towards 2030 to have moderate integrity, as none of them have set a similar short-term target within a five-year interval substantiating their 2030 medium-term commitments. Six other companies (Ahold Delhaize, ArcelorMittal, Foxconn, Holcim, Stellantis, and Thyssenkrupp) come close to meet 1.5°C-aligned decarbonisation milestones in the literature but do not fully meet them. Fifteen out of 24 companies fall way short of meeting such benchmarks. These findings point to the urgent need for companies to better inform and align their corporate targets to sector-specific 1.5°C-aligned decarbonisation milestones identified in the literature. Apart from mostly inadequate targets towards 2030, only 11 of the 24 companies commit to emission reduction targets within a five-year interval; another key recommendation by the UN High-Level Expert Group and the ISO Net Zero Principles (ISO, 2022b; UN HLEG, 2022).



Figure 3: The median commitment to emission reductions between 2019 and 2030 is just 15–21% This chart shows the proportion of full value chain GHG emissions that companies commit to reduce between 2019 and 2030.

The data points in this graphic represent the authors' most optimistic interpretations of companies' absolute emission reduction commitments, including emission intensity targets, based on publicly available information. The median is provided for scenarios with and without the optimistic assumption that intensity targets result in equivalent absolute emission reductions. Targets that are reliant on offsets to an undefined extent are marked as ambiguous. Further details and explanation on the authors' interpretation of companies' targets can be found in the individual company cases in Section B and in Annex II.

Voluntary initiatives and recently published guidelines in principle set high criteria for net-zero pledges to meet, but the lack of mandatory compliance, missing independent scrutiny, and methodological shortcomings to inform 'science-based' target setting requires urgent attention by regulators and civil society.

Several net-zero guidelines and target setting methodologies have been released for corporates over the course of 2022, among others by the UN High-Level Expert Group, the International Standard Organisation (ISO) and the Race to Zero Campaign (ISO, 2022b; Race to Zero, 2022; UN HLEG, 2022). These recent developments point to an emerging consensus on the need for robust, credible, and transparent emissions reduction targets in the short-, medium-, and long-term. After the SBTi released its Net Zero Standard in November 2021, it has verified the netzero targets of around 140 companies' net zero targets as 1.5°C-aligned as of February 2023 (SBTi, 2023). More than 1,600 companies have officially committed to set net-zero targets in line with the SBTi Net Zero Standard, but SBTi has not yet verified their targets.

Our analysis shows that the SBTi's Net Zero Standard may provide a framework for credible long-term targets, but compliance must be checked thoroughly and be continuously monitored in the future. As of February 2023, two of the companies we assessed – H&M Group and Holcim – have had their net-zero pledges certified by SBTi under its Net Zero Standard (see Table 8). We found the netzero pledges of these companies to be of high and moderate integrity, respectively. The requirement of SBTi's Net Zero Standard that net-zero pledges should equate to at least 90% emission reductions across the full value chain, directly addresses the key issue that we identify for most companies' net-zero pledges, which are either ambiguous or do not translate to commitments for deep decarbonisation.

However, the Net Zero Standard is not watertight and depends on honest compliance with the framework: the 2022 Corporate Climate Responsibility Monitor (Day et al., 2022) found that the SBTi-certified net-zero pledge of CVS Health was undermined by highly contentious baseline setting. The credibility and real ambition of **H&M Group** and **Holcim**'s targets contain uncertainties with regards to the companies' potential use of bioenergy and RECs, and CCS technologies, respectively. Another 11 of the 24 companies assessed in this report have officially committed to the Net Zero Standard through SBTi's webpage but SBTi has not yet verified their targets (see Table 8 for company-specific comparisons). We evaluate the net-zero targets of nine of those eleven companies to have low integrity. Table 8: Comparison between target verifications by the Science Based Targets initiative (SBTi) as of February2023 and the integrity assessment as part of the Corporate Climate Responsibility Monitor (CCRM) 2023 for net-zeropledges. Companies listed in alphabetical order.

COMPANY NAME	SBTI VERIFICATION (FOR NET-ZERO TARGETS)	CCRM TARGET INTEGRITY (FOR LONG-TERM TARGETS BEYOND 2030)
Ahold Delhaize	Officially committed (no verification)	Moderate integrity
American Airlines	Officially committed (no verification)	Low integrity
Carrefour	Officially committed (no verification)	Low integrity
Deutsche Post DHL	Officially committed (no verification)	Low integrity
H&M Group	1.5°C	High integrity
Holcim	1.5°C Explanation on difference in assessments: Holcim's net-zero target meets 1.5°C-aligned sectoral benchmarks, but t the heavy reliance on CCUS makes the appropriateness of net-zero term	
Inditex	Officially committed (no verification)	Low integrity
JBS	Officially committed (no verification)	Low integrity
Maersk	Officially committed (no verification)	High integrity
Microsoft	Officially committed (no verification)	Low integrity
Nestlé	Officially committed (no verification)	Low integrity
PepsiCo	Officially committed (no verification)	Low integrity
Walmart	Officially committed (no verification)	Low integrity

SBTi certifications for short- and medium-term targets lend credibility to companies whose targets are highly insufficient. SBTi has certified the 2030 targets of 16 of the 24 companies included in this analysis as aligned with a pathway to limit global temperature increase to 1.5°C, 'well-below 2°C', or 2°C. Another five companies are listed on SBTi's website as 'committed' to science-based targets for 2030. Most of these companies highlight their SBTi certifications prominently in their climate-related communications to defend targets that are highly insufficient in the context of latest available science, and sometimes misleading. The SBTi's verifications of 2030 targets often neglect relevant details leading to the undifferentiated certification of corporate targets, regardless of whether a company is lagging in climate action or can truly be considered a climate leader. Full compliance with existing guidelines and appropriate communication around verified targets appears to be rather an optional avenue for a clear minority of willing companies, rather than a mandatory practice. The comparison between SBTi's verifications of 2030 targets of 2030 targets integrity assessments identifies the following four key common problems (see Table 9 below and Table 19 in the Annex for company-specific comparisons).

• Use of offsetting under the guise of 'insetting' (within value chain removals) by companies operating in the FLAG sector:

The SBTi FLAG guidance and the SBTi Net Zero Standard allow companies operating in the FLAG sector to use 'insetting' to meet their 2030 and net-zero targets (SBTi, 2021c, p. 30, 2022b, pp. 27–28). 'Insetting' refers to approaches under which emissions are offset rather than reduced; moreover, 'insetting' through biological carbon dioxide removals is not an appropriate approach to claim that emissions have been offset, among other reasons, due to high uncertainties regarding the permanence of outcomes (see *Spotlight 4.4.4 What does insetting and climate positive really mean*?). **Nestlé** and **PepsiCo** are examples of companies that rely on offsetting to achieve 2030 pledges under the guise of 'insetting'.

• Continued use of verifications based on methods discontinued by SBTi ('legacy issue'):

SBTi does not sufficiently clarify that certain verifications that continue to be used by companies in their day-to-day communication build on methodologies indefinitely paused by SBTi. Automobile manufacturers such as **Mercedes-Benz** and **Volkswagen** continue to promote their 'well-below 2°C' verification of their scope 3 emissions intensity targets in their annual sustainability reports (Mercedes-Benz Group, 2022e, p. 130, 2022a, p. 23; Volkswagen, 2022c, p. 46), although SBTi has indefinitely paused the use of its methodology for automakers, as the initiative states that the methodology does not reflect a 1.5°C-compatible definition (SBTi, 2022f).

• Target verification not contextualised to emission share covered across a company's entire value chain:

SBTi's temperature ratings for short- and medium-term targets are only provided for targets for scope 1 and 2 emissions (SBTi, 2023b). Those temperature ratings do not – in most cases – apply to scope 3 targets, although companies' scope 3 targets are also listed on SBTi's website alongside the temperature rating, in a way that is very likely to lead to misunderstandings on this issue. For this reason, a 1.5°C verification for a given scope 1 and 2 target might in some cases only cover less than 5% of a company's total footprint. SBTi does not directly put a single verification into context of whether and to which ambition level a company intends to reduce all other emissions. For example, **Microsoft**'s 2030 1.5°C verified renewables target only covers 3% of the company's entire value chain emissions. Certain sector-specific verification methods further leave out specific operational emissions. Our assessments include the significant climate impacts of non-GHG climate forcers associated with aviation – for example in the case of **American Airlines** – while SBTi's aviation guidance exclusively covers GHG emissions from jet fuel (SBTi, 2021d).

• No disclosure of method and underlying data used for specific certification:

SBTi neither provides information on verification methods used nor on underlying data inputs such as base year emissions. On the former, it remains unclear for an external audience whether SBTi uses sector-specific or cross-sectoral methods for specific verifications. Cross-sectoral methods do not generally consider any sector-specific requirements to reduce emissions in line with 1.5°C. For example, we cannot identify whether **Deutsche Post DHL'**s verification uses cross-sectoral methods or sector-specific methods for the aviation, shipping, and road transport sectors. On the latter, SBTi discloses no further information on relevant data inputs used for its verification and whether these have been independently scrutinised. For the 'well-below 2°C' verification for **Carrefour**'s 2030 targets, for example, it remains unclear whether SBTi explicitly allows Carrefour to exclude the vast majority of its stores and global activities from its target coverage, or if such exclusions have been overlooked. An open letter by several scientists called upon SBTi, among others, to strengthen independent scrutiny of baseline emissions data (Carton *et al.*, 2022).

The overlapping issues on undifferentiated verification give cause to increasing concern that these targets do not reflect the urgency to accelerate climate action towards 2030 in line with latest science and require urgent attention by SBTi and regulators.

Table 9: Comparison between target verifications by the Science Based Targets initiative (SBTi) as of February 2023 and the integrity assessment as part of the Corporate Climate Responsibility Monitor (CCRM) 2023 for short- and medium-term targets towards 2030. Companies listed in alphabetical order.

	SBTI TEMPERATURE RATINGS⁺ (NEAR-TERM TARGETS)	CCRM INTEGRITY ASSESSMENT (SHORT- & MEDIUM-TERM TARGETS TOWARDS 2030)	KEY ISSUES EXPLAINING THE DIFFERENCE BETWEEN SBTI TEMPERATURE RATINGS AND CCRM ASSESSMENTS FOR CORPORATE TARGETS TOWARDS 2030				
COMPANY NAME			Use of insetting	Use of discontinued verifications	Exclusion of significant emission share	Lack of disclosure on method and underlying data disclosure	Other reasons (see Annex for further details)
Ahold Delhaize	1.5°C	Moderate integrity			×	×	×
American Airlines	Well-below 2°C**	Poor integrity ***			×		×
Apple	1.5°C	Moderate integrity			×	×	×
Carrefour	Well-below 2°C	Poor integrity			×	×	
Deutsche Post DHL	1.5°C	Poor integrity			×	×	
Fast Retailing	1.5°C	Poor integrity			×		
Holcim	1.5°C	Moderate integrity					×
H&M Group	1.5°C	Moderate integrity			×		×
Inditex	1.5°C	Poor integrity			×		×
PepsiCo	1.5°C	Unclear integrity	×		×	×	
Mercedes-Benz	1.5°C/ Well-below 2°C	Poor integrity		×	×		
Microsoft	1.5°C	Poor integrity			×		
Nestlé	1.5°C	Poor integrity	×		×	×	
Thyssenkrupp	Well-below 2°C	Moderate integrity			×		×
Volkswagen	1.5°C/2°C	Poor integrity		×	×		
Walmart	1.5°C	Poor integrity			×		

* The SBTi temperature ratings for short- and medium-term targets do not cover certain emission scopes, which is not directly obvious from SBTi's presentation and companies' communications. Table 19 in the Annex III provides further information on the verifications' emission coverage.

⁺⁺ SBTi verifies American Airlines 2035 target as a near-term target.

*** Integrity evaluation for 'long-term targets' beyond 2030 given SBTI's 2035 target verification.

3 Reducing emissions

Encompassing measures for deep emission reductions are the backbone of ambitious corporate climate targets. As companies' emission profiles vary widely, there is not a standardised set of measures that all can implement. The integrity and robustness of companies' decarbonisation efforts must be considered against each company's circumstances and emission profile (section 3.2.1).

Electricity-related emissions are relevant for all companies to address and are often a central feature of companies' plans and claims. For this reason, we single out renewable electricity generation and procurement for deeper assessment (section 3.2.2).

Table 10 presents a summary overview of principles for good practice (section 3.2) as well as summary trends, promising examples and bad practice identified from the company assessment (section 3.3).

...........
3.1 Snapshot summary

Table 10: Summary of good practice and trends for emission reduction measures and renewable electricity.

	PRACTICE PRINCIPLES (SECTION 3.2) E FROM THE 24 COMPANIES ASSESSED			
		AVERAGE PERFORMANCE+		
GUUD PRA	CTICE FOR REDUCING EMISSIONS	TRANSPARENCY⁺	INTEGRITY ⁺	
Emission reduction measures	 Implement encompassing and deep decarbonisation measures and disclose details of those measures to support replication and the identification of new solutions. Refrain from using bioenergy where alternatives to combustion exist, and ensure that any bioenergy used does not have negative sustainability implications. 	Moderate 🔺	Low —	
Renewable electricity generation and procurement	 Procure the highest quality renewable electricity available and disclose the full details of that procurement. 	Low —	Very low —	

* Transparency and integrity columns: the bar indicates the distribution of our rating of the 24 companies (**Poor Moderate High n.a.**); the text above the shaded bars represents the average rating across all the companies we assessed, calculated excluding non-applicable cases, on a 5-point scale (**Very low, Low, Moderate, Reasonable, High**); and an indication of progress since the last analysis in 2022 (**M** – **V**), based on the authors' interpretation of progress from the companies that were analysed also in 2022, against the current methodology version.

SUMMARY OF TRENDS, ROLE MODELS AND BAD PRACTICE (SECTION 3.3)

Reduction plans are lengthy but without much substance: Most companies describe their reduction plans in length, but most plans lack detailed information and concrete commitments.

Bioenergy currently plays an important role in the decarbonisation plans of companies in sectors that are more difficult to transition: At least 11 of 24 companies plan to rely on bioenergy, which may have negative sustainability implications.

More companies are taking responsibility for reducing energy-related emissions in their supply chain: At least 4 of 24 companies present plans to reduce energy-related emissions in their supply chain, for instance through collective PPAs.

No significant progress on renewable electricity procurement: We identified limited information on renewable electricity procurement, a relatively accessible measure for most companies. RECs remain in widespread use although they are largely ineffective.

PROMISING EXAMPLES

BAD PRACTICE

Investing in new solutions: Maersk invests in the development of low-carbon fuels.

24/7 matching of renewable electricity generation and consumption: Google and Microsoft are monitoring (a share of) their electricity consumption and RE generation on the local grid 24/7.

Supporting suppliers in reducing their

electricity-related emissions: Walmart sets up collective PPAs for suppliers. Apple connects suppliers with RE project developers and engages with policymakers on supportive regulatory frameworks for RE.

Lack of transparency: Only three companies outline detailed information on their reduction measures and just two on their renewable electricity procurement.

Description of reduction measures is lengthy but lacks clear commitments: For instance, Fast Retailing, the H&M Group and Inditex commit to using 'more sustainable' materials, without explaining what 'more' or 'sustainable' means.

Focus on marginal changes: Most companies focus on efficiency improvements or present innovations that address just a small share of their emissions. Only one company – **Apple** – outlines plans that could transform its business and place it on a 1.5°C-compatible trajectory.

Limited ambition for renewable electricity: On-site installations and high-quality PPAs represent a minor share of RE consumption. 14-20 companies use RECs.

SUPPORTING COMPANIES TO IMPROVE THEIR EMISSION REDUCTION MEASURES AND RENEWABLE ELECTRICITY PROCUREMENT

Sector-specific regulations and carbon pricing can support ambitious first-movers without putting them at an economic disadvantage compared to their less ambitious competitors.

Better guidance on bioenergy use at the national and regional level to help ensure that the limited availability of sustainable bioenergy and biomass is not claimed by a select set of companies, which would push others to use non-sustainable bioenergy resources.

Policies that incentivise, rather than hinder, more renewable energy capacity is key to ensuring that companies can decarbonise their entire supply chain.

Table 11: Summary of developments in 2022 related to emission reduction measures and renewable electricity



+ Authors' interpretation of companies' progress against the current methodology version since the last analysis in 2022 (

NE	W GUIDELINES AND FRAMEWORKS RELATED TO EMISSION REDUCTION MEASURES AND RENEWABLE ELECTRICITY PROCUREMENT IN 2022
UN High-Level Expert Group (UN HLEG, 2022)	Companies must publicly disclose transition plans, specifying what action will be undertaken to achieve targets, including supplier engagement efforts and disclosing how plans and investments are aligned with the targets.
	Companies must update their transition plans every five years and annually report on progress.
	Corporate net-zero pledges must be accompanied by targets for ending the use or support for fossil fuels in line with IPCC and IEA net-zero emissions pathways that limit global warming to 1.5°C with no or limited overshoot.
	Renewable energy procurement targets must be part of corporate net-zero transition plans.
	Companies must avoid the conversion of natural ecosystems in their own operations and supply chain. They must eliminate deforestation and peatland loss by 2025 at the latest, and the conversion of other remaining natural ecosystems by 2030 at the latest.
ISO Net Zero guidelines	Corporates should not delay any mitigation action to achieve their interim or long-term reduction targets.
(ISO, 2022b)	Measures must address scope 1, scope 2 and scope 3 emissions and companies should focus on the full range of emission reduction measures available.
	Companies should aim to use 100% renewable energy and the purchase of renewable energy should lead to the development of further renewable energy capacity. This implies that companies should avoid relying on RECs that 'allocate the renewable portion of a supply that contains a mix of other sources, including fossil fuels', but instead should prioritise renewable energy through power purchase agreements or on-site installations.
	Companies should adjust their consumption in line with the availability of renewable energy. This means they need to minimise consumption when the grid is reliant on high-emission energy.
	Corporates should commit to achieve and maintain operations and supply chains free of deforestation by 2025.
EU Corporate Sustainability Reporting Directive (CSRD) (EU, 2022)	Companies shall report on their plans to ensure that their business model is compatible with the transition to a sustainable economy and with the limiting of global warming to 1.5°C in line with the Paris Agreement (Article 19). In June 2023, the Council of the EU will adopt a set of standards outlining in more detail what information companies need to report on.

Note: The summaries in this table represent the authors' interpretations of the guidelines and developments presented.

38

3.2 Principles for good practice

3.2.1 Emission reduction measures

Corporate actors must implement encompassing and deep decarbonisation measures. Decarbonisation efforts should focus on all relevant emission sources across all three scopes. Adopting readily available measures should be the first priority for companies that claim to be on a decarbonisation pathway, followed by the scaling up of proven flagship projects and—if necessary—investments in research and development to find new decarbonisation solutions. Demonstrated emission reduction measures vary per sector, although electrification and renewable energy are relevant for many sectors. For instance, a switch from combustion engines to electric vehicles in the automobile sector, and e-fuels instead of fossil-based fuels in the shipping sector. In addition, technological and operational efficiency improvements are necessary steps for every company.

Further, companies should have a clear plan to phase out all carbon-intensive infrastructure and products. Net-zero is disingenuous vision for companies that continue to invest in and rely on fossil fuels. Ambitious companies should plan for and implement a set of measures that leads to complete or near decarbonisation of their activities, depending on the sector they are active in.

Transparent disclosure and information sharing can support replication and the identification of new solutions. Companies can show real climate leadership by prioritising transparent exchange on climate change mitigation over industry competition, to support replication of effective measures and to collaborate for the identification of new solutions. Reports that refer to individual flagship projects may potentially inspire readers, but further details are required to support replication and facilitate an assessment of the company's ambition. Companies' planned measures can only be fully appraised if their plans contain details on the scale of planned measures using indicators that demonstrate what proportion of a company's activities will be addressed by the measures, and what the anticipated impacts are for reductions in GHG emissions.

Several companies present bioenergy as a measure to decarbonising their businesses. We have looked at more detail into the issue of bioenergy for this iteration of the Corporate Climate Responsibility Monitor. Full details on the issues around bioenergy can be found section 3 of the methodology document.

Companies demonstrating climate leadership plan and take decarbonisation measures that do not rely on bioenergy when possible; and ensure that any bioenergy they use does not have negative sustainability implications. Some sectors that are difficult to electrify and have limited alternatives to decarbonise might rely on bioenergy to some extent (Calvin et al., 2020; Clarke et al., 2022). However, increasing demand for bioenergy in hard-to-abate industries will lead to competition for limited biomass resources (Pavlenko and Kharina, 2018; ETC, 2021), which is likely to further exacerbate sustainability issues. These include, but are not limited to, deforestation, biodiversity loss and food insecurity (Kline et al., 2015; Hof et al., 2018; Searchinger et al., 2018; Calvin et al., 2020; Ahmed et al., 2021; Clarke et al., 2022; Hanssen et al., 2022). Further, bioenergy is not an emissions-free energy source: cutting down trees or other plants and burning them to generate energy, leads to the release of sequestered carbon. It can take several to hundreds of years to balance out this release of CO₂, depending on the type of trees used (Holsmark, 2012; Mitchell et al., 2012; Ter-Mikaelian et al., 2015; Searchinger et al., 2018). Also, land that is used to grow bioenergy crops cannot be used for other purposes, such as sequestering carbon directly (Searchinger et al., 2022). This carbon opportunity cost of land should be factored in when calculating the net impact of bioenergy.

The potential for sustainable biomass is very limited and outweighed by demand (ETC, 2021). Biomass is not only used to generate energy, but also to produce materials (e.g. timber, paper and bio-feedstocks for the chemicals industry). It is estimated that the potential for sustainable biomass may amount to 40-60EJ by 2050, similar to today's usage of biomass (ETC, 2021). However, demand for biomass to generate energy and produce materials in just four sectors is likely to exceed this supply potential. Companies should therefore use alternative technologies that do not depend on combustion where those exist. If such alternative technologies are likely to emerge in the future, companies should consider using bioenergy only as a temporary solution, ensure it does not have negative sustainability implications, and invest the development of alternative technologies at the same time. Full details on the issues around bioenergy can be found section 3 of the methodology document.

3.2.2 Renewable electricity generation and procurement

40

Companies reduce electricity-related emissions in different ways. How a company goes about sourcing renewable electricity makes a big difference in the actual emission impact and the credibility of renewable electricity consumption claims. Importantly, in most cases, pursuing high-quality procurement constructs does not imply that a company consumes 100% renewable electricity.

Electricity-related emissions are a relevant emissions source for all companies to address and represent a key component of many companies' climate change strategies and pledges. For some companies, those emissions account for the lion's share of their emissions. Other companies may have relatively fewer emissions from electricity consumption today, for instance those in the heavy industry, aviation, and shipping sectors. However, electricity is likely to become increasingly important for those companies, as they move away from fossil fuels to alternatives such as hydrogen and ammonia, for the production of which electricity is needed. As alternative fuels are not yet produced at scale, some companies are investing in new facilities that will produce, for instance, e-methanol or e-hydrogen. Those fuels are only zero carbon if they are based on green electricity.

It is best practice for companies to combine high-quality renewable electricity procurement with the most accurate and transparent emission reporting. Companies have a variety of options for sourcing renewable electricity (see Table 12). Of all the options, own renewable electricity installations are most likely to result in truly additional renewable energy capacity and are therefore usually the highest quality approach that companies can follow. Power Purchase Agreements (PPAs) might result in the development of additional capacity, but the causal link between a certain PPA and additional capacity is often very hard to prove (see Box 1). Renewable Electricity Certificates (RECs) are least likely to result in additional capacity and often have no impact on grid decarbonisation.

As the impact of projects vary and is often unclear, companies should report their electricity-related emissions using the location-based accounting method alongside the market-based accounting method (see section 3 in the methodology document and the Glossary).

RECs and PPAs explained

(adapted from Day et al. (2022))

Renewable Electricity Certificates

Renewable Electricity Certificates (RECs) are used in many countries around the world, sometimes under different names, such as Guarantees of Origin (GOs) or Energy Attribute Certificates (EACs). RECs are certificates that represent the generation of 1 MWh of renewable electricity. Companies – or other actors – procure RECs to match their electricity consumption with the generation of renewable electricity.

RECs come in various forms:

- Bundled RECs supplier generated: a company buys electricity and RECs from one supplier. The supplier generates the RECs themselves.
- Bundled RECs third-party generated: a company buys electricity and RECs from one supplier. The supplier procures the RECs from a third-party and resells them to its own customers.
- Unbundled RECs: a company purchased electricity from one retailer and the RECs from another one.

In all cases, the company consumes electricity from the local grid. The procurement of RECs does not immediately – if at all – change the electricity mix on the grid, nor the electricity mix that a company takes from the grid. This means that if a company purchases RECs, its electricity-related emissions do not actually decrease.

The procurement of RECs is very unlikely to contribute to additional renewable electricity supply capacity. While the purchase of RECs could send a signal to investors that there is demand for renewable energy in theory, there are indications that this is often not the case in practice due to issues including oversupply of certificates and associated low prices, and implicit double counting.

Power Purchase Agreements

Power Purchase Agreements (PPAs) are a long-term contract between an offtaker and an energy supplier. The offtaker agrees to consume a certain amount of electricity against a predetermined price. PPAs are usually, but not necessarily, signed for new renewable energy installations. Companies often also purchase the RECs associated with the installation for which they signed a PPA. This avoids the risk that a third party would buy the RECs, and both that third party and the company that has signed the PPA claim the same renewable electricity.

Unless there is a direct grid line between the renewable energy installation and the company's facilities, companies consume electricity from the local grid. In most cases, there will not be a direct grid line between the company's facilities and the renewable electricity installation. Rather, the renewable electricity installation feeds electricity into the local grid. This implies that while the company may procure 100% renewable electricity through the PPA, it does not consume 100% renewable electricity. For this reason, companies should always report location-based emissions, which are based on the average grid emissions factor.

The causal link between a PPA and additional capacity is often very hard to prove. In some cases, the PPA may influence the energy supplier's investment decision, leading to truly additional renewable electricity generation capacity. However, investments in renewables are attractive in many markets, so a PPA and the long-term financial security it offers to a project developer may not be the decisive factor to go ahead with a certain renewable electricity project.

Table 12: Overview of renewable electricity procurement options.

RENEWABLE ENERGY PROCUREMENT	LIKELIHOOD OF CONTRIBUTING TO ADDITIONAL INSTALLED RENEWABLES CAPACITY		TALLED	
CONSTRUCT	HIGH	MODERATE	LOW	
Own RE installation with storage capacity	×			Constructs ensure the installation of capacity that would not have come online otherwise. New storage solutions in combination with these new installations can help reducing the impact on the local grid and support 24/7 matching of demand and support between the local grid and support 24/7 matching of demand and support between the local grid and support 24/7 matching of demand and support between the local grid and support 24/7 matching of demand and suppor
Own RE installation without storage capacity	×			supply. However, in many cases, companies may still rely on the local grid when their generation and storage does not cover their demand. They should use the location-based emissions factor for the emissions reporting for the energy that is consumed directly from the grid. The emissions factor for the energy that they generate themselves may be zero.
				PPAs can contribute to additional capacity if the PPA is signed with a new RE installation and provides the energy provider with the necessary financial security to go ahead with the construction of the installation. To contribute to reducing a company's energy-related emissions, it is necessary that the PPA is signed for an installation connected to the same electricity grid as the company's facilities. To avoid double claiming of renewable electricity, companies should purchase RECs from the RE installation for which they signed a PPA.
Power Purchase Agreement (PPA)	×	×	×	PPAs are unlikely to contribute to the installation of additional capacity if the PPA is signed for an existing installation (unless the energy provider would need to shut down the installation in the absence of a new PPA). PPAs that are signed for an installation in a different geographical area may lead to additional capacity but do nothing to reduce emissions on the company's local energy grid.
				PPAs do not lead to a direct and immediate reduction of emissions from the consumed electricity at all times of the day. Electricity is still procured from the grid, supplied by a mix of generation technologies. The emission impact is not comparable to a reduction in electricity demand through energy efficiency measures. A location-based emissions factor should be used to accurately indicate the emissions impact associated with electricity consumption.
Capacity premium	×	×	×	The likelihood of a capacity premium leading to additional capacity can be considered high, moderate or low depending on the integrity of the entity that collects the capacity premium and on the construct (see this table's overview) for which the collected funds are invested in.
RECs bundled		×	×	While some claim that RECs may signal to the market that there is demand for renewable electricity, studies have found no evidence that the procurement of RECs leads to the development of additional renewable electricity capacity (Bjørn <i>et al.</i> , 2022).
				Bundled RECs may have a moderate or low likelihood of contributing to additional RE capacity. Likelihood is larger if:
				 RECs are bundled with the energy that a company purchases and generated by the energy supplier (i.e. on the same local grid as the company);
RECs unbundled			×	• RECs are from a new installation. RECs do not lead to a direct and immediate reduction of emissions from the consumed electricity at all times of the day. Electricity is still procured from the grid, supplied by a mix of generation technologies. The emission impact is not comparable to a reduction in electricity demand through energy efficiency measures. A location-based emissions factor should be used to accurately indicate the emissions impact associated with electricity consumption.
Investments in renewable energy installations				Investments in renewable energy capacity are a business case. They can be combined with a PPA or RECs.
24/7 monitoring				24/7 monitoring and matching energy consumption with renewable energy generation can be an add-on to using PPAs or RECs. This improves the quality of those constructs by ensuring that consumption is matched with renewable energy production around the clock.

3.2.3 Good practice assessment criteria

The criteria for good practice in Table 13 form the basis for the company assessments in section B. Full details on the methodology for rating companies' corporate target setting can be found in the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0, section 3.*

Table 13: Good practice for reducing own emissions

	REDUCING EMISSIONS ORPORATE CLIMATE LEADERS EXHIBI	TING GOOD PRACTICE
Α.	Emission reduction measures	 Publish detailed information on the planned emission reduction measures for all relevant emission sources throughout the value chain
		 Outline the expected emission reductions resulting from the implementation of those measures
		 Adopt existing reduction measures and scale up demonstrated flagship projects to mainstream those projects across the organisation
		 Invest in research and development of new technological solutions, where necessary
		✓ Set out a clear plan to phase out all carbon-intensive infrastructure and products
		 Refrain from using bioenergy where alternatives to combustion exist, and ensure that any bioenergy used does not have negative sustainability implications
B.	Renewable electricity	 Pursue the highest quality renewable electricity procurement option
	procurement	 Use the most accurate and transparent accounting method, which reflects emissions from electricity consumed (location-based accounting), rather than the emissions from the electricity bought (market-based accounting)⁺

* We assess the accounting method for electricity-related emissions (i.e. scope 2) under 'tracking and disclosure of emissions' (see section 1).



3.3 Trends, role models, and bad practice

Companies outline their emission reduction plans with a moderate level of transparency in 2022, but we did not identify significant progress on the integrity of those plans. Companies' descriptions of their reduction measures is lengthy but lacks clear commitments. Most companies focus on efficiency improvements or present innovations that address just a small share of their emissions, rather than plans that could sufficiently transform their businesses to place them on a 1.5°C-compatible trajectory within their respective sectors. Almost half of the companies assessed plan to or already rely on bioenergy, despite the negative sustainability implications of this energy source.

We found transparency on renewable electricity procurement to be very low across the board; many companies only report on their procurement approaches in their CDP disclosure. While renewable electricity generation and procurement is a demonstrated and relatively straightforward measure, the large majority of companies in this report continue to rely on Renewable Electricity Certificates that are highly unlikely to have any meaningful impact on grid decarbonisation. A small minority of companies are leading the way with meaningful and replicable measures for their own renewable electricity procurement as well as renewable electricity procurement for suppliers.

While most companies allude to measures targeting scope 1, 2 and 3 emissions, details are often lacking.

The majority of companies in this report present measures targeting emissions across the value chain. Compared to our last report, we find that an increasing number of companies implement measures to reduce emissions in their upstream or downstream value chain (i.e. scope 3): of the 24 companies assessed, 22 present reduction plans for upstream scope 3 emissions and 18 for downstream scope 3.

However, information on planned reduction measures is often vague, which hinders an understanding of whether the companies implement adequate measures and focus on the main emission sources. While 22 of the 24 companies outline their measures with at least a moderate level of detail, in almost all cases, it remains unclear at what scale measures are implemented and what share of current GHG emissions they address. We found that in some cases, companies present their plans in length without making any clear commitments. For instance, all three fashion retailers in this report commit to sourcing materials that are 'more sustainable' than the materials used today, without describing what this means. 'More sustainable' can simply mean a small improvement compared to the baseline and does not necessarily entail a commitment to shift to truly sustainable materials.

Companies' proposed measures are insufficient to catalyse the transformational change that is necessary to limit global warming to 1.5°C.

While most companies state to work on developing nascent zero-carbon technologies, these generally do not address company's key emission sources. Twenty of the 24 companies provide that they invest in new solutions. However, in most cases, these projects do not necessarily address key emission sources. Only few companies are working on innovative technologies that address their most important emission sources. These include, for instance, **Maersk**, which invests in alternative fuels and vessels, **Stellantis**, which works on developing batteries and alternative fuel cells, **Google**, which is pioneering 24/7 monitoring and matching renewable energy generation with consumption, and **Deutsche Post DHL**, which invests in electrifying its fleet and scaling up the production of low-carbon fuels for road transport.

Companies present only limited plans to improve product quality and lifetime, which would lead to lower production levels and substantial emission reductions. The three fashion retailers in this report – **Fast Retailing, H&M Group** and **Inditex** – produce and sell low-cost clothes in large quantities, releasing new product lines with high frequencies. The measures they propose include sourcing 'more sustainable' materials, recycling materials and energy efficiency (see company analyses in Section B). While these can lead to marginal reductions, a *decrease in production* is likely necessary to bring these companies on a 1.5°C trajectory. Likewise, **Foxconn, Microsoft** and **Samsung** present measures targeted at reducing emissions that occur in the production process, but we found no evidence that these companies plan to significantly increase the lifetime of the products they sell. They could so by, for instance, ensure that software updates can be installed on older hardware and selling spare parts, so that consumers can repair their devices, rather than having to buy new ones. **Apple**, on the other hand, does report to take measures to prolong the lifetime of its devices. The tech company's software updates run on ten-year old devices, and the company collects used devices to refurbish them, although still at a small scale compared to the sale of new devices (Apple 2022a, p. 48).

Companies in sectors with particularly limited decarbonisation potential envisage a key role for bioenergy, but this is not an emissions-free energy source and is likely to negatively impact biodiversity, carbon storage and food security, among others.

Companies present bioenergy as part of their emission reduction plans but it is not a credible measure in most cases, given that bioenergy is likely to have a range of negative sustainability implications. At least 11 of the 24 companies plan on using bioenergy to decrease their emissions, including American Airlines, Deutsche Post DHL, Holcim, H&M Group, Maersk, Stellantis and Thyssenkrupp. Bioenergy is not an emissions-free energy source and is very likely to have negative sustainability implications. Companies like H&M Group and Stellantis can rely on alternatives to fossil fuel or biomass combustion to decarbonise their value chains (see section 3.2.1). The shipping and heavy industry sectors also have prospects of decarbonising without using bioenergy in the longer term. Most of the companies in this report claim that they use 'sustainable' bioenergy but even if that would be true, this approach can exacerbate sustainability problems. Although it is commendable that companies set sustainability criteria for the bioenergy that they source, there is the inherent problem that by using sustainable bioenergy – which is and will remain a scarce resource – they push other companies to use non-sustainable biomass. To limit global warming to 1.5°C and protect ecosystems, it is key that *overall demand* of biomass decreases. Companies in sectors with other prospects for decarbonisation should pursue those alternative avenues.

Although sourcing high-quality renewable energy is a relatively accessible measure for many, the majority of companies assessed provide limited information on their renewable energy consumption.

While 10 of the 24 companies state to use renewable energy for more than 50% of their energy consumption, four of them fail to provide sufficient information to back up this claim and another four companies rely on RECs. Amazon, Deutsche Post DHL, Inditex and Mercedes-Benz provide very limited information on the renewable energy they procure, which makes it impossible to assess the integrity of their renewable energy claims. While Mercedes-Benz outlines that it uses on-site installations and PPAs to power its German factories, the car manufacturer does not outline if and how it sources renewable energy for its facilities in other countries. H&M Group, Microsoft, Nestlé and PepsiCo rely on RECs to make their renewable electricity claims, but RECs are extremely unlikely to lead to the installation of additional renewable energy capacity.

RECs remain an important feature of companies' renewable energy procurement approach. At least 14 companies assessed procured RECs in 2021, with another six companies not providing information on their procurement construct. With a few exceptions, those 14 companies provide limited information on the RECs they purchase. RECs typically do not send a meaningful signal to the market that there is demand for additional renewable electricity capacity and unbundled RECs may lead to double counting renewable electricity (See Table 12). Indeed, a study published in 2022 found that the widespread use of RECs leads to an inflated estimate of companies' mitigation efforts (Bjørn *et al.*, 2022).

We identified just two companies that pursue high quality renewable energy procurement constructs today. Apple sources 87% of its electricity consumption from local and newly installed PPAs, while 77% of **Google**'s energy procurement stems from on-site generation or local PPAs. Unbundled RECs were an important component of **Microsoft**'s renewable energy procurement in previous years, but the tech company plans to switch to high-quality PPAs on the local grid by 2025.

Some of the tech companies in this report are pioneering new approaches to securing 100% renewable energy consumption on an hourly basis. On-site installation and PPAs on the local grid are generally the most impactful procurement constructs that companies can pursue. However, there is a mismatch between renewable energy generation and consumption. For instance, a company that has signed a PPA for a wind park cannot claim to use renewable energy on windless days. **Google** and **Microsoft** have both pledged that 100% of their energy generation by local installations by 2030.

We see traction for taking responsibility for energy-related emissions in the value chain.

A handful of companies in this report try to reduce the energy-related emissions of their suppliers, including through guidance on renewable electricity procurement and PPAs. Apple invests in renewable electricity and helps its suppliers sign PPAs (see Box 2) and Walmart sets up collective PPAs through its *Gigatonne PPA* programme. These represent good practice examples that other companies may replicate. Setting up collective PPAs can help unlock greater investments in renewable electricity capacity and may contribute to decarbonising local grids.

Other companies are looking to procure RECs for their suppliers, including **Foxconn** and **H&M Group**. However, procuring RECs for suppliers suffer from the same limitations as RECs for own electricity-related emissions and are unlikely to result in real emission reductions.

Renewable energy policies are underdeveloped in some countries, which makes it difficult for companies and their suppliers to source renewable energy. Large multinationals like the 24 assessed in this report can use their influence to lobby for regulations that allow for and encourage companies to procure renewable energy through PPAs or install their own renewable energy generation capacity. Companies especially mention Asia as a region where regulatory hurdles to renewable electricity procurement exist.

ž2

Apple's strategy for renewable energy in the supply chain

As part of its climate strategy to reach carbon neutrality and reduce 75% of its 2015 emissions footprint by 2030, Apple has set to transition the entire manufacturing supply chain to 100% renewable electricity (Apple, 2022b). If achieved, this goal could significantly impact the company's emissions footprint, as most of Apple's emissions (67%) originate from use of energy to manufacture products in third-party supply factories (Apple, 2022a, p. 84). To implement this vision, the company created the Supplier Clean Energy programme, which seeks to support suppliers to transition to 100% renewable electricity.

Apple launched the Supplier Clean Energy Program in 2015 to transition its 'entire manufacturing supply chain —including material extraction, component manufacturing, and final product assembly— to 100% renewable electricity' by 2030 (Apple, 2022b, p. 1). The programme encourages suppliers to commit to 100% renewable electricity for Apple production, with the current number of suppliers standing at 213, in 25 countries. Apple claims that this number corresponds to about 70% of supplier companies and accounts for 98% of Apple's expenses in materials, manufacturing, and assembly of the company's products worldwide (Apple, 2022a, p. 24). These suppliers report to Apple, and the company tracks their progress in renewable electricity procurement (Apple, 2022b, p. 3).

Apple supports suppliers both technically and financially. It created the Supplier Clean Energy Portal, where it offers internal training and resources that are tailored to each supplier's country and connects them to experts and renewable energy industry associations. Where renewable energy procurement options are limited, Apple has devised constructs such as the Supplier Co-Investment Model, in which Apple and suppliers invest in a common fund that is used to create new renewable electricity capacity for suppliers (Apple, 2022b, p. 3). The China Clean Energy Fund, based on this model, has invested in 1 GW of additional renewable capacity in China. As more renewable energy procurement options become available, the company connects suppliers to buy energy from project developers and utilities; and, to support suppliers further upstream, Apple has directly invested in 500 MW of renewable energy in Japan and China. The company reports that its support efforts have achieved a substantial increase in renewable energy capacity in the supply chain, increasing from about 2 GW operational or committed capacity in 2016 to about 10 GW operational and 6 GW committed capacity in 2021, double the capacity of that in 2020 (Apple, 2022b, p. 2).

Apple and its suppliers report that they strive for high-integrity renewable electricity procurement, where investments likely result in truly additional renewable electricity capacity Apple reports that it applies the same integrity criteria to suppliers' renewable energy that it applies to itself. For suppliers' renewable electricity, power-purchase agreements are prioritised and represent 79% of renewable electricity procurement mechanisms, followed by renewable energy certificates (8%), direct investments (10%), and on-site production (3%) (Apple, 2022b, p. 4). The company claims to advocate in suppliers' countries for regulations that create more opportunities for high-quality renewable electricity procurement. Regarding transparency, there is still some room for improvement in Apple's strategy. Although the company shares the list of suppliers that participate in the Supplier Clean Energy Program (Apple, 2022b, p. 6), it does not disclose details on their regional breakdown, their energy consumption, and their emissions, which are necessary to understand the company's progress in achieving its 2030 target (Wu *et al.*, 2022, p. 23).

Climate contributions and offsetting

Corporate climate leadership includes both implementing ambitious targets for emission reductions in the company's own value chain as well as taking responsibility for unabated emissions. Most companies do not have the ability to immediately eliminate their entire GHG emissions footprint. While more and more companies are charting a pathway to complete decarbonisation, it will usually take years or decades until they are able to entirely achieve this goal, even for the most ambitious entities.

For some companies, taking responsibility for unabated emissions means making climate contributions to support climate change mitigation beyond the company's value chain without making a neutralisation claim, while for others it means claiming to neutralise their emissions through the use of carbon dioxide removals or emission reduction offset credits. Some companies pursue both approaches in parallel. This section discusses emerging issues and trends for the credibility of these two approaches.

4.1 Snapshot summary

Table 14: Summary of good practice and trends for climate contributions and offsetting

SUMMARY OF GOOD PRACTICE PRINCIPLES (SECTION 4.2) AND PERFORMANCE FROM THE 24 COMPANIES ASSESSED GOOD PRACTICE FOR CLIMATE CONTRIBUTIONS AND OFFSETTING TRANSPARENCY⁺ Pursue high transparency and integrity on **Responsibility for** Low Very low climate contributions and any neutralisation unabated emissions claims made today (see criteria below). Climate Provide an ambitious volume of financial support to climate change mitigation activities Very low contributions Low beyond the value chain, without claiming to neutralise the company's own emissions. **Offsetting claims** Clearly disclose offsetting claims and plans; Low Very low avoid misleading pledges and claims; avoid today risk of distraction by also committing to measures for deep emission reductions; commit to procure only high-quality credits **Offsetting plans** from ambitious projects with a permanent Low Very low for the future climate impact; and commit to preventing any form of double-counting of climate impacts.

* Transparency and integrity columns: the bar indicates the distribution of our rating of the 24 companies (**Poor Moderate High n.a.**); the text above the shaded bars represents the average rating across all the companies we assessed, calculated excluding non-applicable cases, on a 5-point scale (**Very low, Low, Moderate, Reasonable, High**); and an indication of progress since the last analysis in 2022 (**M V**), based on the authors' interpretation of progress from the companies that were analysed also in 2022, against the current methodology version.

SUMMARY OF TRENDS, ROLE MODELS AND BAD PRACTICE (SECTION 4.3)

Limited traction for climate contributions, but stage is set for progress in 2023: Only four companies report climate contributions without a neutralisation claim but none of these companies frame this support in terms of taking responsibility for their emissions. Significant developments in guidelines and frameworks for the climate contribution model may facilitate progress in 2023 (see Table 15).

Carbon neutrality claims remain highly contentious: At least 11 of the 24 companies are claiming carbon neutrality in 2022, but the transparency and integrity of those claims remain critically low.

Offsetting through forests: 23 out of 24 companies have offsetting plans for the future, mostly though non-permanent removals from biological carbon dioxide removal and storage, at a scale that would outstrip the planet's resource potential if replicated by others.

PROMISING EXAMPLES BAD PRACTICE Transparency on offset projects: Microsoft Misleading coverage of carbon neutrality claims: Companies' carbon publishes details on all of its offset projects in an neutrality claims cover on average just 3% of their emissions. Microsoft, Apple interactive web-page, including volume of credits and Deutsche Post DHL cover less than 2% of their emissions when they procure carbon credits to tell customers that their business or the service they provide is procured and 'contracted durability' for each project. 'carbon neutral'. Shift to high-hanging fruit: Microsoft plans to neutralise its emissions in the future with more Terminology of 'insetting' is gaining traction: Nestlé and PepsiCo are among the companies successfully lobbying to legitimise the contentious practice of what advanced carbon dioxide removal options with a reasonable degree of permanence, although to date they label as "insetting", (see Spotlight section 4.4.4). the company's portfolio is still heavily focused on 'Mitigating' the non-permanence of biological carbon removals: Most biological carbon dioxide removal.

companies rely on the cheap option of non-permanent and scarce forestry-based carbon dioxide removals. Approaches put forward by **Amazon**, among other companies, to mitigate the issue of non-permanence are implausible (see Spotlight section 4.4.2).

SUPPORTING COMPANIES TO IMPROVE THEIR CLIMATE CONTRIBUTIONS AND OFFSETTING CLAIMS

Regulation on carbon neutrality claims is necessary to make those claims less accessible, and to increase their transparency and integrity. A coalition of companies for climate contributions is necessary to create a critical mass for the transition to this more constructive approach.

Table 15 Summary of developments in 2022 related to corporate climate contributions and offsetting



+ Authors' interpretation of companies' progress against the current methodology version since the last analysis in 2022 (

NEW GUIDELINES AND	FRAMEWORKS RELATED TO CORPORATE CLIMATE CONTRIBUTIONS AND OFFSETTING IN 2022
UN High-Level Expert Group (UN HLEG, 2022)	Carbon credits may be used to claim a contribution to climate change mitigation, but should not be used to claim the <i>neutralisation</i> of emissions towards the achievement of specific 1.5°C-aligned emission reduction targets.
	Additionality and permanence are essential for any carbon credits used beyond these 1.5°C-aligned emission reduction targets to claim net-zero emissions.
	Any credit transactions must be transparently reported, including a specification of whether those credits could be double counted towards any country's own climate pledges.
ISO Net Zero guidelines (ISO, 2022b)	Permanence: Carbon credits should reflect emission reductions or removals that guarantee at least an equivalent degree of permanence to the lifetime of the greenhouse gases they are used to claim to offset. This would effectively rule out the use of carbon dioxide removal with biological storage – for example, forestry, soil sequestration and mangrove restoration projects – for claiming to neutralise any CO ₂ emissions.
	Offset projects should come from the high-hanging fruit of mitigation potential: Activities that address urgent and transformational climate priorities that are beyond the reasonable reach of unilateral action by a single country.
The UNFCCC Conference of Parties (COP27) decision	The climate contribution model appears to have been anchored in the Paris Agreement framework by agreements on the international use of carbon markets, which clarified the suggested uses of two <i>types</i> of unit for future transactions:
(UNFCCC, 2022)	Corresponding adjustments with 'authorised A6.4ERs': If host countries <i>authorise</i> Article 6.4 Emission Reduction credits for the international transfer, then the mitigation outcome can be accounted to the GHG account of the buyer, and a 'corresponding adjustment' is required to ensure that the mitigation outcome is not double counted.
	Climate contributions with 'mitigation contribution A6.4ERs': A 'mitigation contribution' unit is not recognised as a sale of an emission reduction by a country, and hence is counted towards the host country's climate target. Using these credits towards any other emissions target would amount to double counting. This decision represents a clear consensus amongst national governments that when emission reductions are counted by a country, they cannot also be claimed by a company in support of an offset claim. Companies may rather claim a <i>contribution</i> to climate action.
Other developments	Guidance on climate contributions demonstrates an emerging consensus on this new direction of travel for corporate climate responsibility, and ongoing debate on how to evaluate their quality (Höglund and Mitchell-Larson, 2022; WWF Germany, 2022; NewClimate Institute, 2023a).
	Legal complaints against carbon neutrality claims were brought against KLM in the Netherlands and SK Lubricants in South Korea, amongst others. Legal experts ClientEarth published an influential briefing (ClientEarth, 2022), outlining the high risks of legal action associated with claims that companies' emissions are offset through carbon credits.
	Carbon credit provider myclimate announced a shift from offsets to a new impact label in the vein of the climate contribution model. myclimate will no longer offer a carbon neutral label based on the recognition that the current market cannot deliver offsets that can credibly facilitate climate neutral claims in the era of the Paris Agreement (myclimate, 2022).

Note: The summaries in this table represent the authors' interpretations of the guidelines and developments presented.

4.2 Principles for good practice

Climate contributions and offsetting remain a hot topic with divergent views and inconsistent guidance. Despite the continued fragmentation of standards and guidance, a number of developments throughout 2022 underpin a strengthening consensus that there is a limited role for offsetting in credible corporate climate strategies. In the following sections we summarise principles for good practice that we have identified for climate contributions and offsetting claims. Full details on the guiding principles of our analysis can be found in section 4 of the methodology document.

4.2.1 Climate contributions without a neutralisation claim

In recognition of the limitations of offsetting and the need to ramp up financial support for climate action worldwide, some actors – including companies, standard setting initiatives and even providers of carbon offsets – are moving away from the offsetting model to make a climate contribution without any neutralisation claim.

We define climate contributions as finance provided by a company to support climate change action beyond the company's own value chain, without claiming to neutralise its own emissions. A company can claim to contribute to climate change mitigation activities, without claiming ownership of the emission reduction outcomes and without subtracting associated reductions from their own GHG inventory or net-zero target. Climate contributions, which represent an alternative approach to offsetting, are a central feature of NewClimate Institute's Climate Responsibility approach (NewClimate Institute, 2020) and the WWF-BCG Climate Blueprint (WWF and BCG, 2020). The Climate contribution model has been anchored in the framework of the Paris Agreement by the COP27 decision to create a type of unit under Article 6.4 that can be used only for this purpose (a 'Mitigation contribution A6.4ER', see section 4.3). Some established providers of offsets and carbon neutrality labels, including myclimate, are transitioning to this alternative model (see section 4.3).

An internal carbon price on emissions can inform the volume of financial support. This way, climate contributions are linked to a company's responsibility for its own unabated emissions. The volume of financial contributions can serve as a key indicator of climate leadership. Ambitious companies should use the proceeds of an internal carbon price across all scope 1, 2 and 3 emissions, that is set at a high enough level to send a clear incentive signal for embarking on a 1.5° C-compatible decarbonisation trajectory (e.g., usually at least USD 80/tCO₂e, unless otherwise demonstrated by the company; see the accompanying Methodology section 4.2.1).

Companies can channel their climate contributions towards a wide range of activities. Since they are not claiming

to neutralise their emissions, companies making climate contributions are not tied to procuring carbon offset credits and enjoy far greater flexibility in the type of activities they can support to advance global decarbonisation. This could include, for example, support for carbon removals that do not offer sufficient guarantees of permanence to neutralise emissions (see Box 3 section 4.2.2), but which are critical to addressing climate change and require more financial support globally. Other examples include emerging technologies and measures for sectors where the mitigation potential of existing technologies is limited, and where innovation and investment are needed to find new solutions. Uncertainties regarding the specific impacts delivered by more immature technologies and higher-risk investments may make them less attractive to project developers looking to generate offset credits, but a more suitable avenue for those channelling financial support in the form of climate contributions.

Climate contributions without neutralisation claims can provide a transparent, constructive and ambitious approach to take responsibility for unabated emissions:

- More transparent: Targets that are formulated independently from offsetting, without any netting-out of actual climate impacts, are more transparent and provide a clearer signal to decarbonise the company's own value chain.
- More constructive: Developing countries need more financial support to ramp up their mitigation action; voluntary action from companies is a vital channel of such support. A constructive environment is required, where this finance positively reinforces ambition raising, rather than one that provides perverse incentives to limit the ratcheting up of national climate commitments. In contrast to offsetting approaches, if the financial support from voluntary action results in emission reductions that are owned by the actors supported and the host country they operate in, this action will not conflict with the host country's GHG emission reduction target. Instead, it can provide support for reaching and ratcheting up those targets.
- More ambitious: The climate contribution model is aligned with the concept of ratcheting ambition through a race to the top, a concept that underpins the Paris Agreement. If companies are free to self-determine their own ambition for their climate contributions – as countries do through Nationally Determined Contributions – this may result in a race to the top to demonstrate the highest ambition, without limits. This would mark a significant shift from the offsetting approach in which many companies race to the bottom and exploit loopholes to deliver a fixed target at the lowest cost.

Despite these potential advantages, there are still open issues to address with the climate contribution model to ensure that the approach can lead to high quality action. The increased flexibility regarding the types of projects that can be supported under this model can be beneficial for supporting carbon dioxide removals or emerging technologies, but it will also be a challenge to ensure that this flexibility is not used to pursue lower quality projects. In this regard, their remains a significant role for existing market players and standard setting initiatives to contribute to the discussion and tools available for quality assurance.

Companies should disclose details on their climate contributions, including the basis for determining the volume of their financial contributions, the amount that they contribute each year, the recipients and the anticipated or measured impacts. It is critical that communication around these climate contributions avoids any implication that they serve to offset the actual emissions of the company.

4.2.2 Offsetting claims

Companies make an offsetting claim when they assert that GHG emissions within their value chain are 'neutralised', 'nettedout', 'offset', 'inset' or 'counterbalanced' through other emission reduction activities or carbon dioxide removals – inside or outside of their value chain. The practice of claiming to offset emissions is afflicted by controversy and contention due to significant uncertainties in the real impact of offset credit use as well as the suitability of carbon dioxide removals for offsetting emissions. Accordingly, terminology for claiming the offsetting of emissions is highly sensitive and inconsistent. Many actors now avoid the term offsetting entirely; companies and initiatives more often refer to 'neutralisation', 'netting-out', 'compensation', 'reducing the footprint', 'counterbalancing', or other equivalent terminologies. 'Insetting' is also gaining traction as a term to claim the offsetting and *climate positive really mean*?). Some standards and companies propose the use of multiple terminologies to distinguish between offsetting in different circumstances and at different times. We consider that the complication of this single concept creates additional and unnecessary confusion which may detract from the ability of consumers, investors and regulators to critically assess claims made by companies. The *Corporate Climate Responsibility Monitor* assesses all claims that unabated GHG emissions within the value chain are offset as *offsetting claims*, including all synonymous terminologies and project types.

The integrity of an offsetting claim has always depended on various factors, including but not limited to additionality, permanence, avoidance of double counting, avoidance of leakage, and the accuracy of quantified impacts (Carbon Credit Quality Initiative, 2021). In addition to these long-established principles, several new factors are now of key importance to the integrity of an offsetting claim, since the coming into force of the Paris Agreement:

• Avoiding double counting (Methodology Version 3.0 Section 4.3.2.3):

The use of authorised emission reduction credits – or other mechanisms to avoid double counting – alone, does not guarantee the environmental integrity of an offset credit, but is a minimum requirement to uphold integrity. In the era of the Paris Agreement, it is a minimum requirement for offsetting claims that companies procure only carbon credits that are 'authorised' for the international transfer of mitigation outcomes by the governments of the host country. After such an authorised transfer, governments will apply 'corresponding adjustments' to their reporting of progress towards achieving their national targets, to avoid that mitigation outcomes are double counted when the credits are used by another party (e.g. a company) to substantiate an offset claim.

• Avoiding the risk of distraction and delay (Methodology Version 3.0 Section 4.3.2.1):

Offsetting claims must be limited and transparent enough to ensure that they do not distract from the necessity of prioritising immediate emission reductions. To maintain a chance of meeting the 1.5°C temperature limit, all sectors need to embark now on deep decarbonisation trajectories to reach net-zero GHG emissions and eventually net-negative GHG emissions worldwide (IPCC, 2022). In this ever more urgent context, offsetting claims risk distracting from immediate emission reduction measures. If consumers, investors and regulators are led to believe that a company's emissions are lower than they really are, this may lead to a reduction in the extent to which these actors provide further pressure, incentives or support for necessary emission reductions. The relevance of this issue is independent of the quality of the credits used to offset the emissions. To avoid this risk, carbon neutrality and net-zero claims should not rely on offsets for a substantial proportion of the company's emissions: the ISO Net Zero Guidelines recommends that offsets towards net-zero targets should only be used to account for 'residual emissions', which it defines and illustrates as accounting for 0-5% of the 2019 emissions from most sectors (ISO, 2022b). We consider that the same logic applies to other synonymous terminologies, including carbon neutrality claims. Alternatively, this risk may be less relevant, companies communicate targets and emission reduction measures that are clearly sufficient for sector-specific 1.5°C-compatible decarbonisation trajectories.

Additionality in the context of safeguarding Paris ambition (Methodology Version 3.0 Section 4.3.2.2):

Under the global governance framework of the Paris Agreement, offset credits can only provide an appropriate guarantee of additionality if they are generated from high-hanging-fruit mitigation projects, that do not present credit-selling territories with a perverse incentive to limit the extent to which they ratchet up their own ambition during NDC revision cycles (see *Spotlight 4.4.1:* How to assure offset additionality in the context of the Paris Agreement?). This position is aligned with the requirements of the 2022 ISO Guidelines for Net Zero (ISO, 2022b).

• Net-zero compatibility (Methodology Version 3.0 Section 4.3.2.4):

52

To support the objectives of the Paris Agreement, financial support must be channelled to the identification and scaling of long-term solutions. Investments in bridging technologies that represent marginal emission reductions, but which are not compatible with zero-emission technologies, may result in locking-in high carbon infrastructure, and can further delay investment in the cleanest technologies. For sectors that should be fully decarbonised before 2050, the supported technologies and measures must be compatible with a zero-emission sector at the earliest possible point in time. For harderto-abate sectors, the supported technologies should be compatible with other best available or emerging decarbonisation technologies within those sectors.

• Permanence and scarcity of carbon dioxide removals (Methodology Version 3.0 Section 4.3.3; summarised in Box 3): It may be more appropriate for corporates to channel support for carbon dioxide removals through climate contributions without neutralisation claims. It can be good practice for companies to support the development of carbon dioxide removal projects inside or outside their value chain, in parallel to delivering actual emission reductions. It *could in theory* be credible for companies to claim to offset their emissions with carbon dioxide removals under the specific conditions that they only offset residual emissions from highly decarbonised sectors that have no technical prospects for the complete elimination of emissions, and with carbon dioxide removals that have a high likelihood of sufficient permanence. In practice, it is politically challenging to reach a consensus on which sectors these are, and there is very limited availability of carbon dioxide removal projects that can deliver the required guarantee of permanence. Scarce potential and environmental damages mean that CDR measures are not a credible means of offsetting emissions that could feasibly be reduced, while CDR measures that cannot guarantee permanence are not a credible means of offsetting any emissions at all (see **Spotlight 4.4.2**: *Is it possible to mitigate the non-permanence of climate impacts from biological carbon dioxide removals*?). Given these significant limitations, it may be more appropriate for corporates to set separate targets for emission reductions and carbon dioxide removals, and to channel support for carbon dioxide removals through climate contributions without neutralisation claims.



The suitability of carbon dioxide removals for offsetting claims

For further details see the accompanying methodology document, Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0, section 4.3.3.

All scenarios consistent with limiting global warming to a maximum of 1.5°C include a major role for carbon dioxide removals (Rogelj *et al.*, 2018). This includes nature-based solutions for carbon sequestration in forests, soils, peatlands and mangroves, technological solutions such as bioenergy with carbon capture and storage (BECCS) and direct air carbon capture with storage (DACCS), and solutions with mineral storage. Finance is needed to scale up carbon dioxide removal efforts, and corporates could play a key role.

Conditions for credible offsetting with carbon dioxide removals

Credible offsetting of individual companies' GHG emissions through financing carbon dioxide removal initiatives must focus on storage options that provide a sufficient guarantee of **permanence** and are not significantly constrained by technical or physical limitations on the storage **potential**.

CDR permanence: The permanence of carbon-dioxide removals must be guaranteed over a timeframe of centuries to millenniums. The *permanence* of a CDR outcome refers to the degree of certainty that the sequestered carbon will not be released at a later point in time. The release of previously sequestered carbon negates any accrued benefits of the sequestration. A sufficient guarantee of permanence requires a high likelihood that the captured carbon will remain stored over a timeframe of centuries to millenniums. Significant reliance on measures that have a reasonable likelihood of releasing captured carbon over a timeframe of decades present a risk of materially increasing atmospheric carbon concentrations either this century or in the next.

Scarcity of CDR potential: Scarce carbon dioxide removal potential must be reserved for balancing out residual emissions in hard-toabate sectors, for it to remain technically possible to achieve global net-zero emissions. The maximum potential of most carbon dioxide removal measures is technically limited, and further restricted by environmental constraints, such as land requirements, high water consumption, high energy consumption, land degradation and pollution. Many carbon dioxide removal technologies can only be scaled up so far without significantly endangering sustainable development goals including food security. The scarcity of carbon dioxide removal measures is an important consideration when evaluating net-zero claims at the level of individual actors. The use of scarce carbon dioxide removal options must be consistent with achieving net-zero and eventually net-negative emissions at the global level. To align with 1.5°C-compatible pathways at the global level, some sectors with the technical ability to fully decarbonise will need to reach zero emissions, while carbon dioxide removals are likely needed to balance out the residual emissions from other hard-to-abate sectors. Any allocation of rights of ownership to scarce carbon dioxide removals will require international oversight as well as detailed (and likely highly complex) considerations of fairness and appropriate use to ensure efficient and effective efforts to contain and then reduce the atmospheric stock of emissions. Accordingly, it is not appropriate for companies today to make climate pledges, assuming they will have the right to use scarce carbon dioxide removals to neutralise their own emissions decades in the future. If specific companies – for example in the energy industries – claim ownership of scarce carbon dioxide removals now or for a time in the future, then it will not be possible for those removals to balance out residual emissions in hard-to-abate sectors, and it will not be possible to reach net-zero emissions at the economy-

Based on these issues, we conclude that it *could in theory* be credible for companies to claim to offset their emissions with carbon dioxide removals under the specific conditions that they only offset residual emissions from highly decarbonised sectors that have no technical prospects for the complete elimination of emissions, and with carbon dioxide removals that have a high likelihood of sufficient permanence. In practice, it is politically challenging to reach a consensus on which sectors these are, and there is very limited availability of carbon dioxide removal projects that can deliver the required guarantee of permanence. Scarce potential and environmental damages mean that CDR measures cannot be considered a credible offset of emissions that could be feasibly reduced, while CDR measures that cannot guarantee permanence are not a credible means of offsetting any emissions at all. It may be more appropriate for corporates to channel support for carbon dioxide removals through climate contributions without neutralisation claims.

Assessment of specific CDR measures and technologies (according to current best available information)

CDR measures based on biological capture and storage are scarce relative to potential neutralisation needs and do not have the necessary degree of permanence to be credibly considered an equivalent to emission reductions. These measures are also vulnerable to the displacement of emissions to other locations.

For BECCS and DACCS with underground storage, long-term storage is possible, although uncertainty on the risk of leakage remains. The scarce potential of these measures, which may be constrained by environmental concerns and energy system inefficiencies, mean that these measures are not a reasonable equivalent alternative to emission reductions for emissions that could be feasibly reduced.

CDR measures with mineral storage have a reasonable likelihood to meet the criteria of permanence and additional potential to be potentially considered a credible offset of residual emissions from hard-to-abate emission sources. Uncertainties on the environmental limitations mean that the credibility of offset claims for other emissions that could be feasibly reduced remains contentious.

The *Corporate Climate Responsibility Monitor* evaluates the integrity of offsetting claims using the same criteria, regardless of whether the company claims to offset its emissions today or at a point in the future.

It is unlikely that an offsetting claim today can deliver on the criteria necessary for that claim to be credible. These limitations are an important reality, rather than a reason to identify more lenient rules for offsetting claims today. Existing offset market conditions make it far more difficult – potentially unrealistic – for companies to make offsetting claims that can be assessed as having high integrity today. The integrity of offsetting claims today is first and foremost hampered by the reality that there are currently no offset credits available from any markets that can meet all the aforementioned criteria for robust integrity. Although the Paris Agreement is already in force, an accounting mechanism for avoiding **double counting** is yet to be established under any international offsetting standard, though this will be possible through the procurement of authorised A6.4ER credits¹ in the future. There are also currently very few – if any – examples of existing offsetting projects that represent the high-hanging fruit of mitigation potential that can be considered **additional in the context of the Paris Agreement**, given that offsetting markets to date have mainly focused on reaching the most cost-effective mitigation potential. The simple *inability* of the current market to supply offset credits that can credibly underpin carbon neutrality claims was given by myclimate as the reason for their decision to discontinue their carbon neutrality label and transition to an impact label in the vein of a climate contribution (myclimate, 2022).

Credible offsetting plans for the future depend on solid mechanisms and accounting frameworks. Companies can already be transparent about their intentions and how they plan to navigate the many issues that affect the integrity of offsetting claims. Following from the limitations for offsetting claims today, the ability to follow through on high integrity plans for future offsetting will likely depend on the transformation of existing offsetting markets, or the development of new mechanisms that can serve the criteria for credible neutralisation claims. Companies planning to offset their emissions in the future may not be able to identify specific projects today, but they can make an explicit statement of intent to restrict offsetting activity to high-hanging fruit projects with corresponding adjustments, along with other necessary conditions for environmental integrity.



1 A6.4ER credits refer to authorised emission reduction credits established under Article 6.4 of the Paris Agreement.

4.2.3 Good practice assessment criteria

The criteria for good practice in Table 16 forms the basis for the company assessments in section B. Full details on the methodology for rating companies' climate contributions and offsetting claims can be found in the accompanying methodology document, *Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0, section 4.*

Table 16: Good practice for climate contributions and offsetting claims

CLIMATE CONTRIBUTIONS AND CORPORATE CLIMATE LEADERS EXHIB	
C. Responsibility for unabated emissions	 Pursue either a robust approach for climate contributions or neutralisation of emissions, to take adequate responsibility for unabated emissions today.
Climate contributions	 Provide support to projects for climate change mitigation beyond their value chains. Derive the volume of finance from an internal carbon price across all emissions at a price level commensurate with keeping global temperature rise below 1.5°C above pre-industrialised levels. Disclose full details on the volume of finance, the project recipients, and the expected impact. Claim only to make a contribution, without claiming ownership of the reductions for the neutralisation of emissions.
Offsetting claims today	 Ensure that any carbon neutrality claims do not mislead, or distract from the need for emission reductions through the following means: The company presents transparent disclaimers on the dependence on offsetting, and the inherent uncertainties that this entails, alongside carbon neutrality claims. Any carbon neutrality claims apply to all scope 1, 2 and 3 emissions. Carbon neutrality claims involve the offsetting of only residual emissions as defined by the ISO (0-5% of 2019 emissions from most sectors ISO, 2022b); or the company also has 1.5°C-aligned targets for the short-, medium- and long-term. Disclose details on all offsetting activities including the volume of offset credits, details of the projects supported, credit vintages and credit prices paid. Pursue only high-quality offsetting constructs through the following means: Measures are in place to guarantee that the mitigation outcome cannot be double counted (for example through corresponding adjustments). Projects are compatible with net-zero emission technology and infrastructure Carbon dioxide removals measures have a high likelihood of high permanence, the means of storage is not "scarce" and not associated with high environmental costs.
D. Offsetting plans for the future	 Transparently disclose whether or not the company plans to offset any of its emissions in the future, for instance in its target year. If planning to make offsetting claims, the company demonstrates plans and a commitment to meet the aforementioned good practice criteria for offsetting claims.

4.3 Trends, role models, and bad practice

Developments in 2022 charted a clearer path for a transition from offsetting to a climate contribution approach. Misleading offsetting claims are increasingly recognised as a reputational and legal liability. The climate contribution approach gained traction towards wider implementation through developments such as the COP27 decision to create a 'mitigation contribution' unit under Article 6.4, and the announcement by the prominent business consultancy *myclimate* that it will discontinue its climate neutrality label and transition to a climate contribution model.

But companies' climate strategies appear to be behind the curve of these developments. We did not identify any significant progress over the past year on the low level of transparency and integrity of offsetting claims and climate contributions amongst the 24 companies that we assessed, nor did we identify significant momentum for companies to transition to a climate contribution approach. Over reliance on biological carbon dioxide removals, such as forestry projects, remains a critical issue, while the practice of offsetting under the contentious terminology of 'insetting' is gaining momentum and undermining more companies' climate strategies.

Especially critical emerging issues are explored in section 4.4 Spotlight on key questions.

Developments in 2022 charted a clearer path for a transition from offsetting to a climate contribution approach, but companies' climate strategies are behind the curve of developments.

The COP 27 decision to create a 'mitigation contribution' unit under Article 6.4 is a key step towards the wider implementation of the climate contribution model. The UNFCCC Conference of Parties (COP 27) in November 2022 resulted in new agreements on the international use of carbon markets (Article 6 under the Paris Agreement), which clarified the suggested use of two types of unit for future transactions in voluntary and regulatory carbon markets and has anchored the climate contribution model in the rulebook of the Paris Agreement:

- **Corresponding adjustments with 'authorised A6.4ERs' (the offsetting model which limits double counting risks):** Host countries can *authorise* Article 6.4 Emission Reduction credits (authorised A6.4ERs) for the international transfer of mitigation outcomes to the buyer. This means that a mitigation outcome is credited to the GHG account of the buying party and cannot also be counted towards the targets of the host country. A 'corresponding adjustment' is required to ensure that the mitigation outcome is not attributed to two entities.
- Climate contributions with 'mitigation contribution A6.4ERs' (the contribution model without offsetting): This 'mitigation contribution' unit is not directly recognised as a sale of an emission reduction by a country, and hence not accounted for as such by the country. These credits represent emission reductions (or removals) that are counted towards the host country's climate target. The UNFCCC agreement specifies possible uses for such credits, such as results-based finance contributions, and sends a signal that using these credits towards any other emissions target would amount to double counting. This decision represents a clear international consensus amongst national governments that when emission reductions are counted by a country they cannot also be claimed by a company in support of an offset claim. Companies could frame their purchase of these credits as contributions to domestic mitigation, but must not make misleading offset claims. Through the creation of this type of unit, the concept of the 'climate contribution' appears to now be anchored in the rulebook of the Paris Agreement. This sends another signal to companies to start shifting from an offsetting paradigm to one of financing emissions reductions through a climate contribution model.

A transition from offsetting to a climate contribution approach also gained significant traction throughout 2022, with an increasing volume of emerging literature and guidance on its implementation (Höglund and Mitchell-Larson, 2022; WWF Germany, 2022; NewClimate Institute, 2023a). These guidance documents show that while there remains an ongoing debate on how to evaluate the quality and sufficiency of climate contributions, there is an emerging consensus on this new direction of travel for corporate climate responsibility.

Complaints against carbon neutrality claims also gained momentum throughout 2022. Following on from the 2021 decision of the Netherlands' Advertising Code Committee that Shell should cease its "Drive CO_2 neutral" advertising campaigns, legal cases were brought in 2022 against KLM in the Netherlands and SK Lubricants in South Korea, amongst others. Legal experts ClientEarth published an influential briefing on the legal risk of carbon offsets in September 2022 (ClientEarth, 2022), showing that companies are at higher risk of legal action if they claim to compensate their emissions through offsets, rather than pursuing a contribution approach.

Myclimate appears set to be a 'first mover' in

implementation of the climate contribution model. The business consultancy – which has been an internationally recognised provider of offsets and carbon neutrality labels – announced in December 2022 that it will discontinue its climate neutrality label and transition to a new impact label in the vein of the climate contribution model. myclimate will no longer offer a carbon neutral label based on the explicit recognition that the current market cannot deliver offset credits that can credibly facilitate climate neutral claims in the era of the Paris Agreement (myclimate, 2022). Companies' climate strategies are behind the curve of these developments. Just four of the 24 companies we assessed reported activities or donations that could be interpreted as climate contributions without a neutralisation claim. Google makes significant donations to various climate-related projects through its foundation Google.org, while JBS, Stellantis and Walmart report activities or funds to support climate and biodiversity projects. None of these companies link these support activities to their emission footprints or frame the activities in the context of assuming responsibility for their unabated emissions. We found no indication that the climate contribution approach gained any significant traction among the sampled companies over the past year, since the publication of the 2022 Corporate Climate Responsibility Monitor. Aside from the companies sampled in this analysis, a small number of organisations are announcing plans to pursue this approach, including the Dutch multinational chemical company DSM (Mooldijk et al., 2022).

In 2023, companies looking to make climate contributions now have more guidance available and can look to a growing number of examples and service providers (see *Spotlight 4.4.5:* Are companies ready for a shift to climate contributions?).

Many companies are claiming carbon neutrality in 2022, but the transparency and integrity of those claims remains critically low.

At least 11 of the 24 companies assessed claimed in 2022 to have offset their emissions. This is slightly higher than for the proportion of companies sampled in the *2022 Corporate Climate Responsibility Monitor*. Including retailers that sell carbon neutral products, at least 14 of the 24 companies are presenting customers with some form of carbon neutrality claim. Our sample appears to indicate a clear trend that carbon neutrality claims are associated primarily with consumer-facing brands: of the companies we assessed, *all* of the technology and automobile manufacturing companies make some form of carbon neutrality claim, while very few of the heavy industrial companies make any form of carbon neutrality claim.

The degree of transparency in these carbon neutrality claims remains poor in most cases. We did not identify any improvement in the transparency of offsetting claims over the past year. Companies often do not report on details that are relevant for understanding the credibility or even the meaning of the claim, such as its scope coverage, or the precise source (project, vintage, etc.) of the credits on which the offsetting claim is based. **Microsoft** – whose overall climate strategy we found to be reasonably transparent (see Microsoft in Section B) – stands out in the level of transparency that it provides on its offsetting transactions. The company provides a webpage to browse the details of its various carbon crediting projects, including information not only on the projects but also the average prices paid for credits and the 'contracted durability' – or permanence – of the climate impact of each project. The level of detail provided is concise and in a basic format that could very easily be replicated by other companies; there is no clear barrier to the provision of these basic details that facilitate an understanding of a claim's integrity.

The very low degree of integrity in companies' carbon neutrality claims also did not improve over the past year. We found poor integrity of the claims for every one of the companies that we assessed. The integrity of claims is compromised by various issues, including the scope coverage, and the quality of the carbon credits procured, as discussed below.

Carbon neutrality claims cover just a small fraction of the companies' emission footprints.

Of the 11 companies that claim climate neutrality in 2022 in some form, these claims were found to be misleading in their coverage, covering only selected products, services or emission scopes. Of those 11 companies, the average company's carbon neutrality claim covered 3% of their emission footprint, although observers could be misled into understanding that the company is claiming carbon neutrality across its full emission footprint. None of these carbon neutrality claims covered more than 12% of a company's emission footprint:

- Some offsetting claims cover only operational emissions (scope 1 and 2), which usually account for a small minority of a company's overall emissions. Google, Microsoft and Apple claim to be carbon neutral already for several years, while in reality only procuring credits to account for 12%, 2% and less than 1% of their emission footprints, respectively (see Spotlight 4.4.3: Is the tech industry really carbon neutral?).
- Some offsetting claims cover only specific divisions or products, where the demarcation of relevant emission scopes is very unclear. Deutsche Post DHL claims 'climate neutral' deliveries, but procures only enough offset credits to cover transportrelated emissions in specific regions, accounting for less than 1% of the company's full emission footprint. Likewise, all of the automobile manufacturers assessed - Volkswagen, Mercedes-Benz and Stellantis - claim carbon neutrality for a selection of their production lines, portraying a potentially misleading impression to customers about the climate impact of their brands.
- Some companies advertise optional carbon neutral products and services, while in reality this means procuring credits to offset the emissions of the same products, and only for customers who choose to pay a premium. Holcim offers carbon neutral variations of its products under its 'EcoPact Zero' label, while we understand that the only difference from the normal product offering is the acquisition of carbon credits. American Airlines offers the option to customers to offset the emissions associated with their flight, although this covers only the seat of the individual passenger, and does not cover the impacts of non-GHG climate forcers, which account for the majority of the flight's climate impact.

Additionality and permanence of offset credits used is highly contentious

Forestry-related projects account for most offset credit procurement, despite the fundamental unsuitability of these projects for offsetting claims. Most of the companies with carbon neutrality claims do not provide clear details on the projects that they procure offset credits from. However, the information provided is sufficient to understand that at least half of the companies are procuring credits from forestry-related products. This is consistent with research (Donofrio et al., 2022), which found that the volume of forestry and land-use related offset credits transactions quadrupled between 2020 and 2021, representing nearly half of the overall volume of offset credit transactions on the voluntary market in 2021. Offset projects based on the biological storage of carbon - including forestry-related projects are fundamentally unsuitable for offsetting claims due to the non-permanence of the climate impact (see Section 4.2.2 Box 3: The suitability of carbon dioxide removals for offsetting claims and Spotlight 4.4.2: Is it possible to mitigate the non-permanence of biological carbon dioxide removals?).

The additionality of offset projects is highly contentious, for all of the companies that provide information on their offset projects. None of the companies assessed reported the procurement of offset credits from projects that represent the high-hanging fruit of mitigation potential, while some of the projects even compare unfavourably to historic Kyoto-era definitions of additionality. Google explicitly recognises that some of its landfill gas projects may not be strictly additional (see Spotlight 4.4.1: How to assure offset additionality in the context of the Paris Agreement?).

> • .

> > .

.

. .

. .

.

. . . .

•

Offsetting under the guise of 'insetting' is gaining traction, although this practice leads to low credibility GHG emission offsetting claims and the double counting of emission reductions.

Many companies appear to be distancing themselves from the terminology of 'offsetting', a term that is increasingly laden with controversy and liability. Across the 24 companies assessed in this study we found that most companies continue to rely on the practice of offsetting under the guise of other terminologies such as neutralisation, balancing out, netting out, compensating, and the especially problematic terminology 'insetting'.

'Insetting' is a business-driven concept with no universally accepted definition. The examples we have identified of insetting plans all amount to the offsetting of emissions without robust methodologies and the verification steps required from voluntary carbon crediting standards. We highlighted 'insetting' in the 2022 Corporate Climate Responsibility Monitor (Day et al., 2022) as an illegitimate concept with the potential to significantly undermine corporate climate strategies. Since then, the concept has gained considerable traction with increasing evidence on how it is already undermining climate strategies. Deutsche Post DHL and PepsiCo employ the label of insetting to illegitimately claim that their emissions have been offset. Socalled "insetting" also forms a significant component of the medium- and long-term pledges of Nestlé, PepsiCo and JBS, who distance themselves from offsetting but actually plan to offset their emissions through non-permanent land-related carbon dioxide removals under the guise of 'insetting' (see Spotlight 4.4.4: What does insetting really mean?).

Companies are successfully advocating for 'insetting' to be recognised as an appropriate approach; a development with potentially damaging consequences for the credibility of corporate climate strategies. While Nestlé implicitly recognised in their 2021 Net Zero Roadmap that 'insetting' was not yet seen as a legitimate approach, the likes of **Nestlé** and **PepsiCo** have used the prominent roles they hold on advisory committees and the technical working groups of key standard setting initiatives to lobby for the concept. They can now refer to the 2022-published SBTi guidance for forestry, land use and agriculture (FLAG) as evidence of a certain recognition of the terminology. The new SBTi FLAG guidance allows companies to claim the achievement of their emission reduction targets through 'insetting', albeit without clarifying what the term means, nor a method for measuring and independently validating the legitimacy of any claims. This breaks from the long-held SBTi position that emission reduction targets should only be achieved through delivering actual emission reductions. This may have damaging consequences for the credibility of climate targets from companies with significant land use emissions, including not only agri-businesses but also retailers and the fashion industry.

Companies demonstrate excessive reliance on carbon dioxide removals to offset their emissions for net-zero targets.

All but one of the 24 companies assessed have plans to offset their emissions in the future. The only company not to do so – **Walmart** – does not currently set a long-term target for a reduction of its scope 3 emissions.

The role of offsets for achieving long-term net zero targets could fall between 23 and 45% of the companies' combined 2019 emission footprints. This broad range reflects the lack of transparency provided by companies on the role of offsets in the future plans: for the ten companies that are transparent about what their targets mean in terms of emission reductions and offsets, we estimate the reliance on offsetting to be equivalent to 23% of their 2019 emissions, on average. Across all 24 companies, including the companies for which the meaning of their net-zero targets remains ambiguous, we find that the role for offsets could be anywhere in the range from 23% up to the 45% of 2019 emissions that remain unaccounted for in companies' net-zero targets. Nineteen percent of the companies' combined emissions are explicitly excluded from their targets' coverage, while the commitment to the reduction of emissions amounts to just 36% of their combined current GHG emissions (see section 2).

This 23-45% is far in excess of the maximum role for offsetting indicated by SBTi's Net Zero Standard, which stipulates companies' net-zero targets should set out at least a 90% reduction of 2019 emissions, with the role of offsetting limited to a maximum of 10%, and even less in some sectors (SBTi, 2021c). The ISO Guidelines for Net Zero also recommend that offsetting claims should only account for residual emissions which – for most of the sectors indicated in the guidelines – should constitute less than 5% of 2019 emission levels (ISO, 2022b).

Planned reliance on nature-based solutions outstrips the technical potential of the planet's nature resources. Of the companies that offer any information about their offsetting plans for the future, we interpret that at least three-quarters of these companies rely heavily on forestry and land-use related carbon dioxide removal offsets. If three-quarters of the companies follow through on plans to procure offsets for 23-45% of their 2019 emission footprints, this equates to an annual demand of 380-740 MtCO₂e worth of forestry and land-use related carbon dioxide removals offsets. For just 24 companies, which accounted for 4% of global emissions in 2021, this potential offset demand represents a very high proportion of the global environmentally constrained potential for biological carbon dioxide removal and storage measures, estimated at 1.7-6.7 GtCO₂e per year (Fuss *et al.*, 2018; Hepburn *et al.*, 2019 see Methodology Version 3.0 Table 12). Extrapolating the trend of this sample of 24 leading companies to the global scale implies an annual demand for forestry-related carbon dioxide removals of 9.5-18.5 GtCO₂e. This exceeds the mid-point of the estimated global resource potential by at least 2-4 times.

Moreover, these plans demonstrate the widespread lack of awareness that the biological storage of carbon is fundamentally unsuitable for offsetting claims because the climate impact is not permanent. Companies are increasingly proposing plans to overcome the issue of non-permanence, but these plans are implausible (see *Spotlight 4.4.2: Is it possible to mitigate the non-permanence of climate impacts from biological carbon dioxide removal?*).

60

4.4 Spotlight on key questions (commentary)

This spotlight section presents a commentary from the authors on key issues related to current trends for climate contributions and offsetting.

4.4.1 How to assure offset additionality in the context of the Paris Agreement?

Analysis in this section on the offsetting activities of Deutsche Post DHL and Google are adapted from the Corporate Climate Responsibility Monitor 2022 (Day et al., 2022).

Additionality needs redefining in the context of the Paris Agreement to focus on the high-hanging fruit of mitigation projects. We do not consider any of the assessed companies' offset projects as additional in this context. Some of them cannot even be considered additional under old Kyoto-era definitions of additionality.

A key condition for determining the integrity of offset credits is the *additionality* of the emission reductions; that is, the guarantee that credited emission reductions are additional to what could be achieved without the incentives provided by the potential sale of carbon credits. In historical offsetting mechanisms, additionality could be proven by showing that local legislation did not require the activity and that credit revenues could help overcome barriers which would otherwise prevent implementation.

Since the Paris Agreement came into force, the concept of additionality needs to be redefined. The global governance framework of the Paris Agreement represents a different context from the Kyoto-era, under which most existing offsetting mechanisms and standards were developed.

The prospect of offset credit revenue may present a perverse incentive for countries to limit their climate change mitigation ambition. To overcome this potential ambition pitfall, offsetting projects should be sufficiently ambitious that they avoid presenting any conflict with the host country's own ambition.

The high-hanging fruit of mitigation potential refers to the technologies and measures to decarbonise emission sources that remain otherwise unambiguously inaccessible to host country governments in the near- and mid-term future, on account of high adoption costs. Examples of potential projects that have been conceptualised to fulfil this criteria include ground source heat pumps in Mongolia (Nascimento et al., 2020) and near zero energy housing in Colombia (Kachi et al., 2020). The marginal abatement costs of such projects may be in excess of EUR 100 / tCO₂e, demonstrating a significant shift from existing voluntary offsetting markets; the average price of voluntary market credit transactions between 2020 and 2021 was less than EUR 10 for all project types (Donofrio et al., 2022). Further analysis on high hanging fruit projects is currently being undertaken for publication in 2023 (NewClimate Institute, 2023c).

None of the offset projects reported by companies represent the high-hanging fruit of mitigation potential. A shift to high-hanging fruit carbon crediting projects marks a significant transition from historical practices. Emission reduction projects registered under crediting programmes to date have been mostly developed in the context of costsaving, rather than ambition-raising mechanisms. As such there are very few, if any, examples of existing credited projects that represent high-hanging fruits, and which could be considered truly in line with safeguarding and raising ambition in the context of the Paris Agreement.

On the whole, the companies assessed in this report did not provide extensive details about the offsetting projects from which they procure credits. Of the information that exists, companies had procured credits from forestryrelated projects, renewable energy, landfill gas projects and household energy efficiency projects.

Deutsche Post DHL has looked for some of the higher quality credits available on the voluntary carbon markets. Based on its published quality criteria, the company sourced Gold Standard verified offset credits in 2020 from wind power projects in Aruba and in India, cookstove projects in Guatemala and Lesotho, a landfill gas project in Chile, and a biogas programme in Vietnam (Deutsche Post DHL, 2021b). Unlike the majority of projects that supply current voluntary carbon markets, there is a fair chance that some of these seven projects depend on revenue for continuation and that the support provided by Deutsche Post DHL may lead to some additional climate impact. However, in the context global governance framework of the Paris Agreement, which requires all countries to set ambitious emission reduction targets, the additionality of these types of projects is contentious, as they may be accessible to host countries and could be part of host countries' own GHG emission abatement efforts, if the country were to raise their own ambition, either unilaterally or through support. As such, the projects are not a credible equivalent to the reduction of Deutsche Post DHL's own emissions. The same is true for the eight projects that Deutsche Post DHL selected in early 2022 (Deutsche Post DHL, 2021b).

American Airlines purchases carbon offset credits from three programmes: protecting and restoring forests in Mexico, peat swamps in Indonesia, and building improved cookstoves in Honduras (American Airlines, 2022c). None of these projects can be considered unequivocally additional in the context of the Paris Agreement.

Microsoft announced that it will move away from using carbon credits based on reduced emissions – which it recognises are not aligned with the goals set out in the Paris Agreement (Smith, 2020, p. 8) – and aims to increase its share of what it terms 'high durability' carbon removal projects. Further details on what these technologies might consist of are not forthcoming. The portfolio of carbon credits used to claim to offset its emissions in the fiscal year 2021 consists of 99% low-durability carbon dioxide removals with biological storage, such as reforestation projects in the US and soil sequestration projects in Australia. Microsoft reports that these have a 'contracted durability' of just 20-25 years (Microsoft, 2022e, 2022c, p. 13).

Some projects are unlikely to meet even historical Kyoto-era definitions of additionality. Most of the offset credits that **Google** has procured stem from projects in the United States that capture and utilise methane from landfill sites to avoid its release into the atmosphere. The installation of methane capture technology is mandated by local or national government in several industrialised countries. Analysis for other countries where there is no policy mandate for the technologies shows that there is a high economic incentive to implement such projects without support, if the biogas can be used for electricity generation (Warnecke *et al.*, 2017). Accordingly, the additionality of the offset credits from the initial investment on this type of project is contentious. Google notes in a footnote of its carbon offsetting whitepaper that the credibility of offsets from these projects is contended, but that the company prefers to support projects that utilise captured gas (Google, 2011). Making use of the gas is indeed environmentally and economically attractive, and therefore good practice, but it is also the reason in this case why the credit revenue from Google may not lead to any additional climate action.

Google again implicitly recognises the questionable additionality of their offsets when they report that without their support, the 'additional cost for these community programs would have to be borne by local residents and businesses (Google, 2018). It is not clear whether Google is claiming to have subsidised the waste treatment bills of local residents and businesses for projects that were going to happen anyway, or to have financed additional emission reductions from projects that would otherwise not have happened.

4.4.2 Is it possible to mitigate the non-permanence of climate impacts from biological carbon dioxide removals?

More support is needed for carbon dioxide removal and storage measures – including nature-based solutions – but such measures are not suitable for neutralisation claims unless they can guarantee the permanence of the climate impact, among other important criteria. Neutralisation claims based on non-permanent carbon dioxide removals will lead to an increase in the atmospheric concentration of GHGs. The non-permanence of nature-based solutions cannot be mitigated; approaches that claim to do so are implausible.

All scenarios consistent with a 1.5°C temperature increase include a major role for CDR (Rogelj et al., 2018). Finance is needed to scale up carbon dioxide removal efforts, and corporates could play a key role. But carbon sequestrated in soils and forests is highly vulnerable to natural and anthropogenic disturbances such as forest fire or soil erosion; permanence of sequestration is likely over a period of just years to decades. The release of previously sequestered carbon negates any benefits of the initial sequestration: at the point at which the carbon dioxide is released, the atmospheric concentration of carbon dioxide is restored to the same value that it would have been had the CDR activity never taken place. If non-permanent removals are used to claim to neutralise emissions released elsewhere, the global CO₂ concentration will increase as a result (Jeffery et al., 2020). A sufficient guarantee of permanence requires a high likelihood that the captured carbon will remain stored over a timeframe of centuries, to millenniums.

Amid increased awareness that the non-permanence of nature-based carbon dioxide removal projects makes them unsuitable for claiming the neutralisation of emissions, companies and crediting mechanisms are proposing approaches which they claim can mitigate the issue of nonpermanence. The Carbon Credit Quality Initiative (Öko-Institut *et al.*, 2021) categorises these into three approaches. Table 17 describes how each of these approaches are based on flawed assumptions:

Proponents of biological carbon dioxide removals for offsetting often point out the fact that there is not much more that project developers can do beyond these approaches to ensure permanence. This is true, and the reason why these projects are not suitable for claiming the neutralisation of emissions, rather than a justification for leniency.

APPROACH

Temporary credits: Under this approach, offset credits are valid only for a specific period of time, after which they have to be renewed or replaced.

ISSUES

This approach works only in theory, under the implausible assumption that the credit issuing mechanism will continue to operate for the desired duration of permanence, which should be at least centuries. The only experience so far of a mechanism that issues temporary credits in this way was the Clean Development Mechanism. The second commitment period of the CDM is about to come to end in 2023 and a third commitment period is not envisaged. As such, it will become technically impossible to renew or replace temporary credits through this mechanism after 2023. Ensuring permanence through a credit issuing mechanism only works to the extent that there is a guarantee of permanence for the lifetime of the mechanism, which there is not.

Monitoring and compensating

for reversals: A commonly proposed approach is to monitor, report and compensate for reversed mitigation outcomes through the cancellation of issued carbon credits, for a sufficiently long period. For example, the Climate Action Reserve (CAR) requires forestry projects to be monitored and reversals compensated for at least 100 years after credit issuance.

Discounting⁺: Credit issuance from non-permanent projects is discounted at a rate deemed to cover the risk of reversals over the desired time period. A certain proportion of the carbon dioxide removals would not be credited, so that these non-credited outcomes can counterbalance reversals in the future.

Buffer pool: Projects maintain adequate buffer reserves of non-tradable carbon offsets to cover unforeseen losses in carbon stocks. This approach only works under a set of highly implausible assumptions. The example given depends on the assumption that the Climate Action Reserve still exists in 100 years, and that it still follows the same set of rules. The assumption that the corporate governance framework for climate change and offsetting in 100 years from now still resembles the mechanisms in place today is improbable to say the least. An even more unlikely assumption is that the company purchasing the offset credit still exists in 100 years, and that it cares about maintaining the integrity of a carbon neutral claim that it made 100 years ago. This depends, in turn, on a high enough level of public scrutiny on the carbon neutrality claims that companies made 100 years ago. This assumption is entirely untenable. Without fulfilling all of these assumptions, the monitoring and compensation approach offers no value for robustly mitigating the non-permanence of nature-based carbon dioxide removals.

The proposal of a discounting or buffer pool approach to address the issue of the non-permanence of biological carbon dioxide removals is an indication that the issue of non-permanence is not well understood. The non-permanence of forestry and soil related carbon dioxide removal impacts is not merely a possibility; rather the storage of carbon in this form is *likely* to span a timeframe of only *years to decades* (see Methodology Version 3.0 Table 12), and therefore the risk of reversals over a period of centuries can be considered a near-certainty. Discounting to address a risk that can be considered a near-certainty may require a discount rate of up to 100%, resulting in little to no credit issuance. Similarly, a buffer pool would need to be very many times larger – potentially infinitely larger – than the tradable volume of offsets.

* Discounting here refers to the approach to discount issuance to address permanence specifically. We do not discuss discounting approaches to address other issues of credit integrity, such as additionality. Discounting may be a more appropriate mechanism in other contexts.

4.4.3 Is the tech industry really carbon neutral?

Our analysis has shown that despite current and future neutrality claims of tech giants like Google, Microsoft, and Amazon, the ICT sector is far from being 'carbon neutral'. Serious plans for deep emission reduction measures are lacking while companies continue to strongly rely on biological carbon dioxide removals with low durability to meet their claims. While we could identify some good practice examples in the companies' sustainability strategies, in terms of transparency, renewable energy procurement, and innovative emission reduction measures, these good practices do not go nearly far enough to justify the bold and misleading carbon neutrality claims that Google, Amazon, and Microsoft put forward.

One of the main challenges the ICT sector faces is the environmental impact of its operations and products.

The production and use of ICT devices and infrastructure especially constructing and operating datacentres – requires a significant amount of energy. Datacentres and data transmission networks consumed about 1.5% of global electricity and were responsible for about 1% of the global energy related emissions in 2021 (IEA, 2022). The full emissions footprint of the ICT sector is substantially larger, peer-reviewed estimates put ICT emissions at a range between 1.8% to 3.9% of global GHG emissions, depending on the scope of activities included in the calculations (Freitag et al., 2021). Demand for ICT services has grown rapidly over the past decades due to increased digitalisation and rising numbers of internet users, resulting in increased demand for data services, consumer electronic devices, and increased requirements for processing power (IEA, 2022). Similarly, emissions from the sector have also been increasing rapidly; while estimates of the exact global emissions share from the ICT sector vary there is a consensus in literature that ICT emissions have grown faster than global emissions in recent years (Freitag et al., 2021). Despite this, energy efficiency improvements allowed emissions to rise at a slower pace than the sector's energy demand. ICT companies are now faced with the challenge to rapidly decrease emissions while keeping up with the constantly increasing demand for its products and

services. Forecasts estimating the future energy consumption of the ICT sector are uncertain and vary significantly mainly depending on the assumptions regarding energy efficiency gains and limitations. The IEA (IEA, 2022) estimates that energy consumption for datacentres will modestly increase in the immediate future but long-term energy demand remains highly uncertain, but emissions are likely to increase in the absence of new emission reduction measures. Despite these challenges, the ICT sector also offers great opportunities that can help reduce emissions and support the mitigation and adaptation of climate change through the development and deployment of new technologies.

In our report, we analysed the climate strategy of three of the major ICT companies, Amazon, Google, and Microsoft. All three companies belong to the top five companies by market capitalisation (Johnston, 2022). Combined, they control over two thirds of the global cloud infrastructure market and were responsible for around 0.5% of the global GHG emission in 2021 (IEA, 2022; Synergy Research Group, 2022). Accordingly, these companies should be at the forefront and race each other to the fastest and most cost-effective deep decarbonisation to align their business with the 1.5°C Paris-aligned temperature goal. However, the results of our analysis suggest that tech companies are still far from reaching deep decarbonisation and 'carbon neutrality'.

Our findings indicate that ICT companies scramble for the title of 'carbon neutrality' and 'net-zero' in a race to the bottom, focused on speed rather than integrity.

Microsoft, Amazon, and Google are three consumer facing brands directly competing for market share thus, it is not surprising that these companies' follow similar climate communication strategies.

Both Microsoft and Google currently claim to be 'carbon neutral' while only covering 2% and 12% of their full emission print with these claims, respectively. By 2030, Microsoft claims to become 'carbon negative' and Google claims to reach 'net-zero' emissions, covering their full emission footprint. For the consumer it is difficult to distinguish the difference between 'carbon neutral' and 'netzero' and make informed choices based on that information. Microsoft, Amazon, and Google have pledges to reach net-zero emissions, but the insufficiency of their emission reduction targets is strikingly similar. Microsoft's future neutralisation claim entails a commitment to reduce emission by 37%; Google's net-zero target translates to an emission reduction commitment of just 38% compared to 2019 levels. While Amazon does not make any neutralisation claim today, it sets out a target to achieve 'net-zero carbon' by 2040 without making any commitments to actual emission reductions. Offset credits play a major role in the fulfilment of these pledges. Microsoft, Amazon, and Google claim that their emissions are - and will be - 'neutralised' through the purchase of offset credits. While Google provides very limited information on the type of projects they plan to support, to fulfil their neutralisation claim in the future, Amazon and Microsoft will at first focus mainly on forestry-related projects but aim to shift their portfolio towards technological CDR that have a higher contracted durability and a higher likelihood to be additional in the long-run (Amazon, 2022a; Microsoft, 2022c). Microsoft is a best practice example for transparency when it comes to offsetting projects. The company issues an interactive website that allows the user to see all individual offsetting projects that Microsoft supports since 2021 (Microsoft, 2022b).

These neutralisation claims give consumers and investors the misleading impression that Amazon's, Google's, and Microsoft's activities – and the use of their products – do not contribute to climate change. Indulging consumers with the belief that they can use ICT services and products without any significant environmental consequences may lead to adverse effects on the emissions through overconsumption and unnecessary energy use. The extent to which consumer behaviour is influenced by these misleading claims and whether they result in increased consumption and emissions, remains an open question.

The lack of any planned emissions reduction measures and abatement strategies for all three of these major ICT companies beyond their mislabelled net-zero commitments for 2030 or 2040 is striking. Claiming 'netzero' through the purchase of contentious offset credits is not enough to address climate change. Investment decisions made today are likely to impact the years beyond 2030 or even 2040 thus, it is crucial that companies include climate change into their long-term planning to signal to consumers and investors that they are deeply committed to decarbonisation. ICT companies are among the frontrunners for high quality renewable energy procurement, but other emission scopes require closer attention. Google was the first company globally to match its used electricity with renewable energy on an hourly basis, shortly after followed by Microsoft. In 2021, Amazon was the world largest corporate purchaser of renewable energy and Google and Microsoft were under the top four (IEA, 2022). Microsoft and Google are best practice examples on how companies can reduce their scope 2 energy related emissions, tracking both their location- and marketbased emissions and investing in additional renewable energy projects in the local grid that will have the largest impact on overall emission reduction. ICT companies tend to focus on highlighting their efforts procuring renewable energy, an emission source which is highly visible and relatively easy to decarbonise. But direct electricity consumption does not account for ICT companies largest emission source. Indirect scope 3 emission make up for the vast majority of Amazon's, Google's, and Microsoft's emission portfolio; in 2021 scope 3 emission accounted for 59%-77% of the full carbon footprint of those companies. Still, emission reduction measures targeted specifically at emission sources in scope 3 are rather undefined in the companies' sustainability plans.

The IEA (IEA, 2022) status report of the ICT sector makes it clear that ICT companies need scale up ambition to meet the goals of the Paris Agreement and continue to improve their emission monitoring systems, back up their neutrality claims with deep emission reduction targets, continue to purchase local renewable energy, invest in R&D and roll out new energy efficient technologies on a broader scale. Companies need to start raising the bar on serious climate action in the ICT sector.

4.4.4 What does insetting and climate positive really mean?

'Insetting' is a business-driven concept with no universally accepted definition. The approach can lead to low credibility GHG emission offsetting claims and the double counting of emission reductions.

The concept of insetting is promoted by some actors as a better alternative to offsetting, mainly for companies with links to agriculture and land-use sectors in their supply chains. Insetting is sometimes described as offsetting within the value chain. This can mean two different things, both of which are highly contentious:

 Emission reduction projects in the value chain: Here, an emission reduction project – similar to an offsetting project – is implemented within the company's value chain, rather than outside of it. Describing this as insetting is a false concept; this is simply a measure for the reduction of the company's own emissions. In claiming that the reduction of certain emissions neutralises the company's other GHG emissions, the company is either: a) rejecting responsibility for those sources and excluding them from the scope of its target or claim; or b) counting the emission reductions of those measures twice to claim reductions for some emission sources and neutralisation of other emission sources. The credibility of the claim is critically compromised in either case.

In the most extreme case, companies may claim the complete carbon neutrality of their scope 1 and 2 emissions, by claiming the reallocation of marginal reductions from their scope 3 emissions. Given that scope 3 emissions account for the major share of many companies' emissions, such a claim may be possible with only very marginal reductions to scope 3 emissions that could possibly be achieved under business-as-usual trajectories. The possible outcome is that a company claims to be carbon neutral without having taken any action to reduce its scope 1 and 2 emissions. Carbon dioxide removals in the value chain: In this case, measures are taken within a company's value chain to achieve carbon dioxide removal and storage. This may include carbon storage in agricultural soils, and carbon storage in harvested wood and wood-based products. Here, the same environmental integrity issues apply as for any other carbon dioxide removal offsetting projects (see Box 3, section 4.2.2): the suitability of these measures for claiming the neutralisation of GHG emissions is compromised by the lack of permanence of the carbon storage and the scarcity of nature-based solutions for carbon dioxide removals. An apparent key difference between carbon dioxide removals under an 'insetting' approach, as opposed to carbon dioxide removals through certified offsets, is that the companies implementing an insetting approach may not seek independent measurement and verification of the carbon dioxide removals. Most carbon credit standards require projects to go through expert review, stakeholder consultation and independent verification to assess compliance with approved methodologies, and they have rules pertaining to defined crediting periods and the issuance of credits. Those processes do not guarantee a high-quality outcome - and they do not address the fact that biological carbon dioxide removals are fundamentally unsuitable for offsetting claims - but such processes are considered a pre-requisite for environmental integrity. By comparison, we do not see any evidence that companies claiming to inset their emissions are going through such processes. As such, this is simply a weaker variation of an already noncredible offsetting approach.

Several major companies are currently advocating for standards that recognise insetting as valid carbon compensation, including through holding prominent roles on advisory committees and technical working groups of key standard setting initiatives such as GHG Protocol's *Guidance for corporate accounting of land sector emissions and removals* (GHG Protocol, 2021). These companies can now refer to the 2022-published SBTi guidance for forestry, land use and agriculture (FLAG) as evidence of a certain recognition of the terminology. The new SBTi FLAG guidance allows companies to claim the achievement of their emission reduction targets through 'insetting', albeit without clarifying what the term means, nor a method for measuring and independently validating the legitimacy of any claims. This breaks from the long-held SBTi position that emission reduction targets should only be achieved through delivering actual emission reductions. SBTi explicitly acknowledges that the definition of insetting and its suitability towards emission reduction targets remains uncertain, while still allowing its use (SBTi, 2021c, p. 30 Box 3). This may have damaging consequences for the credibility of climate targets from companies with significant land use emissions, including not only agri-businesses but also retailers and the fashion industry.

Climate Positive pledges are based on the principles of insetting and avoided emissions, neither of which is recognised as a legitimate approach for claiming to offset emissions.

In recent years, a small group of companies have started to use the terminology "Climate Positive" for their climate targets. Those companies define climate positive as a state of reducing more greenhouse gas emissions than the value chain emits. We understand that those companies seek to differentiate this approach from offsetting, but we believe that observers are highly likely to interpret the terminology climate positive to mean that unabated emissions have been neutralised.

Companies' *climate positive* targets typically include a combination of insetting measures and claims of *avoided emissions*. 'Avoided emissions' is defined by the ISO Net Zero Guidelines as "a potential effect on greenhouse gas emissions that occurs outside the boundaries of the organization but arising through the use of its products or services, outside scope 1 emissions, scope 2 emissions and scope 3 emissions "(ISO, 2022b). A key difference here from emission reduction offsets is that there is no case for demonstrating the additionality of these avoided emission claims. For example, a company which sells PV modules to its customers may claim avoided emissions

from the customers' use of those PV modules over their expected lifetime. If the sales of these PV systems constitute normal commercial transactions to supply an existing market demand, rather than subsidised interventions from the company, these estimated avoided emissions are in no way additional to what may have occurred had the company not participated in this market. The GHG Protocol already specified in 2004 that any claims of *avoided emission* may not be accounted against scope 1, scope 2 or scope 3 emissions (WBCSD and WRI, 2004). Most recently, the ISO Net Zero Guidelines confirmed this position (ISO, 2022b).

Recognising that neither the concepts of insetting, nor avoided emissions are legitimate approaches for claiming the neutralisation of emissions, we understand that companies using the *climate positive* terminology seek to differentiate this approach from offsetting, by arguing that climate positive does not constitute a neutralisation claim. On the contrary, we believe that observers are very likely to interpret the terminology *climate positive* to mean that unabated emissions have been neutralised and that the company has a net-positive impact on the climate through a net-negative GHG emissions balance.

4.4.5 Are companies ready for a shift to the climate contribution approach?

Offsetting remains the dominant and relatively easy option for companies to wipe away large chunks of their overall climate impacts at low cost and with appealing short-term marketing value. However, acknowledgement of its major limitations in today's climate is on the rise, with court cases highlighting legal risks, a major climate neutral label pulling the plug on their certification, and a recent decision by governments at COP 27 paving the way for greater endorsement of the climate contribution approach. To date, few companies are making climate contributions without neutralisation claims, but despite challenges in its appeal, the tide may be turning.

The findings in our report suggest that many companies only plan to reduce a small share of their full emission footprint, relying instead on offsetting their remaining emissions with contentious carbon credits. Climate neutral, or net-zero, claims backed up by limited actual emission reductions hide the real environmental impact of the company's products and services and risk delaying urgent decarbonisation measures (for more information on the risks associated with offsetting see section 4.2.2). The climate contribution approach provides an alternative route for companies to actively take responsibility for unabated emissions without claiming to neutralise them. This avoids many of the pitfalls associated with offsetting.

There is increased public acknowledgement of the contentious integrity around offsetting claims and over the last year the climate contribution approach has gained some momentum. A growing body of literature and guidance on the approach is emerging (for more information see section 4.3). Still, our analysis does not provide any evidence that the climate contribution model is gaining traction at the company level.

Why is it that the climate contribution approach is still not implemented more broadly by companies despite apparent advantages over offsetting in terms of delivering greater transparency and environmental integrity? Reasons for this implementation gap can be found on both the supply and demand side.

There is currently not enough pressure from consumers, investors, or governments to increase the environmental integrity of neutralisation claims and regulate offsetting claims. Companies are increasingly labelling their products and services as 'carbon neutral' or 'net-zero', to communicate to consumers, investors and even potential regulators that they are not contributing to climate change. Carbon or climate 'neutral' labels are – on the face of it – easy to understand and give the impression that a brand, product or service has no detrimental impact on the climate, enabling consumer-facing brands to effectively market their product as environmentally friendly. Despite the falsehoods that often underpin this impression, many companies today appear to identify significant marketing value from making these claims, whilst avoiding implementing serious emission reduction measures. This can change if governments (and courts) take steps to more stringently regulate claims or if consumers and investors send a signal that they are not willing to pay a premium for environmental credentials that rely heavily on contentious offsetting practices.

A lack of knowledge and awareness by companies may hinder uptake of the climate contribution approach. The concept of funding climate action in return for carbon credits that allow the holder to offset their own impacts has been used as a tool to raise climate finance for at least three decades. Despite historical challenges in attracting private sector interest in the model, a whole infrastructure of methods, tools and in particular professional interests are vested in offsetting markets, along with an array of recent company showcases. Companies today that are exploring how to take responsibility for their emissions may be unaware about alternatives to offsetting or do not fully understand the associated risks it entails. Since adoption of the climate contribution approach is limited to date, there are fewer established best practice examples to draw from. And service offerings - such as off-the-shelf products, or business advisory providers - that support companies to navigate the switch from offsetting towards climate contributions are in short supply.

However, recent developments suggest that there is movement in the market towards the contribution model. At COP 27, in November 2022, a decision was made to label certain credits as 'mitigation contributions' where they support a country to achieve its national target. Shortly after followed an announcement by the business consultancy, myclimate, that it would no longer offer its 'climate neutral' label because it is not aligned with the Paris Agreement. It will instead offer a new 'climate impact' label to its customers which appears more aligned with the climate contribution approach (myclimate, 2022). To our

climate contribution approach (myclimate, 2022). To our knowledge, this is the first label of its kind that provides a clear and accessible pathway for companies to implement the climate contribution approach. These are promising steps towards reducing the implementation barriers of the climate contribution model and presents companies, the consultants that advise them, and the wider stakeholders with the opportunity to play a pivotal role in facilitating the transition to the climate contribution model. There are already a few frontrunners implementing the climate contribution approach, such as Gold Standard, Klarna, and DSM which have distanced themselves from offsetting to increase the environmental integrity of their business. More examples like this can serve as an inspiration for other companies to adopt the climate contribution approach.

SECTION B Company assessments

The Corporate Climate Responsibility Monitor 2023 assesses the integrity of high-profile climate change mitigation pledges from 24 of the world's largest companies. This analysis assesses only companies that have committed to high-profile climate change mitigation pledges under one of the main corporate climate action networks and initiatives. The key objective of the analysis is to identify replicable good practice while assessing the integrity of the most influential global corporate actors that are putting themselves forwards as climate leaders and role models for other companies. Scrutiny of their plans is also necessary to identify whether these influential leaders really are setting the right examples, and whether the guidance and frameworks upon which they are making their plans are sufficient.

We assess the top three global companies for each of the eight following sectors, according to their annual revenue in 2021 (Forbes, 2022): Automotive manufacturers; electronics; fashion retail; food and agriculture; information and communication technology; shipping and aviation; steel and cement, supermarket retail. Our analysis excludes state-owned companies due to our perception that fundamental differences in management structures and decision-making structures for climate change strategy may significantly detract from the comparability of these companies' plans, and the insights that we can draw from the company sample.

An overview of the selected companies and our evaluations is presented in Table 18. The 24 companies covered by this monitor account for approximately USD 3.16 trillion of revenue in 2021, approximately 10% of revenue from the world's largest 500 companies (Forbes, 2022). Their total self-reported GHG emission footprints in 2019, including scope 3 emissions, amount to approximately 2.2 GtCO₂e. This is equivalent to roughly 4% of global GHG emissions.² 10 of the 24 companies selected through the process described above were also assessed in the *2022 Corporate Climate Responsibility Monitor*. The repeat analysis of this small sub-set of companies offers insights into what progress has been made over the past year.

Our company-specific assessments include a rating of the **transparency** and **integrity** of their approaches across the key elements of corporate climate responsibility discussed in section A: **tracking and disclosure of emissions** (section A1), **setting specific and substantiated targets** (section A2), **reducing emissions** (section A3), and **climate contributions and offsetting** (section A4).

Transparency ratings are primarily based upon the extent to which a company publicly discloses the information necessary for an observer to fully understand the integrity of that company's approaches towards the various elements of corporate climate responsibility. **Integrity**, in this context, is a measure of the quality and credibility of those approaches. A full overview of the rating methodology for transparency and integrity of every indicator is presented in the accompanying methodology document, Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0 (NewClimate Institute, 2023b).

The *Corporate Climate Responsibility Monitor* promotes transparency with the philosophy that consumers, regulators, shareholders, and other observers should be able to follow and assess the integrity of companies' claims. Accordingly, the company assessments in this section are based only on publicly available information that could be identified by the authors. Each rating represents the authors' understanding of the publicly available information. In some cases, company information was scattered across different sources (e.g. annual reports, press releases and statements, webpages, and other marketing materials); it is possible in this process that information may have been misinterpreted, or overlooked. Companies should consider how to present information as transparently as possible, to ensure that observers are able to readily identify all the relevant information necessary to understand their climate strategies.

² Some overlap in emission statistics is likely in the cases that one company's scope 3 emissions are included in the scope 1 or 2 emissions of another company in this analysis. We anticipate that any overlap is marginal and of limited significance to the key insights derived from this report. The companies' combined emission footprint may also be higher than this estimate, due to some companies' incomplete emission disclosure. We use 2019 as a base year for analytical purposes, as the most recent year with complete GHG reporting before the COVID-19 pandemic distorted emission trends.

Table 18: Overview of companies assessed in the Corporate Climate Responsibility Monitor 2023

HIGH INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
No companies achieved a high integrity rat	ing			
REASONABLE INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
Maersk	Net zero by 2040		•	p. 100
MODERATE INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
Apple	Carbon neutral by 2030			p. 78
Arcelor Mittal	Net zero by 2050		0	p. 80
Google	Net zero by 2030		0	p. 90
H&M Group	Net zero by 2040	\bigcirc	0	p. 92
Holcim	Net zero by 2050			p. 94
Microsoft	Carbon negative by 2030			p. 104
Stellantis	Net-zero carbon by 2038	\bigcirc		p. 112
Thyssenkrupp	Climate neutral by 2050		0	p. 114
LOW INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
Ahold Delhaize	Net zero by 2050		O	p. 72
Amazon	Net-zero carbon by 2040	C	G	p. 74
Deutsche Post DHL	Net zero by 2050	\bigcirc	G	p. 84
Fast Retailing	2030 emission reduction targets		G	p. 86
Foxconn	Net zero by 2050	G	G	p. 88
Inditex	Net zero by 2040		O	p. 96
Mercedes-Benz	Carbon neutral vehicles by 2039	G	G	p. 102
Nestlé	Net zero by 2050	G	O	p. 106
PepsiCo	Net zero by 2040	G	O	p. 108
Volkswagen	Carbon neutral by 2050		O	p. 116
Walmart	Zero emissions by 2040		C	p. 118
O VERY LOW INTEGRITY	HEADLINE PLEDGE	TRANSPARENCY	INTEGRITY	PAGE
American Airlines	Net zero by 2050		\bigcirc	p. 76
Carrefour	Carbon neutral by 2040	0	0	p. 82
JBS	Net zero by 2040	0	0	p. 98
Samsung Electronics	Net-zero carbon by 2050	C	\bigcirc	p. 110

RATINGS

5-point scale High Reasonable Moderate Low Very low. See individual company analyses. Assessments were made based on public information identified by the authors. A poor rating may not necessarily be an indication that a company's climate strategy is weak, but could also indicate that the information was insufficient to confirm good practice. Ambitious companies can improve their ratings by ensuring that all aspects of their climate responsibility strategies are transparently and accurately disclosed, and in the public domain.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Food, beverages & agriculture	USD 79.5 bn (2021)	70.3 MtCO ₂ e (2021)	Net zero across entire value chain by 2050	O Moderate	Low
1 TRACKING AND D	ISCLOSURE OF EMIS	SIONS		TRANSPARENC	Y & INTEGRITY
70.3 MtCO ₂ e in 2021 purcha agricult Disclos no s3 b		oods and services (upstre activities.		Scope 1 Scope 2 Scope 3 upstream downstream	1. 1. 63. 3.
2 SETTING EMISSIC	IN REDUCTION TARG	ETS		TRANSPARENCY	INTEGRITY
Headline target or pledge	Net zero acro	ss entire value chain by 20	050		
Short- and medium-term (up to 2030) Scope coverage	- Reduce s3 by	s2 by 29% by 2025 and 50% 37% by 2030, compared to 2 argets translate to ~33% 6			D
Own emission reductions (compared to full value chain	33-34 % 2	nd s3 targets are fully alig	e of the separate s1, s2		
Long-term vision (beyond 2030)	S1 and s2: ne	t zero by 2040. S3: net z	ero by 2050.		
Scope coverage Own emission reductions (compared to full value chain	82% ta	et-zero targets presente argets, translating to ~82 2019 baseline). Close to a		\bigcirc	
3 REDUCING OWN E	MISSIONS			TRANSPARENCY	INTEGRITY
Emission reduction measures Comprehensive strategy for s1 and s2. S3 strategy includes various measures, but no clear signs of transformative reductions in significant emission sources.			0	?	
Renewable electricity procurement	,	amount of information in accounts for just 14% of		0	0
4 CLIMATE CONTRI	BUTIONS AND OFFSE	TTING		TRANSPARENCY	INTEGRITY
Responsibility for Neutralisation claim for certain sold products.			0	\bigcirc	
Climate contributions — No climate contributions identified. —			N/A	\bigcirc	
Offsetting claims today		aize's brands sell 'climate-neo nation about the used offsets		\bigcirc	?
Offsetting plans for the fu	ITIILE	2019 emissions to be offse n on project types.	et.		()

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Ahold Delhaize (2022a, 2022b, 2022c, 2022d) and Albert Heijn (2021).
See **Annex II** for more details on our assessments of companies' targets

Ahold Delhaize

Koninklijke Ahold Delhaize N.V. (hereafter: Ahold Delhaize) is the holding company of various supermarkets, chemists and e-commerce retailers. The company's major emissions are related to purchased goods and services, including agricultural activities, which account for 88% of its emissions footprint. Ahold Delhaize has a net-zero target for 2050, covering all its global activities. Alongside this pledge, the company presents emission reduction targets, which translate to roughly 82% reduction of its estimated 2019 emissions. Ahold Delhaize also commits to roughly 33% emission reductions across the value chain by 2030, compared to estimated 2019 levels. Ahold Delhaize's emission reduction strategy for scope 1 and 2 is extensive, but the company provides limited detail on how it wants to reduce scope 3 emissions.

Ahold Delhaize has a net-zero target for 2040 (scope 1 and 2) and for 2050 (scope 3), accompanied with emission reduction targets for the target years, but does not clarify how it will address remaining 18% of emissions. With the emission reduction targets of 90% (2018 baseline) for scope 1 and 2 and 83% (2020 baseline) for scope 3, Ahold Delhaize clarifies what share of emissions it will reduce to reach its netzero pledge (Ahold Delhaize, 2022b, pp. 4; 9). The reduction targets translate to reducing the company's 2019 emissions footprint by roughly 82% by 2050. Ahold Delhaize sees the need to use carbon removals for the remaining 18% of emissions but does not provide any details on the projects or carbon removal measures it intends implement (Ahold Delhaize, 2022c). Without information on the planned offsetting strategy, it is not possible to determine whether the approach to offset these emissions will be credible.

Ahold Delhaize describes its emission reduction strategy for scope 3 in broad terms and focuses mainly on scope 1 and 2. Although scope 3 emissions account for about 95% of Ahold Delhaize's emissions footprint, the company mainly focuses on its scope 1 and 2 emissions in its emission reduction strategy. These measures target the main emission sources of the scopes, are presented with quite some detail and are likely to be in line with the 2030 emission reduction target (Ahold Delhaize, 2022c, 2022d, pp. 4–8). Scope 3 emission reductions are presented in broad terms and mainly depend on actions from others: Ahold Delhaize plans to increasingly engage with suppliers and wants to influence consumers' decisions to choose more sustainable products (Ahold Delhaize, 2022c, 2022d, pp. 4–8). Ahold Delhaize does not provide details on the expected emission reductions from these measures and does not strongly commit to them. Moreover, a large share of measures seems to target operations in Europe, while the majority of emissions occurs in the United States. Ahold Delhaize's climate neutrality claims for its products, including dairy, coffee and bananas, are contentious and potentially misleading. Its Dutch subsidiary Albert Heijn claims that the emissions related to dairy production are compensated by carbon storage in grazing land (Albert Heijn, 2021, p. 31; Ahold Delhaize, 2022a). While carbon storage in grazing lands is a good practice and enhancing it will benefit the climate, claiming this leads to 'climate neutral' dairy products is misleading and inaccurate. The permanence of the carbon storage in grazing lands cannot be guaranteed and grazing lands are not an endless sink. With events such as a change in management practices or extreme weather the carbon is likely to be re-released into the atmosphere within years or decades (see Box 3, Section 4.2.2), and over time, the soil's saturation level is reached and no additional carbon can be stored.

In addition to 'climate neutral' dairy, Albert Heijn sells coffee and bananas branded as climate neutral. Albert Heijn states that the climate impact of its coffee is reduced as much as possible and that residual emissions are compensated with CO_2 credits from forestry and renewable energy projects (Albert Heijn, 2021, p. 27). The emissions footprint of bananas is reduced by 25% and the remaining 75% are offset. We did not find evidence that the offsetting projects are suitable for making a neutralisation claim. One of the major issues related to forestry-based offsetting is the limited permanence (see Box 3 section 4.2.2). Renewable energy projects usually do not represent the high-hanging fruit of mitigation potential and the purchase of offset credits from these projects are unlikely to lead to additional impact (see section 4.4.1).

Ahold Delhaize's reporting on emissions and renewable electricity consumption limits the facilitation of a thorough understanding of recent trends. Ahold Delhaize wants to achieve 100% renewable electricity consumption (Ahold Delhaize, 2022a, p. 60, 2022b, pp. 9–10) and describes the use of PPAs and RECs to claim zero electricity-related emissions (Ahold Delhaize, 2022d, p. 10). The company uses a market-based approach to show a decreasing trend in emissions (Ahold Delhaize, 2022a, p. 58), but the renewable energy constructs do not justify the claim that scope 2 emissions are really zero (see Table 9 Section 3.3.1 in the Methodology). By using the location-based method, which reflects the electricity that is actually consumed by Ahold Delhaize, 2022a, p. 262).

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Technology & Services	USD 469.82 bn (2021)			Low	Low
1 TRACKING AND	DISCLOSURE OF EMIS	SIONS		TRANSPARENC	(& INTEGRITY
		ion sources: The majority otprint comes from s3 emi		Scope 1	12.1
Only financially inte	only market grated emission so	Emission disclosure lacks -based emissions are repo urces in s3 are missing.		Scope 2 Scope 3	4.1 55.4
subsidiaries are cov					
	ION REDUCTION TARG			TRANSPARENCY	INTEGRITY
Headline target or pled Short- and medium-ter Sup to 2030)	m targets - Sourcing 10	oon by 2040 10% of its used electricity fr 1% of Amazon's shipments v			
Scope coverage Own emission reductio (compared to full value ch	ns ain in 2019) 16%	Max. emission reduction of possibly much less: details a use of RECs and offsets to re	re lacking, potential	0	0
Long-term vision (beyond 2030)	by 2030	ges to reach net-zero carbo by 2040.	n across		
Scope coverage Own emission reductio (compared to full value ch	ns ? ⁽	Jnlear scope coverage and i CO ₂ only, or all GHG emissic eduction target alongside t	ons. No emission	\bigcirc	0
3 REDUCING OWN	EMISSIONS			TRANSPARENCY	INTEGRITY
mission reduction mea	implemented	th the potential for deep de I for transport and building se measures is unclear.		0	0
enewable electricity rocurement	transparenc	RE electricty by 2025, targe y on RE procurement at an e of unbundled RECs.		0	?
4 CLIMATE CONTR	RIBUTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for No information on offsetting claims.				\bigcirc	0
Climate contributions Right Now Climate Fund: \$100m for biological CDR. Unclear if related to climate contribution or future neutralisation.				0	0
Offsetting claims today		— No offsetting claims tod	ay identified. ————	N/A	N/A
)ffsetting plans for the	e future Offsetting en	nissions mainly with forestry nat extent Amazon relies on	r-related projects.		()

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Amazon (2021a, 2021b, 2021c, 2022a, 2022b).

Amazon

Amazon.com, Inc. (Amazon) is a major platform for e-commerce and IT services, and retail. Amazon's footprint includes a broad range of emission sources, but the key emission drivers continue to be unclear due to poor granularity of data (Amazon, 2022a, p. 97). Amazon is currently taking proactive steps to test a variety of decarbonisation technologies, especially for renewable electricity and transportation, but medium- and longer-term plans for other emission sources remain unclear (Amazon, 2022a, pp. 20, 22, 24). The company's 'net-zero carbon' by 2040 pledge is unsubstantiated without any explicit reduction target for the company's own emissions, and with a significant role envisaged for offsets. Amazon has made very little progress over the past year to address these significant gaps in their climate strategy.

Key developments over the past year: We could identify only minor changes to Amazon's sustainability strategy since our previous analysis of the case study in the *2022 Corporate Climate Responsibility Monitor* (Day *et al.*, 2022). Accordingly, only minor modifications were made to this case study.

Amazon's 'net-zero carbon' by 2040 pledge currently remains unsubstantiated. Amazon announced its headline target as a cofounder of *The Climate Pledge*, an initiative that mobilises business to commit to 'net-zero carbon' emissions by 2040 (Amazon, 2022a, p. 10). Amazon previously committed to substantiating this 'netzero' pledge with more detailed emission reduction targets in 2022 (Amazon, 2021b), but this has not yet happened. In the meantime, it is unclear whether the target covers only CO_2 or also other GHG emissions and whether it covers emissions across the full value chain. Amazon has also not yet specified to what extent it will reduce its emissions and what the relative importance of offsets will be, although the company stated that offsets will play a role in achieving its target (Amazon, 2022a, p. 10).

Amazon's pledge is weakened by relying on offset credits from CDR with biological capture and storage. Amazon played a major role in the mobilisation of finance for the *Lowering Emissions by Accelerating Forest finance* (LEAF) Coalition, and since 2019 also through the USD 100 million *Right Now Climate Fund* (Amazon, 2022a, p. 18). Through that fund, Amazon provides financial support for the immediate implementation of 'nature-based solutions' that generate credits, that Amazon may plan to use to achieve its 'net-zero' by 2040 pledge (Amazon, 2021a, pp. 2–5). We interpret that the fund is used towards Amazon's 'neutralisation' claim in the future. Although these two initiatives feature a well-considered plan for the provision of long-term support to higher-quality forestry projects, issues related to permanence and additionality of carbon dioxide removals with biological capture and storage mean that such projects are not appropriate sources of credits to support 'neutralisation' claims (see Box 3 Section 4.2.2).

To reduce its emissions, Amazon continues to proactively test a range of decarbonisation technologies, but the lack of granular data on GHG emission sources does not facilitate a thorough understanding of how sufficient these measures are. Amazon's proactive approach for addressing transport emissions continues to include its agreement with Rivian to roll out 100,000 electric vehicles by 2030, investments made in 2020 to test batteryand hydrogen based trucking technologies for longer distance freight, and Amazon's commitment to decarbonise shipping under the *Cargo Owners for Zero Emission Vessels* initiative (Amazon, 2022a, pp. 16, 17, 24). These measures could significantly reduce transport related emissions from scope 1 and scope 3. Amazon is also demonstrating technologies for energy efficiency on flagship sites and is investing in rooftop solar and on-site storage solutions.

Amazon's efforts to reduce emissions from electricity use appear comprehensive at first, but transparency is lacking.

Amazon continues to provide much less detail on how it plans to decarbonise downstream scope 3 emissions. Although several measures are being put in place to reduce material use, improve recycling and extend the lifetime of Amazon branded products, the limited breakdown of scope 3 emissions in its reporting makes it difficult to assess the significance and sufficiency of those measures, as well as the gaps that remain. We did not identify any significant additional information on this issue since our assessment in 2021.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Transport and logistics	USD 29.9 bn (2021)	42.0 MtCO ₂ e (2021)	Net-zero GHG emissions by 2050	D Moderate	C) Very low
1 TRACKING AND	DISCLOSURE OF EMI	SSIONS		TRANSPARENC	Y & INTEGRITY
Major emission sources: Jet fuel accounts for 99% of s1 and 30% of upstream s3 emissions.2.0 MtCO2 e in 2021Disclosure: Various non-GHG emissions disclosed but not contrail cirrus, which is the most relevant non-GHG forcer from aviation. Comprehensive reporting of GHGs across s1, s2 and upstream s3.			Scope 1 Scope 2 Scope 3 upstream downstream	28. 0. 12. 0.	
2 SETTING EMISS	ION REDUCTION TAR	GETS		TRANSPARENCY	INTEGRITY
leadline target or pled	ge Net-zero Gł	IG emissions by 2050			
Chort- and medium-ter up to 2030)	of 50 million	nn GJ of RE by 2025; achie n gallons of jet fuel from fu % SAF by 2030.			0
Scope coverage Own emission reductio (compared to full value ch	ns i	No intensity or absolute re for 2030. The SAF target is what sectoral benchmarks	at the very minimum of		
ong-term vision beyond 2030)		ntensity of jet fuel by 45% and red to 2019; and net-zero GH			
Scope coverage Own emission reductio (compared to full value ch		Targets cover jet fuel but ex emission sources. They do sectoral benchmarks.		\bigcirc	0
3 REDUCING OWN	IEMISSIONS			TRANSPARENCY	INTEGRITY
mission reduction mea	investment	o improve efficiency of airci s in SAF. No commitment to tivities or decreasing dema	phase out carbon-	0	0
Renewable electricity40% of purchased electricity is labelled as 'renewable'; American Airlines purchases unbundled RECs to make this claim.				0	0
4 CLIMATE CONT	RIBUTIONS AND OFFS	SETTING		TRANSPARENCY	INTEGRITY
Responsibility for Inabated emissions		lines offers its customers the ne emissions associated with			0
Climate contributions		No climate contributions identified			0
Offsetting claims today	/	supported do not represent h val projects do not permane			0
) ffsetting plans for the		irlines expects it will need to residual emissions; no furthe		\bigcirc	2

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from American Airlines (2021, 2022a, 2022b, 2022c).

American Airlines

American Airlines Group Inc. (American Airlines) is the world's largest airline in terms of revenue and is headquartered in Texas. Most of its GHG emissions stem from the production and use of jet fuel (about 85%). The company has not committed to any emission reduction target in the next decade. Its net-zero pledge does not include a commitment to reduce *own* emissions. The airline foresees an important role for alternative aviation fuels, but the availability and sustainability of these are uncertain.

American Airlines' offsetting programme has the potential to mislead customers. The company offers its customers the possibility of 'offsetting' the emissions associated with their flight, stating that offsetting reduces 'the impact of air travel on the environment' (American Airlines, 2022c). American Airlines purchases carbon offset credits from three programmes: protecting and restoring forests in Mexico, peat swamps in Indonesia, and building improved cookstoves in Honduras. None of these projects can be considered unequivocally additional in the context of the Paris Agreement, which requires all countries to set ambitious reduction targets and global emissions to move to net zero (see Methodology section 4). Although it is important to scale up financial support for protecting forests and peat lands, the non-permanence of the carbon storage in these measures means that they are not a credible equivalent to emission reductions and do not lead to the CO_2 emissions from flights being 'offset', let alone the complete climate impact for aviation.

American Airlines has not yet presented a reduction target for 2030 and its target for the use of sustainable aviation fuels for that year falls short of what sectoral benchmarks show is necessary. To stand a reasonable chance of limiting global warming to 1.5°C, CO₂ emissions need to be halved between 2019 and 2030 (IPCC, 2022, p. 21) and various studies indicate that the aviation sector should reduce its emissions by at least 20% by 2030 (IEA, 2021b, p. 199; CAT, 2022). American Airlines has not set any emission reduction targets for 2030 or earlier, but committed to use 10% sustainable aviation fuel (SAF) by 2030, to source 2.5 million GJ of renewable energy by 2025, and to achieve absolute reductions of 50 million gallons of jet fuel from fuel-efficiency initiatives by 2025 (American Airlines, 2022a, p. 7). While the SAF target meets some sectoral benchmarks, it falls short on others. Teske (2022, p. 212) and the UNFCCC (UNFCCC, 2021, p. 12) show that SAF must reach a 9-10% share by 2030, but Boehm *et al.* (2022, p. 7) and the IEA (IEA, 2021b, p. 138) found that shares of 13-18% are necessary.

American Airlines has committed to reduce 'well-to-wake' GHG emissions associated with jet fuel by 45% per revenue tonne kilometre (RTK) by 2035, compared to 2019. Although SBTI verified this target as 'below 2°C compatible', the target does not meet the Transition Pathway Initiative's aviation benchmarks aligned with global warming of 1.5°C or below 2°C (TPI, 2022a). The target covers jet fuel but leaves out other emission sources and also does not include non-GHG climate forcers from flying, which account for about two thirds of aviation's climate impact (Lee *et al.*, 2021).

American Airlines net-zero pledge neither entails a clear commitment to reduce own GHG emissions nor does it explicitly cover non-GHG climate forcers from aviation. The airline expects its emissions from burning jet fuel and fuel production to double between 2019 and 2050 in a businessas-usual scenario (American Airlines, 2022a, p. 11). Eighty-five per cent of these expected emissions – or about 70% of 2019 levels - may be reduced through technological and operational improvements, as well as sustainable aviation fuels. The remaining 15% of projected business-as-usual emissions may be 'offset' to claim 'net-zero' (American Airlines, 2022a, p. 11). However, the airline does not make a clear *commitment* to these projected emission reductions. Sectoral benchmarks indicate that a 1.5°C compatible pathway would require the aviation sector to reduce GHG emissions by 80-94% by 2050 and address the non-GHG climate forcing impact from aviation (IEA, 2021b, p. 199; CAT, 2022a; Teske, 2022, p. 216).

American Airlines plans to scale up its use of sustainable aviation fuels (SAF) but these may have negative sustainability implications. SAFs are a critical measure in reducing emissions from aviation (Boehm *et al.*, 2022, p. 91; Jaramillo *et al.*, 2022, p. 60). American Airlines currently uses 0.05% SAF but commits to increase this to 10% by 2030 (American Airlines, 2022a, p. 7, 2022b). The company has partnered with the fuel company Aemetis, which uses wood and vegetable oils to produce SAF (American Airlines, 2021). While it is commendable that American Airlines invests in alternatives to standard jet fuel, reliance on biobased SAFs will very likely contribute – directly or indirectly to problems such as deforestation and destruction of natural habitats (Pavlenko and Searle, 2021, p. 15; P. Jaramillo *et al.*, 2022, pp. 60–61). Further, it is likely that scaling up the production of SAFs made from cooking oil or other waste oils leads to an increase of emissions in those sectors that currently use those waste oils (Pavlenko and Searle, 2021, p. 15). We could not identify evidence that American Airlines invests in the development of synthetic SAF, which requires less water and land resources than bio-based fuels and have a larger abatement potential but are still in the early development stage (Jaramillo *et al.*, 2022, p. 61).

While American Airlines presents a range of decarbonisation measures, it does not address the elephant in the room and the only currently available measure for deep decarbonisation: managing demand for aviation. In addition to its plans to scale up the use of SAFs, the company outlines various other measures, including fleet renewal and operational efficiency (American Airlines, 2022a, pp. 11–18). However, in the absence of new technologies that can be rolled out at scale without negative sustainability implications, the aviation sector can only reach deep levels of decarbonisation through a reduction in activity levels, meaning fewer flights (CAT, 2022a; Graver *et al.*, 2022). We did not identify any clear plans or commitment to phase out all carbon-intensive infrastructure. Accordingly, it is unclear how American Airlines could achieve the deep decarbonisation that its net-zero target implies.

Airlines could support a transition to alternative transport modes, such as rail. It is counterproductive for decarbonisation of the transport sector to lobby against the implementation of taxes and levies. According to InfluenceMap, American Airlines has a mixed track record on climate lobbying (InfluenceMap, 2022a). While the company seems to support federal- and state-level legislation promoting the uptake of SAF, American Airlines also states to 'continue to advocate for [the Carbon Offsetting and Reduction Scheme for International Aviation (CORSIA)] as the single global approach to addressing emissions from international aviation' (American Airlines, 2022a, p. 22). CORSIA is the key instrument to achieve 'carbon neutral growth' in international aviation from 2020, but it is extremely unlikely to do so. The scheme will likely cover less than 50% of international aviation emissions between now and 2035 and the carbon offset credits it allows are unlikely to lead to real emission reductions elsewhere (CAT, 2022a).

SECTOR RE	VENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
	365.8 bn 2021)	24.2 MtCO ₂ e (2021)	Carbon neutral by 2030	Reasonable	D Moderate
1 TRACKING AND DISCLOS	URE OF EMIS	SSIONS		TRANSPARENC	(& INTEGRITY
racking and disclosure 4.2 MtCO ₂ e in 2021		sion sources: Purchase of missions from the use of s		Scope 1 Scope 2	0.1
Subsidiaries are covered.		Detailed disclosure. The r scope 2 emissions is use		Scope 2 upstream downstream	16.9
2 SETTING EMISSION RED	JCTION TAR	GETS		TRANSPARENCY	INTEGRITY
Headline target or pledge	Carbon neut	ral by 2030			
Short- and medium-term targets up to 2030)	Carbon neut	ral by 2030			
Scope coverage		- Full coverage of life-cycle e by 2030 relative to 2015. A		\bigcirc	
Own emission reductions (compared to full value chain in 2019)		or 1.5°C warming. No shor			
Long-term vision (beyond 2030)	No long-tern	n vision for emission reduc	tions beyond 2030.		
Scope coverage Own emission reductions	N/A	No long-term vision comm	unicated.	\bigcirc	\bigcirc
(compared to full value chain in 2019)	by 2050				
3 REDUCING OWN EMISSIC	NS			TRANSPARENCY	INTEGRITY
mission reduction measures	sources, incl	ormation on measures cove uding transition of supply c nd measures to extend proc	chain to renewable	•	
enewable electricity rocurement	renewable,	ormation. 100% of electric mostly supplied through h s in on-site installations wi	igh-quality PPAs.		
4 CLIMATE CONTRIBUTION	S AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	Carbon neutr	ality claim for <1% of emissi	ions.	\bigcirc	\bigcirc
Climate contributions		— No climate contributio		N/A	\bigcirc
Offsetting claims today		utrality claim for s1, s2, emplo set credits from nature-based		\bigcirc	\bigcirc
Offsetting plans for the future		: 37% of 2019 emissions by 2 pased carbon removals.	2030		0

Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Apple (2020, 2022a, 2022b, 2022c).

Apple

Apple Inc. (Apple) is a US-based multinational corporation that specialises in consumer electronics, software, and online services. Most of its emissions stem from the energy used to manufacture products (67%), followed by energyrelated emissions from product use (21%). The company's climate strategy focuses on achieving carbon neutrality by 2030, including the reduction of 63% of its 2019 emissions footprint, with the rest to be offset. Apple implements reasonably comprehensive emission reduction measures that chart a trajectory for deep decarbonisation in the medium term. However, its current and planned carbon neutrality claims are potentially misleading exaggerations of this trajectory, and there is no clear vision for going beyond the 63% reduction by 2030.

Key developments over the past year: We identified no significant changes to Apple's climate strategy since the previous iteration of this analysis was published in February 2022.

Apple commits to achieve carbon neutrality by 2030, including reducing 63% of its 2019 emissions footprint and offsetting the rest; the company does not commit to further emission reductions beyond 2030. In 2020, Apple announced its 2030 *Climate Roadmap* with the aim to achieve carbon neutrality by 2030 (Apple, 2020). The goal includes a commitment to reduce 75% of the company's emissions footprint in 2015 and offset the remaining 25% (Apple, 2022a, p. 12). The 75% emission reduction goal equates to a reduction of 63% of Apple's 2019 emissions footprint, a target approved by SBTi as '1.5°C compatible' (SBTi, 2023a). Although a 63% emission reduction by 2030 represents a steep decarbonisation pathway, it does not represent the degree of decarbonisation that the 'carbon neutral' terminology implies. Apple also does not commit to reductions beyond 2030, even though a continuation of the pledged pathway would put the company on track for significantly deeper decarbonisation by 2035 or 2040. Setting a deep decarbonisation target for 2035 or 2040 might provide a more transparent representation of Apple's ambition and prospects than a carbon neutrality target by 2030.

Apple's offsetting policy could be misleading, as the company currently makes a carbon neutrality claim that covers less than 1% of its emissions footprint in 2021, while relying on naturebased carbon removals for current and future claims. Apple's headline on its environmental website reads: 'We've been carbon neutral since 2020. By 2030, all our products will be too' (Apple, 2022c). Apple's 'carbon neutral' claim today may be misleading, as it only covers offices, retail stores, data centres, employee commuting, and business travel, which together cover less than 1% of the company's emissions footprint in 2021 (Apple, 2022a, p. 15). The carbon neutrality claim is based on the procurement of carbon offset credits from biological carbon dioxide removals in Colombia and Kenya (Apple, 2022a, p. 30). In 2021, the credits amounted to 167,000 tonnes used to offset corporate operations, and Apple procured an additional 500,000 tonnes to offset the increase in scope 3 emissions from 2020 to 2021 (Apple, 2022a, pp. 30, 84). For its 2030 carbon neutrality target, Apple created the Restore Fund to invest USD 200 million in similar projects, aiming to remove 1 MtCO_2 each year (Apple, 2022a, p. 29). Due to the scarcity and limited permanence of nature-based carbon removals, the climate impact of Apple's current and planned support for these measures is not equivalent to the reduction of the company's own emissions. The integrity of Apple's approach would be stronger if the company provided support for these projects as climate contributions, without claiming that its investments 'neutralise' the impact of its emissions.

Apple's emission reduction plans are reasonably comprehensive and target the company's main emissions sources. Apple created the *Supplier Clean Energy Programme* to address emissions from product manufacturing by transitioning the whole manufacturing supply chain to renewable electricity by 2030 (Apple, 2022a, p. 24). The programme trains suppliers, helps them find renewable electricity solutions, and directly invests in renewable electricity projects for suppliers (Apple, 2022a, pp. 24–26) (see Box 2). Other emission reduction measures in product manufacturing include the use of recycled materials (Apple, 2022a, p. 17), innovative aluminium smelting technologies (Apple, 2022a, p. 28), and R&D of lower-emissions integrated circuits (Apple, 2022a, p. 16). Emissions from product use are being tackled through improvements in energy efficiency (Apple, 2022a, p. 18), while the company aims to reduce emissions from product waste through improved designs that allow for higher repairability and extended lifespans (Apple, 2022a, pp. 44–48). The implemented measures have reduced Apple's emissions intensity by unit of revenue by 11% on a yearly average since 2018, although absolute emissions have plateaued due to a significant increase in revenue in 2021.

Apple claims to use 100% renewable electricity since 2018 and discloses information on its high-quality renewable electricity construct, but it could improve the transparency of its energy-related emissions accounting. Apple claims that its facilities use 100% renewable electricity and discloses details on how this electricity is sourced, including location of projects, technologies used, capacities, and supply constructs (Apple, 2022a, pp. 89–99). Ninety percent of renewable electricity comes from 'Apple-created' renewable projects: 9% from directly owned projects, for which the company is also investing in electricity storage; 2.7% from projects in which it owns equity and matches energy generation with use; and 78.3% from local and newly installed PPAs (Apple, 2022a, p. 22). The remaining 10% is procured through utility green energy programs, colocation facility vendors, and RECs, for which Apple applies the same quality standards as for its 'Apple-created' projects. The company also supports community-based renewable electricity projects under the Power for Impact programme, but it is unclear how Apple and other local actors claim the environmental attributes of these projects (Apple, 2022a, p. 27). Regarding indirect energyrelated emissions (scope 2), Apple reports both market- and location-based estimates (Apple, 2022a, p. 109) but only uses the lower market-based value when estimating its total emissions footprint (Apple, 2022a, p. 84). In this case, it would be more ambitious to use the location-based method, which shows that there is still room for energy efficiency improvements to reduce the climate impact of Apple's electricity use.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Steel and cement	USD 76.6 bn (2021)	166.2 MtCO ₂ e (2021)	Net-zero by 2050	D Moderate	D Moderate
1 TRACKING AND	DISCLOSURE OF EMIS	TRANSPARENC	Y & INTEGRITY		
racking and disclosure 66.2 MtCO ₂ e in 2021	processes (s Disclosure:	sion sources: Emissions fro 1) account for over 80% o S1&2 emissions are broke	of reported emissions. en down to emissions	Scope 1 Scope 2 Scope 3 upstream	131.1 7.5
Subsidiaries are cov	Subsidiaries are covered. From mining and steelmaking. Potentially significant share of s3 emissions is not reported on.				28.7 3.0
2 SETTING EMISS	ION REDUCTION TAR	TRANSPARENCY	INTEGRITY		
Headline target or pled	ge Net-zero by 2	2050			
Short- and medium-ter (up to 2030)	m targets (eq. 1.54 tCO	e emissions intensity by 35% b			
Scope coverage Own emission reductio (compared to full value ch	ns by 2030	Supporting measures indica some 1.5°C aligned sectora penchmarks are not directl	l benchmarks, although		
Long-term vision (beyond 2030)	Net-zero by	2050			
Scope coverage Own emission reductio (compared to full value ch	ns ?	We interpret that the targe emissions. Company does r to own emissions reductior	not explicitly commit	0	0
3 REDUCING OWN	EMISSIONS			TRANSPARENCY	INTEGRITY
mission reduction mea	different pot	over s1 and s2, and include t tential low-carbon technolo te technology split in 2030.	ogies without	D	D
enewable electricity rocurement	Renewable	energy procurement const	tructs not disclosed.	0	?
4 CLIMATE CONTI	RIBUTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	No information	on on contributions or offset	ting claims.	0	0
Climate contributions		— No climate contributio	ns identified.	N/A	0
Offsetting claims today	·	— No offsetting claims too	lay identified.	N/A	N/A
Offsetting plans for the	TUTUre	et less than 5% of 2018 s1 an ditionality' credits, to achiev		\bigcirc	\bigcirc

 Rating criteria
 3-point scale
 High
 Moderate
 Poor
 See methodology document for rating criteria.

 Transparency refers to the disclosure of information.
 Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Arcelor Mittal (2021, 2022a, 2022b).

ArcelorMittal

ArcelorMittal, headquartered in Luxembourg, was the world's second biggest steelmaker by volume of crude steel production in 2021. Over 80% of its reported emissions are related to the steelmaking process, which includes emissions from mining. ArcelorMittal has pledged to reach net-zero emissions by 2050. To achieve its goal, the company has set out a decarbonisation roadmap including concrete measures but has not committed to zero or nearzero emissions steelmaking.

Although ArcelorMittal estimates it will offset not more than 5% of its emissions, its 2050 net-zero target remains ambiguous without a clear commitment to reduce its own emissions. In its *Net Zero Roadmap*, ArcelorMittal presents its net-zero target for 2050, complemented with interim targets for 2030. All targets cover scope 1 and 2 emissions, which together represent almost 90% of the total value chain emissions reported by the company in 2019 (ArcelorMittal, 2021b, pp. 58–63). As part of its plan to achieve net zero, ArcelorMittal stated it will offset residual emissions through the purchase of high-quality carbon credits, and currently estimates that these residual emissions (ArcelorMittal, 2021a, p. 12). Although ArcelorMittal estimates that this volume of residual emissions in 2050 will be relatively small, it does not explicitly commit to a specific level of emission reductions. We were unable to find more details on what types of credits will be used regarding the activities generating the credits, the purchase price, and the date when ArcelorMittal plans to start purchasing credits to claim emission reductions.

It remains unclear whether ArcelorMittal's 2030 target requires action beyond Europe. The company has a global target of reducing emission intensity by 25% below 2018 levels by 2030. This target translates to a company-wide emissions intensity of $1.54 \text{ tCO}_2\text{e}/$ tonne steel (ArcelorMittal, 2021a), which falls short of existing 1.5°C compatible benchmarks for the steel sector ($1.13 \text{ to } 1.35 \text{ tCO}_2\text{e}/\text{tonne steel}$) (CAT, 2020b; Boehm *et al.*, 2021; Boehm *et al.*, 2022; Dietz, Gardiner, and Scheer, 2022). However, a direct comparison of company targets to sectoral benchmarks is not possible without further information about reporting boundaries. ArcelorMittal's target also includes emissions from mining, which hinders a direct comparison with available benchmarks. Meanwhile, its European target of 35% intensity reduction would lead to an emission intensity of $1.11 \text{ tCO}_2\text{e}/\text{tonne}$. ArcelorMittal's European operations represent the majority of the company's crude steel production (53%), so it is possible that lowering the carbon intensity of steel production in line with their European target would bring the company's global average intensity close to their global target without action outside of Europe. This could compromise the company's plans to reach net zero globally by 2050, as most of its climate action over the mediumterm may be focused on one specific region. ArcelorMittal presents two main decarbonisation pathways, but it does not set a clear plan to completely phase out its emission-intensive infrastructure. As part of its *Net Zero Roadmap*, ArcelorMittal plans to develop several Direct Reduced Iron -Electric Arc Furnace (DRI-EAF) facilities. DRI-EAF is currently the most promising technology for deep emission reductions in primary steel production towards 2050; it can use natural gas or hydrogen. The company also plans to support the increase of green hydrogen production both through its own assets and in the European market. These plans include collaborations with actors in Germany that would enable green hydrogen supply to some of their plants (ArcelorMittal, 2021a). At the same time, ArcelorMittal also plans to adapt its existing Blast Furnace-Basic Oxygen Furnace facilities to use bioenergy with carbon capture, utilization, or storage (CCU/S). Both technologies are expected to play a role in decarbonising the steel sector over the short- to medium-term (de Villafranca *et al.*, 2022). While the DRI-EAF pathway can lead to scalable, low-risk zero emissions steel when powered by renewables and green hydrogen, the use of bioenergy and CCUS presents several potential challenges that limit its actual emission reductions (de Villafranca *et al.*, 2022), including the scalability of sustainably sourced biomass, and the effectiveness of carbon capture mechanisms, among others.

ArcelorMittal is not clear about its plans for renewable electricity. As part of its *Net Zero Roadmap*, the company acknowledges that a switch to DRI-EAF will substantially increase its power needs, and states that it plans to focus on sourcing renewable electricity through PPAs and RECs to address this (ArcelorMittal, 2021a). However, ArcelorMittal does not provide any details on what specific types of constructs will be used, and how much of their energy will come from renewables and/or be self-generated. More details on the procurement constructs that ArcelorMittal plans to rely on are needed to better understand the potential climate impact of its renewable electricity strategy.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Retail (consumables)	USD 87.8 bn (2021)	136.3 MtCO ₂ e (2021)	Carbon neutrality by 2040	O Very low	O Very low
1 TRACKING AND DIS	CLOSURE OF EMIS	SIONS		TRANSPARENCY	(& INTEGRITY
Iracking and disclosureMajor emission sources: Upstream and packaging (72%), mostly from a use of non-food products (23%).		ng (72%), mostly from agi		Scope 1 Scope 2	0.
Subsidiaries are not cove	reports. Rep	99% of s3 emissions are e ported values include only t for less than 20% of Car	y 'integrated' stores	Scope 3 upstream downstream	113. 21.
2 SETTING EMISSION	REDUCTION TARG	BETS		TRANSPARENCY	INTEGRITY
leadline target or pledge	Carbon neut	ality by 2040			
hort- and medium-term ta up to 2030)		ion of s1 & 2 emissions by 2 ion of s3 emissions by 203			
Scope coverage		Most Carrefour locations a		\bigcirc	\bigcirc
Own emission reductions (compared to full value chain in	6-1 En/	rom the targets; we interp apply to 20-50% of Carrefo			
ong-term vision beyond 2030)	Carbon neut	ality by 2040			
Scope coverage Own emission reductions (compared to full value chain in	2019) less than 0	arget includes a commitme of only s1 and s2 emissions, ocations.		\bigcirc	0
3 REDUCING OWN EM	ву 2050			TRANSPARENCY	INTEGRITY
	Plans addres	s key emission sources but	are limited in		
nission reduction measure	geographica	coverage and lacking conc	crete commitments.		\bigcirc
enewable electricity rocurement		of RE currently. Organisat identified. Geographical c is unclear.		0	0
4 CLIMATE CONTRIBU	JTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for Inabated emissions	No informatio	on on contributions or offset	ting claims.	0	0
Climate contributions		— No climate contribution	ns identified. ———	N/A	0
Offsetting claims today		— No offsetting claims too	lay identified	N/A	N/A
ffsetting plans for the futu		e provided although the 709 ality pledge implies a signific		\bigcirc	?

 Rating criteria
 3-point scale
 High
 Moderate
 Poor
 See methodology document for rating criteria.

 Transparency
 refers to the disclosure of information.
 Integrity
 refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Carrefour (2022a, 2022b, 2022c).

Carrefour

Carrefour S.A. – headquartered in France – is a major global retailer, with around 14,000 convenience stores and supermarkets in over 40 countries. Its supply chain for products and packaging constitutes the major emission source, mostly from agriculture. Carrefour's public reporting of its climate impact and pledges contains significant transparency deficiencies: public GHG emission reporting continues to exclude 98% of the company's emission footprint, and the company's targets and measures appear to exclude more than 80% of Carrefour locations. Due to these scope exclusions and unexplained neutralisation plans, we interpret that Carrefour's 2040 carbon neutrality target entails a commitment to the reduction of less than 1% of its emissions.

Key developments over the past year: We identified no significant changes to Carrefour's climate strategy since the previous iteration of this analysis was published in February 2022. In 2022, the company set out an improved portfolio of emission reduction measures that address key supply chain issues, but concrete commitments remain ambiguous. Carrefour has not provided clarity on the scope exclusions and uncertain neutralisation plans that undermine the company's targets.

Carrefour's carbon neutral by 2040 target entails a commitment to eliminate less than 1% of the company's emission footprint. Carrefour announced a new carbon neutrality target for 2040 on the eve of COP26 in October 2021, accompanied by the more specific target to reduce its scope 1 and 2 emissions by 70% by 2040, compared to 2019 (Carrefour, 2022a, p. 3). This is a direct update of its existing SBTi approved 55% reduction by 2040 target. The significance of these targets is highly limited, since scope 1 and 2 emissions account for less than 2% of Carrefour's emissions. Carrefour's separate interim target of 29% emission reductions from scope 3 emissions account for 98% of the company's footprint, but these emissions are not covered by the 2040 carbon neutrality target. Moreover, Carrefour's targets exclude the majority of Carrefour locations, which are explicitly excluded from the company's GHG reporting in its most recent annual reports (Carrefour, 2022c, p. 163).

The continued inconsistent disclosure of GHG emissions does not facilitate a good understanding of the company's emissions. As with the targets set, Carrefour's disclosure of emissions in its 2020 Annual Report and the 2021 Annual Financial Report includes only a small subset of the company's emission sources and locations: emissions are reported for the company's 'integrated' stores only: this includes less than 11% of Carrefour's 5,799 stores in France, and less than 20% of its 13,894 stores worldwide. Other administrative buildings, warehouses, and supply chain emissions associated with over 80% of Carrefour stores worldwide are excluded (Carrefour, 2022c, p. 163). In addition to these exclusions, Carrefour's reporting of its scope 3 emissions has the potential to be highly misleading. Although Carrefour acknowledges that scope 3 emissions account for 98% of the company's emissions (Carrefour, 2022c, p. 74), and breaks these emissions down with reasonable detail in its CDP responses, the company's main public documentation discloses less than 1% of these scope 3 emissions (Carrefour, 2022c, p. 159). Accordingly, Carrefour reports total company emissions of 1.79 MtCO₂e in its 2021 Annual Financial Report, although full value chain emissions reported to CDP amount to 136 MtCO₃e (Carrefour, 2022b).

Carrefour's emission reduction plan discusses key emission sources throughout the supply chain, but commitments are highly limited in their scope or remain ambiguous. Although Carrefour does not assume responsibility for scope 3 emissions in its GHG emission reporting or its headline carbon neutrality target, the company does have a separate target to reduce these emissions by 29% by 2030 (Carrefour, 2022a, p. 3). Its 2021 Annual Finance *Report* includes a more detailed portfolio of emission reduction measures to achieve this, compared to the 2020 Annual Report. The significance of these measures is limited by the fact that – like the company's emission disclosure and targets – these measures are limited to only its integrated stores in selected countries. Carrefour's strategy to reduce food waste by 50% by 2025 compared to 2016 is substantiated by specific actions for its stores as well as measures to engage consumers. The company also set out a package of measures to transition to less emission intensive products, including measures to increase consumer awareness on lower-carbon product alternatives. These measures represent a step in the right direction for the food and agriculture industry, although they are not complemented by a clear vision for how far that transition should go beyond 2030. While Carrefour sets out individual plans for avoiding deforestation for seven key agricultural products, its 'Zero Deforestation' policy falls short of a commitment to completely eliminate deforestation, as it rather sets targets for all materials to be 'covered by a risk reduction plan' for deforestation by 2025 (Carrefour, 2022a, p. 3). More stringent targets related to deforestation, as well as issues such an organic agriculture, are applied only to selected product lines (Carrefour, 2022c, p. 61).

Carrefour does not yet procure a significant volume of renewable energy but plans for 100% renewable electricity use by 2030, from higher quality sources. Since 2020, Carrefour started to install solar PV on selected stores, supplying a small but undisclosed proportion of the company's electricity demand in France (Carrefour, 2022c, p. 78). On the longer term, Carrefour plans to establish Power Purchase Agreements (PPAs) and to install more renewable energy capacity on its own sites, achieving 100% renewable electricity by 2030 (Carrefour, 2022c, p. 78). It is commendable that Carrefour plans to implement their renewable electricity target with higher quality constructs, but the company is a laggard regarding its lack of action on renewable energy to date, and could be more transparent about the barriers it faces to achieve this transition earlier than 2030.

Carrefour's plans for offsetting emissions and taking responsibility for unabated emissions remain unclear. Carrefour does not currently procure carbon offsets to offset its own emissions. It may, however, intend to do so for its carbon neutrality by 2040 pledge, which was accompanied by a 70% emission reduction target for the same year, but with no further details regarding the remaining emissions (Carrefour, 2022a, p. 3). Carrefour also does not take responsibility for unabated emissions through making a climate contribution.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Transport and logistics	USD 96.6 bn (2021)	45.7 MtCO ₂ e (2021)	Reduce all logistics-related emissions to net zero by 2050	D Moderate	Low
1 TRACKING AND DISC	CLOSURE OF EMIS	SSIONS		TRANSPARENCY	& INTEGRITY
acking and disclosure 5.7 MtCO ₂ e in 2021	(upstream t	ion sources: Emissions fr ransportation and distrib emissions; scope 1 emiss	ution) account for 66%	Scope 1 Scope 2	7. 0.
Subsidiaries are covered.	non-logistic	Aggregate reporting excl s-related emissions (14% full climate impact of avia	6 of total reported	Scope 3 upstream downstream	37.
2 SETTING EMISSION	REDUCTION TARC	BETS		TRANSPARENCY	INTEGRITY
leadline target or pledge	Reduce all log	gistics-related emissions to	o net zero by 2050		
Short- and medium-term tar (up to 2030)		nd s2 by 42% and selected and se	s3 emissions		
Scope coverage		Emissions increased substa argets translate to a redcu		\bullet	\bigcirc
Own emission reductions (compared to full value chain in 2	12%	compared to 2019 emission			
Long-term vision (beyond 2030)	Net-zero logi	stics-related emissions by 2	2050		
Scope coverage Own emission reductions (compared to full value chain in 2	? 1	No clear emission reduction Non-logistics emissions exc		0	\bigcirc
3 REDUCING OWN EMI				TRANSPARENCY	INTEGRITY
		over all emission sources. Th	hev include investments		
mission reduction measures		solutions for electric vehic			
enewable electricity rocurement		ricity consumption comes RECs account for over 95		0	0
4 CLIMATE CONTRIBU	TIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	Offsetting cla	im at product level.		0	0
Climate contributions		— No climate contributio	ns identified. ———	N/A	\bigcirc
Offsetting claims today		eutral delivery' through offset sets) and 'insetting' (reductio	tting (low volume of medium- ons in the value chain).		0
Offsetting plans for the futu		l be 'fully compensated by re etting)' but no explanation a		\bigcirc	\bigcirc

Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Deutsche Post DHL (2019, 2021, 2022a, 2022b, 2022c, 2022d, 2022e, 2022f, 2022g, 2022h, 2022i).

Deutsche Post DHL

Deutsche Post DHL Group (hereafter: Deutsche Post DHL) is the world's largest logistics provider. Its GHG emissions stem mostly from its own and subcontracted transport (scope 1 and upstream scope 3). Deutsche Post DHL commit to netzero logistics-related GHG emissions by 2050 and implements a range of measures to decarbonise its business model. However, it remains unclear whether its net zero pledge is a commitment to deep decarbonisation and the company's 2030 targets fall short of the required emission reductions in the transport and logistics sector.

Key developments over the past year: We identified some positive developments to Deutsche Post DHL's climate strategy since the previous iteration of this analysis was published in February 2022. The logistics provider improved the transparency of its emissions disclosure, although there still is room for improvement. Deutsche Post DHL also clarified that its net-zero target covers logistics-related emissions, but it remains unclear whether this target represents a commitment to deep decarbonisation.

Deutsche Post DHL does not comprehensively disclose all GHG emissions and other climate impacts from its business activities. The logistics provider prominently reports that its 2021 emissions amounted to 39 MtCO₂e in 2021 but this number excludes at least 5.8 MtCO₂e of upstream and downstream scope 3 emissions (Deutsche Post DHL, 2022a, 2022c). Deutsche Post DHL reports on emissions from purchased goods and services, capital goods, and employee commuting in its annual ESG Statbook, but leaves out these emission sources from the aggregate emissions totals in the Statbook and its annual sustainability report (Deutsche Post DHL, 2022c, 2022a). The company does not disclose other scope 3 emission sources, including waste and investments, at all. Seventy per cent of Deutsche Post DHL's logistics-related emissions come from air freight, while road transport accounts for 22% (Deutsche Post DHL, 2022c). The company discloses SOx, NOx and particulate matter emissions but does not report on the full climate impact of its aviation activities, including the radiative forcing impact of contrail cirrus. Deutsche Post DHL could provide a more comprehensive indication of its climate footprint by including the non-GHG climate-forcers from aviation in its annual reporting, since those are responsible for about two thirds of aviation's climate impact (Lee *et al.*, 2021).

It is unclear whether Deutsche Post DHL's net zero target is a commitment to deep decarbonisation of logistic activities. The logistics provider aims to reduce its logistics-related emissions (87% of all emissions disclosed in 2019) to the 'unavoidable minimum' (Deutsche Post DHL, 2022a, p. 18). Deutsche Post DHL could enhance transparency by indicating the order of magnitude for residual emissions that cannot be 'avoided' and what measures the company will take to 'fully compensate' these. Sectoral benchmarks for air freight show reductions of at least 80% by 2050, while road transport and ocean freight needs reduce emissions by over 90% (IEA, 2021b, p. 199; Smith *et al.*, 2021, p. 106; CAT, 2022a, 2022b; SBTi, 2022e, p. 11; Teske *et al.*, 2022, pp. 213, 216). Further, Deutsche Post DHL could improve the integrity of its net-zero pledge by including emissions from purchased goods and services, capital goods and employee commuting, which accounted for 13% of disclosed emissions in 2019.

Deutsche Post DHL's targets for 2030 fall short of sectoral 1.5°C Paris Agreement-aligned benchmarks for aviation, shipping, and road transport. The company committed to reduce scope 1 and 2 emissions by 42% by 2030 and scope 3 emissions from energy and fuel-related activities, upstream transportation and distribution, and business travel by 25% (Deutsche Post DHL, 2022a). Both targets use 2021 as the baseline year and translate to a committed reduction of 12% across the value chain between 2019 and 2030. They replace a target that Deutsche Post DHL had announced in early 2021: to reduce logistics-related emissions to below 29 MtCO₂e by 2030 (Deutsche Post DHL Group, 2021). This target translated to a 11% reduction in full value chain emissions between 2019 and 2030. Deutsche Post DHL is not yet on track to meet its 2030 target. Due to business growth and impacts of the COVID-19 pandemic, the company's emissions increased by almost 20% between 2019 and 2021 (Deutsche Post DHL, 2022c, p. 21). The IEA shows that CO_2 emissions from air and ocean freight need to decrease by 23% and 20%, respectively, between 2019 and 2030, and emissions from road transport by 33% (IEA, 2021b, p. 199). Some other sectoral benchmarks require even larger reductions.

Deutsche Post DHL is pioneering alternatives for internal combustion engines and focuses its reduction measures also on other transport modes. Deutsche Post DHL implements a range of reduction measures that address all relevant emission sources, including sustainable aviation fuels, a shift from road to rail, e-vehicles and e-trikes, and installing on-site renewable energy installations on its German distribution centres (Deutsche Post DHL, 2022a, pp. 25–31). Provided that these alternative transportation modes are based on renewable electricity or other sustainable energy sources, they can significantly reduce scope 1 and upstream scope 3 emissions. Deutsche Post DHL states that 86% of its electricity consumption comes from renewable resources (Deutsche Post DHL, 2022a, p. 22), but the lion's share of this electricity is procured through RECs (Deutsche Post DHL Group, 2022b). In 2020, PPAs and on-site installations accounted for just 0.6% of the company's renewable electricity consumption (Deutsche Post DHL, 2022b). As Deutsche Post DHL's reduction targets depend on the shift to e-vehicles and alternative fuels, it is critical that the company pursues renewable electricity procurement constructs that are likely to contribute to *additional* capacity, including PPAs for new installations and on-site installations (see Section 3.2.2).

Deutsche Post DHL presents biofuels as part of its decarbonisation measures for ocean, air, and road transport but also recognises that these may have negative sustainability implications (Deutsche Post DHL, 2019, pp. 11–12, 2022h, 2022f). While biofuels may play a role in decarbonising air freight, maritime shipping and heavy road transport, overreliance on bio-based fuels risks hindering the transition in other sectors and may indirectly lead to negative sustainability impacts (see Section 3.2.1)

Deutsche Post DHL offers climate neutral delivery through offsetting and so-called insetting practices, which could give customers a false impression of the company's activities' true climate impact. Under the 'GoGreen' label, Deutsche Post DHL has offered its customers 'climate neutral' delivery of mail and parcels since 2011. However, less than 1% of Deutsche Post DHL's total emissions are actually covered by the 'climate neutral' claim and the offset credits that the company procures are highly unlikely to have the same climate impact as emission reductions within the value chain (Deutsche Post DHL, 2022d, 2022e) (see further information in Section 4.2.2).

In 2022, Deutsch Post DHL announced a new label for its deliveries in Germany: 'GoGreen Plus' (Deutsche Post DHL, 2022b). The company promises customers one year of climate neutral deliveries through offsetting under the term 'insetting,' if they pay a premium of EUR 3.79 (USD 4.5). In this context, insetting means that Deutsche Post DHL uses the small sum that each customer pays to implement reduction measures elsewhere in the value chain. We consider 'climate neutral' deliveries based on 'insetting' controversial and potentially misleading to customers. Given that Deutsche Post DHL includes all logistics-related emissions in its 2030 and net-zero targets, it is highly possible that the company could double count the emission reductions achieved through their GoGreenPlus programme: once to claim the 'neutralisation' of the emissions associated with the delivery of specific letters and parcels, and a second time to claim progress towards its scope 3 emission reduction targets. It would be more transparent if Deutsche Post would tell its customers that the company uses the paid premium towards emission reductions in the value chain without making a 'climate neutral delivery' claim.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Fashion retailing	USD 19.4 br (2021)	5.50 MtCO ₂ e (2021)	To reduce scope 1 and 2 emissions by 90% and scope 3 emissions (from the production of raw materials for products, fabric production and garment manufacturing) by 20% by FY2030	D Moderate	Low
1 TRACKING	AND DISCLOSUI	RE OF EMISSIONS		TRANSPARENC	Y & INTEGRITY
racking and discl .50 MtCO ₂ e in 20	121		rocurement of materials in the ally raw material production, ng (upstream scope 3).	Scope 1 Scope 2	0.01
Subsidiaries a		•	losure. No estimate for on and distribution. Market- and provided for scope 2 emissions.	Scope 3 upstream downstream	4.8 0.4
2 SETTING E	MISSION REDUC	TION TARGETS		TRANSPARENCY	INTEGRITY
Headline target o	rplodao	•	ons by 90% and scope 3 emissions (from th manufacturing) by 20% by FY2030	e production of raw materials	for products,
Short- and mediu (up to 2030)	ım-term targets	Emission reductions by 203 - 90% in operations (s1, s2) - 20% in raw materials, fabric (s3, Uniqlo and GU brands)	. ,		
Scope coverage Own emission re	ductions	from brands other scope 3 emissions	isions footprint. Exclusion of emissions r than Uniqlo and GU, and downstream r reductions across the company's whole	•	0
(compared to full v	value chain in 2019)	by 2030 Value chain are ins	sufficient to meet the global benchmark of on reductions by 2030.		
Long-term vision (beyond 2030)	1	Net zero by 2050			
Scope coverage Own emission re (compared to full)	eductions value chain in 2019)	emissions cover	n is unclear. Concrete plans and rage have not been made public. erm vision is set out.	\circ	\bigcirc
	OWN EMISSION			TRANSPARENCY	INTEGRITY
mission reductio	n measures		sources, e.g., energy efficiency cycling, emission reduction plans for gauge impact.	D	?
Renewable electri procurement			ans underway for 100% RE by apan, RECs in other regions.	0	0
4 CLIMATE C	CONTRIBUTIONS	AND OFFSETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissio		No information identified.		0	0
Climate contribu	itions	——— No climate	contributions identified.	N/A	0
Offsetting claims	s today	No offsettin	g claims today identified.	N/A	N/A
Offsetting plans f			ced any plans to offset emissions in the o aim for carbon neutrality by 2050.	\bigcirc	N/A

Rating criteria 3-point scale High Moderate Poor . See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Fast Retailing (2021, 2022a, 2022b).

Fast Retailing

Fast Retailing Co., Ltd. (Fast Retailing) is a Japanese fashion retail multinational company that owns the Uniqlo brand, among others. Most of its emissions stem from the production of clothes in third-party supplying factories. To tackle these emissions, the company provides suppliers with emission reduction plans and related support, but it discloses few details on the goals and ambitions of these plans. The company's headline emission reduction targets fall far short of what is needed to limit global warming to 1.5°C. It wants to reach net-zero emissions in 2050 but has not announced plans for emission reductions after 2030.

Fast Retailing's headline pledges collectively amount to a reduction of 19% of the company's emissions footprint by 2030 compared with 2019, which is not aligned with global efforts to limit global warming to 1.5°C. Fast Retailing has two targets that are to be achieved by 2030, certified to be 1.5°C-compatible by the SBTi. The first one is 20% absolute reduction of supply-chain emissions, specifically from raw materials, fabric, and garment production for the Uniqlo and GU brands, which represent ~95% of the company's revenue (Fast Retailing, 2021, p. 64). The second one is 90% absolute reduction of operational emissions (scopes 1 and 2, under direct company control) (Fast Retailing, 2021, p. 29). These two targets cover 74% of the company's emissions footprint and exclude emission sources like the end-of-life treatment of sold products. They equate to an emissions reduction commitment of just 19% across the full value chain, compared to 2019 levels, falling short of the IPCC's estimate of minimum 43% global GHG emission reductions by 2030 to keep warming below 1.5°C (IPCC, 2022). Furthermore, a sectoral 1.5°C-aligned benchmark indicates that fashion retailers should reduce upstream scope 3 emissions by a minimum of 41% between 2019 and 2030 (Teske, 2022), but Fast Retailing only commits to 16% upstream scope 3 emission

Fast Retailing has expressed the aim to reach net-zero emissions by 2050, but it has not disclosed a concrete plan for emission reductions between 2030 and 2050. Fast Retailing's climate strategy focuses on 2030 emission reduction targets. Although it expresses the intention to 'strengthen our efforts to achieve net zero GHG emissions by 2050' (Fast Retailing, 2021, p. 29), it does not clarify what these efforts would entail in the future. The company does not disclose whether it will reach the net-zero target through further emission reductions or through emissions offsetting in the 2030–2050 period.

Fast Retailing's emission reduction measures focus on emission reduction plans for supplying factories, but details on how the company engages with suppliers—the company's main source of emissions—remain limited. To address emissions from corporate sites, Fast Retailing aims to improve energy efficiency by installing LED lighting, automatic temperature control, and more efficient air conditioning (Fast Retailing, 2021, p. 29, 2022b). However, these measures only target 5% of the company's emissions footprint (scope 2) (Fast Retailing, 2022b). Fast Retailing's most significant source of emissions (76% of its emissions footprint) is the manufacturing of garments in third-party factories, including raw material production, fabric production, and sewing. To address these emissions, the company reports to be recycling more materials and piloting clothing repair stations (Fast Retailing, 2022b; Uniqlo, 2022). While repair stations could potentially extend products' lifetimes, they would have a significant impact in emission reductions only if they would lead to a shift in consumer behaviour and a reduction in the volume of new garments purchased and produced. We did not identify any clear indication that Fast Retailing is preparing to shift away from a fast fashion business model. Fast Retailing also claims to cooperate with suppliers by providing them with suppliertailored emission reduction plans (Fast Retailing, 2021, p. 29, 2022b). These are mentioned to include energy-saving and renewable-energy measures, like eliminating coal energy from manufacturing processes, and Fast Retailing pledges support for their implementation (Fast Retailing, 2021, p. 29). However, the company does not provide specifics on the coverage and depth of the emission reduction plans, what its pledged support entails, or how it aims to enforce them. While this level of supplier interaction may represent good practice, the lack of details does not facilitate a complete understanding of whether Fast Retailing's strategy will be sufficiently effective at reducing supply-chain emissions.

Fast Retailing aims to use 100% renewable electricity by 2030, but it does not commit to procurement options that would likely result in additional renewable electricity capacity. In 2021, less than 10% of Fast Retailing's electricity consumption was sourced from renewable sources (Fast Retailing, 2022a). The company recently installed solar PV capacity at 13 stores in Japan, and it will source renewable electricity for other stores around the world. However, it does not specify the types of constructs that it will pursue to source renewable electricity; the quality of the electricity procurement constructs will determine whether the company's 2030 target will result in truly additional renewable electricity capacity and the abatement of the company's energy-related emissions.

Foxconn	EVENUE	EMIGOLONIO		TRANORADENOV	INTEODITY	
Technology - USE	EVENUE) 214.6 bn (2021)	EMISSIONS 29.4 MtCO ₂ e (2020)	PLEDGE Net-zero emissions in the value chain by 2050	Low		
1 TRACKING AND DISCLO	SURE OF EMIS	SIONS		TRANSPARENC	Y & INTEGRITY	
acking and disclosure 2.4 MtCO ₂ e in 2020 Subsidiaries are covered.	disclosed en breakdown Disclosure: I reported for	ion sources: S3 accounts nissions, but Foxconn pro of emission sources. Incomplete disclosure. S r 2020, no breakdown pr ported for s2 emissions (ovides no 3 emissions only rovided. Highest	Scope 1 Scope 2 Scope 3	0.: 5.: 24.	
2 SETTING EMISSION REI	DUCTION TARG	ETS		TRANSPARENCY	INTEGRITY	
eadline target or pledge	Net-zero emi	ssions in the value chain b	by 2050			
hort- and medium-term target up to 2030)		ions reduction targets, 20). Net-zero emissions in Ta	20 baseline: 21% by 2025, iwanese offices by 2030.			
Scope coverage Own emission reductions (compared to full value chain in 2020)	42%	Inclear whether targets c emissions. The pledged 42 Ilmost aligned with a 1.5°C	% reduction by 2030 is			
ong-term vision beyond 2030)	Net-zero emi	ssions in the value chain	by 2050.			
Scope coverage Own emission reductions (compared to full value chain in 2020	63 %	full coverage of value-cha eductions by 2035 (2020 arget alongside net-zero				
3 REDUCING OWN EMISSI	IONS			TRANSPARENCY	INTEGRITY	
nission reduction measures	suppliers to p	ation. Energy-saving meas procure RE and set targets rered is unclear.		0	?	
enewable electricity ocurement	Details on pr	electricity is renewable, w rocurement constructs n from on-site installations	ot provided, although	0	0	
4 CLIMATE CONTRIBUTIO	NS AND OFFSI	ETTING		TRANSPARENCY	INTEGRITY	
esponsibility for nabated emissions	No informatic	on identified.		0	\bigcirc	
Climate contributions		— No climate contributio	ons identified.	N/A	\bigcirc	
Offsetting claims today		— No offsetting claims to	day identified. ———	N/A	N/A	
ffsetting plans for the future		nat extent Foxconn plans to sets to reach its future targ	0	?		

RATINGS Overall 5-point scale High Reasonable Moderate C Low Very low. Average of sections 1-4. Sections 1-4 5-point scale High Reasonable Moderate C Low Very low. Average of the criteria in each section. Rating criteria 3-point scale High Moderate Poor . See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Foxconn (2021, 2022).

Foxconn

Hon Hai Precision Industry Co., Ltd. (Foxconn) is a Taiwanese multinational electronics corporation that manufactures electronic products for major companies such as Apple, Sony, and Nintendo. It established a net-zero emissions target by 2050 and increasingly stringent emission reduction targets up to 2035. The company does not yet specify the level of emission reductions that it will pursue after 2035 to claim net zero by 2050. Most of Foxconn's emissions are indirect (scope 3, 82%), but the company does not disclose the relevance of specific scope 3 emission sources. Its emission reduction strategy remains unsubstantiated, as the company does not share its complete emissions footprint or concrete, detailed emission reduction measures.

Foxconn commits to a carbon emission reductions path leading to 63% reductions by 2035, compared to 2020. Although this pathway up to 2035 is likely aligned with a 1.5°C warming trajectory, the company does not commit to further emission reductions to reach its 2050 net-zero target. Foxconn's headline pledge is a long-term target to achieve net-zero emissions in the value chain by 2050 (Foxconn, 2022, p. 94). The company commits to reduce carbon emissions by 21% by 2025, by 42% by 2030, and by 63% by 2035, from a 2020 baseline (Foxconn, 2022, p. 94). Foxconn could more transparently state if these targets apply to its complete value chain, and whether they cover all GHGs, as it uses the terms 'carbon emissions' and 'emissions' interchangeably when referring to the targets (Foxconn, 2022, pp. 11, 94). The targets draw a consistent path of reducing emissions up to 2035. If we assume that they apply to the full value chain and to carbon dioxide emissions specifically, Foxconn's goals would be close to meeting the IPCC's benchmarks to keep global warming below 1.5°C: 48% CO₂ emission reductions by 2030 and 80% CO₂ emission reductions by 2040 (IPCC, 2022). Clarifying that its 2050 net-zero target entails further emission reductions, with a limited role for any offsetting of residual emissions, would send a clearer signal for investments and long-term decarbonisation beyond 2035.

Foxconn's emissions disclosure is incomplete and lacks detail, making it difficult to understand the company's overall emissions reduction strategy. Foxconn disclosed a scope 3 emissions estimate for the first time for the year 2020, without providing a breakdown into specific emission sources (Foxconn, 2021, p. 100, 2022, p. 95). The total emissions footprint that year amounted to almost 30 MtCO₂e, of which 82% corresponded to scope 3 emissions. Foxconn has not yet shared a scope 3 emissions estimate for the year 2021 (Foxconn, 2022, p. 95). Due to the lack of a complete and detailed emissions disclosure, we are not able to tell what Foxconn's most relevant emission sources are and whether the company is addressing them sufficiently in its climate strategy.

Foxconn presents limited information on emission reduction measures, which include energy efficiency improvements and the transition of its suppliers to renewable electricity. In its sustainability report for the year 2021, Foxconn outlines a work plan for energy and carbon reduction measures (Foxconn, 2022, p. 96). It includes implementing government regulations for energy efficiency, formulating energy-saving plans, and pursuing the certification of energy management systems. However, Foxconn does not provide more detail on these measures, nor has it shared a timeline for the expected emission reductions, or the share of emissions targeted with each measure. For example, in promoting 'green manufacturing systems' (Foxconn, 2022, p. 96), the company does not explain what 'green' means, and what specific measures it will implement to make its processes 'greener'. Foxconn requires suppliers to commit to achieve net-zero emissions by 2050 and establish long-term emission reduction targets (Foxconn, 2022, pp. 82, 86). The company appraises suppliers' progress and reduces transactions with poor performers. To support suppliers, Foxconn wants to 'guide' them in manufacturing products with 100% renewable electricity (Foxconn, 2022, pp. 87–88). In 2021, Foxconn's guidance included investments in solar PV or procurement of RECs for at least five suppliers, and the company is planning to expand this to over 220 suppliers in the 2022–2050 period. However, Foxconn does not disclose how many suppliers it has in total, the emissions that could be reduced from their transition to renewable electricity, and the extent to which they may rely on RECs.

Foxconn currently consumes a low share of renewable electricity and provides insufficient details on the renewable electricity procurement constructs that it pursues. Foxconn aims for half of its energy consumption (electricity, steam, heat, and fuel) to be renewable by 2030 (Foxconn, 2021, p. 102, 2022, p. 11). The company currently has a 5% share of renewable electricity consumption. Almost two thirds (61%) of this limited renewable consumption is produced on site with solar PV in Chinese campuses and 39% is directly purchased in China and other countries (Foxconn, 2022, p. 99). The lack of information about the contracts that the company signs to purchase renewable power (e.g., whether the electricity bought is generated in the same location where it is consumed) does not allow us to determine if the purchase results in truly additional renewable capacity.

Google	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
	JSD 257.54 bn (2021)	16 MtCO ₂ e (2021)	Net-zero emissions by 2030	Moderate	Moderate
TRACKING AND DIS	TRANSPARENCY & INTEGRITY				
ncking and disclosure MtCO ₂ e in 2021		ion sources: S3 emissions om consumed electricity (Scope 1	0.0
Subsidiaries are covered	location-bas of the two u	63 reporting lacks detail; r ed s2 emissions disclosec sed for aggregated emissi	l, but the lowest	Scope 2 Scope 3	6. 9.
2 SETTING EMISSION	REDUCTION TARG	ETS		TRANSPARENCY	INTEGRITY
eadline target or pledge	Net-zero emi	ssions by 2030			
nort- and medium-term ta p to 2030)		iissions by 2030 % of consumed electricity w	ith 24/7 RE by 2030.		
Scope coverage		Commitment to reduce mar y 50% by 2030, which trans		\bigcirc	\circ
Own emission reductions (compared to full value chain in	37%	eduction of location-based			
ong-term vision eyond 2030)	Google does i	not have any long-term targ	gets beyond 2030.		
Scope coverage Own emission reductions (compared to full value chain in	?	lo emission reduction plans	identified after 2030.	0	0
3 REDUCING OWN EM	ISSIONS			TRANSPARENCY	INTEGRITY
ission reduction measure		uction measures across all emains unclear. Emissions		O	0
enewable electricity66% of consumed electricity is matched with 24/7 RE (100% annually). RE from PPAs (77%), the grid (23%), and on-site installations (0.05%).				•	
4 CLIMATE CONTRIBU	ITIONS AND OFFSI	ETTING		TRANSPARENCY	INTEGRITY
esponsibility for nabated emissions	No informatic	n on contributions or offsett	ing claims.		0
Climate contributions		nces mitigation projects throu missions. Limited detail on pro			?
Offsetting claims today	purchases	claim covers ~12% of 2021 e and offsets. We interpret thi	s claim as misleading.	0	0
fsetting plans for the fut		to offset 67% of its emission tion on the future offset por			\bigcirc

Sections 1-4 5-point scale High Reasonable Moderate Course Average of the criteria in each section. Rating criteria 3-point scale High Moderate Poor See methodology document for rating criteria.

Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Alphabet (2022), Google (2022a, 2022b) and Google.org (2022).

Google

Alphabet Inc., the holding company of Google LCC (hereinafter referred to as Google), is a provider of diverse information technology services and products. Its major emission sources stem from product manufacturing and use, and electricity consumption in data centres. Google's net-zero target for 2030 equates in reality to a 37% emission reduction commitment, with no commitment to deeper decarbonisation beyond 2030. The company's plans for the decarbonisation of its electricityrelated emissions are comprehensive and innovative, but it is unclear if the targets and measures for other emission sources are sufficient, especially for scope 3 emissions, which represent the majority of Google's GHG emission footprint.

Key developments over the past year: We could identify only a few changes to Google's sustainability strategy since our previous analysis of the case study in the 2022 Corporate Climate Responsibility Monitor (Day et al., 2022), including a substantiation of its net-zero pledge with an emission reduction target. Accordingly, only minor modifications were made to this case study.

Google's headline pledge is to reach 'net-zero emissions' by 2030 while keeping its continuous goal of 'carbon neutrality' each year. We consider both claims misleading as they are not substantiated with deep emission reduction commitments. The 'carbon neutrality' claim excludes major scope 3 emission sources that accounted for 58% of the company's GHG emission in 2021 (Google, 2022b, p. 11). Emission sources covered by the target are 'neutralised' through procurement of renewable energy and offset credits that have highly contentious environmental integrity (see Table 3-2, Section 3.2.2). The provided guidance of what Google perceives as 'high-quality' offsets dates back to 2011 (Google, 2011). Google could update that guidance to the newest available standards to improve the integrity of its 'neutralisation' claims. Google's net-zero target for 2030 covers the company's entire operations and value chain emissions. In 2022, Google clarified that this target entails a 50% reduction of its market-based emissions using a location-based accounting method (Google, 2022a, p. 5). This is an improvement compared to a year ago, when the company had not made any commitment alongside its net-zero target (Day *et al.*, 2022). However, a 37% reduction commitment implies that Google will claim to 'neutralise' the majority (67%) of its real emission footprint with carbon offset credits by 2030 or potentially other creative accounting methods. It remains unclear what the portfolio of fister projects will look like, as Google provides only limited information on this. Google acknowledges that a shift towards carbon dioxide removal credits is required to align with the ambition set out in the Paris Agreement but also claims that in the short-and medium-term those credits are not economically feasible at scale (Google, 2022a, p. 12). Google does not transparently disclose whether these carbon dioxide removal measures will be based on biological, geological or mineral carbon storage. Google plans to use 'avoided emission credits'

Google has been developing a comprehensive portfolio of renewable energy generation and procurement since 2017 and demonstrates a good understanding of the limitations of various renewable energy procurement options. The company claims to have 'matched' its energy consumption with renewable energy generation between 2017 and 2021 (Google, 2022b, p. 9). In 2021, 77% of renewable energy procurement stemmed from on-site generation and PPAs within the same grid as the electricity consumption (Google, 2022b, p. 12). Google demonstrates a good understanding of the limitations of various renewable energy procurement options: the company notes that only 66% of its electricity use in 2021 was matched on an hourly basis with regional carbon-free sources, which is one percent less than in 2020 (Google, 2022b, p. 9). Google pledges to increase this share to 100% and achieve 24/7 carbon-free energy by 2030 (Google, 2022b, p. 9). This good practice approach has been adopted by other companies, and the innovative technologies that Google is developing to implement and monitor progress against this target may in the future support other actors and grid operators to optimise their own decarbonisation measures. Google reports both location-based and market-based scope 2 emissions, but the latter is used to discount electricity emissions from aggregated companywide emission statistics. This is somewhat inconsistent with Google's shrewd observation that purchasing enough renewable energy to match annual electricity use may reduce but not eliminate emissions. More prominent reporting of location-based emissions would be more transparent and constructive, given the recognised limitations of its current renewable energy procurement.

Google implements a range of measures to reduce emissions across most emission sources, but more information is needed to judge if they are sufficient. Google reports on a range of emission reduction measures, particularly related to improving energy efficiency in datacentres and office buildings (Google, 2022b, pp. 8–10). Flagship project investments for high-tech and data-driven efficiency in selected office locations can have a positive impact for demonstration purposes and enabling replication of good practice, but we could not identify whether Google has mainstreamed these measures across a large proportion of its locations. Google engages with its suppliers to reduce its scope 3 emissions, invests in renewable energy projects located on the same grid as its main manufacturing regions, and works on improving the energy efficiency of its hardware products (Google, 2022a, p. 11). Still, we could not determine the sufficiency of these measures for scope 3 emissions due to the lack of information on the depth and breadth of measures as well as the poor granularity of GHG emission footprint – into a single data point, for 'business reasons' (Alphabet, 2022, pp. 35, 38). Despite the range of implemented emission reduction measures Google's absolute emission increased by 5% in the last year, mainly driven by an increase in emission from consumed electricity (scope 2 emissions increase in demand for Google's cloud and datacentre services indicated by the increase in revenue.

Google invested around USD 70 million in the advancement of innovative climate solutions through its charity Google.org (Google, 2022a, p. 21). Google.org is a Google-owned charity that provides USD 200 million annually to different non-profits and social organisations to foster innovation and support disadvantaged communities (Google.org, 2022). In the past, Google.org has helped restore ecosystems, enable small businesses to track their emission footprint, and support cities through data-driven climate solutions in areas such as mobility optimisation and mapping solar energy potential (Google, 2022a, p. 21; Google.org, 2022). It is unclear what share of the annual funding is targeted towards climate change mitigation. Increased transparency on the use of the Google.org fund would help to show whether these financial contributions represent good practice

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Fashion retailing	USD 23.2 bn (2021)	8.2 MtCO ₂ e (2021)	Net zero no later than 2040	Moderate	D Moderate
1 TRACKING AND DI	SCLOSURE OF EMIS	SIONS		TRANSPARENC	Y & INTEGRITY
acking and disclosure 2 MtCO ₂ e in 2021		ion sources: Fabric prod (21%) are the most impo		Scope 1 Scope 2	0.0:
Subsidiaries are covere	purchased g	Detailed breakdown of e goods (scope 3, category scope 3 categories are n	1).	Scope 3 upstream downstream	5.5
2 SETTING EMISSIO	N REDUCTION TARG	BETS		TRANSPARENCY	INTEGRITY
eadline target or pledge	Net zero no lo	ater than 2040			
hort- and medium-term up to 2030)		d s2 emissions by 56% by 3 emissions also by 56% by	,		
Scope coverage Own emission reductions (compared to full value chain	56% by 2030	Aligned with global benchr GHG emission reductions, vithin the next 5 years.			
ong-term vision beyond 2030)		rality by 2040			
Scope coverage Own emission reductions (compared to full value chain	by 2040	Commitment to reduce s vell as s3 emissions by 90	,	•	•
3 REDUCING OWN E	MISSIONS			TRANSPARENCY	INTEGRITY
nission reduction measu	res and sustaina recycling an	he production phase has ability implications. Poter d sourcing more sustaina to limited information.	ntial impact of	0	0
enewable electricity rocurement	RECs cover of	over 95% of electricity co	onsumption.	\bullet	0
4 CLIMATE CONTRIE	UTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	H&M Group j	oined the LEAF coalition.		\bigcirc	?
Climate contributions		up joined the LEAF coalitior Ild not identify further deta		\bigcirc	?
Offsetting claims today		— No offsetting claims to	day identified. ————	N/A	N/A
ffsetting plans for the fu	ture DAC removal	emissions will be offset. H& ls, but LEAF coalition meml iological CDR.			0

Sources: Authors' interpretation of identified public documentation from H&M Group (2021, 2022a, 2022b, 2023).

H&M Group

H&M Group is a Sweden-based fast fashion retailer that comprises eight brands, including H&M, COS and Monki. The majority of H&M Group's emissions stem from fabric production, garment manufacturing and raw materials (all scope 3, category 1), which jointly account for over half of reported emissions. Although H&M Group has ambitious emission reduction targets for 2030 and 2040, those may be severely undermined by the lack of a clear decarbonisation plan and reliance on biomass in the production phase.

H&M Group plans to reduce emissions across its value chain by 56% by 2030 and by 90% by 2040. These are ambitious targets that signal the need for immediate climate action, as long as they are not undermined by reliance on RECs and biomass to claim decarbonisation of the supply chain. In 2022, H&M Group replaced its 'climate positive by 2040' target with a commitment to achieve 'net zero CO₂ e emissions by 2040, taking a climate positive approach and having a net positive impact on biodiversity' (H&M Group, 2022b, p. 18). The net-zero target is accompanied by the commitment to reduce emissions across the value chain by 90% (H&M Group, 2023). As part of its net-zero commitment, H&M Group committed to reduce emissions across the entire value chain by 56% between 2019 and 2030. This ambition level goes beyond the global benchmark for a 1.5°C-compatible emission reduction trajectory. However, the true ambition level of H&M Group's targets depends on the measures used to achieve them. We see signals that the company could be planning to rely heavily on RECs and biomass to claim the decarbonisation of its supply chain, which could severely compromise these targets.

While H&M Group refers to a wide range of emission reduction measures, more detailed information is needed to understand the likely reduction impact. Decarbonising the fashion sector requires a diverse set of reduction measures, including reducing overproduction, phasing out coal, switching to renewable energy, maximising material and energy efficiency, and ramping up the development of innovative materials (Berg *et al.*, 2020; Ley *et al.*, 2021; Sadowski *et al.*, 2021). While H&M Group refers to most of these measures in its public communications, it remains unclear to what extent these will reduce emissions across the value chain (H&M Group, 2022b, pp. 31–46). For instance, the company lists various innovative materials it is investing in but does not outline at what scale these could be used in the next decade and what their emission reduction potential is. H&M Group refers to various initiatives through which it procures 'more sustainable' materials, including cotton (H&M Group, 2022b, p. 31). However, more sustainable can simply mean a small improvement compared to the baseline and does not necessarily entail a commitment to shift to truly sustainable materials. More details are needed to understand the mitigation potential of shifting to alternative materials. While H&M Group offers repair and rental services and is expanding its sales of second-hand clothes (H&M Group, 2022b, pp. 41–42), these initiatives can help to significantly reduce emissions *only if* they are scaled up to the point at which they lead to a significant reduction in the number of new clothes produced. However, we could not identify clear plans to reduce (over)production and shift away from a fast fashion business model to a less resource intensive model that focuses on longer-lasting products.

H&M Group committed to phasing out coal in the supply chain, but this pledge is undermined by a heavy reliance on biomass. The company announced it will no longer onboard any new suppliers with on-site coal boilers in their factories and is committed to phase out coal in its supply chain. The company works on addressing barriers to transitioning to biomass boilers in Indonesia and plans for Cambodia to be H&M Group's first production country to use 100% biomass boilers (H&M Group, 2022b, p. 24). While biomass, along with natural gas, are options that are currently available to replace coal in thermal processes, neither are suitable long-term decarbonisation options. Biomass is associated with severe sustainability implications, including deforestation, food insecurity and release of sequestered carbon into the atmosphere (see Section 3 in the Methodology). The very limited potential for sustainable biomass should be used in sectors that have no alternatives for fossil fuel combustion. The fashion sector, however, has these alternatives and needs to transition to zero-carbon options, including green hydrogen and concentrated solar (Ley *et al.*, 2021, p. 21). As those options are not yet viable at commercial scale today, large investments are needed to unlock their decarbonisation potential.

H&M Group commits to 100% renewable electricity in its supply chain by 2030 but it is unclear what share of suppliers are covered by this target and what electricity procurement options H&M Group pursues. H&M Group states in a footnote that the goal of 100% renewable electricity in the supply chain only applies to markets where renewable electricity is available for purchase (H&M Group, 2022b, p. 19). This statement potentially undermines the target's ambition level, as H&M Group provides no information on the share of suppliers that would not be able to source renewable electricity. More details are also needed to understand whether or not H&M Group proactively supports its suppliers in sourcing renewable electricity – and other forms of energy - in countries that have insufficient policies in place to stimulate and allow for renewable energy capacity development. We see a risk that H&M Group abandons any responsibility for electricity-related emissions in its supply chain in countries where there are no renewable electricity certificates already available for sale.

H&M Group also commits to 100% renewable energy in its own operations, but this target will only result in real emission reduction if the renewable energy is sourced from high-quality constructs. H&M Group purchased RECs to claim that 95% from its electricity consumption came from renewable sources in 2021, up from 90% in 2020 (H&M Group, 2022b, pp. 21, 24). RECs do generally not contribute to the installation of additional renewable energy capacity and are not a suitable approach for companies to reduce their electricity-related emissions (see Table 3-2, Section 3.2.2). However, we see some early signs that H&M may move to high-quality procurement options. In December 2022, H&M Group signed a PPA for a new and large solar park in Sweden and stated that 'PPAs play a very important role in the company's climate work to achieve its targets.' (H&M Group, 2022a).

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Steel and cement	USD 29.3 bn (2021)	149.1 MtCO ₂ e (2021)	Net zero by 2050	Reasonable	D Moderate
1 TRACKING AND DISC	CLOSURE OF EMIS	SSIONS		TRANSPARENCY	(& INTEGRITY
racking and disclosure 49.1 MtCO ₂ e in 2021		sion sources: S1 related to 3 purchased goods and ser		Scope 1	119.3
7.1 MtCO ₂ e in 2021		S2 emissions reported usi	•	Scope 2	7.0
Subsidiaries are covered.	uses the hig	-based accounting approa her value to present total 3 sub-categories are exclu	emissions. Unclear	Scope 3 upstream downstream	22. 7.
2 SETTING EMISSION	REDUCTION TARG	GETS		TRANSPARENCY	INTEGRITY
eadline target or pledge	Net-zero by 2	2050			
hort- and medium-term tar up to 2030)	- of cementiciou - of bought clinl - of bought fuel	ce the carbon intensity: Is materials below 2018 by 22.5% ær and cement below 2020 by 20 below 2020 by 20% for s3 (fuel- ts below 2020 by 24% for s3 (dow)% for s3 (purchased goods) and energy related)		
Scope coverage		2030 intensity targets for s			
Own emission reductions (compared to full value chain in 2		1.5°C-aligned sectoral benc emission reductions cannot			
ong-term vision beyond 2030)		50, including: duction in CO2/t cement belov ction in absolute emissions belo			
Scope coverage Own emission reductions (compared to full value chain in 2	? I by 2050	Targets meet 1.5°C-aligner out intensity target and he makes the appropriatenes cerminology contentious.	eavy reliance on CCUS		
3 REDUCING OWN EMI	SSIONS			TRANSPARENCY	INTEGRITY
nission reduction measures	emission red	over all emission sources, in luction estimates and plans it limitations of CCUS are no	to mainstream innovative		0
enewable electricity rocurement		e of energy procured by co onse, showing a high reliar		0	0
4 CLIMATE CONTRIBU	TIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	Product-leve	l offsetting claims.		\bigcirc	0
Climate contributions		— No climate contribution		\bigcirc	\bigcirc
Offsetting claims today	emissions	ict-level, the company offers under its 'ECOPact Zero' lab id volume of credits disclosed	el. No information	0	?
ffsetting plans for the futu		ategy includes a small role fo ive recarbonation of sold pro			

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Holcim (2021, 2022a, 2022b).

Holcim

Holcim Limited is a Swiss-based multinational and one of the world's largest cement manufacturers by production capacity. Most of its emissions stem from cement production (scope 1), especially clinker production and fuel combustion in cement kilns. Holcim has pledged to reach net zero across its entire value chain by 2050, providing clear emission and intensity reduction targets alongside. The company presents measures to address most of its value chain emissions but does not fully clarify the role of emission removals in its strategy.

Holcim set a 2050 net-zero target alongside deep decarbonisation targets covering all its emission sources. Holcim presents its headline pledge of reaching net zero by 2050 alongside *own* emission reduction targets along the value chain: reducing the GHG-intensity of cement by 95% below 2018 for scopes 1 and 2 and reducing absolute scope 3 emissions by 90% below 2020. These targets are in line with existing 1.5°C-aligned benchmarks for the cement sector (see full assessment in Annex II). However, Holcim's interim targets of reducing the GHG intensity of cement by 22.5% for scope 1 and 65% for scope 2 by 2030, both below 2018 levels, only partially align with 1.5°C-compatible benchmarks. Therefore, it remains unclear whether Holcim's 2030 targets are fully in line with a 1.5°C-compatible trajectory to meet its 2050 net-zero target.

Holcim sets out mitigation measures to address all value chain emissions. In its net-zero strategy, Holcim clearly communicates its mitigation plans and measures classified by emission scope, including scope 3 emissions . While the company provides an illustrative pathway indicating each measure's contribution towards net-zero scope 1 and 2 emissions, it stops short of disclosing specific emission reduction estimates for 2030, and the intended amount of removed emissions through passive carbonation. The company showcases many of its initiatives to potentially reduce the GHG intensity of cement products, including recycling construction and demolition waste, reducing the amount of clinker in cement, switching out fossil fuels for waste biofuels, and capturing carbon (Holcim, 2022b, pp. 15–21). Holcim does not clearly acknowledge the challenges associated with such measures, for example related to the availability of waste-fuel supply, or the effectiveness and feasibility of CCUS in cement plants (Lehne and Preston, 2018, pp. 7–9; Bataillle, 2020, pp. 5–8; Fennell Davis and Mohammed 2021 pp. 5–6)

Holcim plans to rely on passive carbonation to reach net zero by 2050, adding uncertainty to the real emission reduction impact of its strategy. As part of its pathway to net zero, Holcim includes emission removals from passive carbonation. Passive carbonation is a *natural* process through which cement reabsorbs CO_2 during its lifecycle and stores it permanently. The actual amounts of sequestered CO_2 depend on many external factors such as the weather, the use of coatings, characteristics of the soil and others (Holcim, 2022b, p. 34), but can reach up to 26% of cement's manufacturing emissions including those from fuel as well as process emissions (Xi *et al.*, 2016). Currently, Holcim does not clearly quantify the emissions removal it plans to achieve through passive carbonation, or its assumptions regarding capture rates per unit of material. Without such information, the credibility of Holcim's carbonation plans remains in question. However, based on Holcim's scope 1 and 2 pathway to net zero (Holcim, 2022b, p. 14) we estimate that the share of the company's emissions expected to be removed through passive carbonation is conservative and not likely to undermine its climate strategy.

Holcim mostly procures renewable electricity through low quality RECs and may have PPAs with fossil-based power producers. Holcim's latest CDP disclosure indicates that lowquality RECs account for the majority of Holcim's renewable electricity purchases (Holcim, 2022a, pp. 29–32). In its public documents, Holcim also recently announced several PPAs with local producers for some of its plants (Holcim, 2022b, pp. 18–19). In 2021, the company signed PPAs for renewable electricity of 500 GWh, which represented around 5% of Holcim's total electricity consumption in the same year (Holcim, 2021, p. 8). The PPAs' impact remains unclear because Holcim fails to provide further information on whether the associated RECs are transferred or cancelled. If Holcim would not purchase certificates alongside its PPAs, it could enable other actors to claim emission reductions using such certificates, leading to the double counting of renewable electricity (see Table 3-2 Section 3.2.2). Holcim provides little information on the emissions impact of its own captive power plants and whether it plans to phase-out all its GHG-intensive power generation. Additionally, Holcim's scope 2 emissions are lower under location-based reporting that market-based reporting. This suggests the company has PPAs with fossil-based power producers in place, making its purchased electricity more carbon-intensive than the grid average.

Holcim makes a contentious product-level claim to offer 'carbon neutral cement' based on the use of offset credits, but the company discloses neither the volume nor the type of credits procured. Under its ECOPact product line, Holcim promotes cement with a 30–90% lower carbon footprint compared to standard (CEM I) concrete (Holcim, 2022b, p. 33). The company further promotes 'carbon neutral' ECOPact^{zero} concrete by offsetting an undefined amount of emissions through offset credits. Holcim neither clarifies the extent nor any sustainability criteria for these credits. The promotion of low-carbon cement based on offsetting can be misleading for end consumers, especially as currently existing technologies in the cement sector are not sufficient for (near-)zero emissions production (Lehne and Preston, 2018, pp. 7–9; Bataillle, 2020, pp. 5–8; Fennell, Davis and Mohammed, 2021, pp. 5–6).

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Fashion retailing	USD 32.8 bn (2021)	13.9 MtCO ₂ e (2021)	Net-zero GHG emissions by 2040	O Moderate	Low
1 TRACKING AND	DISCLOSURE OF EMIS	SIONS		TRANSPARENC	Y & INTEGRITY
Major emission sources: S3 emissions account for 99.6% of the total. Most relevant categories are purchased goods and services (59%) and use of sold products (21%).Disclosure: Almost all s3 reported, except for downstream		Scope 1 Scope 2 Scope 3	0.0:		
Subsidiaries are covered. transport and distribution. S2 reported under market and location-based methods, but lowest estimate used for targets.				upstream downstream	1: 0.4
2 SETTING EMISSI	ION REDUCTION TARG		TRANSPARENCY	INTEGRITY	
leadline target or pledo	Je Net-zero GH	G emissions by 2040			
Short- and medium-terr up to 2030)	n targets -Reduce s3 e	2 emissions by 90% below missions from purchased g v 2018 levels by 2030.			
Scope coverage	S) S2 S3 ₁	argets represents a 10% r	eduction in total		\bigcirc
Own emission reduction (compared to full value cha	10% e	emissions, well below 1.5°C or the fashion industry			
ong-term vision beyond 2030)	Net-zero GH	G emissions by 2040			
Scope coverage Own emission reduction (compared to full value cha	is ?	Net-zero target scope no No own emission reductio Ilongside net-zero pledge	on target presented	0	0
3 REDUCING OWN	EMISSIONS			TRANSPARENCY	INTEGRITY
mission reduction meas		nformation is provided to and sufficiency of emissior		0	?
enewable electricity rocurement		not publicly disclose RE p ses indicate high reliance		0	0
4 CLIMATE CONTR	BUTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for Contributions identified.					0
Climate contributions		ntribution to nature conser I volume (eq. to EUR 0.24/t0			\bigcirc
Offsetting claims today		— No offsetting claims to	-	N/A	N/A
Offsetting plans for the	future target. Memb	plans for emission neutrali pership of the LEAF coalitio carbon dioxide removals.		0	?

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Inditex (2022a, 2022b).

Inditex

Industria de Diseno Textil S.A. (Inditex) is a Spanish-based multinational fashion retailer, better known for its flagship brand, Zara. It is the biggest fast fashion group in the world by revenue, with EUR 27.7 billion in 2021. Most of its emissions stem from its supply chain, especially those related to the sourcing and processing of raw materials. The headline pledge for net zero emissions by 2040 is ambiguous and unsubstantiated; the company does not provide a clear commitment to reduce its emissions by any more than 10% below 2019 levels, and the sufficiency of its emission reduction measures remains unclear.

Due to the lack of transparency around Inditex's net-zero pledge, it is not possible to assess what the commitment entails. Most of Inditex's emissions are scope 3 emissions (~97% in 2021). However, based on its public reports, we cannot independently assess the scope of Inditex's target, or the share of own emissions reduction it would entail. Inditex plans to achieve its net-zero pledge by neutralising residual emissions but does not clarify its approach in terms of the total share of emissions which will be offset, or the types of credits it will purchase (Inditex, 2021a, p. 250). With these ambiguities, the net-zero pledge does not entail any specific commitment regarding the breadth and depth of emission reduction plans.

The company's interim 2030 targets amount to a 10% emissions reduction below 2019 levels, which is not clearly aligned with global efforts to limit global warming to 1.5°C. Inditex has two SBTi-certified targets: to reduce scope 1 and 2 emissions by 90% below 2018 levels by 2030, and to reduce scope 3 emissions from purchased goods and services by 20% below 2018 levels by 2030 (Inditex, 2021b, p. 250), although the latter target appears to fall short of SBTi's own emission reduction benchmark for scope 3 emissions in the apparel sector. This equates to a commitment to reduce emissions across the value chain by just 10-12% compared to 2019 levels. This level of ambition falls well short of sector-specific benchmarks and Inditex is well on track to meet these targets with limited further action. Through its untransparent renewable energy claims, Inditex claims to have already reduced scope 1 and 2 emissions by 86% in 2021 compared to 2018 levels (already achieving 95% of its target). By 2021, the company had also achieved half of its 2030 goal for scope 3 emissions. It is not clear to what extent Inditex' targets really represent an effort that requires significant further action.

Inditex has also committed to sourcing 100% renewable electricity by 2022 but appears to rely on low quality procurement constructs. In 2021, the company reported it covered 91% of its electricity needs with renewable energy (Inditex, 2021b, p. 254), but information from non-public documentation indicates that this is based extensively on the purchase of RECs (Inditex, 2022, pp. 30–37). If Inditex plans to achieve its target largely or exclusively through the procurement of unbundled renewable energy certificates, its actions would be very unlikely to lead to an increase in renewable energy capacity and the abatement of the company's energy-related

97

emissions. Inditex also invests in increasing its own renewable energy generation capacity. In 2021, it reported generating 5,920 MWh from its own solar and wind assets (Inditex, 2021b, p. 255), which remain only a small share (0.3%) of its total electricity consumption.

Inditex' emission reduction measures are not detailed enough to understand their potential significance. The company's sustainability strategy showcases many measures and initiatives targeting different emission sources throughout its value chain (Inditex, 2021b, p. 178,260). It has also committed to mainstream some of its current initiatives, such as sourcing sustainable cotton, but without specifying what that means or how it compares to traditional cotton. Without clear indications of the expected scope and emissions impact of such measures, it is not possible to assess how significant they are. Raw material procurement represents an especially challenging but key emission source for the deep decarbonisation of the sector. While Inditex is working towards improving the durability of their products and extending their life (Inditex, 2021b, p. 194), it is not clear if such measures will achieve the scale required to lead to a significant reduction in their emissions. Reaching net zero through deep decarbonisation will require fashion retailers to move away from a quantity-focused fast fashion business model to a less resource intensive production model. However, Inditex does not set out a clear plan to prepare the business for such a transition.

It is not clear if and how Inditex plans to procure carbon credits to neutralise its emissions. In 2021 the company joined the LEAF coalition (Inditex, 2021b, p. 165), which aims at leveraging finance for tropical forest protection, including through carbon credits. It is unclear whether Inditex' participation in the LEAF coalition represents a climate contribution or a means of claiming offsets. Inditex made no claims of offsetting its own emissions through the purchase of carbon credits in 2021, nor did it state that it will achieve its 2040 target through the offset credits from the LEAF programme. Due to the scarcity of land for forestry-related carbon dioxide removal, as well as the nonpermanence of these removals, such measures cannot represent a credible equivalent to the reduction of emissions, but it would be a more credible approach to support such projects through a contribution claim. In 2022, Inditex partnered with WWF and pledged to provide around EUR 10 million between 2022 and 2024 to support projects focusing on biodiversity and forest protection (Reklev, 2022). We consider this as a form of climate contribution, as Inditex does not clearly plan to purchase or claim any carbon credits through it. However, the volume of finance – equivalent to approximately 0.01% of annual revenue and approximately EUR 0.24/tCO₂e compared to full value chain emissions – is far from a volume that would represent a credible approach for assuming responsibility for emissions.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Food, beverages and agriculture	USD 64.7 bn (2021)	71.1 MtCO ₂ e (2021)	Net-zero GHG emissions by 2040	O Very low) Very low
1 TRACKING AND	DISCLOSURE OF EMIS	TRANSPARENCY	(& INTEGRITY		
acking and disclosure 1 MtCO ₂ e in 2021	no breakdov	ion sources: Major emiss vn provided, we assume t attle are the main emissio	hat emissions related	Scope 1 Scope 2	4.
Subsidiaries are cove	2021 sustai	IBS reports 10 times larg nability report than in pro a clear explanation for th	evious ones. We could	Scope 3	6
2 SETTING EMISS	ION REDUCTION TARG	BETS		TRANSPARENCY	INTEGRITY
leadline target or pledo	ge Net-zero GH	G emissions by 2040			
Short- and medium-tern up to 2030)	m targets Reduce s1&2	emissions by 30% by 203	0 from a 2019 baseline.		
Scope coverage	t	Jnclear language around ir arget. Does not cover s3 (>		\bigcirc	0
Own emission reduction (compared to full value cha	IS 3 %				
eyond 2030) Net-zero emissions by 2040					
Scope coverage Own emission reduction (compared to full value cha	ıs ? a	BS does not specify an emi longside its net-zero emiss ufficiency of that long-terr	sions pledge; the	\circ	?
3 REDUCING OWN	EMISSIONS	TRANSPARENCY	INTEGRITY		
mission reduction mea	potential imp	duction measures described in vague terms; pact remains unclear. Unclear how most mission sources (rearing cattle) will be reduced.		0	0
enewable electricity rocurement	46% of elect	information on RE procur ricity demand is RE-basec nd planned procurement	l, but little to no info	0	?
4 CLIMATE CONTR	IBUTIONS AND OFFS	TRANSPARENCY	INTEGRITY		
Responsibility for unabated emissions	Climate contr	ontributions with insufficient finance volume.			0
Climate contributions Fund for the Amazon projects may have a positive climate impact. Sum of support is not proportional to JBS' GHG footprint.				0	
Offsetting claims today		— No offsetting claims today identified. ————		N/A	N/A
Offsetting plans for the		et emissions towards the ne etails identified.	t-zero target for 2040.	\bigcirc	?

Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from JBS (2021, 2022) and Fund for the Amazon (2022a, 2022b, 2022c).

JBS

JBS S. A. (JBS) is a meat processor headquartered in Brazil. Over 90% of its reported emissions are from farms and feedlots for rearing cattle. The company plans to continue growth in a GHG emission-intensive industry; we did not find evidence of any planned deep decarbonisation measures. JBS does not have an emission reduction target alongside its net-zero emission target for 2040. Its interim targets for 2030 would lead to a 3% emission reduction compared to its reported 2021 emissions.

Key developments over the past year: We could identify only minor changes to JBS's sustainability strategy since our previous analysis of the case study in the 2022 Corporate Climate Responsibility Monitor (Day et al., 2022). Most notably, JBS has increased its share of reported scope 3 emissions, although key emission sources are still excluded. Otherwise, only minor modifications were made to this case study.

JBS does not specify what share of its 2040 net-zero target will be based on emission reductions and what share will rely on offsetting. In its communication on its net-zero target, JBS says that it wants to reduce direct and indirect GHG emissions, while offsetting residual emissions (JBS, 2022, pp. 7, 63). We could not identify an emission reduction target accompanying its net-zero headline pledge. It is therefore unclear what share of JBS's emissions footprint will be offset by 2040. Given the limited detail on emission reduction measures and the expected continuous growth of the company, this share could be significant. JBS does not specify any details regarding what kind of offsetting projects it will procure credits from, the potential volume of credits it envisages needing, nor general criteria for ensuring robust environmental integrity in any offsetting claims it may make.

JBS's formulation of its interim emission intensity target for scope 1 and 2 is not transparent, and the target is highly insufficient, omitting the company's main emission sources. JBS says it wants to reduce scope 1 and 2 emissions intensity by at least 30% by 2030 compared to 2019 (JBS, 2021, p. 11), but is presented as an absolute target in its most recent sustainability report (JBS, 2022, p. 16). This inconsistency undermines the transparency of the target. Moreover, since its reported scope 3 emissions accounted for over 90% of its total emissions footprint in 2021 (JBS, 2022, p. 19), the target is also highly insufficient. We did not identify substantial emission reduction targets for JBS's most important emission sources in scope 3, which are primarily related to cattle rearing, including emissions from enteric fermentation, feed and manure.

We could not find significant details on how JBS intends to achieve its 2030 emission intensity target and 2040 net-zero target. Rather, JBS plans to continue growth in a GHG emission-intensive industry. We did not identify a comprehensive emission reduction strategy: the company provides minimal detail on how it wants to realise its targets. We did not find evidence that JBS intends to substantially innovate or diversify from its current activities. Without major innovations to drastically reduce the emissions footprint of meat production or diversifying away from this highly GHG emissions intensive industry, it is not credible for livestock agribusinesses to claim that they are on a path to deep decarbonisation. JBS describes it is in the process of developing an emission reduction strategy, but this is not expected before late 2023 (JBS, 2022, p. 10).

JBS aims for 60% renewable electricity in its facilities by 2030 and 100% by 2040 but provides little information about current and planned renewable energy supply constructs. The company claims that renewable electricity accounted for a major share of its consumption in 2021 (43%) (JBS, 2022, p. 68). JBS has some renewable energy generation on-site, using solar systems and residue biogas (JBS, 2022, p. 67). The company aims for 60% renewable energy by 2030 and 100% by 2040 (JBS, 2022, p. 16). To achieve this, on-site generation and high-quality energy procurement structures are necessary. However, the company does not specify what procurement constructs it currently uses and what it plans to use. It remains unclear whether these targets are credible.

JBS includes a larger share of emissions in its disclosure, but the data is still not presented transparently, and land-use changes related to rearing cattle are excluded. In its 2022 sustainability report, JBS included more emission sources in its emissions reporting, compared to its 2021 sustainability report (JBS, 2021, p. 43, 2022, pp. 64–65). JBS says that key emission sources including enteric fermentation, feed and manure management are included in its reported scope 3 emissions, but the company does not provide a breakdown of the emissions to these sources (JBS, 2022, pp. 64–65). Moreover, land-use change emissions related to rearing cattle are not covered for its emissions reporting for Brazil (JBS, 2022, p. 65). With the current level of detail, JBS's emissions disclosure does not allow for a thorough understanding of the emission sources and effectiveness of potential mitigation measures.

With its Fund for the Amazon programme, JBS will contribute to local sustainable development without claiming neutralisation, but these contributions are significantly lower than is required to be in proportion to the company's climate footprint, equating to only 0.01% of JBS' annual revenue. With projects such as 'RestaurAmazônia' and 'Release Credit for Forest Bioeconomy', JBS wants to support projects related to sustainable development in the Amazon biome (Fund for the Amazon, 2022b, 2022c). These projects are framed as contributions; we did not find evidence that JBS intends to claim neutralisation of emissions based on the projects' outcome. However, JBS will contribute a maximum USD 93 million to the fund up to 2030 (Fund for the Amazon, 2022a), equal to roughly 0.01% of its revenue. The volume of this financial contribution is equivalent to a carbon price on the company's emissions footprint of approximately just 0.13 USD per tonne CO_2e . This is substantially lower than the range of emerging carbon price recommendations for meaningful climate contributions, that equate to at least 80 USD per tonne of CO_2e (see section 4.2 of the Methodology).

SECTOR R	EVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
• • • • •	D 61.8 bn (2021)	78 MtCO ₂ e (2022)	Net zero across the business and value chain by 2040 with 100% green solutions for customers	Reasonable	Reasonable
TRACKING AND DISCLO	TRANSPARENCY	& INTEGRITY			
acking and disclosure MtCO ₂ e in 2022	accounts for	half of all emissions; up	mption on own vessels (s1) stream transportation and or 17% and 13%, respectively.	Scope 1 Scope 2	34.
Subsidiaries are covered.	separately or including bla	Disclosure of GHGs is c n SOx, NOx and particu ck carbon, but does not ose emissions.		Scope 3 upstream downstream	35.
2 SETTING EMISSION REI	DUCTION TARG	ETS		TRANSPARENCY	INTEGRITY
eadline target or pledge	Net zero acros	ss the business and value	e chain by 2040 with 100% g	reen solutions for custome	ers
hort- and medium-term target p to 2030)	^S 2020) and 25%	r ocean activities: 50% redu of cargo to be transported v rgo, logistics facilities, lands			
Scope coverage Own emission reductions (compared to full value chain in 2019	? T	bsolute emission reduct om the company's emiss he green fuel targets for 5°C-compatible benchm	sion intensity targets. shipping exceed		
ong-term visionNet zero across the business and value chain by 2040beyond 2030)with 100% green solutions for customers					
Scope coverage Own emission reductions (compared to full value chain in 2019	91% N	0% reduction from 2020 ectoral benchmarks. No on-GHG climate forcers ill decrease as Maersk n om fossil fuels to zero-c	explicit mention that are covered, but these nakes the transition	•	•
3 REDUCING OWN EMISSI	TRANSPARENCY	INTEGRITY			
ission reduction measures	and zero-cark	ve plans to shift from for oon alternatives. No com Is as an interim solution	mitment to using		
newable electricity ocurement	energy, but m	terminals are powered nore details are lacking. Installations in the future	Focus on PPAs	0	?
4 CLIMATE CONTRIBUTIO	NS AND OFFSE	TTING		TRANSPARENCY	INTEGRITY
esponsibility for nabated emissions	'Natural climat	e solutions' to sequester	at least 5 MtCO ₂ in 2030.		?
Climate contributions		move and store at least 5 eutralisation claim; furthe		?	
Offsetting claims today		— No offsetting claims to	-	N/A	N/A
Plans to 'neutralise' 10% of 2020 emissions through nature-based climate solutions to make a net-zero claim by 2040.					

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Maersk (2021, 2022a, 2022b, 2022c, 2022d, 2022e, 2022f, 2022g, 2022h, 2023a, 2023b).

See **Annex II** for more details on our assessments of companies' targets

Maersk

Maersk is one of the world's largest container shipping line and vessel operators. Close to half of Maersk's GHG emissions stem from fuel use on their ships; while upstream transportation and distribution accounts for about 30% of the company's GHG footprint. The company committed to net zero by 2040 and specified this includes a reduction of 90% across the value chain, compared to 2020 levels. Maersk could enhance its climate ambition by committing to use biobased fuels as an interim solution only, before making the full switch to zero-carbon technologies, such as e-fuels.

Maersk commits to reduce emissions across the value chain by 90% below 2020 levels but does not yet specify what type of carbon dioxide removals it will use to offset the remaining 10%. Maersk announced its net-zero target for 2040 in January 2022 and has recently specified this target entails a commitment to reduce scope 1, 2 and 3 emissions by 90% below 2020 levels [Maersk, 2023a, p.22]. Compared to 2019 emissions, this means a reduction of 91% across scopes 1, 2 and 3. Maersk's reduction commitment is in line with benchmarks for the maritime sector, most of which show that the shipping sector should fully decarbonise in the 2040s (IRENA, 2021, p. 81; Smith *et al.*, 2021, p. 106; SBTi, 2022e, p. 11; Teske, 2022, p. 213).

Maersk plans to invest in 'robust and third-party verified naturebased CDR projects, such as conservation and restoration projects (Maersk, 2022a, p. 22) but gives no further details on the integrity conditions it would impose on these projects. Although nature-based CDR projects need financial support, issues related to permanence and scarcity make such projects unsuitable to claim the neutralisation of own GHG emissions (see Section 4.4.2).

Maersk's disclosure of GHG emissions and other air pollutants is detailed but could be enhanced by reporting on the climate impact of black carbon emissions. Maersk provides a comprehensive overview of scope 1, 2 and 3 emissions. In its sustainability report over 2022, the company reported for the first time on black carbon emissions, which enhances the comprehensiveness of its disclosure (Maersk, 2023a, p. 36. 2023b). However, the company does not yet report on the climate impact of black carbon and other air pollutants. Black carbon is a climate forcer that warms the Earth through changes in surface albedo when it is deposited on snow and sea ice and by absorbing solar radiation in the atmosphere (CAT, 2022b). The International Maritime Organization estimates that black carbon accounted for about 7% of international shipping's emissions in 2018, making it the sector's second largest emission source (IMO, 2020, p. 110).

Maersk invests heavily in the development of various alternative fuels, but it is unclear if the company considers bio-based fuels only as an interim solution before making the full switch to more sustainable decarbonisation options. The company focuses its efforts on biodiesel from waste feedstocks, bio- and e-methanol, and e-ammonia, preferring not to use LNG and other transitional fuels (Maersk, 2022a, p. 21). Maersk recognises that the limited availability of sustainable feedstock and competing demand from other sectors means that bio-based fuels have limited potential. The company invests in ships that can sail on bio-based and e-fuels (Maersk, 2022d) and has signed various agreements with fuel suppliers to scale the production of e-methanol and biomethanol (European Energy, 2022; Maersk, 2022c; Ørsted, 2022). The suppliers of e-fuels, including Ørsted and European Energy, develop new renewable energy installations to produce those fuels. While it is commendable that Maersk invests in a range of alternative fuels, most of which are not yet available at scale, overreliance on bio-based fuels risks hindering the transition in other sectors and may indirectly lead to negative sustainability impacts. Given that the shipping sector can make the full shift to zero-carbon technologies, including e-fuels, by 2040, biofuels should only be used as a transitional fuel until those e-fuels become available at scale (ICCT, 2018). Maersk could enhance its climate plans by committing to use bio-based fuels only as an interim solution before shifting to zero-carbon fuels that have no sustainability implications.

In addition to its investments in alternative fuels, which will drive down scope 1 emissions, Maersk invests in decarbonising land-side infrastructure and its air transport activities (Maersk, 2022g, 2022e, 2022h). For instance, the company makes the shift to electric trucks, increases its rail transport offering and purchases sustainable aviation fuels for a share of its air freight. Although there is large need for massive investments in decarbonisation technologies for the aviation sector, the potential of SAF is constrained by energy demand in other sectors. The production of synthetic fuels requires large amounts of renewable energy that is currently not available. Maersk commits to decarbonising its terminals, including through the installation of on-site renewable energy capacity, electric container handing equipment and zero-emission shore power for vessels at berth (Maersk, 2021, p. 24). However, more information would be needed to evaluate Maersk's approach to procuring renewable electricity.

Maersk set an ambitious target for green fuel usage in 2030 but could enhance its short-term ambition by also committing to absolute emission reductions. Maersk is committed to transport 25% of all ocean cargo with green fuels that have 'low or very low greenhouse gas emissions on a lifecycle basis' by 2030 (Maersk, 2022b). This likely goes beyond scientific benchmarks showing that 5-17% of maritime fuels need to be zero-emission fuels by 2030 (Smith *et al.*, 2021, p. 11; UNFCCC, 2021, p. 15; Boehm *et al.*, 2022, p. 74). Maersk could further elaborate on this target by specifying the expected share of biobased fuels and assumptions underlying the calculation of fuels' lifecycle emissions. The shipping company is also committed to reduce the carbon intensity of its ocean activities by 50% compared to 2020 emissions, and has set separate targets for air cargo, terminals, logistic facilities and landside activities (see Annex II for a full overview) (Maersk, 2022a, p. 19). The company has not set absolute emission reduction targets for its scope 1 and 3 emissions. These would be needed to send a stronger signal for deep emission reductions in the next seven years.

SECTOR	des-Be	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Automobiles	USD 198.6 bn (2021)	198.6 bn 124.9 MtCO ₂ e CO ₂ -neutral ac		C Low	Low
1 TRACKING AND D	TRANSPARENC	Y & INTEGRITY			
acking and disclosure 24.9 MtCO ₂ e in 2021 Subsidiaries are cover	2021, downs in 2021, ups Disclosure: S prior to spin ed. emissions ca	ion sources: Use phase o stream s3), purchased go tream s3). 53 emissions reported fo -off of Daimler Truck AC ttegories not disclosed a y of downstream s3 disc	oods and services (16% r 2021 only (no data 5 in 2021). Several s3 nd third-party analysis	Scope 1 Scope 2 Scope 3 upstream downstream	0. 1. 23. 10
2 SETTING EMISSIO	IN REDUCTION TARG	ETS		TRANSPARENCY	INTEGRITY
eadline target or pledge	New vehicle f	leet CO₂-neutral across al	l stages of the value chain b	y 2039	
Chort- and medium-term up to 2030) Scope coverage Own emission reductions (compared to full value chain	- 50% absolute	s1 & s2 emissions intensity fee s1 & s2 emissions reduction b lo 1.5°C-aligned phaseout argets cannot be quantifee alue chain due to lack of h	dates for ICEs. 2030 d across the entire	D	0
ong-term vision beyond 2030)	New vehicle f of the value c	leet CO₂-neutral across al hain by 2039	l stages		
Scope coverage Own emission reductions (compared to full value chain	? .	Il emission scopes covere ommitment alongside car Io 1.5°C-aligned phaseout	bon neutrality pledge.	\bigcirc	0
3 REDUCING OWN E	MISSIONS	TRANSPARENCY	INTEGRITY		
nission reduction measu	investments	easures for key emission sources, including s in vehicle electrification and low-carbon steel Limited details on timeline and expected impact.		0	0
enewable electricity ocurement	only. On-site	ormation disclosed for Ge generation and PPAs (so orage to match consumpt	lar, wind, hydro) with	0	0
4 CLIMATE CONTRI	BUTIONS AND OFFSI	TRANSPARENCY	INTEGRITY		
esponsibility for nabated emissions		im for production plants wo h uncertain impact.	orldwide	\bigcirc	0
Climate contributions		No climate contributions identified		N/A	0
Offsetting claims today	Use of offs as of 2022.	Use of offsets to claim 'CO ₂ -neutral' production plants worldwide as of 2022. No information disclosed on volume and project types		\bigcirc	0
setting plans for the future The carbon neutrality target for 2039 target may depend on offsets, but no further details disclosed.			\bigcirc	?	

INGS Overall 5-point scale Ingn C Reasonable Moderate C Low Overy low. Average of sections 1-4.
Sections 1-4 5-point scale High Reasonable Moderate C Low Overy low. Average of the criteria in each section.
Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria.
Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Mercedes-Benz Group (2020a, 2020b, 2021a, 2021b, 2022a, 2022b, 2022c, 2022d, 2022e) and Bonaccorsi et al. (2022)

Mercedes-Benz Group

Mercedes-Benz Group AG is a German automobile manufacturer. Most of Mercedes-Benz Group AG's emissions originate in the use phase of its sold vehicles, accounting for more than 80% of its 2021 emissions. The company pledges to make its entire vehicle fleet ' CO_2 neutral' across all stages of the value chain by 2039 but provides no clarity on what this means for *actual* emission reductions. Mercedes-Benz commits to a wide range of emission reduction measures across all scopes as part of its 'Ambition 2039' programme, and high-quality renewables procurement for its Germany-based operations. Its climate strategy for vehicle electrification and its short-term targets, however, fall short of emission pathways aligned with the Paris Agreement's 1.5°C global warming limit for the automobile industry.

Mercedes-Benz sets a carbon neutrality target for 2039 that covers its entire value chain, but provides no information on what this target means in terms of actual emission reductions. It remains unclear whether and to what extent Mercedes-Benz will rely on offsetting to meet its 2039 target (Mercedes-Benz Group, 2022e, p. 130, 2022a, p. 23). For example, the company neither explains how it defines 'CO₂-neutral production material' as part of its commitment to exclusively purchase such materials by 2039 nor specifies if its suppliers can rely on offsetting to label a given material as 'CO₂-neutral' (Mercedes-Benz Group, 2020b, 2022a, p. 35). Mercedes-Benz claims that it has been purchasing 'CO₂-neutral' battery cells from two suppliers since 2021 (Mercedes-Benz Group, 2022e, p. 146), but these claims are not supported by any clarification on the respective suppliers' production processes.

Mercedes-Benz commits to several interim targets for 2022, 2025 and 2030 that are not aligned with key decarbonisation milestones compatible with the Paris Agreement's 1.5°C temperature limit, especially for use phase emissions of its sold vehicles. Mercedes-Benz vaguely commits to sell only electric vehicles 'wherever market conditions allow' by 2030 (Mercedes-Benz Group, 2022e, p. 130, 2022a, p. 23). The company does not commit to specific phaseout dates for internal combustion engines as part of its electrification roadmap in key markets such as the European Union, the US, or China (Mercedes-Benz Group, 2022a, p. 18). For all these key markets, a 1.5°C compatible trajectory would require that electric vehicles account for 95-100% of all light-duty vehicle sales by 2030 (Climate Action Tracker, 2020b, p. 27; Teske *et al.*, 2022, p. 4). Recent analysis suggests that Mercedes-Benz currently underreports its disclosed life-cycle emissions of sold ICE vehicles by more than 60% due to potentially unrealistic assumptions on vehicle lifetimes (Bonaccorsi *et al.*, 2022, pp. 15–16).

In November 2021 at COP26, Mercedes-Benz was the only German automaker to sign a declaration committing to a fully electric new vehicle fleet in leading markets by 2035 (COP26 Presidency, 2021). Since then, the company has not incorporated specific phase-out dates into its climate strategy. The company's political lobbying supports positions for and against more stringent climate action through its own interactions with regulators and membership in regressive industry associations in the EU and US (InfluenceMap, 2021, 2022b).

Mercedes-Benz implements emission reduction measures for all key emission scopes as part of its 'Ambition 2039' programme, for example, forward-looking actions to facilitate the switch to low-carbon steel procurement in the future. Mercedes-Benz commits to large investments to electrify its vehicles, accelerate the roll-out of charging infrastructure, and support battery technology and low-carbon steel production (Mercedes-Benz Group, 2022e, pp. 132–153, 2022a, pp. 30– 36, 2022c, 2022d). Mercedes-Benz outlines several measures to reduce its upstream supply chain emissions, which represent 19% of all emissions in 2021 (Mercedes-Benz Group, 2022e, pp. 145–147, 2022a, pp. 34–35). The company will procure low-carbon steel from suppliers such as Salzgitter Flachstahl GmbH and SSAB AB, and it purchased an equity stake in a the Swedish start-up H2 Green Steel (H2GS) in 2021 (Mercedes-Benz Group, 2021a, 2022a, p. 34, 2022e, pp. 169–170). H2GS is developing a technology to make steel using green hydrogen and renewable electricity, which will enable Mercedes-Benz as a preferential buyer to use low-carbon steel from 2025 onward. To enhance the transparency and accountability of its plans to procure low-carbon steel, Mercedes-Benz could communicate specific milestones to purchase low-carbon steel over time and the estimated emission reductions impact.

Mercedes-Benz provides limited information on the use of offset credits and renewable electricity procurement that form the basis for its contentious claim of 'carbon-neutral' production plants worldwide as of 2022. Mercedes-Benz uses a combination of site-specific emission reduction measures, procurement of renewable electricity, and offsetting to justify its claim of 'carbon-neutral' production processes (Mercedes-Benz Group, 2022a). The company neither specifies the extent to which it relies on offset credits, nor the type and prices for the credits purchased, making this claim highly ambiguous and contentious (Mercedes-Benz Group, 2022e, p. 150).

As for the procurement of renewable electricity at its production sites, the company installs own renewable electricity generation capacity at many of its sites and uses PPAs and RECs to procure all remaining electricity (Mercedes-Benz Group, 2022e, pp. 149–150, 2022b). In Germany, the Mercedes-Benz Group procures renewable electricity from solar, wind and hydro through PPAs and uses electricity storage facilities that match consumption and generation at all times (Mercedes-Benz Group, 2021b; Statkraft, 2021). The company provides limited information on renewable energy procurement for its production plants outside of Germany.

Microsof						
SECTOR F	EVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY	
Technology - USE Services) 168.09 bn (2021)	19.2 MtCO ₂ e (2021)	D ₂ e Carbon negative by 2030	Reasonable	O Moderate	
TRACKING AND DISCLO	TRANSPARENC	Y & INTEGRITY				
acking and disclosure .2 MtCO ₂ e in 2021	location-bas	ion sources: Emissions fro ed s2), purchased goods a tion of offices/datacentre	nd services (26%), and	Scope 1 Scope 2	0.	
Subsidiaries are covered (ZeniMax Media, acquired in Mar 2021, only included from start of 2022).	scopes, but i	Comprehensive disclosure market-based s2 estimate disclosure and targets.		Scope 3 upstream downstream	10. 4.	
2 SETTING EMISSION RE	DUCTION TARG	ETS		TRANSPARENCY	INTEGRITY	
eadline target or pledge	Carbon nega	tive by 2030				
hort- and medium-term targe p to 2030)		dges to become carbon negat laim of carbon neutrality e				
Scope coverage	SI S2 S3 (Commitment to reduce ma	rket-based emissions		\bigcirc	
Own emission reductions (compared to full value chain in 2019	50/0	ay 50% by 2030, which tran eduction of location-based				
ong-term vision eyond 2030)		move by 2050 all the carbo er directly or by electrical c in 1975'.	0	0		
Scope coverage Own emission reductions (compared to full value chain in 2014	? *	 No clear plans to reduce emissions beyond 2030. We interpret the target as a commitment to offset bictorical s1 and s2 emissions since 1975. 				
3 REDUCING OWN EMISS	TRANSPARENCY	INTEGRITY				
ission reduction measures	r most emission sources, bu : Flagship projects test innc echnologies.		0	0		
		in 2021 through unbundled RECs and PPAs. Target ocal RE with 24/7 matching by 2030.				
4 CLIMATE CONTRIBUTIO	TRANSPARENCY	INTEGRITY				
esponsibility for Microsoft offset less than 2% of its emissions in 2021.				\bigcirc		
Climate contributions		No climate contributions identified			\bigcirc	
Offsetting claims today	Mainly nature-based solutions (99%) with a 'contracted durability' of less than 100 years and an average credit price of 19.4 USD/tCO_2 .				0	
Diffsetting plans for the future Microsoft aims to move away from its strong focus on nature-based solutions, towards an increased share of high durability removals.						

RATINGS Overall 5-point scale High Reasonable Moderate Low Very low Average of sections 1-4. Sections 1-4 5-point scale High Reasonable Moderate Court Very low Average of the criteria in each section. Rating criteria 3-point scale High Moderate Poor See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Microsoft (2021, 2022a, 2022b, 2022c, 2022d, 2022e).

See **Annex II** for more details on our assessments of companies' targets

Microsoft

Microsoft Corporation is a technology company that develops hardware and software products. The company's main emissions stem from the purchase of goods and services (26%), its own electricity consumption (25%), purchased capital goods (22%), and the electricity consumed by the end user of Microsoft's products (21%). Microsoft sets a strong example in its transparency of emissions disclosure and offset credit use, and is proactive in testing innovative technologies for emission reductions. But Microsoft's targets lack integrity: the 'carbon neutral' since 2012 claim and 'carbon negative' by 2030 target are seriously undermined by a large reliance on contentious carbon dioxide removals. In reality, the company makes no commitment to reduce emissions by more than 38% of 2019 levels.

Microsoft's bold claims of 'carbon neutrality' and 'negative carbon' are heavily dependent on carbon dioxide removals that are unlikely permanent. To achieve its 'carbon negative' by 2030 target, Microsoft plans to simply offset half of its current emission footprint. The company has announced it will move away from using carbon credits based on reduced emissions – which it recognises are not aligned with the goals set out in the Paris Agreement (Smith, 2020, p. 8) – and aims to increase its share of what it terms 'high durability' carbon removal projects. However, the portfolio of carbon credits used to claim to offset its emissions in the fiscal year 2021 consists of 99% low-durability carbon dioxide removals with biological storage, such as reforestation projects in the US and soil sequestration projects in Australia. Microsoft reports that these have a 'contracted durability' of just 20-25 years (Microsoft's activities, particularly due to a lack of robustness around the additionality, permanence, and risk of double claiming of the carbon credits used (for more information on Microsoft's offset strategy see Section 4.4.3).

Microsoft's 'carbon neutral' claims exclude most emission sources, and its 'carbon negative' target only entails a 38% emission reduction. Microsoft claims it is 'carbon neutral' since 2012 and pledges to become 'carbon negative' by 2030 while continuously committing to its 'neutrality' target each year. The 'carbon neutrality' claim might be misleading as it only covers around 7% of Microsoft's emissions in 2019; namely the company's direct emissions as well as indirect emissions from electricity, steam, cooling water, and business air travel (Microsoft, 2022a, p. 100). The commitment to become 'carbon negative' by 2030 is much broader, covering Microsoft's total supply and value chain emissions. However, Microsoft uses a GHG emission accounting method to make achieving this target easier without having to substantially decarbonise its purchased electricity. While Microsoft tracks both its location- and market-based electricity consumption emissions, only the market-based values are included in its aggregated emissions disclosure and target coverage (Microsoft, 2022a, p. 104). Microsoft's location-based emissions are almost six times higher than its market-based emissions. As a result, the 2030 target only covers around 76% of Microsoft's full emission footprint. With half of these emissions set be offset, Microsoft's 2030 target entails a commitment to an emission reduction of just 38% of its 2019 location-based emissions (50% of its market-based emissions).

Microsoft currently lacks any significant targets for further emission reductions beyond 2030. The 'carbon negative' terminology of the 2030 pledge may give the impression that this is the company's end-goal, despite only equating to a moderate reduction of the company's emissions. Indeed, Microsoft sets out no further emission reduction commitments beyond this date. Microsoft communicates a longer-term pledge to remove by 2050 all of its direct emissions as well as those from consumed electricity since its foundation in 1975 (Microsoft, 2022a, p. 104). However, this additional offsetting pledge does not commit Microsoft to any further emission reductions. Taking responsibility for historical emissions is good practice but it is far more important to take responsibility for future emissions, through robust and transparent plans for deep decarbonisation.

For its electricity emissions, Microsoft plans to improve on its currently contentious reliance on RECs, through a commitment to 100% local renewable energy with 24/7 matching. Microsoft acknowledges that unbundled RECs – which form the basis of its historical scope 2 emission reduction claims – have limited impact (Microsoft, 2022a, p. 23). The company plans to secure an increased share of high quality long-term PPAs, with the goal to cover 100% of Microsoft's operational electricity demand with additional renewable energy by 2025. Further, Microsoft pledges that by 2030 all of its consumed electricity will come from local renewable energy sources, and be matched on an hourly basis (Microsoft, 2022a, p. 23). This 24/7 matching approach was first applied by Microsoft in 2021 in its Swedish datacentre, where Vattenfall provides RECs to the datacentre on an hourly basis to match consumption (Walsh, 2020, pp. 1–3). This approach is currently being expanded to one of the datacentres in Amsterdam. Microsoft claims that 100% of its electricity consumption in the Amsterdam datacentre will be matched with renewable energy from the Borssele wind park going forward (Microsoft, 2022a, p. 24), though we could not identify how Microsoft still plans to continue to purchase unbundled RECs until 2030 (Microsoft, 2022a, p. 23, 2022d, p. 69). In 2021 around 60% of Microsoft's total energy consumption was covered by higher quality renewable energy (Microsoft, 2022a, p. 101). Microsoft does not clarify what it does with the RECs generated through its PPAs, leaving the door open for double claiming of Microsoft's renewable energy. The share of on-site renewable energy generation has decreased over the last five years and accounts for less than 1% of its total consumed electricity in 2021 (Microsoft, 2022a, p. 101).

Microsoft demonstrates a proactive approach to test innovative solutions for complex emission sources and to engage suppliers, but provides insufficient details to allow us to understand their reduction impact. Emission reduction measures for direct emissions and indirect emissions through consumed electricity (scope 1 and 2) mainly focus on the procurement of renewable energy for its operations (Microsoft, 2022a, pp. 21–24). Yet, it remains unclear how Microsoft plans to reduce most of its scope 3 emissions. Location-based scope 3 emissions account for over 75% of Microsoft's total emissions in 2021. Most of these emissions stem from the extraction of raw materials and manufacturing of products used for datacentres and hardware products, as well as from the use of Microsoft's hard and software products (Microsoft, 2022a, p. 98). Increasing demand for data centre services may drive an increase in scope 3 emissions in the coming years. Microsoft plans to reduce their own operational emissions and by helping its customers to better understand their emission footprint from using Microsoft's services (Microsoft, 2022, p. 11, 15). Microsoft's Supplier Code of Conduct requires all suppliers to commit to an emission reduction of at least 55% by 2030 (Microsoft, 2022e, p. 11). It remains unclear what emission scope that target covers, which base year is applied and if the supplier is allowed to rely on offsets and RECs to achieve its target. For other hard-to-abate emission sources, for instance, embodied emissions in buildings; back-up generators for datacentres; or air travel, Microsoft supports the development of pilot projects that are rolled out company-wide if proven successful. For example, Microsoft partners with CarbonCure to scale up the development low-carbon cement and has applied these new materials in the Linkedln (a subsidiary of Microsoft) headquarters in both Silicon Valley and Dublin (Microsoft, 2022a, pp. 25, 26).

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Food, beverages & agriculture	USD 95.2 bn (2021)	118.7 MtCO ₂ e (2021)	Net zero by 2050	Low	C Low
1 TRACKING AND D	DISCLOSURE OF EMIS	TRANSPARENC	Y & INTEGRITY		
racking and disclosure 18.7 MtCO ₂ e in 2021 Subsidiaries are cover	Major emiss activities and Disclosure: / detail and no red.	Scope 1 Scope 2 Scope 3 upstream downstream	3. 2. 95. 18.		
2 SETTING EMISSIO	ON REDUCTION TARG	ETS		TRANSPARENCY	INTEGRITY
leadline target or pledge	e Net zero by 2	050			
Short- and medium-term (up to 2030)	reduce emiss	uce emissions by 20%. By 2 ions by 50%.	2030,	\cap	0
Scope coverage Own emission reductions (compared to full value chain	16-21% e	argets equate to 16-21% m missions footprint in 2019 neasures to claim that emis			
Long-term vision (beyond 2030)	Net zero by 2	050			
Scope coverage Own emission reductions (compared to full value chain	; e	Indefined role of 'insetting missions are offset; no ass		\circ	0
3 REDUCING OWN E	EMISSIONS	TRANSPARENCY	INTEGRITY		
Emission reduction measures and compared to BAU. Major emission sources are addressed, but no clear signs of transformation in agricultural activities.				0	0
enewable electricityRenewable electricity procurement constructs account for over 60% of consumption, mostly through PPAs. Disclosure of information only in CDP responses.				0	0
4 CLIMATE CONTRI	BUTIONS AND OFFSI	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for Untransparent offset claims through 'insetting'.				\bigcirc	0
Climate contributions	limate contributions ——— No climate contributions identified. ———			N/A	0
Offsetting claims today	Offsetting claims today Offsetting claims under the term 'insetting'. 9.7 MtCO ₂ e removals claimed since 2018; no details provided. Some brands claim climate neutrality.				0
Offsetting plans for the f		pend on offsetting under the s are provided.	e term 'insetting'.	0	0

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Nestlé (2021a, 2021b).

Nestlé

Switzerland-based Nestlé S.A. (Nestlé) is the world's largest food and beverage company by revenue, with brands such as KitKat, Nesquik and Nespresso. The biggest share of Nestlé's emissions is related to agricultural activities. Nestlé's targets are potentially misleading and ambiguous: we interpret that the pledge to reduce emissions by 50% by 2030 translates to emission reductions of just 16–21%, while the meaning of the 2050 net zero pledge is unclear. The company does not have clear plans for the deep decarbonisation of agricultural emissions but rather plans to offset emissions under the guise of 'insetting'.

Key developments over the past year: We did not identify any improvements on the key issues that undermine Nestlé's climate strategy since we published the previous iteration of this analysis in February 2022 (Day *et al.*, 2022). Following the publication of the *2022 Corporate Climate Responsibility Monitor*, Nestlé released a statement to clarify that its 2030 target is compared to 2018 levels rather than a business-asusual trajectory, but the complex and untransparent accounting behind this target leads us to the interpretation that the target remains far weaker than it appears.

Nestlé's emission reduction pledges may be misleading. We interpret that the pledge to reduce emissions by 50% by 2030 translates to only 16–21% emission reductions compared to the company's emissions in 2019. Nestlé's SBTi-certified targets include emission reduction targets for 20% by 2025 and 50% by 2030, compared to a 2018 base year (Nestlé, 2021b, p. 4). In its *Net Zero Roadmap*, Nestlé presents its interim emission reduction targets for each emission source compared to a business-as-usual scenario, showing the targeted emission levels for each emission source for 2030 (Nestlé, 2021b, p. 4). We calculate from the figures presented in the company's *Net Zero Roadmap* that the company's commitments translate to just a 16% reduction of the company's full value chain emissions in 2019, or a maximum of 21% under the most optimistic interpretation (see further details on the target and this calculation in the *Integrity assessment for short- and mediumterm target(s) towards 2030* in Annex II).

Nestlé's 2050 net-zero pledge remains ambiguous and does not entail a clear emission reduction commitment. As for the 2025 and 2030 targets, Nestlé's 2050 net-zero pledge covers only 80% of Nestlé's 2018 emissions footprint; emissions related to use of sold products and purchased services, leased assets, capital goods and investments are excluded (Nestlé, 2021b, p. 7). The pledge does not include any specific emission reduction commitment, while Nestlé states that it plans to claim the neutralisation of emissions. Further clarification is needed to understand whether the 2050 pledge represents a commitment that is credible and comprehensive. Nestlé claims that it will 'not allow' offsetting to achieve its targets, but the company actually claims that a significant portion of emissions are offset under the guise of 'insetting', while some of Nestlé's brands already claim carbon neutrality through offsets. The measures that contribute towards Nestlé's 2030 pledge include biological carbon removal and storage which will offset 12.3 MtCO₂e of Nestlé's 2030 emissions, as well as further so-called 'insetting' measures to offset another 13 MtCO_2 e of its 2030 emissions. 'Insetting' is a business-driven concept with no universally accepted definition. The approach facilitates offsetting claims, under the guise of a different terminology. 'Insetting' is a potentially misleading terminology that can lead to low credibility GHG emission offsetting claims and the double counting of emission reductions (see section 4.4.1). Moreover, individual, consumer-facing Nestlé brands are already claiming 'carbon neutrality', based on carbon offsets (see Day *et al.*, 2022 Box B5).

Nestlé's emissions reporting focuses on a share of emissions and reports progress towards its targets compared to a potentially misleading business-as-usual scenario. In its main sustainability reports, Creating Shared Value, Nestlé reports on emissions without a breakdown (Nestlé, 2021a). Moreover, the company presents its progress towards target realisation compared to a business-as-usual scenario and uses language around cumulative and annual emission reductions and removals interchangeably (Nestlé, 2021a, p. 21). The reporting style does not allow for a thorough understanding of emission sources and trends, and presenting progress as such can be highly misleading.

Nestlé's plans do not include sufficiently transformational measures to achieve deep decarbonisation of agricultural emissions in the long run. The majority of Nestlé's GHG emissions derive from upstream agricultural activities upstream. The agriculture sector faces major challenges for decarbonisation; existing technologies and measures to mitigate the emissions intensity of many agricultural products have limited potential, especially for the livestock sector, which accounts for approximately 30% of Nestlé's emissions. Although Nestlé's range of emission reduction measures are expected to lead to a respectable 48% reduction of manufacturing emissions by 2030, they will reduce emissions from dairy, livestock, soil and forests, which are far more significant emission sources, by just 6% between 2018 and 2030, excluding measures to claim that emissions are offset through non-permanent carbon capture (Nestlé, 2021b, p. 9,12,14,17). These emission sources represent the most significant challenge for agri-businesses. It is not credible for agri-businesses to claim that they are on a path to deep decarbonisation, without major innovations to drastically reduce the emissions footprint of livestock agriculture, or diversifying away from this highly GHG emissions intensive industry.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY	
Food, beverages & agriculture (2021)		63.8 MtCO ₂ e (2021)	Net-zero emissions across value chain by 2040	C Low	C Low	
1 TRACKING AND I	DISCLOSURE OF EMIS	TRANSPARENCY	& INTEGRITY			
63.8 MtCO ₂ e in 2021 agricultura downstrea Subsidiaries appear to be covered, but it remains unclear whether PepsiCo includes Disclosure limited def		sion sources: Major emissions sources are practices (33%), packaging (26%) and up- and n transportation and distribution (20%). Disclosure in public-facing documentation has ail. PepsiCo publishes its CDP disclosures, with down of emission scopes and sources.		Scope 1 Scope 2 Scope 3 upstream downstream	3.6 1.8 44.6 13.8	
2 SETTING EMISSI	ON REDUCTION TARG	ETS		TRANSPARENCY	INTEGRITY	
Headline target or pledg	e Net-zero emis	ssions across value chain b	oy 2040			
Short- and medium-terr (up to 2030) Scope coverage	(2015 baselin under the term (S) (2) (3) T	nd s2 emissions -75% & s3 e). Includes unspecified role n 'insetting'. argets translate to 48% er ut includes offsets claime mission reductions will be	for offsetting nission reductions, d through 'insetting'.	0	?	
Own emission reductions (compared to full value chain in 2019) Image: Compared to full value chain in 2019) Long-term vision (beyond 2030) Net-zero emissions across value chain by 2040						
Scope coverage Own emission reduction (compared to full value cha	S +	PepsiCo does not specify hat it will reduce under i		0	0	
3 REDUCING OWN	EMISSIONS		TRANSPARENCY	INTEGRITY		
mission reduction meas	sures strategy rem	ains unclear. No clear sign	lescribed, but their collective role in the nclear. No clear signs of transformation in es for deep emission reductions.		?	
Renewable electricity procurement	and summar	energy procurement in p y on webpage. 72% of en rces; large share of RE ba	ergy consumption is	0	0	
4 CLIMATE CONTR	IBUTIONS AND OFFSI	TRANSPARENCY	INTEGRITY			
Responsibility for unabated emissions	Offsetting cla	aim for business travel.		\bigcirc	0	
Climate contributions		— No climate contributions identified. ————		N/A	0	
Offsetting claims today Carbon price for business travel. Invested in regenerative agriculation in the supply chain to claim that travel emissions are offset.			\bigcirc	0		
Unserring dialis for the tunne		O targets reliant on offsets ting'. Insufficient informatio		\bigcirc	0	

Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from PepsiCo (2021, 2022a, 2022b, 2022c, 2022d, 2022e).
PepsiCo

PepsiCo, Ltd. (PepsiCo) is a US-based food and beverages company, known for brands such as Pepsi, Lay's, Quaker and Gatorade. Major emissions are from agriculture, packaging, and distribution (all scope 3). PepsiCo has a target to reach net-zero emissions across its value chain by 2040 and says it will need to neutralise residual emissions with carbon offsets, without making any clear emission reduction commitments for the target year. PepsiCo's short- to mid-term climate targets translate to emission reductions of 48% by 2030, but these also depend on an unspecified role for offsetting under the guise of 'insetting'. Although the company wants to increase the share of regenerative agriculture, PepsiCo only describes these and other measures in vague terms. We found limited evidence for transformations that would be necessary for a 1.5°C-aligned food and agriculture sector.

PepsiCo does not specify what share of emissions it will reduce under its 2040 net-zero target. In its Climate Action Strategy, PepsiCo illustrates that it wants to use carbon removals to reach its net-zero targets (PepsiCo, 2021). However, the company does not specify the volume of carbon that it aims to remove and store. To improve the integrity of its net-zero emissions target, it is crucial that PepsiCo specifies which share of emissions will be reduced. Without committing to at least 90% emission reductions, the net-zero target is ambiguous and potentially misleading.

Although PepsiCo explicitly states that the purchase of offset credits does not contribute to realising its 2030 targets, the company partially depends on neutralisation through 'insetting'. PepsiCo aims to reduce its scope 1 and 2 emissions by 75% and its scope 3 emissions by 40% by 2030, compared to 2015 emission levels. These targets translate to 48% emission reductions compared to its 2019 value chain emissions. Although PepsiCo explicitly states that these targets do not depend on the purchase of offset credits (PepsiCo, 2022b), the target realisation partially depends on offsetting emissions under the term 'insetting', as illustrated in PepsiCo's Climate Action Strategy (PepsiCo, 2021). PepsiCo does not specify what kind of 'insetting' practices it will use. The company mentions regenerative agriculture practices but does not specify further. We interpret that 'insetting' is an offsetting claim related to the use of biological carbon storage with low permanence. See Section 4.4.4 for an elaboration on the issues related to 'insetting'. As PepsiCo does not communicate any clear emission reduction commitment for 2030, it is unclear what proportion of its target depends on this form of offsetting. PepsiCo does not commit to emission reductions beyond its interim 2030 targets either.

Although PepsiCo takes several innovative decarbonisation approaches, it remains unclear how these measures will lead to necessary deep emission reductions and the realisation of targets. PepsiCo presents its emission reduction strategy in thematic areas, covering major emission sources such as deforestation, agriculture, transport, energy consumption, and packaging. However, the company shows the current climate impact of only a few of these issues. PepsiCo's agriculturerelated emissions accounted for a third of its 2021 emissions footprint (PepsiCo, 2022b). The company's main strategy to reduce agricultural emissions is implementing regenerative agriculture, though without specifying the actual measures related to this. PepsiCo estimates that regenerative agriculture will lead to a reduction of 3 MtCO₂e by 2030, which would lead to an emissions reduction of only 5% compared to 2021 levels (PepsiCo, 2022a). This falls far short of the radical transformation of the global agriculture sector that would be necessary to align with 1.5°C-compatible decarbonisation trajectories (Boehm *et al.*, 2022, p. 127). For other major emission sources, PepsiCo does not provide estimates of the emission reduction potentials. Since the company's emissions have only further increased from 2015 (PepsiCo, 2022b) and since PepsiCo categorises many of its proposed measures as not yet scalable (PepsiCo, 2021), it seems unlikely that the company's climate strategy is sufficient to bring its emissions in line with what is needed to realise the deep emission reductions that are implied by its mid- and long-term targets.

PepsiCo makes bold claims regarding its renewable electricity consumption, but the lion's share of renewable electricity is procured through RECs. PepsiCo claims that 70% of its 2021 electricity consumption was from renewable sources, and that PepsiCo's operations in 13 of 200 countries were 100% based on renewable electricity (PepsiCo, 2022d, 2022e). However, less than 20% of its electricity consumption is from higher-quality procurement constructs, such as PPAs, or self-generation (PepsiCo, 2022c, pp. 78; 100–131, 2022d). The lion's share of its renewable electricity is procured with unbundled EACs and GOs, both also known as RECs. RECs do not guarantee that the consumed electricity truly stems from additional renewable energy sources (see Box 1, Section 3.2.2). Claiming that its electricity consumption is 70% renewable is therefore highly contentious. Although the company says it wants to finance the development of new wind and solar installations with PPAs, it does not specify the volume of finance or the size of these installations (PepsiCo, 2022d). Stronger commitments to increase the share of renewable power procured with high-quality PPAs or generated on site would make PepsiCo's claims more credible and would have a more meaningful impact in reducing the company's scope 2 emissions.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Technology - Us Electronics	SD 244.4 bn (2021)	143.4 MtCO ₂ e (2021)	Net-zero carbon emissions by 2050	Low	() Very low
1 TRACKING AND DISCI	LOSURE OF EMIS	SIONS		TRANSPARENC	(& INTEGRITY
acking and disclosure 43.4 MtCO ₂ e in 2021 Subsidiaries are covered.	Scope 1 Scope 2 Scope 3 upstream downstream	7. 12. 20. 103.			
2 SETTING EMISSION R	EDUCTION TARG	ETS		TRANSPARENCY	INTEGRITY
leadline target or pledge	Net-zero cark	oon emissions by 2050			
Short- and medium-term targ (up to 2030) Scope coverage	0.0	bon emissions (s1 & s2) for rision by 2030.	r all operations	\bigcirc	\bigcirc
Own emission reductions (compared to full value chain in 20	2% a	he target covers s1 and s2 nd excludes all other emiss		Ý	Ý
Long-term vision (beyond 2030)		ect and indirect carbon em bal operations	issions by 2050		
Scope coverage Own emission reductions (compared to full value chain in 20	20%	Commitment to reduce ~1 i.e., 100% reduction of s1 a n 2021). Scope 3 is not incl	and market-based s2		\bigcirc
3 REDUCING OWN EMIS	SIONS			TRANSPARENCY	INTEGRITY
mission reduction measures	superficially	rget most emission sources and often set to be implem limited clarity on short-ter	ented by 2030 or	0	?
enewable electricity rocurement	on-site insta	for 20% of electricity use, llations, and RECs. 100% l oal in all markets except K	RE in the US and China,	D	0
4 CLIMATE CONTRIBUT	IONS AND OFFSI	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	No informatic	on identified.		\bigcirc	0
Climate contributions		— No climate contribution	ns identified	N/A	
Offsetting claims today	•	Unspecified use of 'emission permits' from CDM projects in Kenya and India; no carbon neutrality claim made.			?
Offsetting plans for the future		ets set for 2030 and 2050; r npany will rely on offsetting		O	?

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Samsung (2022a, 2022b, 2022c).

Samsung Electronics

Samsung Electronics Co., Ltd., (Samsung) is a South Korean multinational corporation that manufactures electronic devices such as smartphones and TVs, and electronic parts such as display panels and semiconductors. Samsung's netzero target for 2050 represents a commitment to reduce just 20% of its 2019 emission footprint. The company's climate strategy currently amounts to medium- and long-term commitments and measures for the decarbonisation of only selected emission sources and product lines, while plans for short-term action remain less clear.

Samsung set a net-zero carbon emissions target by 2050, but it only commits to reduce 20% of its 2019 emissions footprint. Samsung commits to achieve net-zero carbon emissions across all global operations by 2050 and across the company's 'Device Experience (DX) Division' by 2030 (Samsung, 2022b). However, these targets only include scope 1 and 2 emissions, which accounted for just 14% of the company's 2021 emission footprint. Samsung reports that it will set emission reduction targets for its scope 3 emissions in the future (Samsung, 2022b). For the 2050 target, Samsung 'expects to reduce' ~17 MtCO₂e, which correspond to scope 1 and market-based scope 2 emissions in 2021 (Samsung, 2022c, p. 100). A reduction of 17 MtCO₂e would translate to a reduction across the entire value chain of 20% below 2019 levels (Samsung, 2022a). For the 2030 target, the company does not specify the share of scope 1 and 2 emissions that it aims to reduce in the DX Division, but this emission source accounted for only 2% of Samsung's full value-chain emissions footprint in 2019. Samsung's 2030 and 2050 netzero targets can lead consumers and investors to believe that the company aims for deep decarbonisation of its business, while the company has only committed to the decarbonisation of a minor share of its emissions footprint.

Samsung emitted 143 MtCO₂e in 2021, but the company does not share a complete emissions footprint estimate in its public-facing documentation. In its 2022 sustainability report, Samsung discloses emission estimates for scopes 1 and 2, including different GHGs and years, but excludes scope 3 emissions (Samsung, 2022c, p. 94). These account for 86% of the company's 2021 emissions footprint, as we identified in its disclosure to CDP, a document that is only accessible with limitations through registration on the CDP website. Samsung would be more transparent if it shared its complete emissions footprint in its public sustainability strategy. Information on emission sources is highly relevant for Samsung's consumers and investors to understand the meaning and credibility of the company's targets and emission reduction measures. Samsung presents emission reduction measures for most of its emission sources, but many of these are embodied in medium-term targets, while the company's plans for short-term action remain less clear. To reduce emissions in company processes (scope 1), the company will invest in new technologies, including treatment facilities for gases produced in semiconductor manufacturing (Samsung, 2022c, p. 27) and carbon capture in semiconductor lines (Samsung, 2022b). Within its supply chain (scope 3), Samsung aims to improve material recycling by expanding the global presence of its electronic waste collection system, reusing minerals from collected batteries, and using recycled resins (Samsung, 2022b). However, many of these measures lack details on their coverage and expected emission reductions and are set to be implemented by 2030 or beyond, while short-term plans are vaguer. Although the company participates in the CDP Supply Chain programme, it does not disclose concrete examples of technical or financial support for suppliers' emission reduction efforts beyond CDP training seminars (Samsung, 2022c, p. 82). Samsung plans to reduce product use emissions (which account for 70% of its emissions footprint) by improving energy efficiency in semiconductor memories, used in data centres and mobile devices, and reducing power consumption in seven consumer electronic products by 30% by 2025, compared to equivalent products in 2019 (Samsung, 2022c, 2022b). But it is not clear what proportion of Samsung's product sales would be accounted for under these seven products, nor whether there is a target for the remainder of Samsung's product offering.

Samsung does not plan to shift its electricity consumption to 100% renewable energy until 2050 but takes proactive steps towards addressing system-level issues for renewable electricity supply in Asia. Samsung currently has a low share of renewable electricity consumption (15-20%) (Samsung, 2022c, p. 94) and only aims to transition its global operations to 100% renewable electricity by 2050 (Samsung, 2022b). The company claims to procure 100% renewable electricity in the US, China, and Europe, where it aims to replace RECs with higher-quality PPAs (Samsung, 2022b). But the company still relies heavily on RECs in other countries, including Mexico, and plans to do so in the coming years (Samsung, 2022b). Samsung reports to face procurement challenges in other countries including South Korea. To increase procurement options in Asia, Samsung founded the Asia Clean Energy Coalition in 2022 with the objective to bring together influential industry actors in dialogue to address system-level issues in renewable electricity development (ACEC, 2022).

Stellan	tis				
SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Automobiles	USD 176.6 bn (2021)	138.5 MtCO ₂ e (2021)	Carbon net zero by 2038	0 Moderate	D Moderate
1 TRACKING AND DI	SCLOSURE OF EMIS	SIONS		TRANSPARENCY	(& INTEGRITY
acking and disclosure 8.5 MtCO ₂ e in 2021		ion sources: Use phase of stream s3), purchased go eam s3).	Scope 1 Scope 2	1.	
Subsidiaries are covere	third-party a	3 disclosed only in relativ analysis calls integrity of d n. Only market-based esti	ownstream s3 disclosure	Scope 3 upstream downstream	15. 119.
2 SETTING EMISSIO	N REDUCTION TARG	BETS		TRANSPARENCY	INTEGRITY
leadline target or pledge	Carbon net z	ero by 2038			
Short- and medium-term up to 2030)	targets and s3 (below - s1 & s2: 50% - s3: 50% inter - EVs: 100% Bl	e reduction by 2030 acros v 2021) supported by: absolute reduction by 2025 a sisty reduction by 2030, EVs in Europe and 50% BEVs in rts: 40% intensity reduction by	nd 75% by 2030, n US by 2030,		
Scope coverage Own emission reductions (compared to full value chain	50%	Dverarching 2030 target to hain emissions by 50%, co 1.5°C-aligned ICE phaseout	mpared to 2021.		Ŭ
_ong-term vision beyond 2030)	Carbon net z	ero by 2038			
Scope coverage		All emission scopes covered	•		
Own emission reductions (compared to full value chain	> 90 %	eductions comittment of a compared to 2021. 1.5°C-a late for EU.			
3 REDUCING OWN E	MISSIONS			TRANSPARENCY	INTEGRITY
mission reduction measu	res mobility, but	Relevant measures for accelerated transition to electric mobility, but unclear whether measures are sufficient to align with a 1.5°C-aligned decarbonisation pathway.		0	0
enewable electricity rocurement		for <20% of electricity co longside some on-site inst	· · ·	0	0
4 CLIMATE CONTRIE	BUTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for Inabated emissions		Offsetting claims and contributions in Latin America with uncertain climate impact.		0	0
Climate contributions	Biodiversit Details rega	y and reforestation projects in arding the volume of finance n	Brazil. ot identified.	\bigcirc	?
Offsetting claims today		arbon neutral' operations in Sc d on low-quality offset credits.			\bigcirc
)ffsetting plans for the fu	9% of its 202 ture carbon emiss	1 emissions may be offset to ions by 2038. No informatio credits identified.	achieve net zero	\bigcirc	?

RATINGS Overall 5-point scale High Reasonable Moderate Low Very low Average of sections 1-4. Sections 1-4 5-point scale High Reasonable Moderate Low Very low Average of the criteria in each section. Rating criteria 3-point scale High Moderate Poor See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Stellantis (2022a, 2022b, 2022c) and Bonaccorsi et al. (2022)

Stellantis

Stellantis is an automotive company headquartered in the Netherlands. Most of Stellantis' emissions originate in the use phase of its vehicles (85% of 2021 emissions in Europe). The company commits to reach 'carbon net zero' in 2038 by reducing at least 90% of its current CO₂ emissions and plans to offset all remaining emissions. Stellantis' targets for 2025 and 2030 put a distinct focus on short-term emission reductions and vehicle electrification in key markets, but they only partially align with 1.5°C-aligned sectoral pathways for the automobile industry.

Stellantis aims to be 'carbon net zero' by 2038 while limiting the use of offsets to less than 10% of its 2021 emissions. The company set the goal to become carbon net zero by 2038 in 2021 and commits to reduce its own emissions by at least 90% compared to 2021 (Stellantis, 2022a, pp. 39–40). Stellantis plans to compensate the remaining emissions with carbon dioxide removals and other offsetting solutions, but it does not yet provide details on removal technologies or the quality criteria that it would use when procuring offset credits in the future (Stellantis, 2022a, p. 39).

Stellantis commits to an ambitious overarching 2030 target to reduce its emissions by 50% along the entire value chain compared to 2021 levels, but only the 2030 phaseout date for internal combustion engine vehicle sales for the European market meets 1.5°C-aligned sectoral pathways for the automobile industry. The company plans to achieve its overarching 2030 target based on several sub-targets for different emission sources: vehicle production, the vehicle use phase, and its supply chain (Stellantis, 2022a, pp. 34, 37–39). For emissions from the vehicles' use phase, the company aims to sell 100% battery electric vehicles (BEVs) for passenger cars in Europe and 50% BEVs for passenger cars and light-duty trucks in the US by 2030. Whereas Stellantis' target for the EU market is in line with 1.5°C-aligned decarbonisation milestones, its targets for the US market and aspirational sales shares for Brazil, India and China, as outlined in its strategic blueprint *Dare Forward* 2030 (Stellantis, 2022c), are not (see detailed assessment in the Annex). Across all scope 3 emissions, the company expects a 50% emission intensity reduction by 2030 compared to 2021 levels. Recent analysis also suggests that Stellantis currently underreports its disclosed life-cycle emissions of sold vehicles by more than 60% due to optimistic assumptions on vehicle lifetimes ((Bonaccorsi *et al.*, 2022, pp. 15–16).

Stellantis has not signed the clean-vehicle pledge announced at COP26 in November 2021, in which competing automakers from several countries, including the US and Germany, committed to exclusively produce electric vehicles by 2035, or earlier, to support limiting global warming to 1.5°C (COP26 Presidency, 2021). Stellantis further takes mixed and contradictory positions in its political lobbying efforts in the US, UK and EU, both supporting and opposing regulations for stronger climate action (InfluenceMap, 2022c).

Stellantis' range of reduction measures put distinct focus on the rapid transition towards electric mobility, but the company also invests in the development of other technologies, such as e-fuels, hydrogen-based fuel cells, and biofuels of highly uncertain efficiency and sustainability. To support the electrification of its vehicle fleet as part of its effort to reduce downstream scope 3 emissions, Stellantis deploys several vehicle charging solutions such as supporting a public fast charging network that will have 36,000 fast chargers by 2030 across Europe and North America (Stellantis, 2022a, pp. 63–64). The company also investigates measures to reduce emissions in the vehicle use phase, such as fuel efficiency innovations in existing combustion-engine vehicle lines, alternative fuels such as e-fuel produced with hydrogen

energy, and hydrogen-based fuel cells for vehicle propulsion. The latter two drive technologies, however, would require much greater amounts of renewable electricity production than BEVs (Transport & Environment, 2018).

Stellantis further considers biofuels a key measure for emission reductions of its non-electric fleet. In Brazil, for example, the company has deployed bioethanol-compatible vehicles and considers to launch an electric-bioethanol hybrid vehicle model in 2025 (Alerigi Jr., 2022; Stellantis, 2022a, p. 55). However, biofuel production at scale faces the high likelihood of competing with other environmental and social interests, such as food production, biodiversity, and forest protection (Clarke *et al.*, 2022, p. 42). This is especially relevant for the automobile sector for which technological alternatives to biofuel are readily available (see Section 3 of the methodology). None of Stellantis' research into biofuels or alternative fuels should cause a delay in the complete phaseout of ICE vehicles.

Stellantis remains vague on specific measures to decarbonise its upstream scope 3 emissions. For 2021, the company claims that 'more than 55% of our most important suppliers [...] commit to a CO_2 trend which complies with the Paris Agreement'; the company aims to increase this share to 95% by 2030 (Stellantis, 2022a, p. 332). The company does not disclose any further information on which particular measures Stellantis requires from its upstream suppliers beside these vague commitments and whether it takes an active approach in reducing upstream supply chain emissions, for example, through setting procurement targets for low-carbon steel (Stellantis, 2022a, pp. 79–81).

Stellantis plans to rely on offset credits to meet its carbon net zero target but provides no details on the criteria it would set for those credits. Stellantis' 2038 carbon net zero pledge includes a share of carbon offsetting of less than 10% of 2021 emissions (Stellantis, 2022a, p. 39). The company does not disclose what type of offset credits it plans to procure and whether corresponding adjustments would be applied (see Section 4 in the methodology). For 2021, Stellantis claims that 55% of its plants in South America have used offset credits to 'neutralise' their scope 1 and 2 emissions in 2021 (Stellantis, 2022a, p. 22). Stellantis has sourced offset credits from projects such as energy generation from landfill waste, reforestation, and recovering environmentally degraded areas (Stellantis, 2022a, p. 73). Offset credits from such projects are unlikely to represent the high-hanging fruit of mitigation projects that can reasonably be considered additional in the context of the Paris Agreement, and are unsuitable to claim neutralisation of the company's own emissions (see Section 4.4.1). Stellantis could instead adopt a climate contribution approach, providing financial support to climate change action beyond the company's own value chain without claiming to neutralise its own emissions.

Stellantis currently procures predominantly lower-quality RECs to claim that 18% of its electricity consumption is from renewables. The company does not disclose information on how it aims to reach its 50% renewable target in 2025. Renewable electricity and other electricity labelled decarbonised accounted for only 18% and 27% of the company's electricity consumption in 2021 respectively (Stellantis, 2022a, p. 73). The company procures most of its renewable electricity through lower quality RECs with likely very limited impact to foster additional new renewable generation capacity (Stellantis, 2022a, p. 72). Alongside RECs, the company also procures some renewable electricity through higher-quality PPAs and own on-site generation capacity (Stellantis, 2022a, p. 72–74). The company does not further disclose information on the procurement constructs it plans to rely on to meet its renewable targets of 50% in 2025 and 100% in 2030 (Stellantis, 2022a, p. 38).

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Steel and cement	USD 40.2 bn (2021)	60.5 MtCO ₂ e (2021)	Climate neutral by 2050	O Moderate	D Moderate
1 TRACKING AND	DISCLOSURE OF EMIS	TRANSPARENCY & INTEGRITY			
Tracking and disclosure 60.5 MtCO2e in 2021Major emission sources: S1 emissions from with steelmaking represent around 38% of the disclosed emission footprint. Most s3 emissions are from with purchased goods and services.Disclosure: Specific sources or accounting methods for s1 and s2 not disclosed. Only fuel- and energy-related S3 publicly reported. Full s3 emissions reported only to CDP.			Scope 1 Scope 2 Scope 3 upstream downstream	23.2 1.5 34.0 1.8	
2 SETTING EMISSI	ON REDUCTION TARG	GETS		TRANSPARENCY	INTEGRITY
leadline target or pledo	e Climate neut	ral by 2050			
Short- and medium-terr up to 2030) Scope coverage	- To reduce s	1&2 emissions by 30% belo 3 emissions by 16% below 2	2017 levels by 2030	D	
Own emission reduction (compared to full value cha	s 12%	Farget equals a 12% reduct chain. Supporting measures with some 1.5°C-aligned se	s indicate alignment		
Long-term vision (beyond 2030)		&2 by at least 90% below 0% below 2017 levels by 2			
Scope coverage Own emission reduction (compared to full value cha	s 89%	Potentially significant omis Supporting measures indic some 1.5°C aligned sectora	ate alignment with		
3 REDUCING OWN	EMISSIONS			TRANSPARENCY	INTEGRITY
mission reduction meas	production a	se-out its carbon intensive and to reduce scope 3 emis: ans are unclear for other di	sions, but emission	0	D
enewable electricity rocurement	Very limited	information on renewabl	e energy procurement.	\bigcirc	?
4 CLIMATE CONTR	IBUTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	No information	No information on contributions or offsetting claims.		\bigcirc	0
Climate contributions		— No climate contributio	N/A	0	
Offsetting claims today		re being procured to claim ca I service centre from 2024. I		\bigcirc	?
Offsetting plans for the		2019 emissions will be offse fsetting plans unclear.	t toward the 2050 pledge.	\bigcirc	?

Rating criteria 3-point scale High Moderate Poor See methodology document for rating criteria. **Transparency** refers to the disclosure of information. **Integrity** refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Thyssenkrupp (2018, 2021, 2022a, 2022b, 2022c, 2023).

Thyssenkrupp

Thyssenkrupp, based in Germany, is a multinational industrial business group with activities in steelmaking, material logistics, industrial services, and others. 38% of its reported emission footprint are scope 1 emissions related to steelmaking. Thyssenkrupp has pledged to reach climate neutrality by 2050, including a commitment to reduce scope 1, 2 and 3 emissions by at least 89% compared to 2019 levels. To achieve its goal, the company set out a decarbonisation roadmap for its steelmaking operations, transitioning out of coal-based steelmaking by 2045. Uncertainties remain relating to the coverage and relevance of downstream scope 3 emissions in the company's climate strategy.

Thyssenkrupp's targets of reaching climate neutrality by 2050 and climate neutrality for its steelmaking division in 2045 are likely aligned with relevant 1.5°C-compatible benchmarks for the steel sector, although uncertainty remains on scope 3 relevance. The company presents its carbon neutrality pledge for 2050 alongside a commitment to reduce its scope 1 and 2 emissions by 90% below 2018 values, and scope 3 emissions by 90% below 2017 values (Thyssenkrupp, 2023). Compared to Thyssenkrupp's disclosed 2019 whole value chain emissions, this target means a 89% reduction by 2050. The company does not yet provide clear information on how it plans to address or offset the remaining 11% of emissions towards its carbon neutrality claim. The company commits to the same target including the same baselines for its steelmaking division for 2045, alongside clear plans for developing their own near-zero emissions steelmaking facilities, which are planned to reach 100% "climate neutral steel production without coal-based blast furnace" (Thyssenkrupp, 2022b, p. 60). This target appears likely to be in line with 1.5°C compatible benchmarks for steelmaking benchmarks). However, uncertainty remains on the coverage and relevance of downstream scope 3 emissions. There remains a lack of guidance on the appropriate reporting boundaries for scope 3 emission source for steelmaking companies, and Thyssenkrupp's targets do not provide full clarity on this issue. Even if it is determined that scope 3 category 11 emissions are not a relevant emission source for steelmaking companies, and thus for Thyssenkrupp's steelmaking division, Thyssenkrupp is a diversified company and this emission category is likely to be relevant for the other company divisions.

Thyssenkrupp's 2030 interim targets only represent a 12% reduction of its whole value chain GHG emissions below 2019 levels. Without further clarification, it remains unclear whether the 2030 scope 3 target represents a commitment to further emission reductions. The company set out two targets for 2030: a target to reduce scope 1 and 2 emissions by 30% below 2018 levels and a target to reduce scope 3 emissions by 16% below 2017 levels (Thyssenkrupp, 2022a, p. 92). While these two targets cover all of Thyssenkrupp emissions across the value chain, they only translate to a 12% reduction by 2030 compared to 2019 levels. The company has recently adjusted their scope 3 emissions reporting, and subsequently modified their scope 3 target. Previously, Thyssenkrupp reported around

780 MtCO₂e scope 3 emissions in 2017 from the use phase of sold products and committed to reduce them by 16% by 2030 (Thyssenkrupp, 2021, pp. 41–42). However, in its latest documents, Thyssenkrupp stopped reporting on this specific scope 3 category, and adjusted its scope 3 target to cover most of its remaining scope 3 emissions, which in 2017 were around 49 MtCO₂e (Thyssenkrupp, 2018, pp. 48–54). Thyssenkrupp's original target implied an absolute emission reduction of 125 MtCO₂e and its new target implies a reduction of 8 MtCO₂e. Moreover, based on Thyssenkrupp's CDP emissions disclosures both from 2018 and 2022, we estimate that the company's new scope 3 target was already overachieved by 2021 (Thyssenkrupp, 2018, 2022c). However, due to the changes in the scope of Thyssenkrupp's reported emissions in 2022, it remains unclear whether retroactive changes in target baselines would apply, and whether they would affect the absolute emission levels targeted.

Thyssenkrupp provides a clear transition plan for its steelmaking operations, including a transition to 100% hydrogen-based steelmaking in 2045. For the steelmaking division, Thyssenkrupp has set out a clear plan to phase out its carbon-intensive infrastructure, with interim targets showing how the transition will impact overall steel production over time and how emissions will be reduced in the process (Thyssenkrupp, 2022b, p. 60). The plan includes the use of carbon capture, utilization, or storage (CCUS) to reduce emissions in the short term, but clearly identifies this technology as an interim solution before shifting to more reliable and 1.5°C-compatible decarbonisation technologies such as hydrogen-based direct reduction with electricity-powered melting units. While the use of CCUS in blast furnaces may help reduce emissions in the short term, literature suggests that retrofitting current infrastructure will not be enough to reach net zero due to several challenges and potential associated risks (de Villafranca *et al.*, 2022). Hydrogen-based steelmaking on the other hand, can lead to zero-emissions steel when powered by renewables and green hydrogen, and it is potentially scalable without significant risks of adverse effects on other areas for example related to sustainable biomass sourcing (de Villafranca *et al.*, 2022).

Thyssenkrupp is not clear about its procurement of renewable electricity. Based on its publicly available information, we were unable to identify if Thyssenkrupp uses renewable electricity and how the company procures it. When reporting their scope 2 emissions, the company does not specify whether its accounting methodology is location- or market-based.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Automobiles	USD 295.7 bn (2021)	371.3 MtCO ₂ e (2021)	Carbon neutral by 2050	D Moderate	Low
1 TRACKING AND DISCLOSURE OF EMISSIONS				TRANSPARENCY	(& INTEGRITY
Tracking and disclosure Major emission sources: Use phase of sold vehicles (downstream s3, 75%); purchased goods and services (upstream s3, 16%) Subsidiaries are covered. Disclosure: Volkswagen comprehensively discloses emissions across all relevant scopes but third-party analysis calls integrity of downstream s3 disclosure into question.		Scope 1 Scope 2 Scope 3 upstream downstream	4 2 71 292		
2 SETTING EMISSI	ON REDUCTION TARG	, 		TRANSPARENCY	INTEGRITY
eadline target or pledg					
hort- and medium-tern ip to 2030)	s1&s2 - 50% absolute s3 - vehicle li - 30% reductio	-			<u> </u>
Scope coverage Own emission reductions (compared to full value chai	f	lo 1.5°C-aligned phaseout or s1 & s2 equals a 3% redu alue chain, S3 intensity tar	uction across the	Ŭ	Ŷ
ong-term vision beyond 2030)	Carbon neut	Carbon neutral by 2050			
Scope coverage Own emission reductions (compared to full value chai	? 0	II emission scopes covered ommitment alongside carl 5°C-aligned phaseout dat	bon neutrality pledge. No	0	\bigcirc
3 REDUCING OWN EMISSIONS			TRANSPARENCY	INTEGRITY	
nission reduction measures nipoduction. Limited details on timeline and expected impact.		0	0		
enewable electricity ocurement	renewable e	Extensive use of lower-quality RECs to claim 96% of renewable electricty procured in Europe in 2021. Plans to use more PPAs in the future but limited information provided.			\bigcirc
4 CLIMATE CONTRI	BUTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for Inabated emissions	Offsetting cla	etting claims with limited detail.		0	\bigcirc
Climate contributions		No climate contributions identified.		N/A	0
Offsetting claims today			021. No complete information -based projects mentioned.		0
ffsetting plans for the f	uture A joint ventur	2050 targets depend on offsets to unclear extent. venture with ClimatePartner will develop -based carbon offset projects.			0

Rating criteria 3-point scale **High Moderate Poor**. See methodology document for rating criteria. **Transparency** refers to the disclosure of information. **Integrity** refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Volkswagen (2019, 2021, 2022a, 2022b, 2022c), Volkswagen ClimatePartner (2022), VW Kraftwerk (2021) and Bonaccorsi *et al.* (2022)

Volkswagen Group

Volkswagen AG (hereafter: Volkswagen Group) is the world's second-largest and Europe's largest manufacturer of motor vehicles. Most of the company's emissions originate in the use phase of its sold cars and vans (75% of 2021 emissions) and from purchased goods such as steel (16%). The company aims to become carbon neutral by 2050 but remains non-transparent on how many offsets it intends to use to achieve this pledge. Volkswagen launched a joint venture in 2022 to develop its own nature-based carbon offset credits. Despite a wide range of emission reduction measures across all scopes, Volkswagen's climate strategy for vehicle electrification and its 2030 targets fall way short of decarbonisation milestones for the automobile industry to be in line with the Paris Agreement's 1.5°C global warming limit.

Key developments over the past year: We identified few significant changes to Volkswagen's climate strategy since the previous iteration of this analysis was published in February 2022 (Day *et al.*, 2022). In 2022, Volkswagen announced a new target for scope 1 and 2 emission reductions by 2030 and entered a joint venture with ClimatePartner to develop and finance nature-based offset credits from forest conservation and afforestation projects. Volkswagen has neither provided further clarity on its 2050 carbon neutrality target nor committed to specific phaseout dates for internal combustion engines.

The Volkswagen Group's headline carbon neutrality target remains unsubstantiated as the company provides no information on the extent to which it will reduce its own emissions. Apart from its loosely defined intention to comply with the Paris Agreement, the company does not explain how its 2050 carbon neutrality pledge aligns with key 1.5°C-compatible decarbonisation milestones for the automobile industry (see detailed assessment in Annex II). Since announcing its 2050 carbon neutrality pledge in 2019, the company has not disclosed any further information on the extent to which it will *actually* reduce its own emissions by 2050 as part of this pledge. Volkswagen only vaguely communicates its intention to rely on carbon offsets by an undefined amount to meet its 2050 target (Volkswagen, 2022c, pp. 26, 46).

Volkswagen's interim targets for 2025 and 2030 do not include a commitment to phase out internal combustion engines in key markets by 2030, falling way short of 1.5° C-aligned climate action in the automobile sector. Volkswagen's 2030 target to reduce the CO₂ emission intensity per kilometre by 30% by 2030 below 2018 levels for new vehicles is insufficient to limit global warming to 1.5° C (see detailed assessment in Annex II). The company implicitly acknowledges this by committing 'to increase the level of its ambition to 1.5 degrees Celsius' (Volkswagen, 2022c, p. 46), but has yet to present any updated scope 3 emission targets. Recent analysis suggests that Volkswagen currently underreports its disclosed life-cycle emissions of sold vehicles by more than 50% due to unrealistic assumptions on vehicle lifetimes (Bonaccorsi *et al.*, 2022, pp.15-16).

At COP26 in November 2021, Volkswagen opted out of a declaration committing to a totally electric fleet by 2035 to support achieving the 1.5°C target of the Paris Agreement, despite competing manufacturers in the US and Germany signing up to it (COP26 Presidency, 2021). Volkswagen's lobbying efforts through direct interactions with regulators and membership in regressive industry associations in the EU and US represent contradictory positions on supporting the phase out ICE vehicles and CO, targets for cars and vans (InfluenceMap, 2021, 2022d).

The Volkswagen Group implements a far-reaching investment programme aiming at the company's transformation towards electric mobility. Volkswagen commits to large investments to electrify its vehicles, accelerate the rollout of charging infrastructure, and support battery technology development and low-carbon steel production (Volkswagen, 2022c, pp. 41–49). Group-wide investments for the transition to electric vehicles alone will account for a total of EUR 52 billion (ca. USD 61.5 billion) towards 2026. These measures are important to effectively decarbonise the automobile industry's value chain and have significant potential for replication and wider roll-out. The lack of detailed information on their scale, implementation timelines, and projected reduction impacts, however, does not allow us to evaluate their sufficiency to meet the Group's emission reduction targets. Volkswagen continues to use an internal shadow carbon price of EUR 20 (ca USD 22.5) per tonne subject to annual revision (Volkswagen, 2022c, p. 49). this is significantly below a carbon price level that would give a clear incentive for embarking on a 1.5°C-compatible decarbonisation trajectory.

The Volkswagen Group's claim that it operates carbon neutral production lines—as well as its plans for offsetting in the future—are highly contentious. The company used offset credits to claim neutralisation of 6.1 MtCO₂e in 2021 (Volkswagen, 2022c, p. 43), equivalent to around 85% of its scope 1 and 2 emissions in that year. Volkswagen uses these offset credits, among others, to claim neutrality for two vehicle production lines and nine production sites (Volkswagen, 2019, 2021, 2022c, pp. 42–43). The contentious practice of claiming neutrality only for certain business areas and products can mislead consumers and other stakeholders on the vehicles' actual emissions footprint. Volkswagen also intends to achieve its 2025 interim target by using an unspecified amount of offset credits (Volkswagen, 2022c, pp. 26, 46).

The company entered a joint venture with ClimatePartner in 2022 to develop and finance nature-based offset credits from forest conservation and afforestation projects (Volkswagen ClimatePartner, 2022). According to the limited information available, Volkswagen intends to use these offsets to support its climate strategy and meet its 2050 climate neutrality target. ClimatePartner will also sell credits from this joint venture to other companies to claim neutralisation of emissions. While scaling up biological carbon removals requires more financial support, these removals are unlikely permanent and therefore unsuitable to claim the *neutralisation* of the company's own emissions (see Box 3 in Section 4.2.2).

The Volkswagen Group continues to predominately rely on low-quality RECs to claim the company almost entirely procures renewable electricity for its European operations. The company reports to have achieved a 96% and 49% share of externally procured renewable electricity in Europe and worldwide, respectively, in 2021 (Volkswagen, 2022c, p. 42). For Europe, however, the company procured more than 3.5 TWh of renewable electricity through low-quality RECs (Volkswagen, 2022a). RECs generally provide no effective incentives for *additional* renewable energy installations and remain prone to double counting (see Table 3-2, Section 3.2.2). The company aims to accelerate direct investments in new solar and wind capacity across Europe through PPAs, aiming to generate 7 TWh by 2025, up from 1 TWh in 2021 (RWE, 2021; Volkswagen, 2022c, p. 42). The Volkswagen Group's own electricity generation remains highly depended on fossil fuels. The Group's subsidiary VW Kraftwerke GmbH, which is responsible for the electricity and heat supply for German and Czech production plants, reports a share of around 35% of coal and fossil gas in its 2020 electricity mix (VW Kraftwerk, 2021; Volkswagen, 2022c, p. 42). Overall, the Volkswagen Group discloses little information on its renewable electricity procurement strategy.

SECTOR	REVENUE	EMISSIONS	PLEDGE	TRANSPARENCY	INTEGRITY
Food, beverages & agriculture	USD 572.8 bn (2021)	187.7 MtCO ₂ e (2021)	Zero emissions in operations by 2040	O Moderate	Low
1 TRACKING AND DI	TRANSPARENC	(& INTEGRITY			
Tracking and disclosure 187.7 MtCO2 e in 2021Major emission sources: Major emission sources are related to purchased goods and services (upstream s3, 71%).Disclosure: In public-facing documentation, Walmart only reports on s1 and s2 emissions. S3 emissions are only reported in CDP disclosure.		Scope 1 Scope 2 Scope 3 upstream downstream	6.9 10.4 148.6 35.9		
2 SETTING EMISSIO	N REDUCTION TARG	BETS		TRANSPARENCY	INTEGRITY
Headline target or pledge	Zero emissio	ns in operations by 2040			
Short- and medium-term ((up to 2030) Scope coverage Own emission reductions (compared to full value chain i	compared to	emissions by 35% by 202 2015 levels Short- to medium-term targenission reductions compa emissions: not in line with 1	gets translate to 5% red to 2019 value chain	•	0
Long-term vision (beyond 2030)	Zero emissio	ns in operations by 2040			
Scope coverage Own emission reductions (compared to full value chain i	9 % r	ong-term target translate eductions, compared to 20 emissions. This is not in line	019 value chain	•	0
3 REDUCING OWN E	MISSIONS			TRANSPARENCY	INTEGRITY
Emission reduction measures Significant portion of upstream s3 emissions addressed through supplier engagement programme. Measures to reduce s1&2 emissions presented, but only in vague terms.		0	0		
enewable electricity rocurement		her-quality RE constructs RE share remains low.	to reach 100% RE	0	0
4 CLIMATE CONTRIB	UTIONS AND OFFS	ETTING		TRANSPARENCY	INTEGRITY
Responsibility for unabated emissions	Climate contr	ibutions with limited detail.		\bigcirc	?
Climate contributions		o protect/restore 50 million a utralisation claim. Very limite		\bigcirc	?
Offsetting claims today		— No offsetting claims too		N/A	N/A
)ffsetting plans for the fu		es that it will achieve its tar	gets without carbon overs scope 3 emissions.	N/A	N/A

Rating criteria 3-point scale High Moderate Poor. See methodology document for rating criteria. Transparency refers to the disclosure of information. Integrity refers to the quality and credibility of the approach.

Sources: Authors' interpretation of identified public documentation from Walmart (2020a, 2020b, 2021a, 2021b, 2022a, 2022b).

Walmart

Walmart Inc. (Walmart) is a USA-based retail cooperation that operates grocery stores, department stores and hypermarkets. Most of Walmart's emissions (71% of 2019 emissions) originate from the procurement of goods (upstream scope 3). Walmart has set credible targets to take responsibility for its operational scope 1 and 2 emissions. However, its strategy for upstream scope 3 emissions, which account for most of the company's overall climate impact, lacks a clear reduction commitment. Walmart sets no emissions reduction target for scope 3 emissions, but rather builds on a programme to engage with suppliers to voluntarily reduce emissions themselves.

Key developments over the past year: We could identify only minor changes to Walmart's sustainability strategy since our previous analysis of the case study in the **2022 Corporate** *Climate Responsibility Monitor* (Day *et al.*, 2022), including the continued expansion of the supplier engagement programme to reduce upstream scope 3 emissions. Accordingly, only minor modifications were made to this case study.

Walmart's headline target is to reduce its scope 1 and 2 emissions to zero by 2040, complemented by interim targets for 2025 and 2030. Walmart is committed to reducing its scope 1 and 2 emissions (referred to as 'operational emissions') to zero by 2040 (Walmart, 2022a, p. 28) and does not seek to offset emissions. The company has set interim emission reduction targets for its scope 1 and 2 emissions: reductions of 35% by 2025 and 65% by 2030, compared to a 2015 baseline (Walmart, 2022a, p. 28). The targets translate to approximately a 25% emission reduction from scopes 1 and 2 by 2025 and 60% by 2030, from a 2019 baseline. Including scope 3 emissions, the targets translate to only 5% emission reduction by 2030 and 9% by 2050, compared to 2019 levels (Walmart, 2021b, pp. 19–24).

In 2017, Walmart launched Project Gigaton to address scope 3 emissions, which account for 91% of the company's emissions footprint, but the potential impact of the measures remains unclear. Walmart's emission reduction targets only cover scope 1 and 2 emissions, which represented 9% of the company's total GHG emission footprint in 2019 (Walmart, 2020b, pp. 38–47). To address scope 3 emissions, the company launched its Project Gigaton in 2017. Through *Project Gigaton*, Walmart wants to engage suppliers, offering them guidance to reduce their emissions in six areas: energy, product use and design, waste, forestry, agriculture, and packaging (Walmart, 2022b). Suppliers can sign up to the programme and receive access to resources and training that help them setting their own targets and design strategies to tackle their emissions. To increase the share of renewable electricity in its supply chain, Walmart gives suppliers access to collaborative PPAs (Schneider Electric, 2022) – see Box 1, Section 3.2.2. Since 2017, around 4,500 suppliers have joined the programme (Walmart, 2022b); while these suppliers account for 5% of all suppliers, they count for roughly 50% of Walmart's revenue. The reported number of suppliers has almost doubled since the last iteration of our analysis (Day *et al.*, 2022). With *Project Gigaton*, Walmart aims to reduce 1 GtCO. in *cumulative* scope 3 emissions in the period between 2017 and 2030, which is a significant contribution to reducing Walmart's emissions footprint. Although *Project Gigaton* is presented as a central element of Walmart's sustainability strategy, the company did not commit to any targets for scope 3. It remains unclear how the cumulative emission reductions are aligned with a 1.5°C trajectory.

Walmart's public-facing reporting neglects a large share of emissions; Walmart can improve its GHG emissions reporting to ensure transparency and accountability. In its public climate change strategy, the company does not disclose its scope 3 emissions, which account for 91% of the company's total emissions in 2020 (Walmart, 2021b, pp. 19–24). Furthermore, its main reporting of emissions from energy procurement (scope 2) uses a market-based accounting approach. This reduces energy procurement emissions by around 1 MtCO₂e in 2020 compared to a location-based accounting approach. Scope 3 and location-based scope 2 emission estimates are only included in Walmart's disclosure to CDP, which the company publishes on its website (Walmart, 2021b, pp. 19–24); not in its public-facing sustainability documentation.

Walmart commits not to use offsets to reach its target for zero operational emissions, while pledging to make a climate contribution to support nature-based solutions without claiming to neutralise its emissions. Walmart explicitly plans to reduce scope 1 and 2 emissions to zero by 2040, without the use of offsets (Walmart, 2021a). In parallel, Walmart and Walmart Foundation have committed to protect or restore 50 million acres of land by 2030, without linking this contribution to a neutralisation claim (Walmart, 2020a). This could be a credible approach to supporting naturebased solutions for climate change mitigation outside of its value chain. Walmart could improve their transparency on these contributions by disclosing further information on how it determines the volume of support. It remains unclear whether this is linked to assuming responsibility for unabated emissions, particularly given that scope 3 emissions are not included in Walmart's main climate targets.

Glossary and abbreviations

Additional potential (of CDR)	See "Scarcity (of CDR)"
BECCS	Bioenergy with carbon capture and storage
BEV	Battery electric vehicles
Biological capture and storage	See "Nature-based solutions"
CAR	Climate Action Reserve
CCS	Carbon capture and storage
CCU	Carbon capture and utilisation
Climate contribution	We define climate contributions as the financial support provided by a company to support climate change action beyond the company's own value chain, without claiming the neutralisation of its own emissions in return.
Carbon dioxide removals (CDR)	All scenarios consistent with a 1.5°C temperature increase include a major role for carbon dioxide removals.(Rogelj <i>et al.</i> , 2018) This includes nature-based solutions for carbon sequestration in forests, soils, peatlands and mangroves, technological solutions such as BECCS and DACCS with underground storage, and solutions with mineral storage.
Carbon offset credit	A carbon offset credit is a certified unit of a reduction of GHG emissions, or a removal of carbon dioxide (<i>see Carbon dioxide removals</i>), which is used to balance out GHG emissions elsewhere. The practice of offsetting is often contentious (see section 4.1.2).
CDM	Clean Development Mechanism
CDP	Formerly the Carbon Disclosure Project: Many companies report emissions as well as other details of their climate strategies to CDP. CDP provide companies with a certified rating of their level of climate transparency, which is often used in company's marketing materials.
CEO	Chief Executive Officer
CO ₂	Carbon dioxide
СОР	Conference of the Parties (see UNFCCC).
DACCS	Direct Air Carbon Capture and Storage, see also "Carbon dioxide removals (CDR)"
DRI-EAF	Direct reduced iron – Electric arc furnace
ESG	Environmental Social Governance
EU	European Union

EV	Electric vehicle
FLAG	Forest, Land and Agriculture Science Based Target Setting Guidance (a standard by the Science Based Targets initiative for land-based emissions disclosure and target setting).
GHG Protocol	The GHG Protocol is an initiative driven by the World Resources Institute and World Business Council for Sustainable Development, that provides international guidance and standards for GHG emissions accounting.
GHG	Greenhouse gas
Guarantees of origin (GOs)	Other terminology for Renewable Energy Certificates (REC), see "Renewable Energy Certificates (REC)"
High-hanging fruit	The high-hanging fruit of mitigation potential refers to the technologies and measures to decarbonise emission sources that remain otherwise entirely inaccessible to host country governments in the near- and mid-term future, on account of high costs or other insurmountable barriers that cannot reasonably be overcome.
HLEG	The United Nations' High-Level Expert Group on the Net-Zero Emissions Commitments of Non-State Entities
ІСТ	Information and communications technology
IEA	International Energy Agency
Insetting	'Insetting' is a business-driven concept used by a limited number of actors with no universally accepted definition. Insetting is often described as offsetting within the value chain. The approach can lead to low credibility GHG emission offsetting claims and presents a significant risk of double counting the same emission reductions (<i>see section 4.4.4</i>).
Integrity (rating)	The Corporate Climate Responsibility Monitor assesses the transparency and integrity of companies' climate pledges. Integrity, in this context, is a measure of the quality, credibility and comprehensiveness of a company's approaches towards the various elements of corporate climate responsibility.
IPCC	Intergovernmental Panel on Climate Change
ISO	International Organisation for Standardisation
LEV	Low-emission vehicles
LNG	Liquified natural gas
Location-based method (for scope 2 emissions accounting)	The location-based method for scope 2 emissions accounting reflects the average emission intensity of the electricity grid from which the consumer's energy is delivered.
Market-based method (for scope 2 emissions accounting)	The market-based method for scope 2 emissions accounting reflects the emissions from electricity generation specifically procured by the consumer (which may not reflect the electricity they actually consume from a grid that features multiple buyers and sellers). It derives emission factors from contractual renewable electricity procurement instruments.

Nationally determined contributions (NDCs)	Nationally determined contributions (NDCs) are the pledges made by national governments to the United Nations Framework Convention on Climate Change to mitigate climate change. The Paris Agreement requires all Parties to submit and regularly update their NDCs to represent their possible highest level of ambition. Recognising the insufficiency of climate change mitigation commitments in existing NDCs, the Glasgow Pact from COP26 urged all Parties to update their NDCs again ahead of COP27.
Nature-based solutions	Nature-based solutions refer to measures for carbon dioxide removal that involve biological carbon capture and storage in natural ecosystems, such as soils, forests, peatland and mangroves.
Neutralisation	Neutralisation of emissions is usually a term that is synonymous with offsetting and refers to the balancing out of emissions released into the atmosphere with the avoidance, or removal from the atmosphere, of an equivalent volume of emissions elsewhere. Many actors now avoid the term offsetting entirely; companies and initiatives more often refer to "neutralisation", "netting-out", "compensation", "reducing the footprint", while some actors use multiple terminologies to distinguish between offsetting in different circumstances and at different times. We define all claims that unabated GHG emissions within the value chain are offset as <i>offsetting claims</i> , including all synonymous terminologies and all project types.
Non-GHG climate forcers	Non-GHG climate forcers include the emission of gases and aerosols, and processes that change cloud abundance, leading to radiative forcing. Radiative forcing is a change in the balance of radiation in the atmosphere, which contributes to global warming. For example, the non-GHG climate forcers are estimated to increase the climate impact of GHG emissions from the aviation industry by a factor of approximately 3 (Atmosfair, 2016).
Offsetting	See carbon offset.
Permanence (of CDR)	The <i>permanence</i> of a CDR outcome refers to the timescale and degree to which sequestered carbon remains stored and not released into the atmosphere.
Power purchase agreement (PPA)	A PPA is a long-term contract between an electricity provider and an electricity consumer, usually spanning 10-20 years. The consumer agrees to purchase a certain amount of electricity from a specific asset under a pre-determined pricing arrangement. PPAs are generally signed with new renewable energy installations and form part of the project investment decision (NewClimate Institute and Data-Driven EnviroLab, 2020). PPAs can also be signed for existing installations, in which case it is less likely the PPA results in additional renewable electricity capacity. However, it may be that existing installations would cease operations if the operator cannot sign a new PPA.
PV	Photovoltaics
R&D	Research & development

Renewable energy certificate (REC)	Renewable Energy Certificates (RECs) are also known under various names, such as Guarantees of Origin (GOs) or Energy Attribute Certificates (EACs). RECs can be bundled or unbundled with the electricity that a company consumes:
	 Unbundled RECs – the consumer purchases RECs from a third party, separately from their procurement of electricity from another supplier.
	• Bundled RECs – third-party generated: the consumer purchases electricity and RECs from the same supplier, but this supplier has procured the RECs from a third party. In this situation, the supplier may sell electricity generated using fossil fuels but market it as 'low-carbon' electricity by bundling an equivalent volume of RECs into the sale.
	 Bundled RECs – supplier generated: the consumer purchases renewable electricity and associated RECs from the same supplier.
Residual emissions	Residual emissions are the remaining GHG emissions from hard-to-abate emission sources where no known feasible options remain for further decarbonisation. (See also <i>unabated emissions</i>)
Scarcity (of CDR)	The maximum potential of most carbon dioxide removal measures is technically limited, and even further restricted by environmental constraints. Due to issues such as land requirements, high water consumption, high energy consumption, land degradation and pollution, among other environmental costs, carbon dioxide removal technologies can only be scaled-up so far without significantly endangering sustainable development goals, including food security. The scarcity of carbon dioxide removals measures – in terms of their maximum absolute or annual technical potential – is an important consideration when evaluating the feasibility of net-zero claims at the level of individual actors. Robust future use of scarce carbon dioxide removal options must be consistent with achieving net-zero and eventually net-negative emissions at the global level, which is required to avoid the most damaging effects of climate change over the coming decades.
Science Based Targets initiative (SBTi)	SBTi reviews and certifies the climate targets of companies who join the initiative as members. Companies' climate targets are certified as 1.5°C or 2°C compatible if they align with SBTi's own methodology and benchmarks.
Scope (of GHG emissions)	The GHG Protocol Corporate Standard classifies a company's GHG emissions into three 'scopes' (WBCSD and WRI, 2004):
Scope 1 emissions	Scope 1 emissions are direct emissions from owned or controlled sources.
Scope 2 emissions	Scope 2 emissions are indirect emissions from the generation of purchased energy (see also <i>location-based method</i> and <i>market-based method</i>).
Scope 3 emissions	Scope 3 emissions are all indirect emissions (not included in scope 2) that occur in the value chain of the reporting company, including both upstream and downstream emissions (WRI and WBCSD, 2013).
Upstream scope 3 emission sources	Upstream emissions are indirect GHG emissions related to purchased or acquired goods and services (WRI and WBCSD, 2013).
Downstream scope 3 emission sources	Downstream emissions are indirect GHG emissions related to sold goods and services (WRI and WBCSD, 2013).
Normal scope 3 emission sources	The GHG Protocol's Scope 3 Standard identifies 15 distinct reporting categories for scope 3 emission sources, and requires companies to quantify and report scope 3 emissions from each category (WRI and WBCSD, 2013).

Optional scope 3 emission sources (indirect use-phase emissions)	Indirect use-phase emissions are described by the GHG Protocol Scope 3 Standard (WRI and WBCSD, 2013) as an optional reporting component. In contrast to direct use-phase emissions from products, such as the energy consumption of vehicles and appliances, indirect use-phase emissions refer to the emissions that occur indirectly from the use of a product. For example, apparel requires washing and drying; soaps and detergents are often used with heated water.
Sustainable aviation fuels (SAF)	Sustainable aviation fuels are aviation fuels derived from renewables or waste considering certain sustainability criteria.
Transparency (rating)	The Corporate Climate Responsibility Monitor assesses the transparency and integrity of companies' climate pledges. Transparency ratings refer to the extent to which a company publicly discloses the information necessary to fully understand the integrity of that company's approaches towards the various elements of corporate climate responsibility.
UN	United Nations
UNFCCC	United Nations Framework Convention on Climate Change
Unabated emissions	Unabated emissions are GHG emissions from emission sources for which further emission reductions are technically feasible at that point in time. (See also <i>residual emissions</i>)
Value chain emissions	A company's full value chain emissions refers to the entirety of scope 1, scope 2, and scope 3 emissions.
US	United States
Value chain emissions	A company's full value chain emissions refers to the entirety of scope 1, scope 2, and scope 3 emissions.

Annex I Companies assessed in this report

We assess 24 companies in this report. We refer to them using shortened names (see left column) but assess the company and all subsidiaries covered by the full name (see right column).

Shortened name	Full name
Ahold Delhaize	Koninklijke Ahold Delhaize N.V.
Amazon	Amazon.com, Inc.
American Airlines	American Airlines Group Inc.
Apple	Apple Inc.
ArcelorMittal	ArcelorMittal S.A.
Carrefour	Carrefour S.A.
Deutsche Post DHL	Deutsche Post AG (Deutsche Post DHL Group)
Fast Retailing	Fast Retailing Co., Ltd.
Foxconn	Hon Hai Precision Industry Co., Ltd.
Google	Alphabet Inc.
H&M	H & M Hennes & Mauritz AB (H&M Group)
Holcim	Holcim Limited
Inditex	Industria de Diseño Textil, S.A.
JBS	JBS S. A.
Maersk	A.P Møller - Mærsk A/S
Mercedes-Benz	Mercedes-Benz Group AG
Microsoft	Microsoft Corporation
Nestlé	Nestlé S.A.
PepsiCo	PepsiCo, Ltd.
Samsung Electronics /Samsung	Samsung Electronics Co., Ltd.
Stellantis	Stellantis N.V.
Thyssenkrupp	Thyssenkrupp AG
Volkswagen Group	Volkswagen AG
Walmart	Walmart Inc.

Selection criteria

This analysis assesses only companies that have committed to high-profile climate change mitigation pledges under one of the main corporate climate action networks and initiatives. The key objective of the analysis is to identify replicable good practice while assessing the integrity of the most influential global corporate actors that are putting themselves forwards as climate leaders and role models for other companies. Scrutiny of their plans is also necessary to identify whether these influential leaders really are setting the right examples, and whether the guidance and frameworks upon which they are making their plans are sufficient.

We assess the top three global companies for each of the eight following sectors, according to their annual revenue in 2021 (Forbes, 2022): Automotive manufacturers; electronics; fashion retail; food and agriculture; information and communication technology; shipping and aviation; steel and cement, supermarket retail. Our analysis excludes state-owned companies due to our perception that fundamental differences in management structures and decision-making structures for climate change strategy may significantly detract from the comparability of these companies' plans, and the insights that we can draw from the company sample.

The 24 companies covered by this monitor account for approximately USD 3.16 trillion of revenue in 2021³, approximately 10% of revenue from the world's largest 500 companies (Forbes, 2022). Their total self-reported GHG emission footprints in 2019, including scope 3 emissions, amount to approximately 2.2 GtCO₂e. This is equivalent to roughly 4% of global GHG emissions. ⁴10 of the 24 companies selected through the process described above were also assessed in the 2022 Corporate Climate Responsibility Monitor. The repeat analysis of this small sub-set of companies offers insights into what progress has been made over the past year.

³ The currency conversions in this document are based on the yearly average currency exchange rates published by the U.S. Internal Revenue Service (IRS 2023).

⁴ Some overlap in emission statistics is likely in the cases that one company's scope 3 emissions are included in the scope 1 or 2 emissions of another company in this analysis. We anticipate that any overlap is marginal and of limited significance to the key insights derived from this report. The companies' combined emission footprint may also be higher than this estimate, due to some companies' incomplete emission disclosure. We use 2019 as a base year for analytical purposes, as the most recent year with complete GHG reporting before the COVID-19 pandemic distorted emission trends.



Ahold Delhaize

Moderate

Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term? Scope 1 and scope 2 emissions:

- 29% emission reduction by 2025 (2018 baseline)
- 50% emission reduction by 2030 (2018 baseline)

Scope 3:

• 37% emission reduction by 2030 (2020 baseline)

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

We estimate that Ahold Delhaize's short- and mid-term emission reduction targets translate to emission reductions of 33–34% by 2030, compared to 2019 levels.

Ahold Delhaize has not reported on scope 3 emissions for 2019. We estimated a range for these emissions based on the following assumptions:

- The lower part of the range assumes that Ahold Delhaize's 2019 scope 3 emissions over revenue are proportional to the company's emissions over revenue in 2018. Using this method, a 37% reduction below 2020 levels equals a 33% reduction below 2019 levels.
- The upper part of the range assumes that the average of reported scope 3 emissions for 2018 and 2020. Using this method, a 37% reduction below 2020 levels equals a 34% reduction below 2019 levels.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)?

Ahold Delhaize sets one target covering scope 1 and 2 within a five-year timeframe that uses the same metric as its longer-term targets.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We find that Ahold Delhaize's 2030 targets almost align with global cross-sectoral and sector-specific 1.5°C Paris Agreementaligned decarbonisation milestones. Ahold Delhaize's scope 3 emissions account for 95% if its emission footprint. In the absence of available benchmarks from the scientific literature for mixed-good retailers, we compare Ahold Delhaize's scope 3 target to available 1.5°C-aligned benchmarks for agriculture (Dietz, Harvey, *et al.*, 2022, p. 14; Teske, 2022, p. 328), and its reduction across all scopes to cross-sector global benchmarks (IPCC, 2022).

Teske (2022, p. 328) identifies 1.5°C-aligned absolute emission reduction milestones for various emission sources of agricultural activities, which represent the majority of upstream scope 3 emissions for Ahold Delhaize. Emissions from fuel use on farms, heat used for food processing and -packaging need to reduce by 48%, emissions from purchased electricity on a farm-level or used for food processing and -packaging need to reduce 67%, and emissions AFOLU, non-CO₂ GHGs need to reduce by 34% by 2030. In sum, these required reductions mean a reduction of 38% across all scopes, below 2019 levels. Ahold Delhaize's emission reduction target for scope 3 emissions aligns with the latter. However, the company should also reduce scope 3 emission sources that are not covered by these benchmarks, such as upstream transportation and distribution and upstream emissions from non-food products.

The Transition Pathways Initiative (TPI) derives an emission intensity per tonne of agricultural input aligned with '1.5°C' trajectories by 2030: 1.3 tCO₂/tonne agricultural input (Dietz, Harvey, *et al.*, 2022, p. 14). This represents a 52% reduction in intensity compared to 2.8 tCO₂/tonne agricultural input in the 2020 base year. Due to lack of information on intensity and volumes of agricultural input, we cannot directly assess whether Ahold Delhaize's scope 3 target meets these intensity benchmarks. However, the company's targets and measures contribute to the shift that is signalled by the required change in intensities.

Global cross-sectoral benchmarks require GHG and CO_2 emissions to reduce by 43% and 48% between 2019 and 2030, respectively (IPCC, 2022). In the same period, global methane emissions must decrease by 34%. Ahold Delhaize's targeted emission reductions of 33–34% by 2030 across all scopes by 2030 come close to meet these global benchmarks.

Integrity assessment for long-term target(s) (post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

Scope 1 and scope 2 emissions:

- Net zero by 2040, including a 90% emission reduction target (2018 baseline) Scope 3:
 - Net zero by 2050, including an 83% emission reduction target (2020 baseline)

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? We estimate that Ahold Delhaize's long-term emission reduction targets translate to emission reductions of 82% by 2050, compared to 2019 levels.

Ahold Delhaize commits to emission reductions of at least 90% by 2040 below 2018 levels for its scope 1 and 2 emissions and at least 83% by 2050 below 2020 levels for its scope 3 emissions. These reduction targets substantiate its net-zero pledges for 2040 and 2050. Since the net-zero pledges entail a commitment to deep decarbonisation but with varying levels of ambition, we consider that the net-zero terminology is unlikely to be misleading but can be improved to fully understand aggregate emission reduction impact across the entire value chain by 2040 and 2050 respectively. Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which Ahold Delhaize has largely committed itself to do.

Ahold Delhaize has not reported on scope 3 emissions for 2019. We estimated a range for these emissions:

- The lower part of the range assumes that Ahold Delhaize's 2019 scope 3 emissions over revenue are proportional to the company's emissions over revenue in 2018. Using this method, an 83% reduction below 2020 levels equals a 82.2% reduction below 2019 levels.
- The upper part of the range assumes the average of reported scope 3 emissions for 2018 and 2020. Using this method, an 83% reduction below 2020 levels equals an 82.3% reduction below 2019 levels.

We assumed scope 1 and 2 emissions to stay constant after target realisation in 2040. The emission reduction target alongside the net zero pledge for scope 1 and 2 (90%, 2018 baseline) translates to emission reductions of 89% compared to a 2019 baseline.

ls this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We find that Ahold Delhaize's 2050 targets almost align with global cross-sectoral and sector-specific 1.5°C Paris Agreementaligned decarbonisation milestones. Ahold Delhaize's scope 3 emissions account for 95% if its emission footprint. In the absence of available benchmarks from the scientific literature for mixed-good retailers, we compare Ahold Delhaize's scope 3 target to available 1.5°C-aligned benchmarks for agriculture (Dietz, Harvey, *et al.*, 2022, p. 14; Teske, 2022, p. 328), and its reduction across all scopes to cross-sector global benchmarks (IPCC, 2022).

Teske (2022, p. 328) identifies 1.5°C-aligned absolute emission reduction milestones for various emission sources of agricultural activities, which represent upstream scope 3 emissions for Ahold Delhaize. Emissions from fuel use on farms, heat used for food processing and -packaging, purchased electricity on a farm-level or used for food processing and -packaging need to reduce by 100% by 2050, whereas AFOLU emissions and non-CO₂ emissions need to reduce by 42% by 2050 below 2019 levels. In sum, these required reductions mean a reduction of 51% across all scopes, below 2019 levels. Ahold Delhaize's emission reduction target for scope 3 emissions aligns with the latter.

The Transition Pathways Initiative (TPI) derives an emission intensity per tonne of agricultural input aligned with '1.5°C' trajectories by 2050: 0.4 tCO_2 /tonne agricultural input (Dietz, Harvey, *et al.*, 2022, p. 14). This represents an 85% reduction in intensity compared to 2.8 tCO₂/tonne agricultural input in the 2020 base year. Due to lack of information on intensity and volumes of agricultural input, we cannot directly assess whether Ahold Delhaize's scope 3 target meets these intensity benchmarks. However, the company's targets and measures contribute to the shift that is signalled by the required change in intensities.

Global cross-sectoral benchmarks require GHG emissions to reduce by 84% between 2019 and 2050 (IPCC, 2022). Ahold Delhaize's targeted emission reductions of 82% across all scopes by 2050 come close to meet these global benchmarks.

We evaluate the Ahold Delhaize's 2040 and 2050 targets as 'moderate' integrity rather than 'high' because of the use of a net-zero target rather than a long-term emissions reduction target, as well as the fact that not every scope target is fully aligned with available benchmarks.

Amazon

() Poor

Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

- Amazon commits to the following two short- and medium-term targets:
 - 100% share of renewable electricity across all operations by 2025
 - 50% of its shipments to be 'net zero' by 2030

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? We estimate that both targets can jointly result in a **maximum of 16% emission reduction by 2030** below 2019 levels across Amazon's value chain.

The 2025 target on renewable electricity translates into emission reduction of maximum 11% across the value chain by 2030 below 2019 levels. This 11% is based on Amazon's market-based scope 2 emissions in 2019; a location-based estimate is not available. Amazon does not commit to a specific emission reduction target alongside its pledge to make 50% of its shipment 'net zero' by 2030. In 2019, Amazon's scope 1 emissions made up about 10% of the total emission disclosure. If we assume that these scope 1 emissions are entirely attributable to shipments and 'net zero' means reducing to zero, this 2030 target could lead to an additional emission reduction of 5% by 2030 below 2019 emission levels. However, this would be a most optimistic interpretation possible.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

Amazon sets a 100% renewable energy target within a five-year timeframe. The company communicates no interim target within a five-year timeframe for its 50% 'net-zero shipment' by 2030 target.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We are unable to compare Amazon's 2025 and 2030 targets to sectoral 1.5°C Paris Agreement-aligned benchmarks as existing literature provides few specific milestones for the technology service industry. Due to this gap in existing literature, we are not able to assess Amazon's decarbonisation efforts within its value chain against sectoral benchmarks aligned with the Paris Agreement's 1.5°C temperature limit.

Amazon's emission reduction commitment does not comply with global cross-sectoral 1.5°C Paris Agreement-aligned emission trajectories. Amazon commits to no more than 16% emission reduction by 2030 from 2019 levels across the entire value chain, thus falling short of the need to reduce global GHG and CO₂ emissions by 43% and 48% respectively to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022). Given that Amazon is active in a sector with readily accessible decarbonisation options, we consider that Amazon's target for 2030 should meet at least the global benchmark of a 43% GHG reduction below 2019 levels.



Integrity assessment for long-term target(s)(post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long-term beyond 2030? Amazon commits to a 'net-zero' carbon by 2040 target.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Amazon neither commits to any emission reduction target alongside its 'net-zero' carbon by 2040 pledge nor specifies the pledge's emission coverage along the company's value chain. Since Amazon's 'net-zero' carbon pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). 'Net -zero' targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which Amazon does not commit to.

Is this emission reduction commitment in line with 1.5°C compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside Amazon's 'net-zero' carbon by 2040 pledge as highly insufficient. To stand a reasonable chance of limiting global warming to 1.5° C, deep and credible emission reductions in all economic sectors towards mid-century are necessary (IPCC, 2022). A global 1.5° C-aligned pathway across all sectors would require a 100% reduction in CO₂ emissions from 2010 levels by 2050 (2045–2055 interquartile range) and an 84% (73–98%) reduction below 2019 levels of all GHGs by 2050 (IPCC, 2018, 2022). The absence of a long-term reduction commitment in Amazon's climate strategy does not reflect a sense of urgency to decarbonise the company's business.

American Airlines



Integrity assessment for short- and medium-term target(s) towards 2030

${f (i)}\,$ What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

- Achieve an absolute reduction of 50 million gallons of jet fuel from fuel-efficiency initiatives by 2025, using 2019 aircraft as a baseline.
- Source 2.5 million gigajoules of cost-competitive renewable energy by 2025.
- Replace 10% of jet fuel with SAF by 2030.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? The company does not commit to any emission reduction target for 2030, whether in intensity or absolute terms. Its commitments to use SAF and reduce jet fuel consumption may bring no emission reductions at all if an increase in activity outweigh any efficiency improvements.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

Literature provides that CO₂ emissions from aviation should decrease by at least 23% by 2030 to be in line with 1.5°C-aligned trajectories (IEA, 2021b, p. 199; CAT, 2022a). American Airlines does not meet these reduction benchmarks, as it has not set any emission reduction target for 2030. The company also does not commit to reduce the non-GHG radiative forcing impact from its business activities.

The TPI's 1.5°C benchmark for airlines provides that companies should have a carbon intensity of $616 \text{ gCO}_2/\text{RTK}$ by 2030 (Dietz, Byrne and Sheer, 2021, p. 14). To meet this benchmark, American Airlines would need to reduce its carbon intensity by 39% by 2030, compared to 2019. American Airlines' commitment to reduce carbon intensity by 45% by 2035, does not suggest the company is planning – or on track – to achieve a 39% reduction five years earlier.

American Airlines' commitment to use 10% of SAF by 2030 meets the minimum level that various studies show is necessary to bring the aviation sector on a 1.5°C-compatible trajectory. Benchmarks for the share of sustainable aviation fuel by 2030 start at 9%-10 by 2030 (UNFCCC, 2021, p. 12; Teske, 2022, p. 212), while other estimates point to higher shares of 13-18% by 2030 (IEA, 2021b, p. 138; Boehm *et al.*, 2022, p. 74).

Given that American Airlines' SAF target meets only the least ambitious benchmarks, and the company has not committed to any emission reduction targets by 2030, we consider its commitments not aligned with a 1.5°C trajectory.



Integrity assessment for long-term target(s) (post-2030)

$\widehat{\mathbf{i}}$ What do the long-term targets actually mean?

What are the targets for the long-term beyond 2030?

- Reduce the intensity of GHG emissions associated with jet fuel by 45% per revenue tonne kilometre (RTK) operations by 2035, compared to 2019.
- Net-zero GHG emissions by 2050

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Due to uncertainty about how the aviation sector recovers from the COVID-19 pandemic and expected emissions in 2035, we are unable to calculate what American Airlines' intensity target means for emissions across its value chain.

American Airlines' net-zero pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain). We consider that the terminology of the net-zero target may be misleading. Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which American Airlines does not commit to. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b).

Is this emission reduction commitment in line with 1.5°C compatible trajectories or benchmarks for the sector?

American Airlines' commitment to reduce the intensity of GHG emissions from jet fuel by 45% per RTK by 2035 meets the SBTi's benchmark for "well below 2°C" but misses TPI's benchmarks that are compatible with global warming of 'below 2°C' and '1.5°C' (Dietz, Byrne and Sheer, 2021; SBTi, 2021b, 2021d). The latest TPI assessment does not yet include American Airlines 45% intensity reduction commitment: using a carbon intensity of 1016 gCO₂/RTK in 2019 as the baseline, a 45% reduction would bring American Airlines to an emissions intensity of 559 gCO₂/RTK by 2035. This is above the TPI's benchmarks for both 'well below 2°C' and '1.5°C', which are set at 542 gCO₂/RTK and 463 gCO₂/RTK by 2035, respectively.

Both the SBTi's and TPI's benchmarks focus exclusively on jet fuel, not taking into consideration GHG emissions that take place elsewhere in airlines' value chains. The TPI benchmarks also exclude non- CO_2 emissions and radiative forcing impacts; although the TPI notes that its benchmarks 'would almost certainly be tighter' if the non- CO_2 impacts of aviation were taken into account (Dietz, Byrne and Sheer, 2021, p. 14).

In the absence of a commitment to real emission reductions within the value chain, American Airlines' net-zero pledge for 2050 does not meet 1.5°C Paris Agreement aligned milestones for the aviation sector. The IEA Net Zero report shows a reduction of 80% by 2050, compared to 2019; the CAT a reduction of 90%, and Teske a reduction of 100% (IEA, 2021b, p. 199; CAT, 2022a; Teske, 2022, p. 216) as Paris aligned . In addition, the ICCT shows that reducing emissions in the aviation sector by 94% between 2019 and 2050 is compatible with a 1.75°C target (Graver *et al.*, 2022, p. i).

Apple

 \square

Integrity assessment for short- and medium-term target(s) towards 2030

(\mathbf{i}) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Apple pledges to become **carbon neutral by 2030**. Alongside this pledge, Apple commits to an emissions **reduction target of 75% by 2030** below 2015 levels across the entire value chain.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

The emission reduction target equates to a **63% emission reduction by 2030 below 2019 levels across the entire value chain**. The company will offset all remaining emissions in 2030.

Since this reduction target alongside the carbon neutrality pledge falls short of a commitment to deep emission reductions (i.e., at least 90% below 2019 levels), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi's Net Zero Standard (SBTi, 2021c; ISO, 2022b). Carbon neutrality targets can give consumers and investors the impression that the company will reach deep emission reductions by 2030, although the 2030 emission reduction target will only bring Apple part way along a trajectory towards deep decarbonisation, and the company does not currently commit to going further.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? Apple does not commit to earlier interim targets within a five-year time horizon.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We are unable to compare Apple's targets towards 2030 to sectoral 1.5°C-aligned benchmarks as existing literature provides few specific milestones for the electronics industry. This gap in existing literature allows no conclusive assessment of sector-specific decarbonisation efforts across the entire value chain in line with the Paris Agreement's 1.5°C temperature, especially for scope 3 emissions in the electronics industry (representing 96% of Apple's total emissions in 2019).

Apple's absolute emission reduction target for 2030 aligns with 1.5°C-aligned cross-sectoral trajectories. Apple's targeted emission reduction of 63% across its entire value chain by 2030 aligns with the need to reduce global GHG and CO_2 emissions by 43% and 48% respectively to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022). Given that CO_2 is the most relevant GHG in Apple's emission profile and the company operates in a sector with readily accessible decarbonisation options, we consider that Apple's target for 2030 should meet at least the global benchmark of a 48% reduction below 2019 levels.

We evaluate Apple's 2030 target to have 'moderate' rather than 'high' integrity because of the lack of a corresponding short-term target within a five-year timeframe to substantiate it. Such a target represents corporate good practice as it requires immediate action from corporate leadership and makes it easier to hold them accountable.



Integrity assessment for long-term target(s) (post-2030)

${f (i)}\,$ What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? Apple does not commit to any long-term target beyond 2030.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? Apple does not commit to any emission reduction targets after 2030.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target to be highly insufficient considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

ArcelorMittal



Integrity assessment for short- and medium-term target(s) towards 2030

${f i} ight)\,$ What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

ArcelorMittal commits to the following 2030 targets:

- 25% emissions intensity reduction by 2030 for global scope 1 and 2 emissions below 2018 levels (equal to intensity of 1.54 tCO₂/tonne steel in 2030)
- 5% emissions intensity reduction by 2030 for European scope 1 and 2 emissions below 2018 levels (equal to intensity of 1.11 tCO₂/tonne steel in 2030)

How does these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? We cannot independently quantify ArcelorMittal's 2030 emission intensity targets in terms of absolute emission reductions across the value chain compared to a 2019 baseline.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

ArcelorMittal does not commit to earlier interim targets within a five-year time horizon.

Do the short- and medium-term targets align with a 1.5°C trajectory for the sector according to available literature?

We cannot independently assess ArcelorMittal's 2030 intensity targets against existing 1.5°C Paris Agreement-aligned global benchmarks for the steel sector. Existing literature identifies several 2030 benchmarks for steelmakers, focusing mostly on scope 1 and 2 emissions. The carbon intensity of global steel production must reach around 1.13–1.35 tCO₂e/tonne steel (CAT, 2020b; Boehm *et al.*, 2021; Boehm *et al.*, 2022; Dietz, Gardiner, and Scheer, 2022). The Transition Pathways Initiative evaluates ArcelorMittal's global 2030 targets as only aligned with its 'below 2°C' benchmark (TPI, 2022b). However, TPI does not provide any further information on the assumptions underlying its evaluation of ArcelorMittal's 2030 target.

Applying global benchmarks for the steel sector to individual companies requires a significant level of detail on the specific targets. This includes information about the expected share of primary versus secondary steel production in the target year compared to the base year, and the emission boundaries used to calculate baselines. For example, ArcelorMittal's 2030 target for scope 1 and 2 emissions includes some emissions from mining (ArcelorMittal, 2021a, p. 49). Such emissions from mining are possibly not considered in the benchmarks' boundaries. To estimate whether a certain benchmark is compatible with ArcelorMittal's targets in terms of their emission boundaries, we would require more specific information on how the company calculates its emissions intensity baseline.

ArcelorMittal might be sufficiently contributing towards the achievement of other 1.5°C-aligned milestones for the global steel sector's low-carbon technology roadmap. Identified benchmarks in the literature imply that 20 low or near-zero carbon facilities become operational by 2030 (UNFCCC, 2021, p. 15), 70 near-zero emissions primary steel mills by 2030 (Delasalle *et al.*, 2022, p. 69), and most new clean technologies in heavy industry have been demonstrated at scale (IEA, 2021b, pp. 20; 129). ArcelorMittal plans to develop their own near-zero emissions steelmaking facilities, including a zero carbon emissions and green hydrogen powered steel plant in Spain and other zero or near-zero carbon plants in Germany, France and Canada (ArcelorMittal, 2021a, pp. 13–27). Although the company does not provide details on the overall technology split by 2030, we consider they are likely sufficiently contributing to the achievement of these sector-wide benchmarks.

Integrity assessment for long-term target(s)(post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long-term beyond 2030? ArcelorMittal commits to net-zero by 2050.

How does these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

ArcelorMittal does **not firmly commit to a deep emission reduction target** alongside its net zero pledge but estimates its own emission reduction across the entire value chain to be around 95% by 2050 compared to 2018 emission levels (ArcelorMittal, 2021a). This indicative estimate translates to emission reductions of maximum 84% by 2050 across the full value chain compared to 2019 emission levels; however, 84% represents the most optimistic interpretation as ArcelorMittal currently does not fully report on its scope 3 emissions (ArcelorMittal, 2022, p. 15).

Since the company's net-zero pledge does not entail any specific commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which ArcelorMittal does not firmly commit to.

Is the emission reduction commitment as part of the long-term vision in line with 1.5°C compatible trajectories or benchmarks for the sector?

We consider the lack of a firm post-2030 emission reduction target alongside ArcelorMittal's net-zero pledge as highly insufficient. To stand a reasonable chance of limiting global warming to 1.5°C, deep and credible emission reductions towards mid-century are necessary (IPCC, 2022). The lack of a reduction commitment does not signal this sense of urgency. If ArcelorMittal were to adopt its indicative emission reduction estimate as an official target alongside its net zero pledge, the company would likely come close to meet 1.5°C-aligned sectoral benchmarks identified in the literature. Global steel production must reach around zero to 0.13 tCO₂e/tonne steel by 2050 globally (CAT, 2020b; Boehm *et al.*, 2021; Dietz, Hastreiter, and Scheer, 2021; Boehm *et al.*, 2022), representing a reduction in intensity of ar ound 90% compared to 2015 or 2020 levels. Identified benchmarks in the literature also imply all steel facilities in operation by 2050 to be low-carbon (Boehm *et al.*, 2021).

Further assumptions for aggregate analysis in Section A

For our aggregate analysis across all 24 companies in Section A of this report, we make the simplified most optimistic assumption that companies' *intensity* targets equal *absolute* emission reduction targets. For this purpose, we interpret ArcelorMittal's *intensity* targets for 2030 (25% by 2030 below 2018 for scope 1 and 2) to be an *absolute* target assuming constant activity levels. Under this assumption, the target translates to a 20% reduction by 2030 below 2019 levels across the entire value chain. We do not use this estimate for the company-specific integrity assessment of ArcelorMittal in Section B.

Carrefour



Integrity assessment for short- and medium-term target(s) towards 2030

What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

By 2030, Carrefour pledges to reduce scope 1 and 2 emissions by 50%, and scope 3 emissions by 29%, compared to 2019 levels.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? We estimate that Carrefour's targets are equivalent to a commitment to reduce emissions across the value chain by 6–15% by 2030, compared to 2019 levels.

Due to scope 3 emissions accounting for 98% of the company's footprint (Carrefour, 2022c, p. 74), a combination of the targets for a 50% reduction of scope 1 and 2 emissions and a 29% reduction of scope 3 would imply emission reductions of 29% across the full value chain. However, due to the exclusion of the majority of Carrefour locations from its targets, we interpret that these reduction targets only apply to 20–50% of Carrefour output. Carrefour's emission reporting includes only 'integrated stores in integrated countries' (Carrefour, 2022c, p. 163). Stores that Carrefour identifies as 'integrated stores' account for less than 11% of Carrefour's 5,799 stores in France, and less than 20% of Carrefour's 13,894 stores worldwide. Other administrative buildings, warehouses, and supply chain emissions associated with the remaining more than 80% of Carrefour stores worldwide appear to be excluded.

The 6–15% should be understood as an indicative estimate, since we cannot tell what proportion of Carrefour output is covered by the targets. Although the target covers less than 20% of Carrefour's stores worldwide, it is very likely that these stores account for more than 20% of the company's output, since the proportion of hypermarkets covered is considerably higher than the proportion of smaller community stores covered. Accordingly, we offer an indicative estimate that 20-50% of company output is covered by the target. Reducing this 20–50% of the company's emissions footprint by 29% would lead to overall emission reductions in the range of 6–15% compared to the full value chain emissions.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

Carrefour sets a range of short-term targets that use the same metrics as its 2030 medium-term targets (absolute GHG emission reductions) for scope 1 and 2. For scope 3, the 2025 targets focus on non-GHG indicators for specific emission sources within the value chain.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

Carrefour's 2030 medium-term targets neither meet cross-sectoral nor sector-specific 1.5°C Paris Agreement-aligned

decarbonisation milestones. Carrefour's scope 3 emissions account for 98% if its emission footprint. In the absence of available benchmarks from the scientific literature for mixed-good retailers, we compare Carrefour's 6–15% emission reductions by 2030 to available 1.5°C-aligned benchmarks for agriculture, and cross-sector global benchmarks. Global cross-sectoral benchmarks require GHG and CO₂ emissions to reduce by 43% and 48% between 2019 and 2030, respectively (IPCC, 2022). Pathways for global agriculture and food sector in Teske (2022, p. 328) indicate that scope 3 emissions should reduce by at least 34% between 2019 and 2030. The exclusion of most stores from Carrefour's targets means that its overall emission reduction commitment falls far short of any of these benchmarks.



Integrity assessment for long-term target(s) (post-2030)

i)What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

Carrefour pledges to be carbon neutral by 2040. This includes a commitment to 70% reduction of scope 1 and 2 emissions only.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? We estimate that Carrefour's emission reduction target is equivalent to a commitment to reduce less than 1% of its emissions across the value chain by 2040, compared to 2019 levels.

Carrefour's 2040 target covers only scope 1 and scope 2 emissions, which account for approximately 2% of the company's GHG emission footprint in 2019. We interpret that these targets only apply to 20-50% of Carrefour scope 1 and 2 emissions, due to the exclusion of the majority of Carrefour locations from the targets (see assessment of short- and medium-term targets above). These exclusions would mean that, overall, less than 1% of Carrefour's full value chain emissions are covered by the target.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction commitments for scope 3 alongside Carrefour's carbon neutrality pledge as highly insufficient, considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

Deutsche Post DHL



Integrity assessment for short- and medium-term target(s) towards 2030

${f i}ig)$ What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

Deutsche Post committed to the following 2030 targets:

- Reduce scope 1 and 2 emissions by 42% below a 2021 baseline; and
- Reduce scope 3 emissions from fuel- and energy-related activities by 25% below a 2021 baseline.

Deutsche Post DHL also committed to earmark EUR 7 billion (USD 8.3 billion) for sustainable fuels and technologies by 2030. This will be used to achieve:

- At least a 30% share of sustainable fuels in air and ocean freight, and road transport;
- 60% e-vehicles in pick-up and delivery; and
- All newly owned buildings are to be climate neutral (we could not identify a definition of 'climate neutral').

How does these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Deutsche Post DHL's targets equal a **12% emission reduction by 2030 below 2019 levels** across its value chain. Non-logistics-related emissions are excluded from Deutsche Post DHL's 2030 targets. These emissions account for about 6 MtCO₂e every year, or 14% of total emissions. Compared to 2021, the committed reductions equal a reduction of 25% across the entire value chain (i.e., including non-logistics-related emissions). Deutsche Post DHL's emissions across the value chain increased by 18% between 2019 and 2021 (from 38.8 MtCO₂e to 45.7 MtCO₂e). Accordingly, the committed reduction compared to 2019 is lower, namely 12%.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)? Deutsche Post DHL does not commit to earlier interim targets within a five-year time horizon.

Do the short- and medium-term targets align with a 1.5°C trajectory for the sector according to available literature?

Deutsche Post DHL's 2030 targets for sustainable aviation and maritime fuels meet 1.5°C Paris Agreement-aligned decarbonisation milestones. However, its commitment to absolute emission reductions falls short of most sectoral benchmarks for the aviation and shipping sectors.

The majority of Deutsche Post DHL's emissions are from own and subcontracted transport. Air freight accounts for 70% of logistics-related emissions, ground transport for 22%, shipping for 7%, and buildings for 1% (Deutsche Post DHL, 2022d). We could not identify emission reduction targets for the separate business sections, so have compared the 12% reduction commitment by 2030 to sectoral benchmarks for air freight, ocean freight and road transport.

While emission reduction benchmarks for the aviation and shipping sectors vary, most indicate a decline in CO_2 emissions of at least 20% in these sectors between 2019 and 2030. For instance, the IEA Net Zero by 2050 report shows a reduction in aviation's CO_2 emissions of 23% and the CAT of 54% (IEA, 2021b, p. 199; CAT, 2022a). For the shipping sector, the SBTi maritime guidance shows that sectoral emissions decrease by 36% between 2020 and 2030, the IEA Net Zero by 2050 report shows a decrease in CO_2 emissions of 20% between 2019 and 2030, IRENA's decarbonisation pathway provides that CO_2 emissions from the sector are to decrease by about 25% between 2019 and 2030 to be aligned with a 1.5°C trajectory, and the CAT shows a decrease in CO_2 emissions of 51% between 2019 and 2030 (IEA, 2021b, p. 199; IRENA, 2021, p. 81; CAT, 2022b; SBTi, 2022e, p. 11).

Deutsche Post DHL's commitment to use at least 30% sustainable fuels in ocean and air freight by 2030 goes beyond what sectoral benchmarks require. The UNFCCC (2021, p. 15) provides that zero emission fuels make up 5% of international shipping fuels and 15% of national shipping fuels by 2030, Boehm *et al.* (2022, p. 74) outline that 5-17% of maritime fuels need to be zero-emission fuels by 2030 for the shipping sector to be aligned with 1.5°C-aligned pathways, the IEA (2021b, p. 138) finds that ammonia, hydrogen and biofuels will account for 17% of all maritime fuel by 2030, and Teske (2022, p. 213) provides that renewable and synthetic fuels needs to increase rapidly and reach a 30% share by 2030 in shipping freight fuel use by 2030.

Benchmarks for the share of sustainable aviation fuel by 2030 start at 9%–10% by 2030 (UNFCCC, 2021, p. 12; Teske, 2022, p. 212), while other estimates point to higher shares of 13–18% by 2030 (IEA, 2021b, p. 138; Boehm *et al.*, 2022, p. 74).

Deutsche Post DHL's 12% reduction commitment does not meet sectoral benchmarks for road transport. The IEA provides that a 34% reduction is needed for all road transport, and a 12% reduction for trucks between 2019 and 2030 (IEA, 2021b, p. 199). Teske (2022, pp. 214, 333) shows a decrease in road transport emissions of 34%; and a 10% share of renewable fuels for road freight.

The SBTi labelled Deutsche Post DHL's interim targets as '1.5°C compatible' but we could not verify what assumptions went into this assessment and how SBTi came to its conclusion. The SBTi's sectoral guidance for the transport sector is based on an emissions trajectory consistent with a 50% chance of limiting global warming of 1.75°C (SBTi, 2018c, pp. 6, 27–28).

Integrity assessment for long-term target(s) (post-2030)

i) What do the long-term targets actually mean?

 \bigcirc

Poor

What are the targets for the long-term beyond 2030? Deutsche Post DHL commits to net-zero logistics-related emissions by 2050.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

We are unable to estimate what emission reductions across the value chain Deutsche Post DHL commits to. While the company states that it will reduce its emissions to an 'unavoidable minimum' (Deutsche Post DHL, 2022a, p. 18), it does not specify what this means in terms of a specific emission reduction target alongside the net-zero target. We consider the terminology 'net zero' as potentially misleading if it is not accompanied by a clear commitment to deep decarbonisation. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Deutsche Post DHL could enhance clarity of its net-zero pledge by indicating what order of magnitude it expects its residual emissions to be in 2050. The company could further set reduction targets between 2030 and 2050 to chart the pathway to deep decarbonisation.

Is the emission reduction commitment as part of the long-term vision in line with 1.5°C compatible trajectories or benchmarks for the sector?

We consider the lack of a specific post-2030 emission reduction target alongside Deutsche Post DHL's net-zero pledge insufficient, considering the need for deep emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

Sectoral benchmarks show that the shipping sector should be close to fully decarbonised between 2040 and 2050, with very limited room for offsetting unabated emissions (IEA, 2021b, p. 199; Smith *et al.*, 2021, p. 106; SBTi, 2022e, p. 11; Teske *et al.*, 2022, p. 213). Emissions from road transport should decrease by 94% (IEA, 2021b, p. 199; Teske, 2022, p. 216). The benchmarks for CO₂ emissions from aviation are more equivocal. The IEA Net Zero for 2050 report shows a reduction of 80% by 2050, compared to 2019; the CAT a reduction of 90%, and Teske a reduction of 100% (IEA, 2021b, p. 199; CAT, 2022a; Teske, 2022, p. 216). The ICCT shows that reducing emissions in the aviation sector by 94% between 2019 and 2050 is compatible with a 1.75°C target (Graver *et al.*, 2022).

Fast Retailing



Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Fast Retailing's headline pledge is a commitment to reduce **absolute emissions from its own operations (such as stores and main offices) by 90% and emissions from raw materials, fabric, and garment production for the Uniqlo and GU brands by 20% by 2030**, both from a 2019 base year.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? These targets equate to a commitment to reduce Fast Retailing's complete value chain emissions footprint by 19% by 2030, compared to 2019 levels.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? Fast Retailing does not commit to earlier interim targets within a five-year time horizon that entail short-term accountability.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

Fast Retailing's 2030 medium-term targets do not meet cross-sectoral and sector-specific 1.5°C Paris Agreement-aligned milestones identified in existing literature. According to the IPCC's global economy-wide benchmarks to keep warming below 1.5°C, GHG emissions should reduce by 43% by 2030 compared to 2019 levels (IPCC, 2022). Given that emissions in the fashion industry occur in various sectors, including agriculture and energy, we expect the industry to decarbonise at the same speed as this global trajectory. Fast Retailing's targets fall short of this global benchmark.

The company's target also misses sectoral benchmarks. Teske (2022, pp. 322; 327) considers that between 2019 and 2030, the *textile and leather industry* and the *manufactured fibres and synthetic rubber industry* should reduce their GHG emissions by 41% and 46%, respectively. This covers all emissions associated with producing fabrics and other materials and manufacturing the clothes. To be in line with these sectoral benchmarks, Fast Retailing's target for *upstream* scope 3 emissions should be set at a level of *at least* 41%. However, its target for upstream scope 3 emissions represents a 16% reduction below 2019.

Assuming that emissions reduce linearly in the 2019–2030 period, Fast Retailing is currently committed to annual emission reductions of 8.2% for scopes 1 and 2 and 1.3% for scope 3, by 2030. The SBTi guidance requires companies in the apparel industry to commit to annual reductions in their scope 1, 2 and 3 emissions of at least 2.5% to comply with the 'well below 2°C' benchmark and 4.2% to comply with the SBTi's 1.5°C benchmark (SBTi, 2018a, pp. 22; 27). While Fast Retailing's reduction target for scope 1 and 2 emissions is likely aligned with the SBTi's 1.5°C, its target for scope 3 does not even meet the 'well below 2°C' benchmark.



Integrity assessment for long-term target(s)(post-2030)

${f i}ig)$ What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? Fast Retailing pledges **net-zero emissions by 2050**.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Fast Retailing does not commit to a deep emissions reduction target alongside its net-zero pledge by 2050. Since Fast Retailing's net-zero pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which the company does not commit to.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside Fast Retailing's net-zero pledge as highly insufficient considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

Foxconn



Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Foxconn commits to the following short- and medium-term targets:

- To reduce carbon emissions by 21% by 2025, from a 2020 baseline
- To reduce carbon emissions by 42% by 2030, from a 2020 baseline
- To achieve net-zero emissions in the offices of its Taiwanese campuses in 2030.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

We cannot independently estimate the intended emission reductions compared to a 2019 baseline because of the incomplete emissions disclosure for 2019. For this reason, we can only use Foxconn's target of 42% carbon emission reductions by 2030 compared to a 2020 baseline. We assume that Foxconn's targets apply to the company's full value-chain emissions unless stated otherwise.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? Foxconn sets a target within a five-year timeframe that uses the same metric as its 2030 medium-term emissions reduction target.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We are unable to compare Foxconn's targets to sectoral 1.5°C-aligned benchmarks as existing literature provides few specific milestones for the electronics industry. This gap in existing literature allows no conclusive assessment of sector-specific decarbonisation efforts across the entire value chain in line with the Paris Agreement's 1.5°C temperature, especially for scope 3 emissions in the electronics industry.

Foxconn's 2030 medium-term target comes close to meeting 1.5°C Paris Agreement-aligned cross-sectoral global milestones identified in existing literature. Given that CO₂ is the most relevant GHG in Foxconn's emission profile, we consider that Foxconn's target of 42% carbon emission reductions by 2030 from 2020 levels does not fully reach the 48% reduction from 2019 levels required to limit global warming to 1.5°C (IPCC, 2022).

Integrity assessment for long-term target(s) (post-2030)

${f i}ig)$ What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

Foxconn does not commit to any emission reduction target alongside its pledge to achieve **net-zero emissions across its value chain by 2050**. However, the company commits to a **63% emissions reduction target by 2035, compared to a 2020 baseline**.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

We are not able to estimate the emission reductions from a 2019 baseline because of the incomplete emissions disclosure for 2019. For this reason, we can only use Foxconn's target of 63% carbon emission reductions by 2035 compared to a 2020 baseline. We assume that Foxconn's targets apply to the company's full value-chain emissions unless stated otherwise.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

Foxconn's targeted emission reductions towards 2035 may be aligned with global 1.5°C-compatible trajectories up to 2035, but the company does not specify any further emission reduction targets alongside its 2050 net-zero pledge. The IPCC finds that global 1.5°C-aligned pathways across all sectors require an 80% reduction in CO_2 emissions below 2019 levels by 2040 and an 84% reduction below 2019 levels of all GHGs by 2050 (IPCC, 2022). Foxconn's target to reduce 63% of its carbon emissions by 2035 likely aligns with this 1.5°C-compatible trajectory for global CO_2 emissions. As Foxconn sets no emission reduction targets alongside its 2050 net-zero pledge, we evaluate the integrity of Foxconn's long-term vision only as 'moderate'.

Google

() Poor

Integrity assessment for short- and medium-term target(s) towards 2030

${f i}ig)$ What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

- Google commits to the following three short-and medium-term targets:
 - 'Staying carbon neutral' each year going forward
 - 24/7 renewable electricity across all operations sourced by 2030
 - Achieving 'net-zero emissions' by 2030

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Alongside its 'net-zero emissions' pledge, Google commits to an **emission reduction target of 50% from its market-based emissions** below 2019 levels across the entire value chain. This translates into a **37% reduction of its location-based emissions** by 2030 below 2019 levels across all scopes. Since this reduction target alongside the net-zero pledge falls well short of a commitment to deep emission reductions (i.e., at least 90% below 2019 levels), we consider that the terminology 'net-zero emissions' may be misleading. Our position is in line with the ISO Guidelines for Net Zero and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net -zero targets can give consumers and investors the impression that the company aims to reach deep level of emission reductions, which Google does not commit to.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

Google does not commit to earlier interim targets within a five-year time horizon. Google's pledge to be 'carbon neutral' each year is not a target but rather a 'neutralisation' claim, that is not related to its target of 'net-zero emissions' by 2030.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We are unable to compare Google's targets towards 2030 to sectoral 1.5°C-aligned benchmarks as existing literature provides few specific milestones for the technology services industry. Due to this gap in existing literature, we are not able to assess Google's decarbonisation efforts within its value chain against sectoral benchmarks aligned with the Paris Agreement's 1.5°C temperature limit.

Google's emission reduction commitment does not comply with global cross-sectoral 1.5°C Paris Agreement-aligned emission trajectories. Google's goal to reduce 37% of its location-based emissions across its entire value chain by 2030 falls short of the need to reduce global GHG and CO_2 emissions by 43% and 48%, respectively, to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022). Given that CO_2 is the most relevant GHG in Google's emission profile and the company operates in a sector with readily accessible decarbonisation options, we consider that Google's target for 2030 should meet at least the global benchmark of a 48% reduction below 2019 levels.



Integrity assessment for long-term target(s) (post-2030)

What do the long-term targets actually mean?

What are the targets for the long-term beyond 2030? Google does not commit to any long-term target beyond 2030.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? Google does not commit to any emission reduction targets after 2030.

Is this emission reduction commitment in line with 1.5°C compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside Google's medium-term 'net-zero' emissions target by 2030—accompanied by an emission reduction of only 37% by 2030 compared to 2019—as highly insufficient. To stand a reasonable chance of limiting global warming to 1.5°C, deep and credible emission reductions in all economic sectors towards mid-century are necessary (IPCC, 2022). A global 1.5°C-aligned pathway across all sectors would require a 100% reduction in CO_2 emissions from 2010 levels by 2050 (2045–2055 interquartile range) and an 84% (73–98%) reduction below 2019 levels of all GHGs by 2050 (IPCC, 2018, 2022). The absence of any long-term reduction commitment in Google's climate strategy does not reflect a sense of urgency to decarbonise the company's business.

H&M Group



Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term? How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

H&M Group commits to reducing GHG emissions across the entire value chain by 56% by 2030, compared to 2019.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

Although H&M Group commits to several targets related to sustainable materials and renewable energy for 2030, the company does not commit to earlier GHG emission reduction targets within a five-year time horizon that entail short-term accountability.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

H&M Group's GHG emission reduction target for 2030 meets global cross-sectoral and sector-specific 1.5°C Paris Agreementaligned benchmarks. Given that emissions in the fashion industry occur in various sectors, including agriculture and energy, we consider this target aligned with global benchmarks that require GHG and CO₂ emissions to reduce by 43% and 48% by 2030, respectively (IPCC, 2022).

Teske (2022, pp. 322; 327) considers that between 2019 and 2030, the textile and leather industry and the manufactured fibres and synthetic rubber industry should reduce their GHG emissions in absolute terms by 41% and 46%, respectively. This covers all emissions associated with producing fabrics and other materials and manufacturing the clothes. To be in line with these sectoral benchmarks, H&M Group's target for upstream scope 3 emissions should be set at a level of at least 41%. The company's target goes beyond this reduction level.

We evaluate H&M Group's 2030 target to have 'moderate' integrity rather than 'high' because of the lack of a corresponding short-term target within a five-year timeframe to substantiate it. Specific short-term interim targets requiring immediate action and accountability are of primary importance for credible corporate commitments to fight climate change (UN HLEG, 2022).

High

Integrity assessment for long-term target(s)(post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

H&M Group commits to an emissions reduction target of **90% by 2040 below 2019 levels** across the entire value chain, alongside its net-zero pledge.

Is this emission reduction commitment in line with 1.5°C compatible trajectories or benchmarks for the sector?

H&M Group's 2040 emission reduction target meets global cross-sectoral and sector-specific 1.5°C Paris Agreement-aligned benchmarks. Given that emissions in the fashion industry occur in various sectors, including agriculture and energy, we consider this target aligned with global benchmarks that require a deep emission reduction of 84% reduction below 2019 levels of all GHG by 2050 (IPCC, 2022). The target also meets benchmarks for fashion retailers' upstream scope 3 emissions. According to Teske (2022, pp. 322; 327), emissions in the manufactured fibres and synthetic rubber industry need to reduce by 76% and emissions in the textile and leather sector by 74% by 2040, below 2019 levels. These emissions form part of H&M Group's upstream scope 3 emissions.

Holcim

Moderate

Integrity assessment for short- and medium-term target(s) towards 2030

${f i}$ What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

Holcim commits to emissions intensity targets by 2030 for all its reported emissions:

- Scope 1: reduce the carbon intensity of cement (expressed in kgCO₂/tonne cement) to 520 kgCO₂/tonne of cement by 2025.
- Scope 1: reduce the carbon intensity of cement (expressed in kgCO₂/tonne cement) by 22.5% below 2018 levels by 2030.
- Scope 2: reduce the carbon intensity of cement (expressed in kgCO₂/tonne cement) by 65% below 2018 levels by 2030.
- Scope 3 purchased goods: reduce the carbon intensity of clinker and cement bought (expressed in kgCO₂/tonne of clinker and cement bought) by 20% below 2020 levels by 2030.
- Scope 3 fuel and energy: reduce carbon intensity of bought fuel (expressed in kgCO₂/tonne of fuel bought) by 20% below 2020 levels by 2030.
- Scope 3 downstream transport and distribution: reduce carbon intensity of sold products (expressed in kgCO₂/tonne of product sold) by 24% below 2020 levels by 2030.

How does these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

We cannot quantify Holcim's intensity targets in terms of absolute emission reductions below 2019 levels due to a lack of publicly available information.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)? Holcim sets a short-term target within a five-year timeframe that uses the same metric as its 2030 medium-term targets for scope 1 emissions.

😑 Do the short- and medium-term targets align with a 1.5°C trajectory for the sector according to available literature?

Holcim's 2030 targets come close to meet 1.5°C Paris Agreement-aligned decarbonisation milestones for the cement industry identified in existing literature. For 2030, existing 1.5°C-aligned benchmarks for scope 1 and 2 emissions identify a range of 360–463 kgCO₂/tonne cement (CAT, 2020b; Boehm *et al.*, 2022; SBTi, 2022a, 2022d). Holcim's 2030 target to reach an emissions intensity of 454 kgCO₂/tonne cement across its scope 1 and 2 emissions meets the benchmarks at the upper end (SBTi, 2022a, 2022d), but clearly misses those at the lower end of the range (CAT, 2020b; Boehm *et al.*, 2022).

Holcim sets a separate 2025 target targeting its scope 1 emissions (520 kgCO_2 /tonne) within a five-year timeframe that use the same metric as its 2030 target (441 kgCO_2 /tonne). The TPI found that both targets are compatible with its 'below 2°C' benchmarks, falling short of their ' 1.5° C' benchmarks (Dietz, Hastreiter and Jahn, 2021, p. 9; TPI, 2022c).

We cannot independently compare Holcim's range of scope 3 intensity reduction targets to 1.5°C-aligned sectoral benchmarks as literature provides few specific milestones for the cement industry's scope 3 emissions.

Moderate

Integrity assessment for long-term target(s)(post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long-term beyond 2030?

Holcim commits to the following emission reduction targets alongside its pledge to reach **net zero by 2050** across its entire value chain emissions:

- Scope 1: reduce the carbon intensity of cement (expressed in kgCO₂/tonne cement) by 95% below 2018 levels by 2050
- Scope 2: reduce the carbon intensity of cement (expressed in kgCO₂/tonne cement) by 95% below 2018 levels by 2050
- Scope 3: reduce emissions by 90% below 2020 levels by 2050.

How does these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Holcim substantiates its net-zero pledge by committing to an absolute reduction of scope 3 emissions of 90% by 2050 below 2019 levels; and to reduce the carbon intensity of scope 1 and 2 emissions by 95%. Given that this is an intensity target – and that the target comprises not only emission reductions but through claims to have neutralised emissions through CCUS – it is not clear what the absolute emission reduction impact of these targets would be. This uncertainty makes the appropriateness of the net zero terminology contentious, since we interpret that it is unlikely that the targets would lead to at least 90% emission reductions, excluding measures to neutralise emissions.

We cannot quantify Holcim's scope 1 and 2 *intensity* targets in terms of *absolute* emission reductions along Holcim's value chain compared to 2019 levels due to a lack of publicly available information. Holcim's scope 3 target is equivalent to an absolute reduction of 26.2 MtCO₂e. We cannot estimate what share of Holcim's 2019 emissions this represents as the company updated their scope 3 emissions accounting methodology in 2020 without providing updated emission disclosures for previous years, leading to substantially higher year-on-year emissions in 2020 compared to previous years (Holcim, 2021, p. 16).

Is the emission reduction commitment as part of the long-term vision in line with 1.5°C compatible trajectories or benchmarks for the sector?

Holcim's 2050 intensity reduction targets for scope 1 and 2 emissions meet 1.5°C-aligned milestones for cement producers. For scope 1 and 2 emissions, existing literature identifies a 1.5°C-compatible benchmarks range of 30-90 kgCO₂/tonne cement (CAT, 2020b; SBTi, 2021a, 2021c, 2022c; Boehm *et al.*, 2022). Holcim's intensity reduction target of 30.7 kgCO₂/tonne cement for scope 1 and 2 emissions meets the most ambitious end of this benchmark range. The TPI also evaluates the company's intensity reduction targets for scope 1 emissions as compatible with their interpretation of a 1.5°C benchmark (Dietz, Hastreiter and Jahn, 2021, p. 9; TPI, 2022c).

Holcim's absolute 90% emission reduction target for scope 3 emissions by 2050, below 2020 levels, comes close to aligning with the range of available sectoral benchmarks. For scope 3 emissions, available literature suggests an absolute emissions reduction of more than 90% (SBTi, 2021c, p. 27) and 100% (Teske, 2022, p. 323) of all scope 3 emissions by 2050, compared to a 2019 or 2020 base year respectively.

Further assumptions for aggregate analysis in Section A

For our aggregate analysis across all 24 companies in Section A of this report, we make the simplified most optimistic assumption that companies' intensity targets equal absolute emission reduction targets. For this purpose, we interpret Holcim's multiple intensity targets for 2030 to be absolute targets assuming constant activity levels. Under this assumption, these targets jointly translate into a 11% absolute reduction by 2030 below 2019 levels across the entire value chain. We do not use this estimate for the company-specific integrity assessment of Holcim in Section B.

Inditex

() Poor

Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Inditex has set out two interim GHG emissions targets: first, to reduce scope 1 and 2 emissions by 90% below 2018 levels by 2030, and second, to reduce scope 3 emissions from purchased goods and services by 20% below 2018 levels by 2030.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

These targets jointly represent a reduction of **10–12% by 2030 across the entire value chain** below 2019 levels. To calculate this value, we have assumed that the distribution of scope 3 emissions between different sub-categories in 2018 and 2019 have remained constant. This is because Inditex does not provide detailed information on its full scope 3 emissions for 2018, which is the baseline year for its SBTi target. The range provided comes from the existing uncertainty regarding how Inditex plans to reduce its scope 2 emissions. If the company would reduce its scope 2 emissions largely or exclusively through the purchase of low-quality renewable energy certificates, we would not consider those emissions reduced. This would mean the total targeted reduction would amount to 10% of Inditex's total emissions. If instead it would reduce its emissions largely or exclusively through energy efficiency measures that reduce electricity demand alongside higher quality renewable energy procurement constructs, the emission reductions would represent 12% of the total value chain emissions in 2019.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

Inditex does not commit to earlier interim emission reduction targets within a five-year time horizon that require immediate action. The company has a 2022 renewable energy target which it is on track to meet, although it does not provide details as to the procurement constructs it uses to reach its target (Inditex, 2022a).

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

Inditex's reduction commitment for reductions across the value chain does not meet cross-sectoral and sector-specific 1.5°C-aligned milestones identified in existing literature. According to the IPCC's global economy-wide benchmarks to keep warming below 1.5°C, GHG emissions should reduce by 43% by 2030, compared to 2019 (IPCC, 2022). Given that emissions in the fashion industry occur in various sectors, including agriculture and energy, we expect the industry to decarbonise at the same speed as this global trajectory. Inditex's targets falls short of this global benchmark.

The company's 2030 target for scope 1 and 2 emissions meets sectoral benchmarks. The SBTi guidance for the apparel and footwear sector requires companies to commit to annual reductions of at least 2.5% to comply with the 'well below 2°C' benchmark and 4.2% to comply with the SBTi's '1.5°C' benchmark (SBTi, 2018a, pp. 22; 27). These reductions are based on global trajectories consistent with limiting global warming to well below 2°C and 1.5°C, respectively. Inditex meets the SBTi's 1.5°C benchmark for its target for scopes 1 and 2, which represent less than 3% of Inditex's total emissions in 2019.

Inditex's 2030 target for scope 3 emissions is not aligned with SBTi's 'well below 2°C' and '1.5°C' benchmarks for the apparel sector. The SBTi guidance requires companies in the apparel industry to commit to annual reductions of at least 2.5% to comply with the 'well below 2°C' benchmark and 4.2% to comply with the SBTi's '1.5°C' benchmark (SBTi, 2018a). Assuming a constant emissions reduction rate from the base year to the target year, Inditex's target for scope 3 translates to a 1.7% annual reduction.

The company's scope 3 2030 target is also not aligned with other 1.5°C-compatible sectoral benchmarks. Teske (2022, pp. 322; 327) considers that between 2019 and 2030, the *textile and leather industry* and the *manufactured fibres and synthetic rubber industry* should reduce their GHG emissions by 41% and 46%, respectively. This covers all emissions associated with producing fabrics and other materials and manufacturing the clothes. To be in line with these sectoral benchmarks, Inditex's target for upstream scope 3 emissions should be set at a level of at least 41%. However, its target for upstream and downstream scope 3 emissions represents a 10% reduction below 2019. Scope 3 emissions represented 97% of Inditex's total emissions in 2021.
i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

Inditex's headline pledge includes a commitment to reach net zero GHG emissions by 2040.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Inditex neither commits to a deep emissions reduction target alongside its 2050 net-zero pledge nor specifies emission scopes covered. Since the company's net-zero pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Guidelines for Net Zero and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net -zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which Inditex does not commit to.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside Inditex's net-zero pledge as highly insufficient considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

JBS



Integrity assessment for short- and medium-term target(s) towards 2030

i)What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

JBS has a target to reduce its scope 1 and 2 emission intensity by 30% by 2030 from a 2019 baseline.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? We cannot quantify JBS's intensity target in terms of absolute emission reductions below 2019 levels due to a lack of publicly available information.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? JBS does not commit to earlier interim targets within a five-year time horizon.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

JBS's 2030 targets are not aligned with any of the few 1.5°C Paris Agreement-aligned milestones for the livestock sector available in the literature.

Agricultural emissions, including those from enteric fermentation and manure management, need to reduce significantly before 2030 (Boehm *et al.*, 2022, p. 120). Teske (2022, p. 328) identifies 1.5°C-aligned absolute emission reduction milestones for various emission sources of agricultural activities. Emissions from fuel use on farms, heat used for food processing and -packaging need to reduce by 48%, emissions from purchased electricity on a farm-level or used for food processing and -packaging need to reduce 67%, and emissions AFOLU, non-CO₂ GHGs need to reduce by 34% by 2030. In sum, these required reductions mean a reduction of 38% across all scopes, below 2019 levels. We did not identify any information indicating that JBS's emission intensity target aligns with the latter.

We did not identify any targets for a reduction of scope 3 emissions in the short- and medium-term in JBS' public communications.



Integrity assessment for long-term target(s) (post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? JBS pledges to have net-zero emissions by 2040.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

JBS does not commit to a deep emissions reduction target alongside its 2040 net-zero pledge. The company only makes a vague statement that it intends to offset residual emissions in the target year to achieve the target. Since the company's net-zero pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. This position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which JBS does not commit to.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside JBS's net-zero pledge as highly insufficient considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

Further assumptions for aggregate analysis in Section A

For our aggregate analysis across all 24 companies in Section A of this report, we make the simplified most optimistic assumption that companies' *intensity* targets equal *absolute* emission reduction targets. For this purpose, we interpret JBS's *intensity* target (50% by 2030 below 2019 levels for scope 1 and 2) to be an *absolute* target assuming constant activity levels. Under this assumption, the target translates to a 3% reduction by 2030 below 2019 levels across the entire value chain. We do not use this estimate for the company-specific integrity assessment of JBS in Section B.

Maersk

Moderate

Integrity assessment for short- and medium-term target(s) towards 2030

${f i}$ What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Maersk committed to the following targets for 2030 (Maersk, 2022f, p. 12):

- Ocean: reduce the intensity of its ocean activities by 50% per transported container, compared to 2020.
- Ocean: transport 25% of its ocean cargo by green fuels.
- Air: transport minimum 30% of cargo with SAF.
- Logistics facilities: minimum 90% of green operations ('green operations' are defined as operated on renewable energy and/or green fuels).
- Landside: minimum 20% of moves of customers' cargo on low/zero emissions technology.
- Terminals: reduce scope 1 and 2 emissions of own terminals by about 70%, compared to 2020.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Maersk's intensity target for ocean activities is in line with sectoral intensity benchmarks. However, depending on the increase of the company's activities in the next decade, it is possible that the company's absolute emissions could increase towards 2030 if growth in activity outpaces reductions in emissions.

Maersk's target for emission reductions at its terminals covers about 0.03% of 2020 emissions and will not lead to substantial reductions across the entire value chain. The company's commitments to use lower- and zero-carbon fuels for 25% of its ocean activities and SAF for 30% of air cargo by 2030 could substantially reduce the company's emissions footprint, but only if they are not outpaced by a substantial increase in the total amount of fuel used.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? Maersk does not commit to earlier interim targets within a five-year time horizon.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

Maersk's 2030 interim targets meet some 1.5°C Paris Agreement-aligned milestones for the shipping sector. The company's emissions intensity goal for ocean activities by 2030 is in line with TPI's 1.5°C-aligned intensity benchmarks for the sector (Dietz, Byrne, Hastreiter *et al.*, 2021), but meeting the Paris Agreement's temperature goal requires absolute emission reductions . The CAT shows that a decrease of 51% would be necessary between 2019 and 2030; the SBTi guidance for the maritime sector indicates a necessary emission reduction of 36% below 2020 levels; IRENA's decarbonisation pathway provides that CO_2 emissions from the sector are to decrease by about 25% between 2019 and 2030 to be aligned with a 1.5°C trajectory; and the IEA Net Zero report shows a decrease in CO_2 emissions of 20% (IEA, 2021b, p. 199; IRENA, 2021, p. 81; CAT, 2022b; SBTi, 2022e, p. 11).

Maersk committed to transport 25% of its ocean cargo by green fuels by 2030. These are fuels with 'low or very low greenhouse gas emissions on a lifecycle basis' (Maersk, 2022b). This likely goes beyond benchmarks showing that 5-17% of maritime fuels need to be zero-emission fuels by 2030 for the shipping sector to be aligned with 1.5°C-aligned pathways (IEA, 2021a, p. 138; Smith *et al.*, 2021, p. 11; UNFCCC, 2021, p. 15; Boehm *et al.*, 2022, p. 74). Teske (2022, p. 212) provides that renewable and synthetic fuels needs to increase rapidly and reach a 30% share in fuel use by 2030. We cannot fully evaluate how Maersk's target aligns with these benchmarks given the little information disclosed by Maersk on the fuels' lifecycle emission calculations and the share of bio-based fuels vis-à-vis e-fuels.

(i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? Maersk commits to **net-zero emissions by 2040**.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? Alongside its net-zero pledge, Maersk commits to reduce emissions across scopes 1, 2 and 3 by 90% below 2020 levels. This equals a 91% reduction from 2019 emission levels.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

A commitment to reduce emissions by 90% below 2020 levels by 2040 is in line with Paris Agreement-aligned milestones for the shipping sector. Various studies show that the sector should be fully decarbonised between 2040 and 2050 with no room for offsetting unabated emissions (IRENA, 2021, p. 81; Smith *et al.*, 2021, p. 106; SBTi, 2022e, p. 11; Teske, 2022, p. 213).

Further assumptions for aggregate analysis in Section A

For our aggregate analysis across all 24 companies in Section A of this report, we make the simplified most optimistic assumption that companies' *intensity* targets equal *absolute* emission reduction targets. For this purpose, we interpret Maersk's *intensity* target for ocean activities (50% per transported container below 2020 levels) to be an *absolute* target assuming constant activity levels. Under this assumption, the target translates to a maximum 50% reduction by 2030 below 2019 levels across the entire value chain. We did not account for emissions related to other transport modes due to limited data availability. We do not use this estimate for the company-specific integrity assessment of Maersk in Section B.

Mercedes-Benz



Integrity assessment for short- and medium-term target(s) towards 2030

${f i}ig)$ What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Mercedes-Benz commits to the following two emission reduction targets by 2030, both below 2018 levels:

- >40% CO₂ emissions intensity reduction per kilometre for new vehicles ('well-to-wheel' use phase, scope 3 category 11)
- 50% absolute emissions reduction across scope 1 and scope 2 emissions

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

We are unable to quantify the extent to which the interim targets reduce emissions across the value chain compared to a **2019 baseline**, due to a lack of emissions reporting for Mercedes-Benz before **2021**. The company provides emissions data for the year 2019, this includes emissions from Daimler Truck AG, which spun off from Mercedes-Benz Group AG in December 2021.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)?

Mercedes-Benz does not commit to earlier emission reduction targets within a five-year time horizon.

Do the short- and medium-term targets align with a 1.5°C trajectory for the sector according to available literature?

Mercedes-Benz's range of 2030 interim targets do not meet 1.5°C Paris Agreement-aligned milestones for automobile

manufacturers' downstream scope 3 emissions (SBTi, 2018b, 2018c; CAT, 2020b; Dietz *et al.*, 2020; IEA, 2021b; UNFCCC, 2021; Boehm *et al.*, 2022; Teske *et al.*, 2022). Sixty to ninety-five percent of all light-duty vehicle (LDVs) sales globally should be electric that is, have zero tailpipe emissions—by 2030 to stay below the Paris Agreement's temperature limit of 1.5°C (CAT, 2020b; IEA, 2021b). Ninety-five to one-hundred percent of LDV sales should be electric in Mercedes-Benz's main markets, such as the European Union, China and the US (CAT, 2020b; Teske *et al.*, 2022), and reach 100% by 2035 in all leading markets (IEA, 2021b, pp. 20; 138; UNFCCC, 2021, pp. 10–11). Mercede s-Benz would not meet these benchmarks by 2030, as it presently provides no market-specific phase-out dates for internal combustion engines. In this context, German NGO Deutsche Umwelthilfe filled a court action against Mercedes-Benz in 2021 to enforce a legally binding commitment to stop producing combustion-engine cars by November 2030 (Metz, Müller-Kraenner and Resch, 2021; Setzer and Higham, 2022; Waldersee, 2022).

Mercedes-Benz instead still refers to its SBTi 'well-below 2°C' verification of its scope 3 emissions intensity target (Mercedes-Benz Group, 2022e, p. 130, 2022a, p. 23). SBTi has indefinitely paused the use of its methodology for automakers, as the initiative states that the methodology does not reflect a 1.5°C-compatible definition (SBTi, 2022f).



Integrity assessment for long-term target(s)(post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

The Mercedes-Benz Group commits to CO₂-neutral new vehicle fleet along all stages of the value chain by 2039.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

The Mercedes-Benz Group does **not commit to a deep emission reduction target** alongside its carbon neutrality pledge. Since the company's carbon neutrality pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Guidelines for Net Zero and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Carbon neutrality targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which the Mercedes-Benz Group does not commit to.

Is the emission reduction commitment as part of the long-term vision in line with 1.5°C compatible trajectories or benchmarks for the sector?

The Mercedes-Benz Group does not explain why it considers the 2039 carbon neutrality target aligned with 1.5°C-aligned decarbonisation milestones for the automobile sector. The automobile industry should sell only electric vehicles by 2030–2035 globally to comply with 1.5°C-compatible decarbonisation milestones (CAT, 2020b; IEA, 2021b; UNFCCC, 2021; Boehm *et al.*, 2022; Teske *et al.*, 2022). Alongside its headline pledge, Mercedes-Benz aims to sell only electric vehicles by the end of the decade 'wherever market conditions allow' (Mercedes-Benz Group, 2022e, p. 130, 2022a, p. 23). The company, however, stops short of committing to specific phase-out dates for internal combustion engines in key markets such as the European Union, the US, or China. Accordingly, if Mercedes-Benz considers market conditions unsuitable for a phase-out of internal combustion engines, the company could potentially continue to sell combustion-engine vehicles in key markets way beyond 2030.

Further assumptions for aggregate analysis in Section A

For our aggregate analysis across all 24 companies in Section A of this report, we make the simplified most optimistic assumption that companies' *intensity* targets equal *absolute* emission reduction targets. For this purpose, we interpret Mercedes-Benz's *intensity* targets for 2030 (at least 40% by 2030 below 2018 for downstream scope 3 use phase emissions) to be an *absolute* target assuming constant activity levels. Together with Mercedes-Benz's absolute target for scope 1 and 2 (50% by 2030 below 2021), these targets jointly translate into a 32% absolute reduction by 2030 below 2021 levels across the entire value chain. We do not use this estimate for the company-specific integrity assessment of Mercedes-Benz in Section B.

Microsoft



Integrity assessment for short- and medium-term target(s) towards 2030

) What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

Microsoft commits to the following three short-and medium-term targets:

- 'Staying carbon neutral' each year since 2012
- 24/7 renewable electricity across all operations sourced by 2030
- Becoming 'carbon negative' across its entire value chain by 2030

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Microsoft's 'carbon negative' target for 2030 entails a commitment to reduce **50% of its market-based emissions, by 2030** across the entire value chain, compared to 2020 levels. This translates into a location-based **emission reduction of 38% by 2030** across the entire value chain, compared to 2019 levels. Since this reduction target alongside the 'carbon negative' pledge falls well short of a commitment to deep emission reductions (i.e., at least 90% below 2019 levels), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). 'Carbon negative' targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which the company does not commit to.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

Microsoft does not commit to earlier interim targets within a five-year time horizon. Microsoft's pledge to be 'carbon neutral' each year is not a target but rather a neutralisation claim that is not related to its 'carbon negative' by 2030 target.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We are unable to compare Microsoft's 2030 emission reduction target to sectoral 1.5°C Paris Agreement-aligned benchmarks as existing literature provides few specific milestones for the technology service industry. Microsoft's emission reduction targets for scope 1 and scope 2 market-based emissions, both of which are verified by SBTi as aligned with its interpretation of the 1.5°C temperature limit (SBTi, 2023a), only cover around 3% of Microsoft's total emissions in 2021. The main source of scope 1 and 2 emissions for information and communications technology (ICT) companies stems from consumed electricity (SBTi, 2020, p. 15). We could not identify any sectoral 1.5°C-aligned benchmark in the literature for the largest share of Microsoft's emissions, namely scope 3 emissions, which account for 75% of its 2021 location-based emissions. Due to this gap in existing literature, we are not able to assess Microsoft's decarbonisation efforts within its value chain against 1.5°C-aligned sectoral benchmarks.

Microsoft's emission reduction commitment does not comply with cross-sectoral 1.5°C Paris Agreement-aligned emission

trajectories. Microsoft's commitment to reduce 38% of its location-based emissions across its entire value chain by 2030 falls short of the need to reduce global GHG and CO_2 emissions by 43% and 48%, respectively, to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022). Given that CO_2 is the most relevant GHG in Microsoft's emission profile and the company operates in a sector with readily accessible decarbonisation options, we consider that Microsoft's target for 2030 should meet at least the global benchmark of a 48% reduction below 2019 levels.



Integrity assessment for long-term target(s) (post-2030)

(i) What do the long-term targets actually mean?

What are the targets for the long-term beyond 2030?

Microsoft commits to remove all direct emissions or those from consumed electricity since its foundation in 1975 by 2050 (Microsoft, 2022a, p. 104).

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? Microsoft does not commit to any further emission reductions target alongside its long-term pledge.

Is this emission reduction commitment in line with 1.5°C compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside Microsoft's medium-term 'carbon negative' by 2030 target—the latter accompanied by an emission reduction of only 38% by 2030 compared to 2019—as highly insufficient. To stand a reasonable chance of limiting global warming to 1.5°C, deep and credible emission reductions in all economic sectors towards mid-century are necessary (IPCC, 2022). A global 1.5°C-aligned pathway across all sectors would require a deep emission reduction of 100% CO₂ from 2010 levels by 2050 (2045–2055 interquartile range) and an 84% (73–98%) reduction below 2019 levels of all GHG by 2050 (IPCC, 2018, 2022). The absence of a long-term reduction commitment in Microsoft's climate strategy does not reflect a sense of urgency to decarbonise the company's business.

Nestlé

) Poor

Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Nestlé pledges to reduce its emissions by 20% by 2025 and 50% by 2030, compared to a 2018 baseline.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? We interpret that the pledge to reduce emissions by 50% by 2030 translates to only 16–21% emission reductions compared to the company's full value chain emissions in 2019.

In its *Net Zero Roadmap* (Nestlé, 2021b), Nestlé presents its interim emission reduction targets compared to a business-as-usual scenario and shows the targeted emission levels for each emission source for 2030. We calculate from the figures presented in the company's *Net Zero Roadmap* (Nestlé, 2021b) that the company actually commits to reduce its full emission footprint from 116 MtCO₂e in 2019 to between 91-97 MtCO₂e in 2030. This would represent a reduction of just 16-21% of the company's full value chain emissions. The stark difference between this range and the 50% target communicated by Nestlé lies in the company's exclusion of various emission sources from its pledge, as well as the inclusion of measures for biological and non-permanent carbon removal and storage which Nestlé accounts as negative emissions to claim that its real emissions are cancelled out.

The range of 16–21% is due to different potential interpretations of the poorly defined measures to 'transform the product portfolio', through which Nestlé states that it plans to 'reduce future emissions growth' by 6 MtCO₂e (Nestlé, 2021b, p. 21). The upper end of the 16–21% range is based on the optimistic assumption that the 6 MtCO₂e refers to further emission reductions across a number of emission sources and compared to 2019 levels. However, the ambiguous presentation of this measure in the *Net Zero Roadmap* could also refer to a 6 MtCO₂e reduction compared to unspecified future emissions growth, which may not lead to any further emission reductions compared to 2019 levels.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? Nestlé sets a target within a five-year timeframe that uses the same metric as its 2030 medium-term target.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

Nestlé's 2030 medium-term targets neither meet cross-sectoral nor sector-specific 1.5°C Paris Agreement-aligned

decarbonisation milestones. The majority of Nestlé's emission footprint derives from agricultural emissions. In the absence of available benchmarks from the scientific literature for food and beverage companies specifically, we compare Nestlé's 16–21% emission reductions by 2030 to available 1.5°C-aligned benchmarks for agriculture, and cross-sector global benchmarks. Global cross-sectoral benchmarks require GHG and CO₂ emissions to reduce by 43% and 48% between 2019 and 2030, respectively (IPCC, 2022). Pathways for global agriculture and food sector in Teske (2022, p. 328) indicate that scope 3 emissions should reduce by at least 34% between 2019 and 2030. Although the 50% reduction communicated by Nestlé would appear to align the company with these benchmarks, our interpretation Nestlé's target translates just a 16-21% reduction of the full value chain emissions in 2019 means that Nestlé's plans fall far short of any of these benchmarks.



Integrity assessment for long-term target(s) (post-2030)

What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? Nestlé pledges to achieve net zero emissions by 2050 (Nestlé, 2021b, p. 3).

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Nestlé does not commit to an emissions reduction target alongside its 2050 net zero pledge. Since the company's net zero pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Guidelines for Net Zero and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which Nestlé does not commit to.

ls this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside Nestlé's net-zero pledge as highly insufficient considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

PepsiCo

? Unclear

Integrity assessment for short- and medium-term target(s) towards 2030

${f i}ig)$ What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

PepsiCo has the following medium-term targets:

- To reduce scope 1 and 2 emissions by 75%, by 2030, compared to a 2015 baseline.
- To reduce scope 3 emissions by 40%, by 2030, compared to a 2015 baseline.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

PepsiCo explicitly states that its targets do not depend on carbon offsets, but its Climate Action Strategy includes 'insetting' to realise them. The targets translate to emission reductions of **maximum 48% by 2030 across the entire value chain**, compared to a 2019 baseline. However, it remains unclear what share of the targets would be realised through offsets under the term 'insetting'. **Emission reductions will be lower than 48% if neutralisation is claimed through 'insetting'**.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? PepsiCo does not commit to earlier interim targets within a five-year time horizon.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We are unable to compare PepsiCo's medium-term targets to sectoral 1.5°C Paris Agreement-aligned benchmarks as it remains unclear what share of the targets will be met without offsetting practices such as 'insetting'. Global cross sector benchmarks require GHG and CO₂ emissions to reduce by 43% and 48% between 2019 and 2030, respectively (IPCC, 2022). Pathways for global agriculture and food sector in Teske (2022, p. 328) indicate that scope 3 emissions should reduce by at least 34% between 2019 and 2030. Without specifying what share of the target will be met through reducing emissions and what share will depend on offsets (termed 'insetting'), we cannot compare PepsiCo's targets to these benchmarks.

Integrity assessment for long-term target(s)(post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

PepsiCo commits to net-zero emissions across its value chain by 2040.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

PepsiCo does not commit to a deep emissions reduction target alongside its 2040 net zero pledge. Since the company's net zero pledge does not entail any commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which PepsiCo does not commit to.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside PepsiCo's net zero pledge as highly insufficient considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

Samsung Electronics



Integrity assessment for short- and medium-term target(s) towards 2030

) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Samsung commits to a net-zero carbon emissions target by 2030 that covers scopes 1 and 2 of the Device Experience (DX) Division. The company does not specify an emissions reduction target alongside its net-zero pledge.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Samsung's net-zero target for its DX Division equates to an emissions reduction target of only 2% across the full value chain by 2030, under the assumption that the company will reduce scope 1 and 2 emissions from its DX Division by 100% by the target year. Since Samsung's net-zero pledge does not entail a commitment to deep decarbonisation (i.e., reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which the company does not commit to.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? Samsung does not commit to earlier interim targets within a five-year time horizon.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

We consider that Samsung's net-zero pledge, which entails a reduction of only 2% of emissions across its full value chain, is highly insufficient considering the need for deep and credible emission reductions towards 2030 to stand a reasonable chance of limiting global warming to 1.5° C (IPCC, 2022).

() Poor

Integrity assessment for long-term target(s) (post-2030)

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

Samsung commits to a net-zero carbon emissions target by 2050 that covers scopes 1 and 2. The company aims to reduce 17 $MtCO_2e$ by 2050, which correspond to its scope 1 and 2 emissions footprint in 2021.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Samsung's 2050 target of 17 MtCO₂e of scope 1 and 2 emission reductions alongside its net-zero pledge equates to **20% emission reduction by 2030 along the entire value chain**, compared to its 2019 emissions footprint. Since this reduction target alongside the net-zero pledge falls well short of a commitment to deep emission reductions (i.e., at least 90% below 2019 levels), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Net Zero Guidelines and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Net-zero targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which the company does not commit to.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

Samsung's targeted emission reductions of 20% by 2050, compared to 2019 levels, fall way short from global efforts required to limit global warming to 1.5°C. The IPCC finds that global 1.5°C-aligned pathways across all sectors require a 100% reduction in CO_2 emissions from 2010 levels by 2050 and an 84% reduction below 2019 levels of all GHGs by 2050 (IPCC, 2018, 2022). Samsung's 2050 target falls way short of these global efforts.

We evaluate the integrity of Samsung's emissions reduction target using global economy-wide benchmarks as we did not identify 1.5°C-aligned benchmarks for the electronics industry in existing literature. This gap in existing literature allows no conclusive assessment of sector-specific decarbonisation efforts across the entire value chain in line with the Paris Agreement's 1.5°C temperature, especially for scope 3 emissions in the electronics industry (representing 86% of Samsung's total emissions in 2019).

Stellantis

Moderate

Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Stellantis commits to an overarching emission reduction target of 50% by 2030 across the entire value chain, compared to 2021 levels. The overarching interim target is supported by the following targets, all using a 2021 base year:

- To reduce absolute scope 1 and 2 emissions by 50% by 2025 and by 75% by 2030.
- To reduce scope 3 emission intensity by 50% by 2030.
- To sell 100% BEVs for passenger cars in Europe and 50% BEVs for passenger cars and light-duty trucks in the US by 2030.
- To reduce the emission intensity of purchased parts per EV by 40% by 2030.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Stellantis commits to an overarching interim emission reduction target of 50% by 2030 across the entire value chain, compared to 2021 levels. Due to the company's recent formation, we cannot recalculate the targeted emission reductions compared to a 2019 base year.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)?

Stellantis sets a range of short-term targets within a five-year timeframe by 2025 that use the same metrics as its 2030 medium-term targets.

Do the short- and medium-term targets align with a 1.5°C trajectory for the sector according to available literature?

Stellantis' range of 2030 targets meet some of the 1.5°C Paris Agreement-aligned milestones for automobile manufacturers' downstream scope 3 emissions identified in existing literature (CAT, 2020b; Dietz *et al.*, 2020; IEA, 2021b; UNFCCC, 2021; Boehm *et al.*, 2022; Teske *et al.*, 2022). Sixty to ninety-five percent of all light-duty vehicle (LDVs) sales should be electric—that is, have zero tailpipe emissions—by 2030 globally to stay below the Paris Agreement's warming limit of 1.5°C (CAT, 2020b; IEA, 2021b). Ninety -five to one-hundred percent of LDV sales should be electric in Stellantis' main markets, including the European Union, China and the US by 2030 (CAT, 2020b; Teske *et al.*, 2022), and reach 100% by 2035 in all leading markets (IEA, 2021b, pp. 20; 138; UNFCCC, 2021, pp. 10–11). The company only meets these benchmarks for its target to sell 100% EVs in the EU by 2030, but not for its target to sell 50% in the US. In the latter market, Stellantis aims to reach 100% by 2038.

Apart from the core targets for the EU and US markets, the company has not committed to any further phaseout dates for internal combustion engines in other sales markets or other vehicle categories, such as light commercial vehicles. Stellantis presented aspirational EV sale shares by 2030 for the Middle East and Africa regions, Brazil, India and the Asia Pacific region, and China, as part of its strategic blueprint Dare Forward 2030 (Stellantis, 2022c). These indicative targets miss 1.5°C-compatible sectoral benchmarks for Brazil, India and China (CAT, 2020b). The targeted sale shares are:

- >25% share of LEVs in the Middle East and Africa regions.
- ~20% share of LEVs in Brazil (compared to 45-95% for all LDV sales being electric by 2030 in 1.5°C-aligned scenarios).
- ~50% share of BEVs in India and the Asia Pacific region (compared to 80-95% for all LDV sales being electric by 2030 in 1.5°C-aligned scenarios, including two- and three-wheelers).
- 60% of share of passenger car BEVs in China (compared to 95-100% for all LDV sales being electric by 2030 in 1.5°C-aligned scenarios, including two- and three-wheelers).

Stellantis claims that its 2030 interim targets align with the 'vParis Climate Agreement and [a] 1.5°C scenario, referring to the SBTi's guidance for the transport sector (Stellantis, 2022a, p. 39). However, the SBTi only provides benchmarks for downstream scope 3 emissions from the use of sold products as part of their interpretation of a 'well below 2°C' scenario as of February 2023 (SBTi, 2018b, 2018c), but it does not provide similar benchmarks for a 1.5°C scenario. SBTi has indefinitely paused the use of its methodology for automakers, as the initiative states that it does not reflect a 1.5°C-compatible definition (SBTi, 2022f).

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030?

Stellantis' headline pledge of **carbon net zero by 2038** includes a commitment to reduce emission by at least 90% across its entire value chain, compared to 2021 levels. Stellantis will offset less than 10% of its 2021 emissions by 2038. Due to the company's recent formation, we cannot recalculate the targeted emission reductions to a 2019 base year.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Stellantis commits to an emissions reduction target of at least 90% by 2038 below 2021 levels across the entire value chain alongside its carbon net-zero pledge. Since the net-zero pledge entails a commitment to deep decarbonisation across the entire value chain, we consider that the net-zero terminology is unlikely to be misleading.

Is the emission reduction commitment as part of the long-term vision in line with 1.5°C compatible trajectories or benchmarks for the sector?

Stellantis' 2038 targets meet some of the 1.5°C Paris Agreement-aligned milestones for automobile manufactures. The automobile industry should only sell electric vehicles by 2030–2035 globally to comply with 1.5°C-compatible decarbonisation milestones (CAT, 2020b; IEA, 2021b; UNFCCC, 2021; Boehm *et al.*, 2022; Teske *et al.*, 2022). Stellantis does not explicitly commit to this specific benchmark for its entire vehicle fleet sold globally by 2040, only for certain markets, such as the EU by 2030 and the US by 2038, responsible for 75% of all sales in 2021 (Stellantis, 2022b, pp. 29–39). In this context, Stellantis also announced its intention to 'refine emission reduction trajectories aligned with 1.5°C scenario per region' throughout 2022 (Stellantis, 2022a, p. 37).

Thyssenkrupp

O Moderate

Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short- to medium term?

Thyssenkrupp presents two targets for 2030: a 30% absolute reduction of its scope 1 and 2 emissions below 2018 levels, and a 16% absolute reduction of scope 3 emissions below 2017 levels.

Thyssenkrupp also sets out a target for the fiscal year 2022/23: to reduce its CO_2 emissions by 1 t CO_2 / EUR million in sales, excluding all steel-related activities.

How does these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? The two 2030 targets translate to a **12% emissions reduction by 2030** below 2019 emissions across the entire value chain.

To calculate what Thyssenkrupp's scope 3 target means compared to 2019 whole value chain emissions, we assumed a retroactive implementation of the company's latest emission reporting boundaries. In 2022, Thyssenkrupp stopped reporting on their processing of sold products and use of sold products emissions. Before this change, Thyssenkrupp's previously communicated scope 3 target only covered use of sold products emissions, which were reported at 780 MtCO₂e in the target's 2017 base year. However, the company's new scope 3 target no longer covers this category and the company no longer reports emissions from it. Because of these reporting boundaries changes, emissions from the latest historical year and previous years are not directly comparable. For this reason, we assumed a similar reporting boundary as the one the company currently uses to estimate the share of 2019 emissions covered by Thyssenkrupp's 2030 targets.

Because the target scope is not clear, we are not able to assess what Thyssenkrupp's 2022/23 target means in terms of own emission reductions.

Do these targets cover both the short-term (within 5 years) and medium-term (up to 2030)?

Thyssenkrupp has set out an annual interim target of reducing its CO_2 emissions by $1 tCO_2$ per EUR million in sales excluding its steel activities. We cannot evaluate whether the target requires any significant action by the company given its unclear scope.

Do the short- and medium-term targets align with a 1.5°C trajectory for the sector according to available literature?

We cannot independently assess Thyssenkrupp's cross-divisional 2030 targets against existing 1.5°C-aligned benchmarks for the steel sector. For scope 1 emissions, Teske (2022, p. 326) finds that a 54% reduction by 2030 below 2019 levels would be necessary to stay below the 1.5° temperature limit, while scope 2 and 3 emissions would need to be reduced by 66% and 38% respectively. Other literature finds that the carbon intensity of global steel production must reach around 1.13–1.35 tCO₂e/tonne steel (CAT, 2020b; Boehm *et al.*, 2021, 2022; Dietz, Gardiner and Scheer, 2022). We cannot independently evaluate the company's existing targets to these steel sector-specific benchmarks due to a lack of information on how Thyssenkrupp's different business divisions contribute to the cross-divisional 2030 targets.

The TPI evaluates Thyssenkrupp's targets in line with their interpretation of ' 1.5° C' benchmark by 2030 (TPI, 2022d). According to TPI, Thyssenkrupp's targets translate to carbon intensity of 1.10 tCO_2 /tonne steel by 2030. TPI provides no further information on how it derives Thyssenkrupp's targeted emission intensity by 2030.

Thyssenkrupp might be sufficiently contributing towards the achievement of other 1.5°C-aligned milestones for the steel

sector. Identified milestones for 1.5°C-compatible pathways in the literature would require that 20 low or near-zero carbon facilities and 70 near-zero emissions primary steel mills become operational by 2030 (UNFCCC, 2021, p. 15; Delasalle *et al.*, 2022, p. 69), and that most new clean technologies in heavy industry have been demonstrated at scale (IEA, 2021b, pp. 20; 129). Thyssenkrupp plans to transition to 100% hydrogen-based steelmaking by 2045 and presents clear plans in terms of developing its own near-zero emissions steelmaking facilities (Thyssenkrupp, 2022b, pp. 60–62).

What do the long-term targets actually mean?

What are the targets for the long-term beyond 2030?

Thyssenkrupp pledges to become carbon neutral across all its business divisions by 2050, including at least a 90% reduction of scope 1 and 2 emissions (2018 baseline) and a 90% reduction of scope 3 emissions (2017 baseline).

The company commits to the same target including the same baselines for its steel division by 2045, alongside a plan to transition to 100% hydrogen-based steelmaking in 2045.

How does these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Thyssenkrupp commits to a 90% reduction of scope 1, 2 emissions (below 2018), and scope 3 emissions (below 2017) by 2050. Compared to the whole value chain emissions that the company discloses for 2019, these targets would mean an 89% reduction. Since the carbon neutrality pledge entails a commitment to deep decarbonisation across the entire value chain, we consider that the terminology is unlikely to be misleading. Carbon neutrality targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which the company has committed itself to do.

Uncertainty remains regarding the targets' scope 3 coverage. Thyssenkrupp only reported downstream scope 3 emissions from product use up to 2020, after which it has not done so further. There is a lack of definitive guidance from any major standard setting initiatives on whether scope 3 category 11 is a relevant emissions category for steelmaking companies. Most emission sources from the use of steel likely fall under the category of indirect product use phase emissions that may be considered an optional reporting scope, as opposed to the direct product use phase emissions which form part of the normal emissions reporting scope under scope 3 category 11. Thyssenkrupp indicates that they will defer to and comply with SBTi guidance on this issue once the SBTi's guidance for steelmaking companies has been published in 2023. Although this uncertainty arises partly due to a gap in the available guidance on this issue, the transparency of the company's communication could be improved by clearly specifying which scope 3 accounting approach the target covers. This would be especially relevant, given that the target base year is 2017, when the company was using a different accounting approach for its scope 3 emissions.

Is the emission reduction commitment as part of the long-term vision in line with 1.5°C compatible trajectories or benchmarks for the sector?

Thyssenkrupp's targets may align with 1.5°C Paris Agreement-aligned milestones for steelmaking, but the sufficiency of the target for other divisions cannot be confirmed due to uncertainties regarding the relevance of downstream scope 3 emissions. The target to reduce emissions from the steel making division by 90% by 2045 appears likely to be in line with available benchmarks from SBTi Net Zero Standard (SBTi, 2021c, pp. 18; 27). Version 1.0 of the SBTi Net Zero Standard specifies a reduction of 91% by 2050 (SBTi, 2021c, pp. 18; 27); the corresponding tool version 1.0.3 indicates a reduction of 93% by 2050 (SBTi, 2022c). Thyssenkrupp's targets are presented alongside clear plans for developing their own near-zero emissions steelmaking facilities, which are planned to reach 100% "climate neutral steel production without coal-based blast furnace" (Thyssenkrupp, 2022b, p. 60). This could be in line with global benchmarks of reducing the carbon intensity of steel production to around zero to 0.13 tCO₂e/tonne steel by 2050 globally (CAT, 2020b; Boehm *et al.*, 2021, 2022; Dietz, Hastreiter and Scheer, 2021). However, uncertainty remains on the relevance of downstream scope 3 emissions. Even if it is determined that scope 3 category 11 emissions are not a relevant emission source for a steelmaking companies, and thus for Thyssenkrupp's steelmaking division, this emission category is likely to be relevant for other company divisions of Thyssenkrupp as a diversified company. Due to this uncertainty and this potentially significant omission, we assess the long-term targets to be of moderate integrity despite the company's targets likely being in line with 1.5°C Paris Agreement-compatible milestones for steelmaking.

Volkswagen Group



Integrity assessment for short- and medium-term target(s) towards 2030

i) What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

Volkswagen Group commits to the following emission reduction targets by 2025 and 2030:

- 30% CO₂ emissions intensity reduction per kilometre for new passenger cars and light duty vehicles by 2025 (below 2015 levels, allowing for offsets to meet the target).
- 30% CO₂ emissions intensity reduction per kilometre for new passenger cars and light duty vehicles by 2030 (below 2018 levels, not allowing for offsets to meet the target).
- 50% absolute emissions reduction by 2030 across scope 1 and scope 2 emissions (below 2018 levels, not allowing for
 offsets to meet the target).

The Volkswagen Group states that these targets represent 'minimum requirements for the Group brands' (Volkswagen, 2022c, pp. 26, 46). Single brands of the Volkswagen Group, for instance Porsche AG and SCANIA AB, set additional targets for this timeframe.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

Volkswagen's 2030 absolute emissions reduction target for scopes 1 and 2 is equivalent to 3% emission reductions by 2030 below 2019 levels across the entire value chain. We cannot independently quantify Volkswagen's interim intensity targets for scope 3 emissions.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)?

The company sets a 2025 intensity reduction target for CO_2 emission of new vehicles. We do not consider this target clearly aligned with its 2030 intensity reduction target as the target for 2025 explicitly relies on an unspecified amount of offsets, while the company will not use offsets towards its 2030 target (Volkswagen, 2022c, pp. 26, 46).

Do the short- and medium-term targets align with a 1.5°C trajectory for the sector according to available literature?

Volkswagen's range of 2030 interim targets do not meet 1.5°C Paris Agreement-aligned milestones for automobile manufacturers' downstream scope 3 emissions identified in existing literature (SBTi, 2018b, 2018c; CAT, 2020b; Dietz *et al.*, 2020; IEA, 2021b; UNFCCC, 2021; Boehm *et al.*, 2022; Teske *et al.*, 2022). Sixty to ninety-five percent of all light-duty vehicle (LDVs) sales should be electric—that is, have zero tailpipe emissions—by 2030 globally to stay below the Paris Agreement's warming limit of 1.5°C (CAT, 2020b; IEA, 2021b). Ninety-five to one-hundred percent of LDV sales should be electric in Volkswagen's main markets such as the European Union, China and the US by 2030 (CAT, 2020b; Teske *et al.*, 2022), and reach 100% by 2035 in all leading markets (IEA, 2021b, pp. 20; 138; UNFCCC, 2021, pp. 10–11).

The Volkswagen Group commits to reach at least a 50% electric vehicle share by 2030 in its three key markets (Volkswagen, 2022c, p. 41). These are the European Union, the US and China, which jointly represent 83% of all vehicles sales in 2021 (Volkswagen, 2022b, p. 109), For the EU, the Volkswagen Group intends to achieve a 60% share, while aiming to reach 70% for its Volkswagen Passenger Cars brand. The company significantly falls short to meet 1.5°C-aligned benchmarks by 2030 as it provides no market-specific phase-out dates for internal combustion engines. In this context, claimants associated with Greenpeace Germany and Fridays for Future Germany filled a court action against Volkswagen Group in 2021 to enforce a legally binding commitment to stop producing combustion-engine cars by the end of 2029 (Kaiser *et al.*, 2021; Setzer and Higham, 2022).

The Volkswagen Group still refers to the SBTi's 'well-below 2°C' verification of its scope 3 emissions intensity target (Volkswagen, 2022c, p. 46). However, SBTi has indefinitely paused the use of its methodology for automakers, as the initiative states that it does not reflect a 1.5°C-compatible definition (SBTi, 2022f). The Transition Pathway Initiative (TPI) considers the company's 2030 targets for downstream scope 3 emissions not aligned with its definition of a 2 Degrees Scenario (TPI, 2022e). The TPI's assessment compares its 2 Degree Scenario benchmark range of 40–77 gCO₂e /vkm to its interpretation of Volkswagen's 2030 target of 97 gCO₂e /vkm.

i) What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? The Volkswagen Group aims to achieve **carbon neutrality by 2050**.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

The Volkswagen Group does **not commit to a deep emissions reduction target** alongside its 2050 carbon neutrality pledge. Since the company's carbon neutrality pledge does not entail any commitment to deep decarbonisation (i.e., a reduction of at least 90% of 2019 emissions across the entire value chain), we consider that the terminology of this target may be misleading. Our position is in line with the ISO Guidelines for Net Zero and the SBTi Net Zero Standard (SBTi, 2021c; ISO, 2022b). Carbon neutrality targets can give consumers and investors the impression that the company aims to reach deep levels of emission reductions, which the Volkswagen Group does not commit to.

Is the emission reduction commitment as part of the long-term vision in line with 1.5°C compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction target alongside Volkswagen Group's carbon neutrality as highly insufficient, considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022). The automobile industry should sell only electric vehicles by 2030–2035 globally to comply with 1.5°C-compatible decarbonisation milestones (CAT, 2020b; IEA, 2021b; UNFCCC, 2021; Boehm *et al.*, 2022; Teske *et al.*, 2022). Downstream emissions from the use of sold vehicles amounted to around 75% of the Volkswagen's emissions in 2021, but Volkswagen has not committed to any phase-out dates for internal combustion engines across its different brands in the context of its carbon neutrality pledge.

Further assumptions for aggregate analysis in Section A

For our aggregate analysis across all 24 companies in Section A of this report, we make the simplified most optimistic assumption that companies' *intensity* targets equal *absolute* emission reduction targets. For this purpose, we interpret Volkswagen's *intensity* targets for 2030 (30% by 2030 below 2018 for downstream scope 3 use phase emissions) to be an *absolute* target assuming constant activity levels. Together with Volkswagen's absolute target for scope 1 and 2 (50% by 2030 below 2018), these targets jointly translate into a 25% absolute reduction by 2030 below 2019 levels across the entire value chain. We do not use this estimate for the company-specific integrity assessment of Volkswagen in Section B.

Walmart

() Poor

Integrity assessment for short- and medium-term target(s) towards 2030

${f i}$ What do the short- and medium-term targets actually mean?

What are the targets for the short to medium term?

- Scope 1 and scope 2: 35% emission reduction by 2025 from 2015.
- Scope 1 and scope 2: 65% emission reduction by 2030 from 2015.

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)? Walmart's short- and medium-targets translate to reducing emissions by 5% by 2030, compared to 2019 value chain emissions.

Do these targets cover both the short term (within 5 years) and medium term (up to 2030)? Walmart sets a target within a five-year timeframe that use the same metric as its 2030 target and long-term target.

Do these emission reduction commitments align with a 1.5°C trajectory for the sector according to available literature?

Walmart's 2030 medium-term targets neither meet cross-sectoral nor sector-specific 1.5°C Paris Agreement-aligned decarbonisation milestones. Walmart's scope 3 emissions account for 91% if its emission footprint. In the absence of available benchmarks from the scientific literature for mixed-good retailers, we compare Walmart's 5% emission reductions by 2030 to available 1.5°C-aligned benchmarks for agriculture, and cross-sector global benchmarks. Global cross-sector benchmarks require GHG and CO₂ emissions to reduce by 43% and 48% between 2019 and 2030, respectively (IPCC, 2022). Pathways for global agriculture and food sector in Teske (2022, p. 328) indicate that scope 3 emissions should reduce by at least 34% between 2019 and 2030. The exclusion of the lion's share of emissions from Walmart's targets means that its overall emission reduction commitment falls far short of any of these benchmarks.

() Poor

Integrity assessment for long-term target(s)(post-2030)

${f i}$ What do the long-term targets actually mean?

What are the targets for the long term beyond 2030? Walmart committed to zero emissions in operations by 2040 (scope 1 and 2).

How do these targets equate to emission reductions across the value chain (compared to a 2019 baseline)?

We estimate that Walmart's emission reduction target is equivalent to a commitment to reduce **around 9% of its emissions across the value chain by 2040**, compared to 2019 levels. Walmart's 2040 target covers only scope 1 and scope 2 emissions, which account for approximately 9% of the company's GHG emission footprint in 2019.

Is this emission reduction commitment in line with 1.5°C-compatible trajectories or benchmarks for the sector?

We consider the lack of any post-2030 emission reduction commitments for scope 3 alongside Walmart's targets for scope 1 and 2 as highly insufficient, considering the need for deep and credible emission reductions towards mid-century to stand a reasonable chance of limiting global warming to 1.5°C (IPCC, 2022).

Annex III Comparison of SBTi verifications with CCRM assessment.

Table 19: Comparison between target verifications by the Science Based Targets initiative (SBTi) as of February 2023 and the integrity assessment as part of the Corporate Climate Responsibility Monitor (CCRM) 2023 for short- and medium-term targets towards 2030. Companies listed in alphabetical order.

COMPANY NAME	SBTI TEMPERATURE RATINGS (near-term targets)	CCRM INTEGRITY ASSESSMENT (SHORT- & MEDIUM-TERM TARGETS TOWARDS 2030)	KEY ISSUES EXPLAINING THE DIFFERENCE BETWEEN SBTI VERIFICATIONS AND CCRM ASSESSMENTS FOR CORPORATE TARGETS TOWARDS 2030 (E.G., DUE TO ISSUES THAT UNDERMINE THE TARGET INTEGRITY)
Ahold Delhaize	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Moderate integrity covering the entire value chain emissions (s1, s2, s3)	SBTi does not disclose which methods have been used to verify Ahold Delhaize's targets. The targets currently published on SBTi's website as of February 2023 are outdated; in November 2022, Ahold Delhaize published new targets. We found that these targets almost align with 1.5°C-aligned global cross-sector and sector-specific benchmarks.
American Airlines	Well-below 2°C temperature alignment provided for s1 and s2 targets only, excluding non-GHG climate impacts Note: SBTi verifies American Airlines 2035 target as a near-term target.	Poor integrity covering the entire value chain emissions (s1, s2, s3) including non-GHG climate impacts Note: Integrity evaluation for long-term targets' beyond 2030 given SBTI's 2035 target verification.	The SBTi methodology for aviation covers emissions from jet fuel but leaves out other emission sources and non-GHG climate effects from aviation. In addition, SBTi only provides 'well-below 2°C'-aligned certifications but does not provide '1.5°C'-aligned certifications. American Airlines' 2035 intensity target does not meet TPI's benchmarks that are compatible with global warming of 'below 2°C' and '1.5°C'.
Apple	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Moderate integrity covering the entire value chain emissions (s1, s2, s3)	Apple's 2030 emission reduction target aligns with 1.5°C-aligned cross-sectoral global emission trajectories while no sector-specific benchmarks have been identified in the literature. The company does not set a short-term target for within a five-year interval to substantiate its 2030 target.
Carrefour	Well-below 2°C temperature alignment provided for s1 and s2 targets only (s3 target covering cat 1 purchased goods and services and cat 11 use of sold products emissions listed on SBTi webpage but not covered by provided temperature alignment for s1 & s2)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	A footnote of the company's Annual Report reveals through potentially misleading language that the majority of Carrefour's stores and global activities are excluded from the company's emissions disclosure and targets. Accordingly, the 2030 targets equate to a commitment to reduce 2019 emissions by an estimated 6-14%, which falls way short of sector-specific and cross-sectoral 1.5°C Paris Agreement-aligned decarbonisation milestones.
Deutsche Post DHL	1.5°C temperature alignment provided for \$1 and \$2 targets only excluding non-GHG climate impacts (\$3 target listed on SBTi webpage, but not covered by provided temperature alignment for \$1 & \$2)	Poor integrity covering the entire value chain emissions (s1, s2, s3) including non-GHG climate impacts	SBTi does not communicate what assumptions and methodologies the verification is based on. 70% of Deutsche Post's emissions are from air transport, but SBTi does not have a 1.5°C methodology for airlines; the maritime guidance was released after SBTi verified Deutsche Post's target. The baseline year 2021 is much higher than previous years, making the target sound more ambitious than it is.
Fast Retailing	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target covering cat 1 purchased goods and services listed on SBTi webpage but not covered by provided temperature alignment for s1 & s2)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	Targets verified by SBTi only cover 74% of the company's emissions 2019 footprint, excluding downstream scope 3 and a share of upstream scope 3 emissions. When taken together, the 2030 targets equate to a commitment to reduce 2019 full value-chain emissions by an estimated 19% and fall short of cross-sectoral and sector-specific 1.5°C-aligned benchmarks.
Holcim	1.5°C temperature alignment provided for s1 and s2 targets. It remains unclear whether s3 target covering cat 1, cat 3, cat 4 and cat 9 emissions verified with the temperature alignment or not	Moderate integrity covering the entire value chain emissions (s1, s2, s3)	Targets for scope 1 and 2 only meet the upper range of available 1.5°C-aligned benchmarks (SBTi), but clearly miss the lower range (CAT, State of Climate Action). Target for scope 1 only meet 'below 2°C' benchmarks by TPI. Substantial range of 2030 benchmarks due to underlying assumptions about the technical potential of some mitigation levers.

COMPANY NAME	SBTI TEMPERATURE RATINGS (NEAR-TERM TARGETS)	CCRM INTEGRITY ASSESSMENT (SHORT- & MEDIUM-TERM TARGETS TOWARDS 2030)	KEY ISSUES EXPLAINING THE DIFFERENCE BETWEEN SBTI VERIFICATIONS AND CCRM ASSESSMENTS FOR CORPORATE TARGETS TOWARDS 2030 (E.G., DUE TO ISSUES THAT UNDERMINE THE TARGET INTEGRITY)
H&M Group	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Moderate integrity covering the entire value chain emissions (s1, s2, s3)	The company does not set a short-term target for within a five-year interval to substantiate its 2030 target.
Inditex	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	SBTi verification of scope 1 and 2 target as '1.5°C' covers only 3% of 2021 total emissions. Scope 3 target misses all 1.5°C-aligned benchmarks identified in the literature.
PepsiCo	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	covering the entire value chain emissions (s1, s2, s3)	SBTi does not disclose which methods have been used to verify PepsiCo's targets. SBTi allows offsetting under the guise of 'insetting' to realise targets as part of the FLAG guidance; we do not consider this an adequate approach to claim neutralisation of emissions, among other reasons, due to high uncertainties regarding permanence and potential. Since the role of insetting remains unknown, we cannot do an assessment of PepsiCo's reduction targets.
Mercedes-Benz	1.5°C temperature alignment provided for s1 and s2 targets only Well-below 2°C temperature alignment provided for s3 target covering emissions from the use of sold products (cat 11)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	Targets for 2030 do not meet 1.5°C-aligned phaseout dates for internal combustion engines for key markets. SBTi presents intensity target for scope 3 emissions from use of sold products of light duty vehicles as 'well-below 2°C compatible' on the website, but methodology for automobile manufactures has been paused indefinitely by SBTi. SBTi verification of scope 1 and 2 target as '1.5°C' covers only 1% of 2021 total emissions (verification method not disclosed).
Microsoft	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	SBTi's 1.5°C verification for Microsoft's renewables target for 2030 only covers around 3% of Microsoft's 2021 emissions. Scope 3 intensity target announced by Microsoft in 2017 still verified by SBTi but covered under a more recent target to become carbon negative by 2030. 38% emission reduction by Microsoft across all scopes next to carbon negative target falls way short of global cross-sector 1.5°C-aligned benchmarks.
Nestlé	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	The targets for 2025 and 2030 depend on offsetting under the guise of 'insetting'. SBTi FLAG methodologies allows companies to depend on 'insetting' as part of its '1.5°C' verification. We do not consider this an adequate approach to claim neutralisation of emissions, among other reasons, due to high uncertainties regarding permanence and potential.
Thyssenkrupp	Well-below 2°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Moderate integrity covering the entire value chain emissions (s1, s2, s3)	We cannot independently assess Thyssenkrupp's cross-divisional 2030 targets against existing 1.5°C benchmarks available in the literature for the steel sector (SBTi, TPI, CAT, SoCA). However, Thyssenkrupp might be sufficiently contributing towards the achievement of other 1.5°C-aligned milestones for the steel sector, namely the operationalisation of low or near-zero carbon facilities by 2030.
Volkswagen	1.5°C temperature alignment provided for s1 and s2 targets only 2°C temperature alignment provided for s3 target covering emissions from the use of sold products (cat 11)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	Targets for 2030 do not meet 1.5°C-aligned phaseout dates for internal combustion engines for key markets. SBTi presents intensity target for scope 3 emissions from use of sold products of light duty vehicles as '2°C compatible' on the website, but methodology for automobile manufactures has been paused indefinitely by SBTi. SBTi verification of scope 1 and 2 target as '1.5°C' covers only 2% of 2021 total emissions (verification method not disclosed).
Walmart	1.5°C temperature alignment provided for s1 and s2 targets only (s3 target listed on SBTi webpage, but not covered by provided temperature alignment for s1 & s2)	Poor integrity covering the entire value chain emissions (s1, s2, s3)	SBTi verification of scope 1 and 2 target as '1.5°C' covers only 9% of 2019 total emissions (verification method not disclosed). No 2030 target for scope 3 emissions.

LEGENDS: s1:scope1 s2:scope2 s3:scope3 cat:category

Annex IV Assumptions for aggregated analysis

The aggregated impact analysis across the 24 companies' climate strategies assessed in Section A of the *Corporate Climate Responsibility Monitor 2023* contains several assumptions that are additional to the assumptions presented in the individual company assessments in Section B. These additional assumptions concern the interpretation of corporate intensity targets, the minimum emission reduction levels assumed for net-zero target years, and the overlap between companies' value chain emissions.

Intensity targets

Several companies commit to intensity targets for the period up to 2030 that cannot directly be translated into absolute emission reduction commitments. To give a most optimistic scenario of the emission reductions that companies' emission reductions could lead to, we present aggregated findings with the scenario that companies' *intensity* targets will lead to an equivalent emissions reduction in *absolute* terms. In other words, we present the scenario that activity levels remain constant until 2030. For example, Mercedes-Benz commits to at least 40% CO₂ emissions intensity reduction per kilometre for new vehicles by 2030 below 2018 levels ('well-to-wheel' use phase, scope 3 category 11). For the aggregate analysis in Section A, we present the optimistic scenario that this is an absolute reduction target for scope 3 category 11 emission of 40% by 2030 below 2018 levels assuming constant activity levels. We provide further explanation on the interpretation of these targets for all six companies (ArcelorMittal, Holcim, Maersk, Mercedes-Benz, JBS, and Volkswagen) in respective company sections in Annex II. We consider that this aggregated scenario is highly optimistic and unlikely in some cases; accordingly, we do not use this optimistic scenario for the company-specific integrity assessments in Section B, where we evaluate companies' real commitments.

Minimum emission reduction levels assumed for net-zero target years

Half of the companies do not commit to any emission reduction target next to their net-zero pledge; they do not specify what share of their net-zero pledge will be achieved through real reductions within the value chain and what share through offsetting. For these companies, we take absolute emission reduction commitments closest to the net-zero target year as minimum assumption of reductions in the net-zero target year. For example, Foxconn commits to 63% carbon emission reductions by 2035 compared to a 2020 baseline but does *not* commit to any emission reduction target alongside its net-zero pledge for 2050. We assume a 63% emission reduction by 2050 compared to a 2020 baseline as a minimum reduction for the net-zero target year as well. We make such assumptions for eleven companies (Amazon, American Airlines, ArcelorMittal, Deutsche Post DHL, Fast Retailing, Foxconn, Mercedes-Benz, Nestlé, Inditex, JBS, and Volkswagen). We do not use this estimate for the company-specific integrity assessment in Section B.

Emissions overlap between companies' value chains

Certain emission sources might overlap between companies' value chains, for example between steel companies' production emissions (ArcelorMittal, Thyssenkrupp) and automobile manufactures' upstream scope 3 emissions from procuring steel for their car production (Mercedes-Benz, Stellantis, Volkswagen). We do not consider these potential overlaps for the aggregate analysis given the small sample size and a lack of publicly-available information. Our aggregate results in Figure S1 and Figure 2 might represent an upper bound (optimistic) estimate given that companies likely consider each other's specific emission reduction commitments in their respective climate pledges. For example, Apple actively engages with Foxconn on its decarbonisation strategy as one of Apple's key own mitigation measures. For this reason, emission sources with solid decarbonisation plans might be counted twice as part of the total joint footprint of 2.2 GtCO₂e across 24 companies. Even if the joint footprint across all 24 companies is less than 2.2 GtCO₂e due to overlaps between the companies, we do not anticipate that this would have any significant impact on the aggregated findings.

References

- ACEC (2022) Driving corporate renewable electricity procurement in Asia. Asia Clean Energy Coalition. Available at: https://asiacleanenergycoalition.com/en/ (Accessed: 10 January 2023).
- Ahmed, S., Warne, T., Smith, E., et al. (2021) 'Systematic review on effects of bioenergy from edible versus inedible feedstocks on food security', npj Science of Food, 5(9). doi:10.1038/s41538-021-00091-6.
- Ahold Delhaize (2022a) Ahold Delhaize Annual Report 2021. Zaandam, The Netherlands: Koninklijke Ahold Delhaize N.V. Available at: https://media.aholddelhaize.com/media/2zdi1gsi/aholddelhaize_ar21_interactive.pdf?t=637819090356130000 (Accessed: 9 November 2022).
- Ahold Delhaize (2022b) Climate Plan | Update November 2022. Koninklijke Ahold Delhaize N.V. Available at: https://www.ahold-delhaize.com/news/ahold-delhaize-sets-updated-co2-emissions-reductions-targets-for-its-entire-value-chain-in-line-with-ungoal-of-keeping-global-warming-below-1-5-c/ (Accessed: 9 December 2022).
- Ahold Delhaize (2022c) Climate update. Koninklijke Ahold Delhaize N.V. Available at: https://www.aholddelhaize.com/en/news/ ahold-delhaize-shares-climate-update-with-dutch-ngo-milieudefensie/ (Accessed: 14 December 2022).
- Albert Heijn (2021) Steeds beter eten. Duurzaamheidsverslag 2020 Albert Heijn. Zaandam, The Netherlands. Available at: https:// nieuws.ah.nl/steeds-beter-eten/ (Accessed: 8 November 2022).
- Alerigi Jr., A. (2022) 'Stellantis weighs producing ethanol hybrid vehicle in Brazil', Reuters, 4 March. Available at: https://www.reuters.com/business/autos-transportation/stellantis-weighs-producing-ethanol-hybrid-vehicle-brazil-2022-03-04/ (Accessed: 1 November 2022).
- Alphabet (2022) Alphabet CDP Climate Change Response. Alphabet, Inc. Available at: https://kingfisher.com/en/media/Features_ Blogs/sustainability-and-communities-/cdp-climate-change-rating-.html (Accessed: 28 November 2022).
- Amazon (2021a) Amazon pledges support to forests and communities in the Brazilian Amazon, Amazon.com, Inc. Available at: https://www.aboutamazon.com/news/sustainability/amazon-pledges-support-to-forests-and-communities-in-the-brazilianamazon (Accessed: 28 November 2022).
- Amazon (2021b) Carbon Footprint: Driving Carbon Out of Our Business. Available at: https://sustainability.aboutamazon.com/environment/sustainable-operations/carbon-footprint (Accessed: 5 January 2022).
- Amazon (2021c) Renewable Energy Methodology. Amazon.com, Inc. Available at: https://sustainability.aboutamazon.com/renewable-energy-methodology.pdf (Accessed: 28 November 2022).
- Amazon (2022a) Delivering Progress Every Day Amazon's 2021 Sustainability Report. Amazon.com, Inc. Available at: https://sustainability.aboutamazon.com/2021-sustainability-report.pdf (Accessed: 28 November 2022).
- Amazon (2022b) Renewable Energy, Amazon.com, Inc. Available at: https://sustainability.aboutamazon.com/environment/renewable-energy?energyType=true (Accessed: 28 November 2022).
- American Airlines (2021) 'American Airlines Makes Additional Commitment to Sustainable Aviation Fuel', 1 December. Available at: https://news.aa.com/news/news-details/2021/American-Airlines-Makes-Additional-Commitment-to-Sustainable-Aviation-Fuel-CORP-OTH-12/default.aspx (Accessed: 25 November 2022).
- American Airlines (2022a) ESG Report 2021. Fort Worth, Texas, United States of America: American Airlines Group Inc. Available at: https://news.aa.com/esg/ (Accessed: 25 November 2022).
- American Airlines (2022b) Sustainable aviation fuel. Available at: https://news.aa.com/esg/climate-change/sustainable-avia-tion-fuel/ (Accessed: 30 November 2022).
- American Airlines (2022c) Together We Can Help Protect Our Planet, Cool Effect. Available at: https://www.cooleffect.org/american-airlines (Accessed: 30 November 2022).
- Apple (2020) 'Apple commits to be 100 percent carbon neutral for its supply chain and products by 2030', *Apple Newsroom*, 21 July. Available at: https://www.apple.com/newsroom/2020/07/apple-commits-to-be-100-percent-carbon-neutral-for-its-sup-ply-chain-and-products-by-2030/ (Accessed: 1 November 2022).
- Apple (2022a) Environmental Progress Report. Apple Inc. Available at: https://www.apple.com/environment/pdf/Apple_Environmental_Progress_Report_2022.pdf (Accessed: 5 October 2022).
- Apple (2022b) Supplier Clean Energy 2022 Program Update. Cupertino, United States of America: Apple Inc. Available at: https://www.apple.com/environment/pdf/Apple_Supplier_Clean_Energy_Program_Update_2022.pdf (Accessed: 16 November 2022).
- Apple (2022c) The products you love also love the planet. Apple Inc. Available at: https://www.apple.com/environment/ (Accessed: 16 November 2022).
- ArcelorMittal (2021a) *Climate Action Report 2*. Luxembourg City, Luxembourg: ArcelorMittal. Available at: https://constructalia.arcelormittal.com/files/Climate_Action_Report_2_July_2021--94aa5d83ef86cd03ec059ef8d1728966.pdf (Accessed: 1 December 2022).
- ArcelorMittal (2021b) Climate Change 2021, CDP. CDP.

- ArcelorMittal (2022) Basis of Reporting 2021. Luxembourg: ArcelorMittal. Available at: https://corporate.arcelormittal.com/media/hlhnrqpk/arcelor-mittal-basis-of-reporting-2021.pdf (Accessed: 1 December 2022).
- Atmosfair (2016) Atmosfair Flight Emissions Calculator. Berlin: Atmosfair gGmbH. Available at: https://www.atmosfair.de/wp-content/uploads/atmosfair-flight-emissions-calculator-englisch-1.pdf (Accessed: 22 December 2022).
- Bataillle, C. (2020) Low and zero emissions in the steel and cement industries. Barriers, technologies and policies, OECD Green Growth Papers. Paris: OECD Publishing. Available at: www.oecd.org/greengrowth (Accessed: 29 August 2022).
- Berg, A., Magnus, K.-H., Kappelmark, S., et al. (2020) Fashion on climate: How the fashion industry can urgently act to reduce its greenhouse-gas emissions. McKinsey & Company and Global Fashion Agenda. Available at: https://www.mckinsey.com/industries/ retail/our-insights/fashion-on-climate (Accessed: 28 October 2022).
- Bjørn, A., Lloyd, S.M., Brander, M. and Matthews, H.D. (2022) 'Renewable energy certificates threaten the integrity of corporate science-based targets', *Nature Climate Change*, 12, pp. 539–546. doi:https://doi.org/10.1038/s41558-022-01379-5.
- Boehm, S., Lebling, K., Levin, L., et al. (2021) State of Climate Action 2021: Systems Transformations Required to Limit Global Warming to 1.5°C. Washington, DC: World Resources Institute (WRI). Available at: https://www.wri.org/research/state-climate-action-2021 (Accessed: 1 December 2022).
- Boehm, S., Jeffery, L., Levin, K., *et al.* (2022) *State of Climate Action 2022*. Berlin, Germany and Washington, D.C., United States of America: Bezos Earth Fund, Climate Action Tracker, World Resources Institute. Available at: https://www.wri.org/research/state-climate-action-2022 (Accessed: 2 November 2022).
- Bonaccorsi, L., Ferraro, E. and Massuama, M. (2022) Oil companies in disguise On a ticking 'carbon bomb' called 'Scope 3 emissions' mandatory reporting'. And why investors should avoid car stocks and cars' ESG ratings. Transport & Environment. Available at: https://www.transportenvironment.org/wp-content/uploads/2022/09/Carbon-ESG-finance-Car-Study-V6.pdf (Accessed: 28 December 2022).
- Calvin, K., Cowie, A., Berndes, G., *et al.* (2020) 'Bioenergy for climate change mitigation: Scale and sustainability', *GCB Bioenergy*, 13, pp. 1346–1371. doi:10.1111/gcbb.12863.
- Carbon Credit Quality Initiative (2021) The Carbon Credit Quality Initiative. Available at: https://carboncreditquality.org/ (Accessed: 20 January 2022).
- Carrefour (2022a) Carrefour CDP Climate Change Response 2022. CDP.
- Carrefour (2022b) Commitments for climate. Available at: https://www.carrefour.com/en/csr/climat-commitments (Accessed: 15 November 2022).
- Carrefour (2022c) Universal Registration Document 2021, Annual Financial Report. Available at: https://www.carrefour.com/sites/ default/files/2022-05/Carrefour_-2021_Universal_Registration_Document_1.pdf (Accessed: 15 November 2022).
- Carton, W., Stabinsky, D., Lewis, S., et al. (2022) 'Re: Science community concerns over SBTi approach'. Available at: https://forestsandfinance.org/wp-content/uploads/2022/10/Scientist-letter-to-SBTi-2.pdf (Accessed: 16 December 2022).
- CAT (2020a) EU. Berlin, Germany: Climate Action Tracker (Climate Analytics, NewClimate Institute). Available at: https://climateactiontracker.org/countries/eu/ (Accessed: 22 December 2022).
- CAT (2020b) Paris Agreement Compatible Sectoral Benchmarks: Elaborating the decarbonisation roadmap. Berlin, Germany: Climate Action Tracker (Climate Analytics, NewClimate Institute). Available at: https://climateactiontracker.org/documents/753/ CAT_2020-07-10_ParisAgreementBenchmarks_FullReport.pdf (Accessed: 1 December 2022).
- CAT (2022a) Sector Assessment: International aviation. Berlin, Germany: Climate Action Tracker (Climate Analytics, NewClimate Institute). Available at: https://climateactiontracker.org/sectors/aviation/ (Accessed: 15 May 2022).
- CAT (2022b) Sector assessment: International shipping. Berlin, Germany: Climate Action Tracker (Climate Analytics, NewClimate Institute). Available at: https://climateactiontracker.org/sectors/shipping/ (Accessed: 15 May 2022).
- Clarke, L., Wei, Y.-M., Navarro, A. de la V., et al. (2022) 'Chapter 6 Energy Systems', in Shukla, P.R. et al. (eds) IPCC, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK and New York, NY, USA: Cambridge University Press. doi:10.1017/9781009157926.008.
- ClientEarth (2022) Legal risks of carbon offsets. ClientEarth. Available at: https://www.clientearth.org/latest/documents/legal-risks-of-carbon-offsets/ (Accessed: 12 January 2023).
- Congress.gov (2021) H.R.2570 117th Congress (2021-2022): Climate Risk Disclosure Act of 2021. United States. Available at: https://www.congress.gov/bill/117th-congress/house-bill/2570.
- COP26 Presidency (2021) Policy paper: COP26 declaration on accelerating the transition to 100% zero emission cars and vans. UK Government. Available at: https://www.gov.uk/government/publications/cop26-declaration-zero-emission-cars-and-vans#-full-publication-update-history (Accessed: 2 November 2022).
- Day, T., Mooldijk, S., Smit, S., Posada, E., Hans, F., Fearnehough, H., Kachi, A., Warnecke, C., Kuramochi, T. and Höhne, N. (2022) Corporate Climate Responsibility Monitor 2022: Assessing the Transparency and Integrity of Companies' Emission Reduction and Net-Zero Targets. Berlin and Cologne, Germany: NewClimate Institute, Carbon Market Watch. Available at: https://newclimate. org/2022/02/07/corporate-climate-responsibility-monitor-2022/ (Accessed: 23 December 2022).
- Delasalle, F., Speelman, E., Graham, A., et al. (2022) Making net-zero steel possible An industry-backed, 1.5°C-aligned transition strategy. Mission Possible Partnership. Available at: https://missionpossiblepartnership.org/wp-content/uploads/2022/09/ SteelTSExecutiveSummary.pdf (Accessed: 5 October 2022).

- Deutsche Post DHL (2019) Sustainable Fuels for Logistics: Status. Opportunities. Solutions. Bonn, Germany: Deutsche Post DHL. Available at: https://www.dpdhl.com/en/trends-in-logistics/studies/sustainable-fuels.html (Accessed: 20 December 2022).
- Deutsche Post DHL (2021a) 'Accelerated Roadmap to decarbonization: Deutsche Post DHL Group decides on Science Based Targets and invests EUR 7 billion in climate-neutral logistics until 2030', Press Release, 22 March. Available at: https://www.dhl.com/ global-en/home/press/press-archive/2021/accelerated-roadmap-to-decarbonization.html (Accessed: 20 December 2022).
- Deutsche Post DHL (2021b) Climate protection projects. Available at: https://www.dpdhl.com/en/sustainability/environment/ green-product-portfolio/climate-protection-projects.html (Accessed: 20 December 2021).
- Deutsche Post DHL (2022a) 2021 ESG Presentation: ESG roadmap progress to date. Bonn, Germany: Deutsche Post DHL Group. Available at: https://www.dpdhl.com/content/dam/dpdhl/en/media-center/investors/documents/presentations/2021/ DPDHL-ESG-Presentation-2021.pdf (Accessed: 12 November 2022).
- Deutsche Post DHL (2022b) Deutsche Post DHL Group, CDP Climate Change Questionnaire 2022. CDP.
- Deutsche Post DHL (2022c) 'DHL bietet Privatkunden neuen Service GoGreen Plus zur aktiven Vermeidung von Emissionen beim Paketempfang an', 5 October. Available at: https://www.dpdhl.com/de/presse/pressemitteilungen/2022/dhl-bietet-privatkundenneuen-service-gogreen-plus.html (Accessed: 29 November 2022).
- Deutsche Post DHL (2022d) 'ESG Statbook Geschäftsjahr 2021'. Bonn, Germany: Deutsche Post DHL Group. Available at: https://www.dpdhl.com/content/dam/dpdhl/en/media-center/investors/documents/statbooks/2021/DPDHL-ESG-Statbook-2021.xlsx (Accessed: 29 November 2022).
- Deutsche Post DHL (2022e) 'Greenhouse Gas Verification Statement Number CCP.Vol0027-2021'. Deutsche Post DHL Group. Available at: https://www.dpdhl.com/content/dam/dpdhl/en/media-center/responsibility/dpdhl-group-sgs-verification-statement-gogreen-offsetting-04-2022.pdf (Accessed: 29 November 2022).
- Deutsche Post DHL (2022f) Klimaschutzprojekte. Available at: https://www.dpdhl.com/de/nachhaltigkeit/umwelt/gruenes-produktportfolio/klimaschutzprojekte.html (Accessed: 29 November 2022).
- Deutsche Post DHL (2022g) 'Pressemitteilung', 7 July. Available at: https://www.dpdhl.com/de/presse/pressemitteilungen/2022/ dhl-global-forwarding-und-hapag-lloyd-setzen-mit-modernem-biokraftstoff-ein-zeichen-fuer-nachhaltigen-seetransport.html (Accessed: 29 November 2022).
- Deutsche Post DHL (2022h) 'Science Based Target Initiative bestätigt Klimaziele von Deutsche Post DHL Group / Science Based Target Initiative confirms climate targets of Deutsche Post DHL Group', *Press Release*, 17 November. Available at: https://www. dpdhl.com/de/presse/pressemitteilungen/2022/science-based-target-initiative-bestaetigt-klimaziele-von-deutsche-post-dhlgroup.html (Accessed: 24 November 2022).
- Deutsche Post DHL (2022i) Sustainability. Available at: https://www.deutschepost.de/en/n/nachhaltigkeit.html (Accessed: 30 November 2022).
- Dietz, S., Irwin, W., Rauis, B., et al. (2020) Carbon performance assessment of automobile manufacturers: Note on methodology [December 2020]. London, United Kingdom. Available at: https://www.transitionpathwayinitiative.org/publications/70. pdf?type=Publication (Accessed: 22 December 2022).
- Dietz, S., Byrne, R., Hastreiter, N., et al. (2021) Carbon Performance assessment of international shipping: note on methodology [November 2021]. London, United Kingdom: Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/publications/102.pdf?type=Publication (Accessed: 13 December 2022).
- Dietz, S., Byrne, R. and Sheer, A. (2021) Carbon Performance assessments of airlines: Notes on Methodology [November 2021]. London, United Kingdom: Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/publications/101.pdf?type=Publication (Accessed: 13 December 2022).
- Dietz, S., Gardiner, D. and Scheer, A. (2022) Carbon performance assessment of steelmakers: note on methodology [March 2022]. London, United Kingdom: Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/publications/103.pdf?type=Publication (Accessed: 26 September 2022).
- Dietz, S., Harvey, E., Jahn, V. and Keiller, A.N. (2022) TPI Carbon Performance Assessment of Food Producers: Discussion paper. London, United Kingdom: Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/publications/109.pdf?type=Publication (Accessed: 13 December 2022).
- Dietz, S., Hastreiter, N. and Jahn, V. (2021) Carbon performance assessment of cement producers: note on methodology [updated version, November 2021]. London, United Kingdom: Transition Pathway Initiative (TPI). Available at: https://www.transi-tionpathwayinitiative.org/publications/104.pdf?type=Publication (Accessed: 23 September 2022).
- Dietz, S., Hastreiter, Nikolaus and Scheer, A. (2021) Carbon performance assessment of Other Industrials companies: discussion paper [November 2021]. London, United Kingdom: Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/publications/97.pdf?type=Publication (Accessed: 26 September 2022).
- Donofrio, S., Maguire, P., Daley, C., et al. (2022) The Art of Integrity State of the voluntary carbon markets 2022 Q3. Ecosystem Marketplace. Available at: https://www.ecosystemmarketplace.com/publications/state-of-the-voluntary-carbon-markets-2021/#:~:text=Data from the State of,%24472.9M)%2C and growth in (Accessed: 13 December 2022).
- ETC (2021) Bioresources within a Net-Zero Emissions Economy: Making a Sustainable Approach Possible. Energy Transitions Commission. Available at: https://www.energy-transitions.org/wp-content/uploads/2022/07/ETC-Bioresources-Report-Final.pdf (Accessed: 20 October 2022).

- EU (2022) 'Directive (EU) 2022/2464 of the European Parliament and of the Council of 14 December 2022 amending Regulation (EU) No 537/2014, Directive 2004/109/EC, Directive 2006/43/EC and Directive 2013/34/EU, as regards corporate sustainability reporting'. EUR-Lex: Access to European Union law. Available at: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CE-LEX%3A32022L2464 (Accessed: 18 January 2023).
- European Energy (2022) 'E-methanol for Maersk's first vessel', Press Release, 14 March. Available at: https://europeanenergy. com/2022/03/14/e-methanol-for-maersks-first-vessel/.
- Fast Retailing (2021) Integrated Report 2021: LifeWear, Changing the World. Tokyo, Japan: Fast Retailing Co., Ltd. Available at: https://www.fastretailing.com/eng/ir/library/annual.html (Accessed: 29 September 2022).
- Fast Retailing (2022a) Improved Energy Efficiency. Fast Retailing Co., Ltd. Available at: https://www.fastretailing.com/eng/sustainability/environment/energy.html (Accessed: 30 September 2022).
- Fast Retailing (2022b) Response to Climate Change. Fast Retailing Co., Ltd. Available at: https://www.fastretailing.com/eng/sustainability/environment/climatechange.html (Accessed: 30 September 2022).
- Fennell, P.S., Davis, S.J. and Mohammed, A. (2021) 'Decarbonizing cement production', Joule, 5(6), pp. 1305–1311. Available at: https://www.cell.com/joule/fulltext/S2542-4351(21)00197-5?_returnURL=https%3A%2F%2Flinkinghub.elsevier.com%2Fretrieve%2Fpii%2FS2542435121001975%3Fshowall%3Dtrue (Accessed: 1 December 2022).
- Forbes (2022) The Global 2000 2021. Available at: https://www.forbes.com/lists/global2000/ (Accessed: 1 June 2022).
- Foxconn (2021) 2020 Hon Hai Corporate Sustainability Report. New Taipei City, Taiwan: Hon Hai Precision Industry Co., Ltd. Available at: https://www.foxconn.com/s3/reports/CSR/EN/2020/2020) 3 CSR report-英文版.pdf (Accessed: 1 November 2022).
- Foxconn (2022) 2021 Hon Hai Sustainability Report. New Taipei City, Taiwan: Hon Hai Precision Industry Co., Ltd. Available at: https://www.foxconn.com/s3/reports/CSR/EN/2021/HonHai_2021_Sustainability Report_EN.pdf (Accessed: 11 October 2022).
- Freitag, C., Berners-Lee, M., Widdicks, K., et al. (2021) 'The real climate and transformative impact of ICT: A critique of estimates, trends, and regulations', Patterns, 2(9). doi:10.1016/j.patter.2021.100340.
- Fund for the Amazon (2022a) How We Operate. Available at: https://fundojbsamazonia.org/en/about-us/how-we-operate/ (Accessed: 22 November 2022).
- Fund for the Amazon (2022b) Supported Projects. Available at: https://fundojbsamazonia.org/en/projects/supported-projects/ (Accessed: 22 November 2022).
- Fund for the Amazon (2022c) The Fund. Available at: https://fundojbsamazonia.org/en/about-us/the-fund/ (Accessed: 22 November 2022).
- Fuss, S., Lamb, W.F., Callaghan, M.W., et al. (2018) 'Negative emissions—Part 2: Costs, potentials and side effects', Environmental Research Letters, 13(6), p. 063002. doi:10.1088/1748-9326/aabf9f.
- GHG Protocol (2015) 'An amendment to the GHG Protocol Corporate Standard GHG Protocol Scope 2 Guidance'. World Resources Institute and World Business Council for Sustainable Development. Available at: https://ghgprotocol.org/sites/default/files/ Scope2_ExecSum_Final.pdf.
- GHG Protocol (2021) Update Greenhouse Gas Protocol Carbon Removals and Land Sector Initiative. Available at: https://ghgprotocol.org/blog/update-greenhouse-gas-protocol-carbon-removals-and-land-sector-initiative (Accessed: 18 January 2022).
- Google.org (2022) Philanthropic Funding & Nonprofit Fellowships Google.org Our approach. Available at: https://www.google.org/our-approach/(Accessed: 28 November 2022).
- Google (2011) Google's Carbon Offsets: Collaboration and Due Diligence. Google Inc. Available at: https://static.googleusercontent.com/media/www.google.com/en//green/pdfs/google-carbon-offsets.pdf (Accessed: 30 November 2022).
- Google (2018) Capturing value from waste in upstate New York Sustainability. Available at: https://sustainability.google/progress/ projects/landfill-NewYork/ (Accessed: 1 November 2021).
- Google (2022a) Accelerating Climate Action at Google and Beyond: A Progress Update. Google LLC. Available at: https://www.gstatic.com/gumdrop/sustainability/google-2022-climate-action-progress-update.pdf (Accessed: 28 November 2022).
- Google (2022b) Environmental report. Google LLC. Available at: https://www.gstatic.com/gumdrop/sustainability/google-2022-environmental-report.pdf (Accessed: 28 November 2022).
- Graver, B., Zheng, X.S., Rutherford, D., et al. (2022) Vision 2050: Aligning aviation with the Paris Agreement. Washington DC, USA: International Council on Clean Transportation (ICCT). Available at: https://theicct.org/publication/global-aviation-vision-2050-align-aviation-paris-jun22/ (Accessed: 30 November 2022).
- H&M Group (2021) Sustainability Performance Report 2020. Stockholm, Sweden: H&M Group. Available at: https://hmgroup.com/ news/hm-group-sustainability-performance-report-2020/ (Accessed: 28 October 2022).
- H&M Group (2022a) 'H&M group signs a power purchase agreement with Neoen and Alight for a new solar park in Sweden (90 MWp)', 13 December. Available at: https://hmgroup.com/news/hm-group-signs-power-purchase-agreement-with-neoen-and-alight-for-new-solar-park-in-sweden/.
- H&M Group (2022b) Sustainability Disclosure 2021. Stockholm, Sweden: H&M Group. Available at: https://hmgroup.com/sustainability/sustainability-reporting/ (Accessed: 21 October 2022).
- H&M Group (2023) Climate. Available at: https://hmgroup.com/sustainability/circularity-and-climate/climate/ (Accessed: 10 January 2023).

- Hanssen, S. V., Steinmann, Z.J.N., Daioglou, V., *et al.* (2022) 'Global implications of crop-based bioenergy with carbon capture and storage for terrestrial vertebrate biodiversity', GCB *Bioenergy*, 14(3), pp. 307–321. doi:10.1111/gcbb.12911.
- Hepburn, C., Adlen, E., Beddington, J., et al. (2019) 'The technological and economic prospects for CO2 utilization and removal', Nature, 575, pp. 87–97. doi:https://doi.org/10.1038/s41586-019-1681-6.
- Higham, C. and Kerry, H. (2022) Taking companies to court over climate change: who is being targeted?, Commentary. London, United Kingdom: Grantham Research Institute on climate change and the environment. Available at: https://www.lse.ac.uk/granthaminstitute/news/taking-companies-to-court-over-climate-change-who-is-being-targeted/ (Accessed: 16 December 2022).
- Hof, C., Voskamp, A., Biber, M.F., et al. (2018) 'Bioenergy cropland expansion may offset positive effects of climate change mitigation for global vertebrate diversity', PNAS, 115(52), pp. 13294–13299. doi:10.1073/pnas.1807745115.
- Höglund, R. and Mitchell-Larson, E. (2022) *Bridging the corporate ambition gap*. Carbon Gap. Available at: https://carbongap.org/ bridging-the-corporate-ambition-gap/ (Accessed: 12 January 2023).
- Holcim (2021) 2021 Sustainability Performance Report. Holcim. Available at: https://www.holcim.com/sites/holcim/ files/2022-04/25022022-sustainability-performance_fy_2021_report-en.pdf (Accessed: 1 December 2022).
- Holcim (2022a) Climate Change 2022, CDP. CDP.
- Holcim (2022b) Holcim's Net-Zero Journey Climate Report 2022. Holcim. Available at: https://www.holcim.com/sites/holcim/files/2022-04/08042022-holcim-climate-report-2022.pdf (Accessed: 1 December 2022).
- Holsmark, B. (2012) 'Harvesting in boreal forests and the biofuel carbon debt', *Climate Change*, 112, pp. 415–428. doi:10.1007/ s10584-011-0222-6.
- ICCT (2018) China's Stage VI emission standard for heavy-duty vehicles (final rule), The ICCT Policy Update. International Council on Clean Transportation (ICCT). Available at: www.theicct.org/sites/default/files/publications/China_VI_Policy_Update_20180720. pdf (Accessed: 18 January 2023).
- IEA (2021a) Ammonia Technology Roadmap: Towards more sustainable nitrogen fertiliser production. Paris, France: International Energy Agency (IEA) and OECD Publishing. Available at: https://doi.org/10.1787/f6daa4a0-en (Accessed: 23 January 2023).
- IEA (2021b) Net Zero by 2050: A Roadmap for the Global Energy Sector. Paris, France: International Energy Agency (IEA). Available at: https://www.iea.org/reports/net-zero-by-2050 (Accessed: 10 August 2022).
- IEA (2022) Data Centres and Data Transmission Networks. Paris, France: International Energy Agency (IEA). Available at: https:// www.iea.org/reports/data-centres-and-data-transmission-networks (Accessed: 17 January 2023).
- IMO (2020) Fourth IMO Greenhouse Gas Study. London, United Kingdom: International Maritime Organization. Available at: https://www.imo.org/en/OurWork/Environment/Pages/Fourth-IMO-Greenhouse-Gas-Study-2020.aspx (Accessed: 7 December 2022).
- Inditex (2022a) 2021 Statement of Non-Financial Information. Madrid, Spain: Inditex. Available at: https://www.inditex.com/ itxcomweb/api/media/8cae4f70-40ea-4152-9688-732c4fdd8d91/2021_inditex_statement_of_non_financial_information. pdf?t=1655306399672 (Accessed: 1 December 2022).
- Inditex (2022b) Climate Change 2022, CDP. CDP.
- InfluenceMap (2021) German Automakers and Climate Policy. London, United Kingdom: InfluenceMap. Available at: https:// influencemap.org/EN/report/German-Automakers-And-Climate-Policy-a3edf15c64b2e258c29f83beb93337f6 (Accessed: 29 November 2022).
- InfluenceMap (2022a) American Airlines Group, LobbyMap. Available at: https://lobbymap.org/company/American-Airlines-Group-Inc-fe5d4859f8a0fc026a6485b948854f40 (Accessed: 30 November 2022).
- InfluenceMap (2022b) Mercedes-Benz Group Climate Policy Engagement [Status: 29.11.2022]. InfluenceMap. Available at: https://lobbymap.org/company/Mercedes-Benz-7c1efd951fa2a6dfc2b58e9f311f6d7b/projectlink/Mercedes-Benz-in-Climate-Change-a17c0588f166e2db8b67bd67fb195b0a (Accessed: 29 November 2022).
- InfluenceMap (2022c) Stellantis Climate Policy Engagement [Status: 29.11.2022]. InfluenceMap. Available at: https://lobbymap. org/company/Stellantis-NV-019d8501313bd6e981bf5591457e9d00/projectlink/Stellantis-NV-in-Climate-Change-daa5cbba18af1355160ebb29be091493 (Accessed: 29 November 2022).
- InfluenceMap (2022d) Volkswagen Group Climate Policy Engagement [Status: 29.11.2022]. InfluenceMap. Available at: https://lobbymap.org/company/Volkswagen-9e7f6038049cce3caa35490440a6a54b/projectlink/Volkswagen-In-Climate-Change (Accessed: 28 November 2022).
- IPCC (2018) 'Summary for Policymakers', in Masson-Delmotte, V. et al. (eds) Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. Cambridge, UK and New York, NY, USA: Cambridge University Press, pp. 3–24. doi:10.1017/9781009157940.001.
- IPCC (2022) 'Summary for Policymakers', in Shukla, P.R. et al. (eds) Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK and New York, NY, USA: Cambridge University Press. doi:10.1017/9781009157926.002.
- IRENA (2021) A pathway to decarbonise the shipping sector by 2050. Abu Dhabi, United Arab Emirates: International Renewable Energy Agency. Available at: https://www.irena.org/publications/2021/Oct/A-Pathway-to-Decarbonise-the-Shipping-Sector-by-2050 (Accessed: 21 October 2022).

- IRS (2023) Yearly Average Currency Exchange Rates: U.S. Internal Revenue Service. Available at https://www.irs.gov/individuals/ international-taxpayers/yearly-average-currency-exchange-rates (Accessed: 8 February 2023)
- ISO (2022a) Defining net zero. International Organization for Standardization (ISO). Available at: https://www.iso.org/contents/ news/2022/06/defining-net-zero.html (Accessed: 26 August 2022).
- ISO (2022b) Net Zero Guidelines Accelerating the transition to net zero. Geneva, Switzerland: International Organization for Standardization (ISO). Available at: https://www.iso.org/obp/ui/en/#iso:std:iso:iwa:42:ed-1:v1:en (Accessed: 25 November 2022).
- Jaramillo, P., Kahn Ribeiro, S., Newman, P., et al. (2022) 'Chapter 10 Transport', in Shukla, P.R. et al. (eds) IPCC, 2022: Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK and New York, NY, USA: Cambridge University Press. doi:10.1017/9781009157926.012.
- JBS (2021a) JBS Annual Sustainability Report 2021. JBS S.A. Available at: https://jbs.com.br/wp-content/uploads/2022/08/-sustainability-in-report-jbs-2021.pdf (Accessed: 22 November 2022).
- JBS (2021b) Sustainability Report Environmental, Social and Governance JBS 2020. São Paulo, Brazil: JBS S.A. Available at: https://jbs.com.br/wp-content/uploads/2021/08/-sustainability-in-report-jbs-2020.pdf (Accessed: 22 November 2022).
- Jeffery, L., Höhne, N., Moisio, M., et al. (2020) Options for supporting Carbon Dioxide Removal. Cologne and Berlin, Germany: NewClimate Institute. Available at: https://newclimate.org/wp-content/uploads/2020/07/Options-for-supporting-Carbon-Dioxide-Removal_July_2020.pdf (Accessed: 22 December 2022).
- Johnston, M. (2022) Biggest Companies in the World by Market Cap. Investopedia. Available at: https://www.investopedia.com/ biggest-companies-in-the-world-by-market-cap-5212784#:~:text=As of 2022%2C Apple%2C Saudi,companies in the world today. (Accessed: 17 January 2023).
- Kachi, A., Warnecke, C., Hagemann, M., et al. (2020) Net-zero energy housing virtual Article 6 pilot. Cologne and Berlin: NewClimate Institute. Available at: https://newclimate.org/wp-content/uploads/2020/01/Colombia-Art-6-Virtual-Pilot-15-Janua-ry-2020.pdf.
- Kaiser, M., Hipp, R. and Mayer, C. (2021) 'Kaiser, et al. v. Volkswagen AG [Petition]'. Braunschweig, Germany: Rechtsanwälte Günther. Available at: http://climatecasechart.com/wp-content/uploads/sites/16/non-us-case-documents/2021/2021111_16019_petition.pdf (Accessed: 29 November 2022).
- Kline, K.L., Martinelli, F.S., Mayer, A.L., *et al.* (2015) 'Bioenergy and Biodiversity: Key Lessons from the Pan American Region', *Environmental Management*, 56, pp. 1377–1396. doi:10.1007/s00267-015-0559-0.
- Lee, D.S., Fahey, D.W., Skowron, A., *et al.* (2021) 'The contribution of global aviation to anthropogenic climate forcing for 2000 to 2018', *Atmospheric Environment*, 244, p. 117834. doi:10.1016/J.ATMOSENV.2020.117834.
- Lehne, J. and Preston, F. (2018) Making Concrete Change: Innovation in Low-carbon Cement and Concrete, Chatham House Report. London, United Kingdom: The Royal Institute of International Affairs. Available at: https://www.chathamhouse.org/2018/06/ making-concrete-change-innovation-low-carbon-cement-and-concrete (Accessed: 21 December 2022).
- Ley, K., Mazijk, R. van, Hugill, R., et al. (2021) Unlocking the trillion-dollar fashion decarbonisation opportunity: Existing and innovative solutions. Apparel Impact Institute (Aii) and Fashion for Good. Available at: https://apparelimpact.org/reports/unlocking-the-trillion-dollar-fashion-decarbonisation-opportunity-report/ (Accessed: 28 October 2022).
- Maersk (2021) 2020 Sustainability Report. Copenhagen, Denmark: A.P Møller Mærsk A/S. Available at: https://www.maersk. com/sustainability/reports-and-resources (Accessed: 27 September 2022).
- Maersk (2022a) 2021 Sustainability Report. Copenhagen, Denmark: A.P. Møller Mærsk A/S. Available at: https://www.maersk. com/sustainability/reports-and-resources (Accessed: 27 September 2022).
- Maersk (2022b) A.P. Moller Maersk accelerates Net Zero emission targets to 2040 and sets milestone 2030 targets. Available at: https://www.maersk.com/news/articles/2022/01/12/apmm-accelerates-net-zero-emission-targets-to-2040-and-sets-milestone-2030-targets (Accessed: 11 November 2022).
- Maersk (2022c) 'A.P. Moller Maersk and Carbon Sink sign strategic partnership to accelerate green marine fuels production', *Press Release*, 16 November. Available at: https://www.maersk.com/news/articles/2022/11/16/maersk-and-carbon-sink-sign-strategic-partnership (Accessed: 21 December 2022).
- Maersk (2022d) 'A.P. Moller Maersk continues green transformation with six additional large container vessels', Press Release, 5 October. Available at: https://www.maersk.com/news/articles/2022/10/05/maersk-continues-green-transformation (Accessed: 11 November 2022).
- Maersk (2022e) 'A.P. Moller Maersk to operate first of its kind green warehouse in Denmark', Press Release, 30 August. Available at: https://www.maersk.com/news/articles/2022/08/30/maersk-to-operate-first-of-its-kind-green-warehouse-in-denmark (Accessed: 11 November 2022).
- Maersk (2022f) Investor Day 2022 Presentation. Copenhagen, Denmark: A.P. Moller Maersk. Available at: https://investor. maersk.com/esg-day (Accessed: 7 December 2022).
- Maersk (2022g) 'Maersk orders 110 Volvo VNR Electric trucks for North America', Press Release, 29 March. Available at: https://www.maersk.com/news/articles/2022/03/29/maersk-orders-110-volvo-vnr-electric-trucks-for-north-america (Accessed: 11 November 2022).
- Maersk (2022h) 'Maersk to sign Sustainable Aviation Fuel contract with Air France KLM', *Press Release*, 23 February. Available at: https://www.maersk.com/news/articles/2022/02/23/maersk-to-sign-sustainable-aviation-fuel-contract-with-air-france-klm (Accessed: 11 November 2022).

- Maersk (2023a) 2022 Sustainability Report. Copenhagen, Denmark: A.P. Møller Mærsk A/S. Available at: https://www.maersk. com/sustainability/reports-and-resources (Accessed: 8 February 2023).
- Maersk (2023b) ESG Factbook 2022. Copenhagen, Denmark: A.P. Møller Mærsk A/S. Available at: https://www.maersk.com/ sustainability/reports-and-resources (Accessed: 8 February 2023).
- Mercedes-Benz Group (2020a) 'Mercedes-Benz plants will produce CO₂-neutral worldwide from 2022', Press Release, 24 May. Available at: https://media.mercedes-benz.com/article/c11e72b6-44b6-46c2-8733-a53bdf6eed1f (Accessed: 12 October 2022).
- Mercedes-Benz Group (2020b) 'Next Milestone Ambition 2039: The global Mercedes-Benz supply chain is becoming CO2 neutral', Press Release, 7 December. Available at: https://group-media.mercedes-benz.com/marsMediaSite/en/instance/ko.xht-ml?oid=48369319 (Accessed: 12 October 2022).
- Mercedes-Benz Group (2021a) 'Mercedes-Benz to use green steel in vehicles in 2025, reducing its carbon footprint', Press Release, 24 May. Available at: https://group-media.mercedes-benz.com/marsMediaSite/en/instance/ko/Mercedes-Benz-to-usegreen-steel-in-vehicles-in-2025-reducing-its-carbon-footprint.xhtml?oid=50048886&ls=L3NIYXJjaHJlc3VsdC9zZWFyY2hy-ZXN1bHQueGh0bWw_c2VhcmNoU3RyaW5nPWdyZWVuK3N0ZWVsJnNIYXJ (Accessed: 12 October 2022).
- Mercedes-Benz Group (2021b) 'Unique green power concept: Mercedes-Benz to source CO2-free electricity from solar, wind and hydropower from 2022 onwards', Press Release, 26 March. Available at: https://group-media.mercedes-benz.com/marsMediaSite/en/instance/ko/Unique-green-power-concept-Mercedes-Benz-to-source-CO2-free-electricity-from-solar-wind-and-hydropower-from-2022-onwards.xhtml?oid=49400046 (Accessed: 12 October 2022).
- Mercedes-Benz Group (2022a) Mercedes-Benz Group Climate Policy Report. Stuttgart, Germany: Mercedes-Benz Group AG. Available at: https://group.mercedes-benz.com/dokumente/investoren/berichte/geschaeftsberichte/mercedes-benz/mercedesbenz-ir-climate-policy-report-fy-2021.pdf (Accessed: 11 October 2022).
- Mercedes-Benz Group (2022b) 'Mercedes-Benz plans to build wind farm at its test track in Papenburg, northern Germany', Press Release, 19 September. Available at: https://group-media.mercedes-benz.com/marsMediaSite/en/instance/ko/Mercedes-Benzplans-to-build-wind-farm-at-its-test-track-in-Papenburg-northern-Germany.xhtml?oid=54134101&ls=L2RIL2luc3RhbmNIL2tvL1VudGVybmVobWVuLnhodG1sP29pZD05MjY1NjY0JnJlbElkPTYwODI5JmZ (Accessed: 12 October 2022).
- Mercedes-Benz Group (2022c) 'Mercedes-Benz prepares car production network for new electric portfolio', Press Release, 29 July. Available at: https://group-media.mercedes-benz.com/marsMediaSite/en/instance/ko/Mercedes-Benz-preparescar-production-network-for-new-electric-portfolio.xhtml?oid=53631676&ls=L2RIL2luc3RhbmNIL2tvL1VudGVybmVobW-VuLnhodG1sP29pZD05MjY1NjY0JnJlbElkPTYwODI5JmZyb21PaWQ9OTI2N (Accessed: 14 October 2022).
- Mercedes-Benz Group (2022d) 'Mercedes-Benz signs Memorandum of Understanding with Government of Canada to strengthen cooperation across the electric vehicle value chain, including natural resources development', *Press Release*, 23 August. Available at: https://group-media.mercedes-benz.com/marsMediaSite/en/instance/ko/Mercedes-Benz-signs-Memorandum-of-Understanding-with-Government-of-Canada.xhtml?oid=53952099&ls=L2RIL2luc3RhbmNIL2tvL1VudGVybmVobW-VuLnhodG1sP29pZD05MjY1NjY0JnJlbElkPTYwODI5JmZyb21PaWQ9OTI2 (Accessed: 14 October 2022).
- Mercedes-Benz Group (2022e) Sustainability Report 2021 Mercedes-Benz Group. Stuttgart, Germany: Mercedes-Benz Group AG. Available at: https://sustainabilityreport.mercedes-benz.com/2021/servicepages/downloads/files/entire-mercedes-benz-sr21.pdf (Accessed: 1 October 2022).
- Metz, B., Müller-Kraenner, S. and Resch, J. (2021) 'Deutsche Umwelthilfe (DUH) v. Mercedes-Benz AG [Statement of Claim]'. Stuttgart, Germany: Deutsche Umwelthilfe e.V. Available at: http://climatecasechart.com/wp-content/uploads/sites/16/non-uscase-documents/2021/20210920_17804_petition.pdf (Accessed: 13 October 2022).
- Microsoft (2021) A new approach for Scope 3 emissions transparency. Microsoft Corporation. Available at: https://www.microsoft.com/en-au/sustainability/emissions-impact-dashboard?activetab=pivot_2%3Aprimaryr12 (Accessed: 28 November 2022).
- Microsoft (2022a) 2021 Environmental Sustainability Report. Microsoft Corporation. Available at: https://query.prod.cms.rt.microsoft.com/cms/api/am/binary/RE4RwfV (Accessed: 20 October 2022).
- Microsoft (2022b) FY22 additions to Microsoft's carbon removal portfolio. Microsoft Corporation. Available at: https://app.powerbi.com/view?r=eyJrljoiZTU5OTYwN2EtOTI3Ni00NGE0LThjNWItZTUzZTFINWIxNzFhliwidCl6ImMxMzZIZWMwLWZIO-TItNDVIMC1iZWFILTQ2OTg0OTczZTIzMilsImMiOjF9 (Accessed: 28 November 2022)
- Microsoft (2022c) Microsoft carbon removal. An update with lessons learned in our second year. Microsoft Corporation. Available at: https://query.prod.cms.rt.microsoft.com/cms/api/am/binary/RE4QOOD (Accessed: 28 November 2022).
- Microsoft (2022d) Microsoft Corporation CDP Climate Change Questionnaire. Microsoft Corporation and CDP. Available at: https://query.prod.cms.rt.microsoft.com/cms/api/am/binary/RE2EWBx (Accessed: 28 November 2022).
- Microsoft (2022e) Microsoft's FY21 Carbon Removal Portfolio, Microsoft Corporation. Available at: https://app.powerbi.com/ view?r=eyJrljoiOGM2MGFINGYtMGNINy00YzY5LWEyMTAtOTA0ODEyNzEzYTczliwidCl6ImMxMzZIZWMwLWZIOTItND-VIMC1iZWFILTQ2OTg0OTczZTIzMiIsImMiOjF9 (Accessed: 28 November 2022).
- Microsoft (2022f) Microsoft Supplier Code of Conduct. Microsoft Corporation. Available at: https://www.microsoft.com/en-us/ procurement/supplier-conduct.aspx?activetab=pivot%3Aprimaryr7 (Accessed: 28 November 2022).
- Milne, R. (2022) 'Maersk accelerates net zero push in journey to cleaner shipping', *Financial Times*, 12 January. Available at: https://www.ft.com/content/080b2b09-9923-4e7d-a264-e149ec118764.
- Mitchell, S.R., Harmon, M.E. and O'Connell, K.E.B. (2012) 'Carbon debt and carbon sequestration parity in forest bioenergy production', GCB Bioenergy, 4, pp. 818–827. doi:10.1111/j.1757-1707.2012.01173.x.

- Mooldijk, S., Hans, F., Marquardt, M., et al. (2022) Evaluating corporate target setting in the setting in the Netherlands. Berlin and Cologne, Germany: NewClimate Institute. Available at: https://newclimate.org/sites/default/files/2022-07/NewClimate_Evaluating_corporate_target_setting_in_the_Netherlands_Report_July22.pdf (Accessed: 14 July 2022).
- myclimate (2022) myclimate presents climate protection label of the future. Available at: https://de.myclimate.org/en/information/ news-press/news/newsdetail/myclimate-presents-climate-protection-label-of-the-future/ (Accessed: 16 December 2022).
- Nascimento, L., Kachi, A., Mooldijk, S. and Warnecke, C. (2020) Renewable heating virtual Article 6 pilot. Ground source heat pumps in Khovd, Mongolia. Cologne and Berlin, Germany: NewClimate Institute. Available at: https://newclimate.org/2020/01/29/renewable-heating-virtual-article-6-pilot-ground-source-heat-pumps-in-khovd-mongolia/ (Accessed: 19 January 2023).
- Nestlé (2021a) Creating Shared Value and Sustainability Report 2021. Nestlé S.A. Available at: https://www.themomentum.com/ reports/nestle-2021-shared-value-and-sustainability-report (Accessed: 3 January 2023).
- Nestlé (2021b) 'Nestlé's Net Zero Roadmap'. Vevey, Switzerland: Nestlé S.A. Available at: https://www.nestle.com/sites/default/ files/2020-12/nestle-net-zero-roadmap-en.pdf.
- NewClimate Institute (2020) Our climate responsibility approach A new approach for organisations to take responsibility for their climate impact. Cologne, Germany: NewClimate Institute. Available at: https://newclimate.org/climateresponsibility (Accessed: 5 December 2022).
- NewClimate Institute (2023a) A guide to climate contributions: taking responsibility for emissions without offsetting. Cologne and Berlin, Germany: NewClimate Institute. Available at: https://newclimate.org/resources/publications/a-guide-to-climate-contributions-taking-responsibility-for-emissions-without (Accessed: 31 January 2023).
- NewClimate Institute (2023b) Guidance and assessment criteria for good practice corporate emission reduction and net-zero targets: Version 3.0. Cologne and Berlin, Germany: NewClimate Institute. Available at: https://newclimate.org/resources/publications/ corporate-climate-responsibility-monitor-2023 (Accessed: 13 February 2023).
- NewClimate Institute (2023c) The high hanging fruit of mitigation potential. Cologne and Berlin, Germany: NewClimate Institute. Available at: https://newclimate.org/resources/publications/the-high-hanging-fruit-of-mitigation-potential (Accessed: 31 January 2023).
- NewClimate Institute and Data-Driven EnviroLab (2020) Navigating the nuances of net-zero targets. Thomas Day, Silke Mooldijk and Takeshi Kuramochi (NewClimate Institute) and Angel Hsu, Zhi Yi Yeo, Amy Weinfurter, Yin Xi Tan, Ian French, Vasu Namdeo, Odele Tan, Sowmya Raghavan, Elwin Lim, and Ajay Nair (Data-Driven EnviroLab). Available at: https://newclimate. org/2020/10/22/navigating-the-nuances-of-net-zero-targets/ (Accessed: 19 January 2023).
- Öko-Institut, WWF-US and EDF (2021) Methodology for assessing the quality of carbon credits. Carbon Credit Quality Initiative. Available at: https://carboncreditquality.org/download/MethodologyForAssessingTheQualityOfCarbonCredits.pdf (Accessed: 19 January 2023).
- Ørsted (2022) 'Ørsted and Maersk sign landmark green fuels agreement, as Ørsted enters the US Power-to-X market', *Newsroom*, 10 March. Available at: https://orsted.com/en/media/newsroom/news/2022/03/20220310491311 (Accessed: 25 November 2022).
- Pavlenko, N. and Kharina, A. (2018) Policy and Environmental Implications of using HEFA+ for aviation. Washington D.C., USA. International Council on Clean Transportation (ICCT). Available at: https://www.theicct.org/sites/default/files/publications/ Green-Diesel-Aviation_ICCT-Working-Paper_20180321_vF.pdf (Accessed: 25 November 2022).
- Pavlenko, N. and Searle, S. (2021) Assessing the sustainability implications of alternative aviation fuels. 11. International Council on Clean Transportation (ICCT). Available at: https://theicct.org/sites/default/files/publications/Alt-aviation-fuel-sustainability-mar2021.pdf (Accessed: 25 November 2022).
- PepsiCo (2021) 'PepsiCo's Climate Action Strategy'. Available at: https://www.pepsico.com/docs/default-source/sustainability-and-esg-topics/pepsico's-climate-action-strategy.pdf (Accessed: 1 December 2022).
- PepsiCo (2022a) Agriculture. Available at: https://www.pepsico.com/our-impact/esg-topics-a-z/agriculture (Accessed: 1 December 2022).
- PepsiCo (2022b) Climate Change. Available at: https://www.pepsico.com/our-impact/esg-topics-a-z/climate-change (Accessed: 30 November 2022).
- PepsiCo (2022c) PepsiCo, Inc. CDP Climate Change Questionnaire 2022. CDP. Available at: https://www.pepsico.com/docs/default-source/sustainability-and-esg-topics/2022-cdp-climate-response.pdf?sfvrsn=e4a89bea_5 (Accessed: 1 December 2022).
- PepsiCo (2022d) Renewable Energy. Available at: https://www.pepsico.com/our-impact/esg-topics-a-z/renewable-energy (Accessed: 1 December 2022).
- PepsiCo (2022e) Who we are | About PepsiCo. Available at: https://www.pepsico.com/who-we-are/about-pepsico (Accessed: 20 December 2022).
- Race to Zero (2022) Starting Line and Leadership Practices 3.0 Minimum criteria required for participation in the Race to Zero campaign. UNFCCC. Available at: https://climatechampions.unfccc.int/wp-content/uploads/2022/06/Race-to-Zero-Criteria-3.0-4. pdf (Accessed: 27 July 2022).
- Rekley, S. (2022) Spanish fashion group signs 3-year nature protection deal with WWF, Carbon Pulse. Carbon Pulse. Available at: https://carbon-pulse.com/185745/?utm_source=CP+Daily&utm_campaign=63a3b75c90-CPdaily21122022&utm_medium=e-mail&utm_term=0_a9d8834f72-63a3b75c90-110277897 (Accessed: 22 December 2022).
- Rogelj, J., Shindell, D., Jiang, K., et al. (2018) 'Mitigation Pathways Compatible with 1.5°C in the Context of Sustainable Development', in Masson-Delmotte, V. et al. (eds) Global Warming of 1.5°C. An IPCC Special Report. Cambridge, UK and New York, NY, USA: Cambridge University Press, pp. 93–174. Available at: https://doi.org/10.1017/9781009157940.004.

- RWE (2021) Green electricity for green vehicles RWE signs long-term contract with VW and Luxcara, Press Release. RWE AG. Available at: https://www.rwe.com/en/press/rwe-ag/2021-04-29-green-electricity-for-green-vehicles (Accessed: 26 October 2022).
- Sadowski, M., Perkins, L. and McGarvey, E. (2021) Roadmap to Net Zero: Delivering Science-Based Targets in the Apparel Sector, World Resources Institute. Washington D.C., United States of America: World Resources Institute (WRI) and Apparel Impact Institute (Aii). doi:10.46830/wriwp.20.00004.
- Samsung (2022a) Samsung Electronics Climate Change 2022. CDP.
- Samsung (2022b) 'Samsung Electronics Announces New Environmental Strategy', <u>Samsung Newsroom</u>, 15 September. Available at: https://news.samsung.com/global/samsung-electronics-announces-new-environmental-strategy.
- Samsung (2022c) <u>Samsung Electronics Sustainability Report 2022: A Journey Towards a Sustainable Future</u>. Suwon-si, South Korea: Samsung Electronics Co., Ltd. Available at: https://images.samsung.com/is/content/samsung/assets/uk/sustainability/overview/ Samsung_Electronics_Sustainability_Report_2022.pdf (Accessed: 11 November 2022).
- SBTi (2018a) Apparel and footwear sector Science-based targets guidance. Science Based Targets Initative (SBTi); CDP; United Nations Global Compact; World Resources Institute (WRI); World Wide Fund for Nature (WWF). Available at: http://www.jstor. org/stable/j.ctt16gznzj.10 (Accessed: 26 September 2022).
- SBTi (2018b) 'Sectoral Decarbonization Approach Transport Tool, Version 1.1'. Science Based Targets Initative (SBTi); CDP; World Resources Institute (WRI); World Wide Fund for Nature (WWF). Available at: https://view.officeapps.live.com/op/view.aspx?src=https%3A%2F%2Fsciencebasedtargets.org%2Fresources%2Flegacy%2F2018%2F10%2FSDA-Transport-tool_v1.1_locked. xlsx&wdOrigin=BROWSELINK (Accessed: 14 March 2022).
- SBTi (2018c) Transport Science-Based Target Setting Guidance. Science Based Targets Initative (SBTi). Available at: https://sciencebasedtargets.org/wp-content/uploads/2018/05/SBT-transport-guidance-Final.pdf (Accessed: 30 November 2022).
- SBTi (2020) Guidance for ICT Companies Setting Science Based Targets: Mobile networks operators, fixed networks operators, and data centres operators. Science Based Targets Initative (SBTi). Available at: https://sciencebasedtargets.org/resources/lega-cy/2020/04/GSMA_IP_SBT-report_WEB-SINGLE.pdf (Accessed: 28 November 2022).
- SBTi (2021a) Pathway to Net-Zero: SBTi Technical Summary (Version 1.0). Science Based Targets Initative (SBTi). Available at: https://sciencebasedtargets.org/resources/files/Pathway-to-Net-Zero.pdf (Accessed: 23 December 2022).
- SBTi (2021b) 'SBT Aviation Tool (Version 1.1)'. Science Based Targets Initiative (SBTi). Available at: https://view.officeapps.live. com/op/view.aspx?src=https%3A%2F%2Fsciencebasedtargets.org%2Fresources%2Ffiles%2FSBTi_Aviation_Tool_v1.1_locked. xlsx&wdOrigin=BROWSELINK (Accessed: 23 December 2022).
- SBTi (2021c) SBTi Corporate Net-Zero Standard, Version 1.0. Science Based Targets Initative (SBTi); CDP; United Nations Global Compact; World Resources Institute (WRI); World Wide Fund for Nature (WWF). Available at: https://sciencebasedtargets.org/ resources/files/Net-Zero-Standard.pdf (Accessed: 1 December 2022).
- SBTi (2021d) Science-Based Target Setting for the Aviation Sector (Version 1.0). Science Based Targets Initative (SBTi); CDP; United Nations Global Compact; World Ressources Institute (WRI); World Wide Fund for Nature (WWF). Available at: https://sciencebasedtargets.org/resources/files/SBTi_AviationGuidanceAug2021.pdf (Accessed: 23 December 2022).
- SBTi (2022a) Cement Science Based Target Setting Guidance (Version 1.0, September 2022). Science Based Targets Initative (SBTi); CDP; United Nations Global Compact; World Resources Institute (WRI); World Wide Fund for Nature (WWF). Available at: https://sciencebasedtargets.org/resources/files/SBTi-Cement-Guidance.pdf (Accessed: 5 October 2022).
- SBTi (2022b) Forest, Land and Agriculture Science Based Target-Setting Guidance. Science Based Targets Initative (SBTi); CDP; United Nations Global Compact; World Resources Institute (WRI); World Wide Fund for Nature (WWF). Available at: https:// sciencebasedtargets.org/resources/files/SBTiFLAGGuidance.pdf (Accessed: 28 September 2022).
- SBTi (2022c) 'Net-Zero Tool (v1.0.3, April 2022)'. Science Based Targets Initative (SBTi); CDP; United Nations Global Compact; World Resources Institute (WRI); World Wide Fund for Nature (WWF). Available at: https://view.officeapps.live.com/op/view. aspx?src=https%3A%2F%2Fsciencebasedtargets.org%2Fresources%2Ffiles%2FNet-Zero-tool.xlsx&wdOrigin=BROWSELINK (Accessed: 5 October 2022).
- SBTi (2022d) 'Science-based Target Setting Tool (v2.1, July 2022)'. Science Based Targets Initative (SBTi). Available at: https://view. officeapps.live.com/op/view.aspx?src=https%3A%2F%2Fsciencebasedtargets.org%2Fresources%2Ffiles%2FSBTi-target-setting-tool.xlsx&wdOrigin=BROWSELINK (Accessed: 5 October 2022).
- SBTi (2022e) Science Based Target Setting for the Maritime Transport Sector (v1.0). Science Based Targets Initative (SBTi); CDP; United Nations Global Compact; World Resources Institute (WRI); World Wide Fund for Nature (WWF). Available at: https:// sciencebasedtargets.org/resources/files/SBTi-Maritime-Guidance.pdf (Accessed: 6 December 2022).
- SBTi (2022f) Transport Our updated OEMS policy (Status: October 2022). Science Based Targets Initative (SBTi). Available at: https://sciencebasedtargets.org/sectors/transport#our-updated-oems-policy (Accessed: 11 October 2022).
- SBTi (2023a) Companies taking action (status as of January 2023). Science Based Targets Initative (SBTi). Available at: https:// sciencebasedtargets.org/companies-taking-action#dashboard (Accessed: 19 January 2023).
- SBTi (2023b) Why is temperature alignment given for scope 1 and 2 targets only, not scope 3? Science Based Targets Initative (SBTi). Available at: https://sciencebasedtargets.org/companies-taking-action#why-is-temperature-alignment-given-for-scope-1-and-2-targets-only-not-scope-3 (Accessed: 24 January 2023).
- Schneider Electric (2022) Gigaton PPA. Available at: https://gigatonppa.com/ (Accessed: 6 December 2022).

- Searchinger, A.T., James, O. and Dumas, P. (2022) *Europe's Land Future? Opportunities to use Europe's land to fight climate change and improve biodiversity and why proposed policies could undermine both.* Princeton, New Jersey, United States of America: Center for Policy Research on Energy and the Environment (C-PREE) at Princeton University. Available at: https://scholar.princeton.edu/tsearchi/publications/europes-land-future (Accessed: 4 November 2022).
- Searchinger, T.D., Beringer, T., Holtsmark, B., et al. (2018) 'Europe's renewable energy directive poised to harm global forests', Nature Communications, 9, p. 3474. doi:10.1038/s41467-018-06175-4.
- Setzer, J. and Higham, C. (2022) Global trends in climate change litigation: 2022 snapshot. London, United Kingdom: Grantham Research Institute on Climate Change and the Environment; Centre for Climate Change Economics and Policy. Available at: https://www.lse.ac.uk/granthaminstitute/wp-content/uploads/2022/08/Global-trends-in-climate-change-litigation-2022-snapshot.pdf (Accessed: 16 December 2022).
- Smith, B. (2020) Microsoft will be carbon neutral by 2030. Microsoft Corporation. Available at: https://blogs.microsoft.com/ blog/2020/01/16/microsoft-will-be-carbon-negative-by-2030/ (Accessed: 28 November 2022).
- Smith, T., Baresic, D., Fahnestock, J., et al. (2021) A Strategy for the Transition to Zero-Emission Shipping: An analysis of transition pathways, scenarios, and levers for change. London, United Kingdom: UMAS and Getting to Zero Coalition. Available at: https:// www.u-mas.co.uk/wp-content/uploads/2021/10/Transition-Strategy-Report.pdf (Accessed: 29 November 2022).
- Statkraft (2021) Statkraft supplies solar and wind power on a large scale. Available at: https://www.statkraft.com/newsroom/ news-and-stories/archive/2021/statkraft-supplies-solar-and-wind-power-on-a-large-scale/ (Accessed: 12 October 2022).
- Stellantis (2022a) 2021 Corporate Social Responsibility Report: Powered By Our Diversity, We Lead The Way The World Moves. Hoofddorp, The Netherlands: Stellantis N.V. Available at: https://www.stellantis.com/content/dam/stellantis-corporate/sustainability/csr-disclosure/stellantis/2021/Stellantis_2021_CSR_Report.pdf.
- Stellantis (2022b) Annual Report and Form 20-F for the year ended December 31, 2021. Stellantis N.V. Available at: https://www.stellantis.com/content/dam/stellantis-corporate/investors/financial-reports/Stellantis_NV_2021_Annual_Report.pdf (Accessed: 14 November 2022).
- Stellantis (2022c) 'Dare Forward 2030: Stellantis Long-Term Strategic Plan'. Amsterdam, the Netherlands: Stellantis N.V., p. 54. Available at: https://www.stellantis.com/content/dam/stellantis-corporate/investors/events/strategic-plan-2030/2022_03_01_ Strategic_Plan.pdf (Accessed: 1 November 2022).
- Synergy Research Group (2022) 'Q3 Cloud Spending Up Over \$11 Billion from 2021 Despite Major Headwinds; Google Increases its Market Share', Press Release, 27 October. Available at: https://www.srgresearch.com/articles/q3-cloud-spending-up-over-11-billion-from-2021-despite-major-headwinds-google-increases-its-market-share (Accessed: 17 January 2023).
- Ter-Mikaelian, M.T., Colombo, S.J., Lovekin, D., *et al.* (2015) 'Carbon debt repayment or carbon sequestration parity? Lessons from a forest bioenergy case study in Ontario, Canada', *GCB Bioenergy*, 7, pp. 704–716. doi:10.1111/gcbb.12198.
- Teske, S. (2022) Achieving the Paris Climate Agreement Goals. Part 2: Science-based Target Setting for the Finance industry Net-Zero Sectoral 1.5 °C Pathways for Real Economy Sectors. Cham, Switzerland: Springer. Available at: https://link.springer.com/content/ pdf/10.1007/978-3-030-99177-7.pdf (Accessed: 24 August 2022).
- Teske, S., Bratzel, S., Tellermann, R., et al. (2022) The Internal Combustion Engine Bubble. Hamburg, Germany: Greenpeace. Available at: https://www.greenpeace.de/publikationen/ICE-Bubble_2.pdf (Accessed: 11 November 2022).
- The Hague District Court (2021) Milieudefensie et al. v. Royal Dutch Shell Plc. (Judgment of 26 May 2021, court-issued English translation). Den Haag, the Netherlands: The Hague District Court. Available at: http://climatecasechart.com/wp-content/uploads/sites/16/non-us-case-documents/2021/20210526_8918_judgment-1.pdf (Accessed: 23 December 2022).
- The White House (2022) 'FACT SHEET: Biden-Harris Administration Proposes Plan to Protect Federal Supply Chain from Climate-Related Risks', *Press Release*, 10 November. Available at: https://www.whitehouse.gov/briefing-room/statements-re-leases/2022/11/10/fact-sheet-biden-harris-administration-proposes-plan-to-protect-federal-supply-chain-from-climate-re-lated-risks/ (Accessed: 12 January 2023).
- Thyssenkrupp (2018) Climate Change 2018. CDP
- Thyssenkrupp (2021) Climate Change 2021. CDP.
- Thyssenkrupp (2022a) 2021-2022 Annual Report. Thyssenkrupp AG. Available at: https://d2zo35mdb530wx.cloudfront. net/_binary/UCPthyssenkruppAG/a6f9d5ed-1610-402e-97d3-325f3c35f3fb/Annual-report-2021_2022-thyssenkrupp-AG.pdf (Accessed: 1 December 2022).
- Thyssenkrupp (2022b) Charts on Q4 and FY 2021/22. Thyssenkrupp AG. Available at: https://d2zo35mdb530wx.cloudfront. net/_binary/UCPthyssenkruppAG/bfff4df7-23fb-44fa-931a-fb326128e082/221129_Charts-Q4_FY_21_22_Facts-and-Figures_ vFINAL.pdf (Accessed: 1 December 2022).
- Thyssenkrupp (2022c) Climate Change 2022. CDP.
- Thyssenkrupp (2023) Climate strategy and targets. Thyssenkrupp AG. Available at: https://www.thyssenkrupp.com/en/company/ sustainability/climate-strategy-and-targets (Accessed: 20 January 2023).
- TPI (2022a) American Airlines (Assessment of 13 January 2022). Transition Pathway Initiative (TPI). Available at: https://www. transitionpathwayinitiative.org/companies/american-airlines (Accessed: 30 November 2022).
- TPI (2022b) ArcelorMittal (Assessment of 1 April 2022). Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/companies/arcelor-mittal (Accessed: 30 November 2022).
- TPI (2022c) Holcim (Assessment of 1 April 2022). Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/companies/holcim (Accessed: 18 November 2022).

- TPI (2022d) Thyssenkrupp (Assessment of 1 April 2022). Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/companies/thyssenkrupp (Accessed: 30 November 2022).
- TPI (2022e) Volkswagen (Assessment of 13 January 2022). Transition Pathway Initiative (TPI). Available at: https://www.transitionpathwayinitiative.org/companies/volkswagen (Accessed: 30 November 2022).
- Transport & Environment (2018) How to decarbonise European transport by 2050. Brussels, Belgium: Transport & Environment. Available at: https://www.transportenvironment.org/discover/how-decarbonise-european-transport-2050/ (Accessed: 26 October 2022).
- UN HLEG (2022) Integrity Matters: Net Zero Commitments by Businesses, Financial Institutions, Cities and Regions. United Nations' High-Level Expert Group on the Net Zero Emissions Commitments of Non-State Entities. Available at: https://www.un.org/sites/ un2.un.org/files/high-level_expert_group_n7b.pdf (Accessed: 29 November 2022).
- UNFCCC (2021) Upgrading our systems together: A global challenge to accelerate sector breakthroughs for COP26 and beyond. UNFCCC Climate Champions. Available at: https://racetozero.unfccc.int/wp-content/uploads/2021/09/2030-breakthroughsupgrading-our-systems-together.pdf (Accessed: 1 December 2022).
- UNFCCC (2022) Rules, modalities and procedures for the mechanism established by Article 6, paragraph 4, of the Paris Agreement. FCCC/PA/CMA/2022/L.14. UNFCCC. Available at: https://unfccc.int/documents/621906 (Accessed: 12 January 2023).
- UNFCCC (2023) NAZCA Global Climate Action portal. Available at: https://climateaction.unfccc.int/ (Accessed: 19 January 2023).
- Uniqlo (2022) UNIQLO Launches RE.UNIQLO STUDIO in London Regent Street Store. Fast Retailing Co., Ltd. Available at: https:// www.fastretailing.com/eng/group/news/2209091300.html (Accessed: 24 November 2022).
- de Villafranca, M.J., Nilsson, A., Smit, S., et al. (2022) Decarbonisation in the global steel sector: tracking the progress. Cologne and Berlin, Germany: NewClimate Institute. Available at: https://newclimate.org/sites/default/files/2022-12/steel_final.pdf (Accessed: 2 December 2022).
- Volkswagen (2019) Borneo, Green Energy & Co.: What Volkswagen is doing for the environment. Volkswagen AG. Available at: https://www.volkswagenag.com/en/news/stories/2019/12/what-volkswagen-is-doing-for-the-environment.html (Accessed: 5 January 2022).
- Volkswagen (2021) How Volkswagen is becoming a climate-neutral company. Volkswagen AG. Available at: https://www.volk-swagenag.com/en/news/stories/2021/02/how-volkswagen-is-becoming-a-climate-neutral-company.html (Accessed: 26 October 2022).
- Volkswagen (2022a) Volkswagen AG Climate Change 2022. CDP.
- Volkswagen (2022b) Volkswagen Aktiengesellschaft: Annual Report 2021. Wolfsburg, Germany: Volkswagen AG. Available at: https://annualreport2021.volkswagenag.com/ (Accessed: 14 November 2022).
- Volkswagen (2022c) Volkswagen Aktiengesellschaft: Sustainability Report 2021. Wolfsburg, Germany: Volkswagen AG. Available at: https://www.volkswagenag.com/presence/nachhaltigkeit/documents/sustainability-report/2021/Nonfinancial_Report_2021_e.pdf (Accessed: 14 October 2022).
- Volkswagen ClimatePartner (2022) Volkswagen ClimatePartner: Joint venture for climate action started, Press Release. Volkswagen ClimatePartner GmbH. Available at: https://volkswagen-climatepartner.com/en/joint-venture-climate-action-started (Accessed: 26 October 2022).
- VW Kraftwerk (2021) Energieträgermix für 2020. VW Kraftwerk GmbH. Available at: https://www.vw-kraftwerk.de/presence/ dokumente/transparenz/stromkennzeichnung/VWK_Stromkennzeichnung 2020.pdf (Accessed: 26 October 2022).
- Waldersee, V. (2022) 'Mercedes-Benz climate case dropped by German court, appeal planned', Reuters, 13 September. Available at: https://www.reuters.com/business/autos-transportation/german-district-court-drops-climate-change-case-against-merce-des-benz-2022-09-13/ (Accessed: 14 October 2022).
- Walmart (2020a) Frequently asked questions. Bentonville, United States of America. Available at: https://corporate.walmart.com/ media-library/document/frequently-asked-questions/_proxyDocument?id=00000174-add4-dcf3-a7fc-afd4d41f0000 (Accessed: 2 December 2022).
- Walmart (2020b) Walmart, Inc. CDP Climate Change Questionnaire 2020. Walmart Inc. and CDP. Available at: https://corporate. walmart.com/esgreport/media-library/document/walmart-inc-cdp-climate-change-questionnaire-2020-wednesday-september-30-2020/_proxyDocument?id=0000017a-5896-de8c-a17a-7ab7a1b40000 (Accessed: 2 December 2022).
- Walmart (2021a) *Climate change*. Available at: https://corporate.walmart.com/esgreport/esg-issues/climate-change (Accessed: 2 December 2022).
- Walmart (2021b) Walmart, Inc. Climate Change 2021. CDP; Walmart, Inc. Available at: https://corporate.walmart.com/esgre-port/cdp-response-archive (Accessed: 2 December 2022).
- Walmart (2022a) Environmental, Social and Governance Summary Report. Walmart, Inc. Available at: https://corporate.walmart. com/esgreport/media-library/document/walmart-fy2022-esg-summary/_proxyDocument?id=00000182-21ec-d591-afe2-2bf-cb4df0000 (Accessed: 2 December 2022).
- Walmart (2022b) Project Gigaton. Available at: https://www.walmartsustainabilityhub.com/climate/project-gigaton (Accessed: 2 December 2022).
- Walsh, N. (2020) Achieving 100 percent renewable energy with 24/7 monitoring in Microsoft Sweden, Microsoft Corporation. Available at: https://azure.microsoft.com/en-us/blog/achieving-100-percent-renewable-energy-with-247-monitoring-in-micro-soft-sweden/ (Accessed: 28 November 2022).

- Warnecke, C., Day, T., Schneider, L., et al. (2017) Vulnerability of CDM projects for discontinuation of mitigation activities: Assessment of project vulnerability and options to support continued mitigation. Cologne, Germany: NewClimate Institute. Available at: https://newclimate.org/2017/05/11/vulnerability-of-cdm-projects-for-discontinuation-of-mitigation-activities/ (Accessed: 23 December 2022).
- WBA (2022) Automotive Methodology [Status: 11.10.2022]. World Benchmarking Alliance; CDP; French Environment and Energy Management Agency (ADAME). Available at: https://www.worldbenchmarkingalliance.org/publication/automotive/me-thodology/ (Accessed: 11 October 2022).
- WBCSD and WRI (2004) Greenhouse Gas Protocol A Corporate Accounting and Reporting Standard. Revised Edition. World Resources Institute and World Business Council for Sustainable Development. Available at: https://ghgprotocol.org/corporate-standard (Accessed: 25 May 2018).
- WRI and WBCSD (2013) Technical Guidance for Calculating Scope 3 Emissions. World Resources Institute, World Business Council for Sustainable Development. Available at: http://www.ghgprotocol.org/sites/default/files/ghgp/standards/Scope3_Calculation_Guidance_0.pdf (Accessed: 6 June 2019).
- Wu, K., Kitchin, R. and Yu, J. (2022) Supply Change: Climate commitments and renewable energy progress by consumer electronics brands and their top suppliers. Taipei, Taiwan: Greenpeace, Stand.earth. Available at: https://www.greenpeace.org/static/pla-net4-eastasia-stateless/2022/10/89382b33-supplychange.pdf (Accessed: 16 November 2022).
- WWF and BCG (2020) *Beyond Science-Based Targets: A Blueprint for Corporate Action on Climate and Nature*. World Wide Fund For Nature, Gland, Switzerland and Boston Consulting Group, Boston MA, USA. Available at: https://wwfint.awsassets.panda.org/downloads/beyond_science_based_targets__a_blueprint_for_corporate_action_on_climate_and_nature.pdf (Accessed: 23 December 2022).
- WWF Germany (2022) Ein Nachfolgemodell f
 ür die CO₂-Kompensation: wie Unternehmen zus
 ätzlichen Klimaschutz finanzieren sollten. WWF Germany. Available at: https://www.wwf.de/fileadmin/fm-wwf/Publikationen-PDF/Unternehmen/ WWF-Fit-fuer-Paris-Nachfolgemodell-CO2-Kompensation.pdf (Accessed: 12 January 2023).
- Xi, F., Davis, S.J., Ciais, P., et al. (2016) 'Substantial global carbon uptake by cement carbonation', *Nature Geoscience*, 9(12), pp. 880–883. doi:10.1038/ngeo2840.

The rapid acceleration in the volume of corporate climate pledges, combined with the fragmentation of approaches and the general lack of regulation or oversight, means that it is more difficult than ever to distinguish between real climate leadership and unsubstantiated greenwashing.

The Corporate Climate Responsibility Monitor 2023 evaluates the climate strategies of 24 major corporations. It critically analyses the transparency and integrity of corporate pledges and claims to identify replicable good practice and areas for improvement.



